Euclidean structures and operator theory in Banach spaces

Nigel J. Kalton

Emiel Lorist

Lutz Weis

Author addresses:

Department of Mathematics, University of Missouri, Colombia, MO 65201, USA

DELFT INSTITUTE OF APPLIED MATHEMATICS, DELFT UNIVERSITY OF TECH-NOLOGY, P.O. BOX 5031, 2600 GA DELFT, THE NETHERLANDS *Email address*: emiellorist@gmail.com

INSTITUTE FOR ANALYSIS, KARLSRUHE INSTITUTE FOR TECHNOLOGY, EN-GLERSTRASSE 2, 76128 KARLSRUHE, GERMANY *Email address*: lutz.weis@kit.edu

Contents

Introduction	1
Notation and conventions	9
 Chapter 1. Euclidean structures and α-bounded operator families 1.1. Euclidean structures 1.2. α-bounded operator families 1.3. The representation of α-bounded operator families on a Hilbert space 1.4. The equivalence of α-boundedness and C*-boundedness 	11 12 20 e 25 26
Chapter 2. Factorization of α -bounded operator families 2.1. Factorization of γ - and π_2 -bounded operator families 2.2. α -bounded operator families on Banach function spaces 2.3. Factorization of ℓ^2 -bounded operator families through $L^2(S, w)$ 2.4. Banach function space-valued extensions of operators	31 31 36 39 43
Chapter 3. Vector-valued function spaces and interpolation 3.1. The spaces $\alpha(H, X)$ and $\alpha(S; X)$ 3.2. Function space properties of $\alpha(S; X)$ 3.3. The α -interpolation method 3.4. A comparison with real and complex interpolation	53 53 60 65 69
Chapter 4. Sectorial operators and H^{∞} -calculus 4.1. The Dunford calculus 4.2. (Almost) α -sectorial operators 4.3. α -bounded H^{∞} -calculus 4.4. Operator-valued and joint H^{∞} -calculus 4.5. α -bounded imaginary powers	77 78 80 84 89 97
 Chapter 5. Sectorial operators and generalized square functions 5.1. Generalized square function estimates 5.2. Dilations of sectorial operators 5.3. A scale of generalized square function spaces 5.4. Generalized square function spaces without almost α-sectoriality 	103 104 111 115 122
Chapter 6. Some counterexamples 6.1. Schauder multiplier operators 6.2. Sectorial operators which are not almost α -sectorial 6.3. Almost α -sectorial operators which are not α -sectorial 6.4. Sectorial operators with $\omega_{H^{\infty}}(A) > \omega(A)$	129 129 135 137 141
Acknowledgements	149

iii

iv

CONTENTS

Bibliography

Abstract

We present a general method to extend results on Hilbert space operators to the Banach space setting by representing certain sets of Banach space operators Γ on a Hilbert space. Our assumption on Γ is expressed in terms of α -boundedness for a Euclidean structure α on the underlying Banach space X. This notion is originally motivated by \mathcal{R} - or γ -boundedness of sets of operators, but for example any operator ideal from the Euclidean space ℓ_n^2 to X defines such a structure. Therefore, our method is quite flexible. Conversely we show that Γ has to be α -bounded for some Euclidean structure α to be representable on a Hilbert space.

By choosing the Euclidean structure α accordingly, we get a unified and more general approach to the Kwapień–Maurey factorization theorem and the factorization theory of Maurey, Nikišin and Rubio de Francia. This leads to an improved version of the Banach function space-valued extension theorem of Rubio de Francia and a quantitative proof of the boundedness of the lattice Hardy–Littlewood maximal operator. Furthermore, we use these Euclidean structures to build vectorvalued function spaces. These enjoy the nice property that any bounded operator on L^2 extends to a bounded operator on these vector-valued function spaces, which is in stark contrast to the extension problem for Bochner spaces. With these spaces we define an interpolation method, which has formulations modelled after both the real and the complex interpolation method.

Using our representation theorem, we prove a transference principle for sectorial operators on a Banach space, enabling us to extend Hilbert space results for sectorial operators to the Banach space setting. We for example extend and refine the known theory based on \mathcal{R} -boundedness for the joint and operator-valued H^{∞} -calculus. Moreover, we extend the classical characterization of the boundedness of the H^{∞} -calculus on Hilbert spaces in terms of BIP, square functions and dilations to the Banach space setting. Furthermore we establish, via the H^{∞} -calculus, a version of Littlewood–Paley theory and associated spaces of fractional smoothness for a rather large class of sectorial operators. Our abstract setup allows us to reduce assumptions on the geometry of X, such as (co)type and UMD. We conclude with some sophisticated counterexamples for sectorial operators, with as a highlight the construction of a sectorial operator of angle 0 on a closed subspace of L^p for $1 with a bounded <math>H^{\infty}$ -calculus with optimal angle $\omega_{H^{\infty}}(A) > 0$.

²⁰²⁰ Mathematics Subject Classification. Primary: 47A60; Secondary: 47A68, 42B25, 47A56, 46B30, 46B20, 46B70.

Key words and phrases. Euclidean structure, R-boundedness, factorization, sectorial operator, H^{∞} -calculus, Littlewood–Paley theory, BIP, operator ideal.

The second author is supported by the VIDI subsidy 639.032.427 of the Netherlands Organisation for Scientific Research (NWO). The third author is supported by the Deutsche Forschungsgemeinschaft (DFG) through CRC 1173.

Introduction

Hilbert spaces, with their inner product and orthogonal decompositions, are the natural framework for operator and spectral theory and many Hilbert space results fail in more general Banach spaces, even L^p -spaces for $p \neq 2$. However, one may be able to recover versions of Hilbert space results for Banach space operators that are in some sense "close" to Hilbert space operators. For example, for operators on an L^p -scale the Calderon–Zygmund theory, the A_p -extrapolation method of Rubio de Francia and Gaussian kernel estimates are well-known and successful techniques to extrapolate L^2 -results to the L^p -scale.

A further approach to extend Hilbert space results to the Banach space setting is to replace uniform boundedness assumptions on certain families of operators by stronger boundedness assumptions such as γ -boundedness or \mathcal{R} -boundedness. Recall that a set Γ of bounded operators on a Banach space X is γ -bounded if there is a constant such that for all $(x_1, \dots, x_n) \in X^n, T_1, \dots, T_n \in \Gamma$ and $n \in \mathbb{N}$ we have

(1)
$$\left\| (T_1 x_1, \cdots, T_n x_n) \right\|_{\gamma} \le C \left\| (x_1, \cdots, x_n) \right\|_{\gamma},$$

where $\|(x_k)_{k=1}^n\|_{\gamma} := (\mathbb{E}\|\sum_{k=1}^n \gamma_k x_k\|_X^2)^{\frac{1}{2}}$ with $(\gamma_k)_{k=1}^n$ a sequence of independent standard Gaussian random variables. If X has finite cotype, then γ -boundedness is equivalent to the better known \mathcal{R} -boundedness and in an L^p -space with $1 \le p < \infty$ γ -boundedness is equivalent to the discrete square function estimate

(2)
$$\|(T_1x_1,\cdots,T_nx_n)\|_{\ell^2} \le C \|(x_1,\cdots,x_n)\|_{\ell^2}$$

where $\|(x_k)_{k=1}^n\|_{\ell^2} := \|(\sum_{k=1}^n |x_k|^2)^{1/2}\|_{L^p}$. Examples of the extension of Hilbert space results to the Banach space setting under γ -boundedness assumptions include:

- (i) On a Hilbert space the generator of a bounded analytic semigroup (T_z)_{z∈Σσ} has L^p-maximal regularity, whereas on a UMD Banach space this holds if and only if (T_z)_{z∈Σσ} is γ-bounded (see [Wei01b]).
- (ii) If A and B are commuting sectorial operators on a Hilbert space H with $\omega(A) + \omega(B) < \pi$, then A + B is closed on $D(A) \cap D(B)$ and

$$||Ax||_H + ||Bx||_H \lesssim ||Ax + Bx||_H, \quad x \in D(A) \cap D(B).$$

On a UMD Banach space this is still true if A is γ -sectorial and B has a bounded H^{∞} -calculus (see [**KW01**]).

(iii) A sectorial operator A on a Hilbert space H has a bounded H^{∞} -calculus if and only if it has bounded imaginary powers $(A^{it})_{t \in \mathbb{R}}$. On a Banach space X with Pisier's contraction property, one can characterize the boundedness of the H^{∞} -calculus of a sectorial operator A on X by the γ -boundedness of the set $\{A^{it} : t \in [-1, 1]\}$ (see [**KW16a**]).

INTRODUCTION

These results follow an active line of research, which lift Hilbert space results to the Banach space setting. Typically one has to find the "right" proof in the Hilbert space setting and combine it with γ -boundedness and Banach space geometry assumptions in a nontrivial way.

In this memoir we will vastly extend these approaches by introducing Euclidean structures as a more flexible way to check the enhanced boundedness assumptions such as (1) and (2) and as a tool to transfer Hilbert space results to the Banach space setting without reworking the proof in the Hilbert space case. Our methods reduce the need for assumptions on the geometry of the underlying Banach space X such as (co)type and the UMD property and we also reach out to further applications of the method such as factorization and extension theorems.

We start from the observation that the family of norms $\|\cdot\|_{\gamma}$ (and $\|\cdot\|_{\ell^2}$) on X^n for $n \in \mathbb{N}$ has the following basic properties:

$$\|(x)\|_{\gamma} = \|x\|_X, \qquad x \in X$$

(4) $\|\mathbf{A}\mathbf{x}\|_{\gamma} \le \|\mathbf{A}\| \|\mathbf{x}\|_{\gamma}, \qquad \mathbf{x} \in X^{n},$

where the matrix $\mathbf{A} \colon \mathbb{C}^n \to \mathbb{C}^m$ acts on the vector $\mathbf{x} = (x_1, \cdots, x_n) \in X^n$ in the canonical way and $\|\mathbf{A}\|$ is the operator norm of \mathbf{A} with respect to the Euclidean norm. A *Euclidean structure* α on X is now any family of norms $\|\cdot\|_{\alpha}$ on X^n for $n \in \mathbb{N}$, satisfying (3) and (4) for $\|\cdot\|_{\alpha}$. A family of bounded operators Γ on X is called α -bounded if an estimate similar to (1) and (2) holds for $\|\cdot\|_{\alpha}$. This notion of α -boundedness captures the essence of what is needed to represent Γ on a Hilbert space. Indeed, denote by Γ_0 the absolute convex hull of the closure of Γ in the strong operator topology and let $\mathcal{L}_{\Gamma}(X)$ be the linear span of Γ_0 normed by the Minkowski functional

$$||T||_{\Gamma} = \inf \{\lambda > 0 : \lambda^{-1}T \in \Gamma_0\}.$$

Then Γ is α -bounded for some Euclidean structure α if and only if we have the following "representation" of Γ : there is a Hilbert space H, a closed subalgebra \mathcal{B} of $\mathcal{L}(H)$, bounded algebra homomorphisms $\tau \colon \mathcal{L}_{\Gamma}(X) \to \mathcal{B}$ and $\rho \colon \mathcal{B} \to \mathcal{L}(X)$ such that $\rho\tau(T) = T$ for all $T \in \mathcal{L}_{\Gamma}(X)$, i.e.

$$\Gamma \subseteq \mathcal{L}_{\Gamma}(X) \xrightarrow{\tau} \mathcal{L}(X)$$

$$\xrightarrow{\tau} \overset{\rho}{\underset{\mathcal{B}}{\longrightarrow}} \mathcal{L}(H)$$

This theorem (see Theorems 1.3.2 and 1.4.6) is one of our main results. It reveals the deeper reason why results for bounded sets of operators on a Hilbert space extend to results for α -bounded sets of operators on a Banach space.

On the one hand α -boundedness is a strong notion, since it allows one to represent α -bounded sets of Banach space operators as Hilbert space operators, but on the other hand it is a minor miracle that large classes of operators, which are of interest in applications, are α -bounded. Partially this is explained by the flexibility we have to create a Euclidean structure:

(i) The choices $\|\cdot\|_{\gamma}$ and $\|\cdot\|_{\ell^2}$ that appeared in (1) and (2) are the "classical" choices.

(ii) Every operator ideal $\mathcal{A} \subseteq \mathcal{L}(\ell^2, X)$ defines a Euclidean structure $\|\cdot\|_{\mathcal{A}}$

$$\|(x_1,\cdots,x_n)\|_{\mathcal{A}} := \left\|\sum_{k=1}^n e_k \otimes x_k\right\|_{\mathcal{A}}, \qquad x_1,\cdots,x_n \in X,$$

where $(e_k)_{k=1}^{\infty}$ is an orthonormal basis for ℓ^2 .

(iii) Let \mathcal{B} be a closed unital subalgebra of a C^* -algebra. If $\rho: \mathcal{B} \to \mathcal{L}(X)$ is a bounded algebra homomorphism, then one can construct a Euclidean structure α so that for every bounded subset $\Gamma \subseteq \mathcal{B}$ the set $\rho(\Gamma) \subseteq \mathcal{L}(X)$ is α -bounded.

The choice $\alpha = \gamma$ and the connection to \mathcal{R} -boundedness leads to the theory presented e.g. in [**DHP03**, **KW04**] and [**HNVW17**, Chapter 8]. The choice $\alpha = \ell^2$ connects us with square function estimates, essential in the theory of singular integral operators in harmonic analysis. With a little bit of additional work, boundedness theorems for such operators, e.g. Calderón–Zygmund operators or Fourier multiplier operators, show the ℓ^2 -boundedness of large classes of such operators. Moreover ℓ^2 -boundedness of a family of operators can be deduced from uniform weighted L^p -estimates using Rubio de Francia's A_p -extrapolation theory. See e.g. [**CMP11**, **GR85**] and [**HNVW17**, Section 8.2].

After proving these abstract theorems in Chapter 1, we make them more concrete by recasting them as factorization theorems for specific choices of the Euclidean structure α in Chapter 2. In particular choosing $\alpha = \gamma$ we can show a γ -bounded generalization of the classical Kwapień–Maurey factorization theorem (Theorem 2.1.2) and taking α the Euclidean structure induced by the 2-summing operator ideal we can characterize α -boundedness in terms of factorization through, rather than representability on, a Hilbert space (Theorem 2.1.3). Zooming in on the case that X is a Banach function space on some measure space (S, μ) , we show that the ℓ^2 -structure is the canonical structure to consider and that we can actually factor an ℓ^2 -bounded family $\Gamma \subseteq \mathcal{L}(X)$ through the Hilbert space $L^2(S, w)$ for some weight w (Theorem 2.3.1). Important to observe is that this is our first result where we actually have control over the Hilbert space H. Moreover it resembles the work of Maurey, Nikišin and Rubio de Francia [Mau73, Nik70, Rub82] on weighted versus vector-valued inequalities, but has the key advantage that no geometric properties of the Banach function space are used. Capitalizing on these observations we deduce a Banach function space-valued extension theorem (Theorem 2.4.1) with milder assumptions than the one in the work of Rubio de Francia [**Rub86**]. This extension theorem implies the following new results related to the so-called UMD property for a Banach function space X:

- A quantitative proof of the boundedness of the lattice Hardy-Littlewood maximal function if X has the UMD property.
- The equivalence of the dyadic UMD⁺ property and the UMD property.
- The necessity of the UMD property for the ℓ^2 -sectoriality of certain differentiation operators on $L^p(\mathbb{R}^d; X)$.

Besides the discrete α -boundedness estimates as in (1) and (2) for a sequence of operators $(T_k)_{k=1}^n$, we also introduce continuous estimates for functions of operators $T \colon \mathbb{R} \to \mathcal{L}(X)$ with α -bounded range, generalizing the well-known square function

estimates for $\alpha = \ell^2$ on $X = L^p$ given by

$$\left\| \left(\int_{\mathbb{R}} |T(t)f(t)|^2 \, \mathrm{d}t \right)^{1/2} \right\|_{L^p} \le C \left\| \left(\int_{\mathbb{R}} |f(t)|^2 \, \mathrm{d}t \right)^{1/2} \right\|_{L^p}.$$

To this end we introduce "function spaces" $\alpha(\mathbb{R}; X)$ and study their properties in Chapter 3. The space $\alpha(\mathbb{R}; X)$ can be thought of as the completion of the step functions

$$f(t) = \sum_{k=1}^{n} x_k \, \mathbf{1}_{(a_{k-1}, a_k)}(t),$$

for $x_1, \dots, x_n \in X$ and $a_0 < \dots < a_n$, with respect to the norm

$$||f||_{\alpha} = \left\| \left((a_k - a_{k-1})^{-1/2} x_k \right)_{k=1}^n \right\|_{\alpha}.$$

The most striking property of these spaces is that any bounded operator $T: L^2(\mathbb{R}) \to L^2(\mathbb{R})$ can be extended to a bounded operator $\widetilde{T}: \alpha(\mathbb{R}; X) \to \alpha(\mathbb{R}; X)$ with the same norm as T. As the Fourier transform is bounded on $L^2(\mathbb{R})$ one can therefore quite easily develop Fourier analysis for X-valued functions without assumptions on X. For example boundedness of Fourier multiplier operators simplifies to the study of pointwise multipliers, for which we establish boundedness in Theorem 3.2.6 under an α -boundedness assumption. This is in stark contrast to the Bochner space case, as the extension problem for bounded operators $T: L^2(\mathbb{R}) \to L^2(\mathbb{R})$ to the Bochner spaces $L^p(\mathbb{R}; X)$ is precisely the reason for limiting assumptions such as (co)type, Fourier type and UMD. We bypass these assumptions by working in $\alpha(\mathbb{R}; X)$.

With these vector-valued function spaces we define an interpolation method based on a Euclidean structure, the so-called α -interpolation method. A charming feature of this α -interpolation method is that its formulations modelled after the real and the complex interpolation method turn out to be equivalent. For the γ -and ℓ^2 -structures this new interpolation method can be related to the real and complex interpolation methods under geometric assumptions on the interpolation couple of Banach spaces, see Theorem 3.4.4.

In Chapter 4 and 5 we apply Euclidean structures to the H^{∞} -calculus of a sectorial operator A. This is feasible since a bounded H^{∞} -calculus for A defines a bounded algebra homomorphism

$$\rho \colon H^{\infty}(\Sigma_{\sigma}) \to \mathcal{L}(X)$$

given by $f \mapsto f(A)$. Therefore our theory yields the α -boundedness of

$$\{f(A): \|f\|_{H^{\infty}(\Sigma_{\sigma})} \le 1\}$$

for some Euclidean structure α , which provides a wealth of α -bounded sets. Conversely, α -bounded variants of notions like sectoriality and BIP allow us to transfer Hilbert space results to the Banach space setting, at the heart of which lies a transference result (Theorem 4.4.1) based on our representation theorems. With our techniques we generalize and refine the known results on the operator-valued and joint H^{∞} -calculus and the "sum of operators" theorem for commuting sectorial operators on a Banach space. We also extend the classical characterization of the boundedness of the H^{∞} -calculus in Hilbert spaces to the Banach space setting. Recall that for a sectorial operator A on a Hilbert space H the following are equivalent (see [McI86, AMN97, LM98])

(i) A has a bounded H^{∞} -calculus.

INTRODUCTION

- (ii) A has bounded imaginary powers $(A^{it})_{t \in \mathbb{R}}$.
- (iii) For one (all) $0 \neq \psi \in H^1(\Sigma_{\sigma})$ with $\omega(A) < \sigma < \pi$ we have

$$||x||_H \simeq \int_0^\infty \left(||\psi(tA)x||_H^2 \frac{\mathrm{d}t}{t} \right)^{1/2}, \qquad x \in D(A) \cap R(A).$$

- (iv) $[X, D(A)]_{1/2} = D(A^{1/2})$ with equivalence of norms, where $[\cdot, \cdot]_{\theta}$ denotes the complex interpolation method.
- (v) A has a dilation to a normal operator on a larger Hilbert space H.

Now let A be a sectorial operator on a general Banach space X. If α is a Euclidean structure on X satisfying some mild assumptions and A is *almost* α -sectorial, i.e.

$$\{\lambda AR(\lambda, A)^2 : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_\sigma\}$$

is α -bounded for some $\omega(A) < \sigma < \pi$, then the following are equivalent (see Theorems 4.5.6, 5.1.6, 5.1.8, 5.2.1 and Corollary 5.3.9)

- (i) A has a bounded H^{∞} -calculus.
- (ii) A has α -BIP, i.e. $\{A^{it} : t \in [-1, 1]\}$ is α -bounded.
- (iii) For one (all) $0 \neq \psi \in H^1(\Sigma_{\nu})$ with $\sigma < \nu < \pi$ we have the generalized square function estimates

(5)
$$\|x\|_X \simeq \|t \mapsto \psi(tA)x\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)}, \qquad x \in D(A) \cap R(A).$$

- (iv) $(X, D(A))_{1/2}^{\alpha} = D(A^{1/2})$ with equivalence of norms, where we use the α -interpolation method from Chapter 3.
- (v) A has a dilation to the "multiplication operator" \mathcal{M}_s with $\sigma < s < \pi$ on $\alpha(\mathbb{R}; X)$ given by

$$\mathcal{M}g(t) := (it)^{\frac{2}{\pi}s} \cdot g(t), \qquad t \in \mathbb{R}.$$

For these results we could also use the stronger notion of α -sectoriality, i.e. the α -boundedness of

$$\{\lambda R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma}\}\$$

for some $\omega(A) < \sigma < \pi$, which is thoroughly studied for the γ - and ℓ^2 -structure through the equivalence with \mathcal{R} -sectoriality. However, we opt for the weaker notion of almost α -sectoriality to avoid additional assumptions on both α and X.

We note that the generalized square function estimates as in (5) and their discrete counterparts

$$\|x\|_X \simeq \sup_{t \in [1,2]} \left\| (\psi(2^n t A) x)_{n \in \mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)}, \qquad x \in X,$$

provide a version of Littlewood–Paley theory, which allows us to carry ideas from harmonic analysis to quite general situations. This idea is developed in Section 5.3, where we introduce a scale of intermediate spaces, which are close to the homogeneous fractional domain spaces $\dot{D}(A^{\theta})$ for $\theta \in \mathbb{R}$ and are defined in terms of the generalized square functions

$$\|x\|_{H^{\alpha}_{\theta,A}} := \|t \mapsto \psi(tA)A^{\theta}x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}.$$

If A is almost α -sectorial, we show that A always has a bounded H^{∞} -calculus on the spaces $H^{\alpha}_{\theta,A}$ and that A has a bounded H^{∞} -calculus on X if and only if $\dot{D}(A^{\theta}) = H^{\alpha}_{\theta,A}$ with equivalence of norms (Theorem 5.3.6). If A is not almost α bounded, then our results on the generalized square function spaces break down. We analyse this situation carefully in Section 5.4 as a preparation for the final chapter.

The final chapter, Chapter 6, is devoted to some counterexamples related to the notions studied in Chapter 4 and 5. In particular we use Schauder multiplier operators to show that almost α -sectoriality does not come for free for a sectorial operator A, i.e. that almost α -sectoriality is not a consequence of the sectoriality of A for any reasonable Euclidean structure α . This result is modelled after a similar statement for \mathcal{R} -sectoriality by Lancien and the first author [**KL00**]. Furthermore, in Section 6.3 we show that almost α -sectoriality is strictly weaker than α -sectoriality, i.e. that there exists an almost α -sectorial operator A which is not α -sectorial.

Throughout Chapter 4 and 5 we prove that the angles related to the various properties of a sectorial operator, like the angle of (almost) α -sectoriality, (α -)bounded H^{∞} -calculus and (α -)BIP, are equal. Strikingly absent in that list is the angle of sectoriality of A. By an example of Haase it is known that it is possible to have $\omega_{\text{BIP}}(A) \geq \pi$ and thus $\omega_{\text{BIP}}(A) > \omega(A)$, see [Haa03, Corollary 5.3]. Moreover in [Kal03] it was shown by the first author that it is also possible to have $\omega_{H^{\infty}}(A) > \omega(A)$. Using the generalized square function spaces and their unruly behaviour if A is not almost α -sectorial, we provide a more natural example of this situation, i.e. we construct a sectorial operator on a closed subspace of L^p such that $\omega_{H^{\infty}}(A) > \omega(A)$ in Section 6.4.

The history of Euclidean structures

The γ -structure was first introduced by Linde and Pietsch [LP74] and discovered for the theory of Banach spaces by Figiel and Tomczak–Jaegermann in [FT79], where it was used in the context of estimates for the projection constants of finite dimensional Euclidean subspaces of a Banach space. In [FT79] the norms $\|\cdot\|_{\gamma}$ were called ℓ -norms.

Our definition of a Euclidean structure is partially inspired by the similar idea of a lattice structure on a Banach space studied by Marcolino Nhani [Mar01], following ideas of Pisier. In his work c_0 plays the role of ℓ^2 . Other, related research building upon the work of Marcolino Nhani includes:

- Lambert, Neufang and Runde introduced operator sequence spaces in [LNR04], which use norms satisfying the basic properties of a Euclidean structure and an additional 2-convexity assumption. They use these operator sequence spaces to study Figá–Talamanca–Herz algebras from an operator-theoretic viewpoint.
- Dales, Laustsen, Oikhberg and Troitsky [DLOT17] introduced *p-multi-norms*, building upon the work by Dales and Polyakov [DP12] on 1- and ∞-multinorms. They show that a strongly *p*-multinormed Banach space which is *p*-convex can be represented as a closed subspace of a Banach lattice. This representation was subsequently generalized by Oikhberg [Oik18]. The definition of a 2-multinorm is exactly the same as our definition of a Euclidean structure.

Further inspiration for the constructions in Section 1.4 comes from the theory of operator spaces and completely bounded maps, see e.g. [BL04, ER00, Pau02, Pis03].

In the article by Giannopoulos and Milman [GM01] the term "Euclidean structure" is used to indicate the appearance of the Euclidean space \mathbb{R}^n in the Grassmannian manifold of finite dimensional subspaces of a Banach space, as e.g. spelled out in Dvoretzky's theorem. This article strongly emphasizes the connection with convex geometry and the so-called "local theory" of Banach spaces and does not treat operator theoretic questions. For further results in this direction see [MS86, Pis89, Tom89].

Our project started as early as 2003 as a joint effort of N.J. Kalton and L. Weis and since then a partial draft-manuscript called "Euclidean structures" was circulated privately. The project suffered many delays, one of them caused by the untimely death of N.J. Kalton. Only when E. Lorist injected new results and energy the project was revived and finally completed. E. Lorist and L. Weis would like to dedicate this expanded version to N.J. Kalton, in thankful memory. Some results concerning generalized square function estimates with respect to the γ -structure have in the mean time been published in [KW16a].

Structure of the memoir

This memoir is structured as follows: In Chapter 1 we give the definitions, a few examples and prove the basic properties of a Euclidean structure α . Moreover we prove our main representation results for α -bounded families of operators, which will play an important role in the rest of the memoir. Afterwards, Chapters 2-4 can be read (mostly) independent of each other:

- In Chapter 2 we highlight some special cases in which the representation results of Chapter 1 can be made more explicit in the form of factorization theorems.
- In Chapter 3 we introduce vector-valued function spaces and interpolation with respect to a Euclidean structure.
- In Chapter 4 we study the relation between Euclidean structures and the H[∞]-calculus for a sectorial operator.

Chapter 5 treats generalized square function estimates and spaces and relies heavily on the theory developed in Chapter 3 and 4. Finally in Chapter 6 we treat counterexamples related to sectorial operators, which use the theory from Chapter 4 and 5.

Notation and conventions

Throughout this memoir X will be a complex Banach space. For $n \in \mathbb{N}$ we let X^n be the space of *n*-column vectors with entries in X. For $m, n \in \mathbb{N}$ we denote the space $m \times n$ matrices with complex entries by $M_{m,n}(\mathbb{C})$ and endow it with the operator norm. We will often denote elements of X^n by \mathbf{x} and use x_k for $1 \leq k \leq n$ to refer to the *k*th-coordinate of \mathbf{x} . We use the same convention for a matrix in $M_{m,n}(\mathbb{C})$ and its entries.

The space of bounded linear operators on X will be denoted by $\mathcal{L}(X)$ and we will write $\|\cdot\|$ for the operator norm $\|\cdot\|_{\mathcal{L}(X)}$. For a Hilbert space H we will always let its dual H^* be its Banach space dual, i.e. using a bilinear pairing instead of the usual sesquilinear pairing.

By \leq we mean that there is a constant C > 0 such that inequality holds and by \simeq we mean that both \leq and \gtrsim hold.

CHAPTER 1

Euclidean structures and α -bounded operator families

In this first chapter we will start with the definition, a few examples and some basic properties of a Euclidean structure α . Afterwards will study the boundedness of families of bounded operators on a Banach space with respect to a Euclidean structure in Section 1.2. The second halve of this chapter is devoted to one of our main theorems, which characterizes which families of bounded operators on a Banach space can be represented on a Hilbert space. In particular, in Section 1.3 we prove a representation theorem for α -bounded families of operators. Then, given a family of operators Γ that is representable on a Hilbert space, we construct a Euclidean structure α such that Γ is α -bounded in Section 1.4.

1.0.1. Random sums in Banach spaces. Before we start, let us introduce random sums in Banach spaces. A random variable ε on a probability space (Ω, \mathbb{P}) is called a *Rademacher* if it is uniformly distributed in $\{z \in \mathbb{C} : |z| = 1\}$. A random variable γ on (Ω, \mathbb{P}) is called a *Gaussian* if its distribution has density

$$f(z) = \frac{1}{\pi} e^{-|z|^2}, \qquad z \in \mathbb{C}$$

with respect to the Lebesgue measure on \mathbb{C} . A *Rademacher sequence* (respectively *Gaussian sequence*) is a sequence of independent Rademachers (respectively Gaussians). For all our purposes we could equivalently use real-valued Rademacher and Gaussians, see e.g. [HNVW17, Section 6.1.c].

Two important notions in Banach space geometry are type and cotype. Let $p \in [1,2]$ and $q \in [2,\infty]$ and let $(\varepsilon_k)_{k=1}^{\infty}$ be a Rademacher sequence. The space X has type p if there exists a constant C > 0 such that

$$\left\|\sum_{k=1}^{n}\varepsilon_{n}x_{n}\right\|_{L^{p}(\Omega;X)} \leq C\left(\sum_{k=1}^{n}\|x_{k}\|_{X}^{p}\right)^{\frac{1}{p}}, \qquad \mathbf{x} \in X^{n}.$$

The space X has cotype q if there exists a constant C > 0 such that

$$\left(\sum_{k=1}^{n} \|x_k\|_X^q\right)^{\frac{1}{q}} \le C \left\|\sum_{k=1}^{n} \varepsilon_n x_n\right\|_{L^q(\Omega;X)}, \qquad \mathbf{x} \in X^n,$$

with the obvious modification for $q = \infty$. We say X has *nontrivial type* if it has type p > 1 and we say X has *finite cotype* if it has cotype $q < \infty$. Any Banach space has type 1 and cotype ∞ . Moreover nontrivial type implies finite cotype (see [**HNVW17**, Theorem 7.1.14]).

We can compare Rademachers sums with Gaussians sums and if X is a Banach lattice with $\ell^2\text{-}\mathrm{sums}.$

PROPOSITION 1.0.1. Let $1 \leq p, q \leq \infty$, let $(\varepsilon_k)_{k=1}^{\infty}$ be a Rademacher sequence and let $(\gamma_k)_{k=1}^{\infty}$ be a Gaussian sequence. Then for all $\mathbf{x} \in X^n$ we have

$$\left\|\left(\sum_{k=1}^{n} |x_k|^2\right)^{1/2}\right\|_X \lesssim \left\|\sum_{k=1}^{n} \varepsilon_k x_k\right\|_{L^p(\Omega;X)} \lesssim \left\|\sum_{k=1}^{n} \gamma_k x_k\right\|_{L^q(\Omega;X)}$$

where the first expression is only valid if X is a Banach lattice. If X has finite cotype, then for all $\mathbf{x} \in X^n$ we have

$$\left\|\sum_{k=1}^{n} \gamma_k x_k\right\|_{L^p(\Omega;X)} \lesssim \left\|\sum_{k=1}^{n} \varepsilon_k x_k\right\|_{L^p(\Omega;X)} \lesssim \left\|\left(\sum_{k=1}^{n} |x_k|^2\right)^{1/2}\right\|_X$$

where the last expression is only valid if X is a Banach lattice.

For the proof we refer to [HNVW17, Theorem 6.2.4, Corollary 7.2.10 and Theorem 7.2.13].

1.1. Euclidean structures

A Euclidean structure on X is a family of norms $\|\cdot\|_{\alpha}$ on X^n for all $n \in \mathbb{N}$ such that

(1.1)
$$\|(x)\|_{\alpha} = \|x\|_{X}, \qquad x \in X$$

(1.2)
$$\|\mathbf{A}\mathbf{x}\|_{\alpha} \le \|\mathbf{A}\| \|\mathbf{x}\|_{\alpha}, \qquad \mathbf{x} \in X^{n}, \quad \mathbf{A} \in M_{m,n}(\mathbb{C}), \quad m \in \mathbb{N}.$$

It will be notationally convenient to define $\|\mathbf{x}\|_{\alpha} := \|\mathbf{x}^T\|_{\alpha}$ for a row vector \mathbf{x} with entries in X. Alternatively a Euclidean structure can be defined as a norm on the space of finite rank operators from ℓ^2 to X, which we denote by $\mathcal{F}(\ell^2, X)$. For $e \in \ell^2$ and $x \in X$ we write $e \otimes x$ for the rank-one operator $f \mapsto \langle f, e \rangle x$. Clearly we

have $||e \otimes x|| = ||e||_{\ell^2} ||x||_X$ and any element $T \in \mathcal{F}(\ell^2, X)$ can be represented as

$$T = \sum_{k=1}^{n} e_k \otimes x_k$$

with $(e_k)_{k=1}^n$ an orthonormal sequence in ℓ^2 and $\mathbf{x} \in X^n$. If α is a Euclidean structure on X and $T \in \mathcal{F}(\ell^2, X)$ we define

$$||T||_{\alpha} := ||\mathbf{x}||_{\alpha},$$

where \mathbf{x} is such that T is representable in this form. This definition is independent of the chosen orthonormal sequence by (1.2) and this norm satisfies

(1.1')
$$||f \otimes x||_{\alpha} = ||f||_{\ell^2} ||x||_X \qquad f \in \ell^2, \quad x \in X,$$

(1.2')
$$||TA||_{\alpha} \le ||T||_{\alpha} ||A|| \qquad T \in \mathcal{F}(\ell^2, X), \quad A \in \mathcal{L}(\ell^2).$$

Conversely a norm α on $\mathcal{F}(\ell^2, X)$ satisfying (1.1') and (1.2') induces a unique Euclidean structure by

$$\|\mathbf{x}\|_{\alpha} := \left\| f \mapsto \sum_{k=1}^{n} \langle f, e_k \rangle x_k \right\|_{\alpha}, \qquad \mathbf{x} \in X^n,$$

where $(e_k)_{k=1}^n$ is a orthonormal system in ℓ^2 .

Conditions (1.2) and (1.2') express the right-ideal property of a Euclidean structure. We will call a Euclidean structure α *ideal* if it also has the left-ideal condition

(1.3)
$$\|(Sx_1,\ldots,Sx_n)\|_{\alpha} \leq C \|S\| \|\mathbf{x}\|_{\alpha}, \qquad \mathbf{x} \in X^n, \quad S \in \mathcal{L}(X),$$

which in terms of the induced norm on $\mathcal{F}(\ell^2, X)$ is given by

(1.3')
$$\|ST\|_{\alpha} \le C \|S\| \|T\|_{\alpha} \qquad T \in \mathcal{F}(\ell^2, X), \quad S \in \mathcal{L}(X)$$

If we can take C = 1 we will call α isometrically ideal.

A global Euclidean structure α is an assignment of a Euclidean structure α_X to any Banach space X. If it can cause no confusion we will denote the induced structure α_X by α . A global Euclidean structure is called *ideal* if we have

(1.4)
$$\|(Sx_1,\ldots,Sx_n)\|_{\alpha_Y} \le \|S\| \|\mathbf{x}\|_{\alpha_X}, \quad \mathbf{x} \in X^n, \quad S \in \mathcal{L}(X,Y)$$

for any Banach spaces X and Y. In terms of the induced norm on $\mathcal{F}(\ell^2, X)$ this assumption is given by

(1.4')
$$\|ST\|_{\alpha_Y} \le \|S\| \|T\|_{\alpha_X} \qquad T \in \mathcal{F}(\ell^2, X), \quad S \in \mathcal{L}(X, Y).$$

Note that if α is an ideal global Euclidean structure then α_X is isometrically ideal, which can be seen by taking Y = X in the definition. Many natural examples of Euclidean structures are in fact isometrically ideal and are inspired by the theory of operator ideals, see [**Pie80**].

For two Euclidean structures α and β we write $\alpha \leq \beta$ if there is a constant C > 0 such that $\|\mathbf{x}\|_{\alpha} \leq C \|\mathbf{x}\|_{\beta}$ for all $\mathbf{x} \in X^n$. If C can be taken equal to 1 we write $\alpha \leq \beta$.

PROPOSITION 1.1.1. Let β be an ideal Euclidean structure on a Banach space X. Then there exists an ideal global Euclidean structure α such that $\alpha_X \simeq \beta$. Moreover, if β is isometrically ideal, then $\alpha_X = \beta$.

PROOF. Define α_Y on a Banach space Y as

$$\|\mathbf{y}\|_{\alpha_Y} = \sup\{\|(Ty_1, \dots, Ty_n)\|_{\beta} : T \in \mathcal{L}(Y, X), \|T\| \le 1\}, \quad \mathbf{y} \in Y^n.$$

Then (1.1) and (1.2) for α_Y follow directly from the same properties of β and (1.4) is trivial, so α is an ideal global Euclidean structure. Furthermore, by the ideal property of β , we have

$$\|\mathbf{x}\|_{\alpha_{\mathbf{X}}} \le C \, \|\mathbf{x}\|_{\beta} \le C \, \|\mathbf{x}\|_{\alpha_{\mathbf{X}}}, \qquad \mathbf{x} \in X^n$$

so α_X and β are equivalent. Moreover, they are equal if β is isometrically ideal. \Box

Although our definition of a Euclidean structure is isometric in nature, we will mostly be interested in results stable under isomorphisms. If α is a Euclidean structure on a Banach space X and we equip X with an equivalent norm $\|\cdot\|_1$, then α is not necessarily a Euclidean structure on $(X, \|\cdot\|_1)$. However, this is easily fixed. Indeed, if $C^{-1} \|\cdot\|_X \leq \|\cdot\|_1 \leq C \|\cdot\|_X$, we define

$$\|\mathbf{x}\|_{\alpha_1} := \max\{\|\mathbf{x}\|_{\mathrm{op}_1}, C^{-1} \,\|\mathbf{x}\|_{\alpha}\}, \qquad \mathbf{x} \in X^n,$$

where op₁ denotes the Euclidean structure on $(X, \|\cdot\|_1)$ induced by the operator norm on $\mathcal{F}(\ell^2, X)$. Then α_1 is a Euclidean structure on $(X, \|\cdot\|_1)$ such that $\alpha \simeq \alpha_1$.

Examples of Euclidean structures. As already noted in the previous section, the operator norm induces an ideal global Euclidean structure, as it trivially satisfies (1.1'), (1.2') and (1.4'). For $\mathbf{x} \in X^n$ the induced Euclidean structure is given by

$$\|\mathbf{x}\|_{\rm op} = \sup\left\{\left\|\sum_{k=1}^n a_k x_k\right\|_X : \sum_{k=1}^n |a_k|^2 \le 1\right\} = \sup_{\|x^*\|_{X^*} \le 1} \left(\sum_{k=1}^n |x^*(x_k)|^2\right)^{1/2}.$$

Another example is induced by the *nuclear norm* on $\mathcal{F}(X,Y)$, which for $T \in \mathcal{F}(X,Y)$ is defined by

$$||T||_{\nu} := \inf \left\{ \sum_{k=1}^{n} ||e_k|| ||x_k||_X : T = \sum_{k=1}^{n} e_k \otimes x_k \right\}$$

in which the infimum is taken over all finite representations of T, see e.g. [Jam87, Chapter 1]. Again this norm satisfies (1.1'), (1.2') and (1.4') and for $\mathbf{x} \in X^n$ the induced Euclidean structure is given by

$$\|\mathbf{x}\|_{\nu} = \inf \Big\{ \sum_{j=1}^{m} \|y_j\|_X : \mathbf{x} = \mathbf{A}\mathbf{y}, \mathbf{A} \in M_{n,m}(\mathbb{C}), \max_{1 \le j \le m} \sum_{k=1}^{n} |A_{kj}|^2 \le 1, \Big\}.$$

The operator and nuclear Euclidean structures are actually the maximal and minimal Euclidean structures.

PROPOSITION 1.1.2. For any Euclidean structure α on X we have

$$op \leq \alpha \leq \nu$$
.

PROOF. Fix $\mathbf{x} \in X^n$. For the operator norm structure we have

$$\|\mathbf{x}\|_{\mathrm{op}} = \sup_{\substack{\mathbf{A} \in M_{1,n}(\mathbb{C}) \\ \|\mathbf{A}\| \le 1}} \|\mathbf{A}\mathbf{x}\|_X = \sup_{\substack{\mathbf{A} \in M_{1,n}(\mathbb{C}) \\ \|\mathbf{A}\| \le 1}} \|\mathbf{A}\mathbf{x}\|_{\alpha} \le \|\mathbf{x}\|_{\alpha}.$$

For the nuclear structure take $\mathbf{y} \in X^m$ such that $\mathbf{x} = \mathbf{A}\mathbf{y}$ with $\mathbf{A} \in M_{n,m}(\mathbb{C})$ and $\sum_{k=1}^n |A_{kj}|^2 \leq 1$ for $1 \leq j \leq m$. Then we have

$$\|\mathbf{x}\|_{\alpha} = \|\mathbf{A}\mathbf{y}\|_{\alpha} \le \sum_{j=1}^{m} \|(A_{1j}y_j, \dots, A_{nj}y_j)\|_{\alpha} \le \sum_{j=1}^{m} \|(y_j)\|_{\alpha} = \sum_{j=1}^{m} \|y_j\|_X,$$

so taking the infimum over all such \mathbf{y} gives $\|\mathbf{x}\|_{\alpha} \leq \|\mathbf{x}\|_{\nu}$.

The most important Euclidean structure for our purposes is the *Gaussian struc*ture, induced by a norm on $\mathcal{F}(\ell^2, X)$ first introduced by Linde and Pietsch [LP74] and discovered for the theory of Banach spaces by Figiel and Tomczak–Jaegermann [**FT79**]. It is defined by

$$||T||_{\gamma} := \sup \left(\mathbb{E} \left\| \sum_{k=1}^{n} \gamma_k T e_k \right\|_X^2 \right)^{1/2}, \qquad T \in \mathcal{F}(\ell^2, X),$$

where the supremum is taken over all finite orthonormal sequences $(e_k)_{k=1}^n$ in ℓ^2 . For $\mathbf{x} \in X^n$ the induced Euclidean structure is given by

$$\|\mathbf{x}\|_{\gamma} := \left(\mathbb{E}\left\|\sum_{k=1}^{n} \gamma_k x_k\right\|_X^2\right)^{\frac{1}{2}}, \qquad \mathbf{x} \in X^n,$$

where $(\gamma_k)_{k=1}^n$ is Gaussian sequence (see e.g. [HNVW17, Proposition 9.1.3]). Properties (1.1') and (1.4') are trivial, and (1.2') is proven in [HNVW17, Theorem 9.1.10]. Therefore the Gaussian structure is an ideal global Euclidean structure.

Another structure of importance is the π_2 -structure induced by the 2-summing operator ideal, which will be studied more thoroughly in Section 2.1. The π_2 -norm

is defined for $T \in \mathcal{F}(\ell^2, X)$ as

$$||T||_{\pi_2} := \sup \Big\{ \Big(\sum_{k=1}^n ||TAe_k||_X^2 \Big)^{1/2} : A \in \mathcal{L}(\ell^2), ||A|| \le 1 \Big\},\$$

where $(e_k)_{k=1}^{\infty}$ is an orthonormal basis for ℓ^2 . The induced Euclidean structure for $\mathbf{x} \in X^n$ is

$$\|\mathbf{x}\|_{\pi_{2}} := \sup \Big\{ \Big(\sum_{j=1}^{m} \|y_{j}\|_{X}^{2} \Big)^{1/2} : \mathbf{y} = \mathbf{A}\mathbf{x}, \mathbf{A} \in M_{m,n}(\mathbb{C}), \|\mathbf{A}\| \le 1 \Big\}.$$

Properties (1.1), (1.2) and (1.4) are easily checked, so the π_2 -structure is an ideal global Euclidean structure as well. If X is a Hilbert space, the π_2 -summing norm coincides with the Hilbert-Schmidt norm, which is given by

$$||T||_{\mathrm{HS}} := \left(\sum_{k=1}^{\infty} ||Te_k||^2\right)^{1/2}, \qquad T \in \mathcal{F}(\ell^2, X)$$

for any orthonormal basis $(e_k)_{k=1}^{\infty}$ of ℓ^2 . For an introduction to the theory of *p*-summing operators we refer to [**DJT95**].

If X is a Banach lattice, there is an additional important Euclidean structure, the ℓ^2 -structure. It is given by

$$\|\mathbf{x}\|_{\ell^2} := \left\| \left(\sum_{k=1}^n |x_k|^2 \right)^{1/2} \right\|_X, \quad \mathbf{x} \in X^n.$$

Again (1.1) is trivial and (1.2) follows directly from

(1.5)
$$\left(\sum_{k=1}^{n} |x_k|^2\right)^{1/2} = \sup\left\{\left|\sum_{k=1}^{n} a_k x_k\right| : \sum_{k=1}^{n} |a_k|^2 \le 1\right\},$$

where the supremum is taken in the lattice sense, see [LT79, Section 1.d].

By the Krivine-Grothendieck theorem [LT79, Theorem 1.f.14] we get for $S \in \mathcal{L}(X)$ and $\mathbf{x} \in X^n$ that

$$||(Sx_1,\ldots,Sx_n)||_{\ell^2} \le K_G ||S|| ||\mathbf{x}||_{\ell^2},$$

where K_G is the complex Grothendieck constant. Therefore the ℓ^2 -structure is ideal. The Krivine-Grothendieck theorem also implies that if X is a Banach space that can be represented as a Banach lattice in different ways, then the corresponding ℓ^2 -structures are equivalent. This follows directly by taking T the identity operator on X. An example of such a situation is $L^p(\mathbb{R})$ for $p \in (1, \infty)$, for which the Haar basis is unconditional and induces a lattice structure different from the canonical one.

The ℓ^2 -structure is not a global Euclidean structure, as it is only defined for Banach lattices. However, starting from the ℓ^2 -structure on some Banach lattice X, Proposition 1.1.1 says that there is an ideal global Euclidean structure, which is equivalent to the ℓ^2 -structure on X. We define the ℓ^g -structure as the structure obtained in this way starting from the lattice L^1 . So for $\mathbf{x} \in X^n$ we define

$$\|\mathbf{x}\|_{\ell^{\mathsf{g}}} := \sup_{T} \|(Tx_1, \dots, Tx_n)\|_{\ell^2},$$

where the supremum is taken over all $T: X \to L^1(S)$ with $||T|| \leq 1$ for any measure space (S, μ) . By definition this is a global, ideal Euclidean structure.

Let us compare the Euclidean structures we have introduced.

PROPOSITION 1.1.3. We have on X

(i) $\gamma \leq \pi_2$. Moreover $\pi_2 \lesssim \gamma$ if and only if X has cotype 2.

Suppose that X is a Banach lattice, then we have on X

(ii) $\ell^2 \lesssim \gamma$. Moreover $\gamma \lesssim \ell^2$ if and only if X has finite cotype. (iii) $\ell^2 \leq \ell^g \lesssim \ell^2$.

PROOF. For (i) let $(\gamma_k)_{k=1}^n$ be a Gaussian sequence on a probability space $(\Omega, \mathscr{F}, \mathbb{P})$. Let $f_1, \ldots, f_n \in L^2(\Omega)$ be simple functions of the form $f_k = \sum_{j=1}^m t_{jk} \mathbf{1}_{A_j}$ with $t_{jk} \in \mathbb{C}$ and $A_j \in \mathscr{F}$ for $1 \leq j \leq m$ and $1 \leq k \leq n$. Define

$$\mathbf{A} := \left(\mathbb{P}(A_j)^{1/2} t_{jk} \right)_{j,k=1}^{m,n}.$$

Then we have for $\mathbf{x} \in X^n$ and $\mathbf{y} := \mathbf{A}\mathbf{x}$

$$\left\|\sum_{k=1}^{n} f_k x_k\right\|_{L^2(\Omega;X)} = \left(\sum_{j=1}^{m} \|y_j\|^2\right)^{1/2} \le \|\mathbf{x}\|_{\pi_2} \|\mathbf{A}\|$$

and

$$\|\mathbf{A}\| = \sup_{\|\mathbf{b}\|_{\ell_m^2} \le 1} \left(\sum_{j=1}^m \left| \sum_{k=1}^n \mathbb{P}(A_j)^{1/2} t_{jk} b_k \right|^2 \right)^{1/2} = \sup_{\|\mathbf{b}\|_{\ell_m^2} \le 1} \left\| \sum_{k=1}^n b_k f_k \right\|_{L^2(\Omega)}.$$

Thus approximating $(\gamma_k)_{k=1}^n$ by such simple functions in $L^2(\Omega)$, we deduce

$$\|\mathbf{x}\|_{\gamma} \leq \|\mathbf{x}\|_{\pi_{2}} \sup_{\|\mathbf{b}\|_{\ell_{m}^{2}} \leq 1} \left\|\sum_{k=1}^{n} b_{k} \gamma_{k}\right\|_{L^{2}(\Omega)} = \|\mathbf{x}\|_{\pi_{2}}.$$

Suppose that X has cotype 2. By [HNVW17, Corollary 7.2.11] and the right ideal property of the γ -structure, we have for all $\mathbf{x} \in X^n$, $\mathbf{A} \in M_{m,n}(\mathbb{C})$ with $\|\mathbf{A}\| \leq 1$ and $\mathbf{y} = \mathbf{A}\mathbf{x}$ that

$$\left(\sum_{k=1}^{n} \|y_k\|_X^2\right)^{1/2} \lesssim \left(\mathbb{E}\left\|\sum_{k=1}^{n} \gamma_k y_k\right\|_X^2\right)^{1/2} = \|\mathbf{A}\mathbf{x}\|_{\gamma} \le \|\mathbf{x}\|_{\gamma},$$

which implies that $\|\mathbf{x}\|_{\pi_2} \lesssim \|\mathbf{x}\|_{\gamma}$. Conversely, suppose that the γ -structure is equivalent to π_2 -structure, then we have for all $\mathbf{x} \in X^n$

$$\left(\sum_{k=1}^{n} \|x_k\|_X^2\right)^{1/2} \le \|\mathbf{x}\|_{\pi_2} \lesssim \|\mathbf{x}\|_{\gamma} = \left(\mathbb{E}\left\|\sum_{k=1}^{n} \gamma_k x_k\right\|_X^2\right)^{1/2}$$

So by [HNVW17, Corollary 7.2.11] we know that X has cotype 2.

For (ii) assume that X is a Banach lattice. By Proposition 1.0.1 we have $\|\mathbf{x}\|_{\ell^2} \leq \|\mathbf{x}\|_{\gamma}$. If X has finite cotype we also have $\|\mathbf{x}\|_{\gamma} \leq \|\mathbf{x}\|_{\ell^2}$. Conversely, if the ℓ^2 -structure is equivalent to γ -structure, then we have again by Proposition 1.0.1 that

$$\left(\mathbb{E}\left\|\sum_{k=1}^{n}\varepsilon_{k}x_{k}\right\|_{X}^{2}\right)^{\frac{1}{2}} \gtrsim \left\|\left(\sum_{k=1}^{n}|x_{k}|^{2}\right)^{1/2}\right\|_{X} \gtrsim \left(\mathbb{E}\left\|\sum_{k=1}^{n}\gamma_{k}x_{k}\right\|_{X}^{2}\right)^{\frac{1}{2}}$$

where $(\varepsilon_k)_{k=1}^n$ is a Rademacher sequence. This implies that X has finite cotype by [**HNVW17**, Corollary 7.3.10].

For (iii) note that by the Krivine-Grothendieck theorem [LT79, Theorem 1.f.14] we have $\|\mathbf{x}\|_{\ell^{g}} \leq K_{G} \|\mathbf{x}\|_{\ell^{2}}$. Conversely take a positive $x^{*} \in X^{*}$ of norm one such that

$$\left\langle \left(\sum_{k=1}^{n} |x_k|^2\right)^{1/2}, x^* \right\rangle = \|\mathbf{x}\|_{\ell^2}.$$

Let L be the completion of X under the seminorm $||x||_L := x^*(|x|)$. Then L is an abstract L^1 -space and is therefore order isometric to $L^1(S)$ for some measure space (S, μ) , see [LT79, Theorem 1.b.2]. Let $T : X \to L$ be the natural norm one lattice homomorphism. Then we have

$$\|\mathbf{x}\|_{\ell^2} = \left\| \left(\sum_{k=1}^n |Tx_k|^2 \right)^{1/2} \right\|_L \le \|\mathbf{x}\|_{\ell^g}.$$

Duality of Euclidean structures. We will now consider duality for Euclidean structures. If α is a Euclidean structure on a Banach space X, then there is a natural *dual Euclidean structure* α^* on X^* defined by

$$\|\mathbf{x}^*\|_{\alpha^*} := \sup \Big\{ \sum_{k=1}^n |x_k^*(x_k)| : \mathbf{x} \in X^n, \|\mathbf{x}\|_{\alpha} \le 1 \Big\}, \qquad \mathbf{x}^* \in (X^*)^n.$$

This is indeed a Euclidean structure, as (1.1) and (1.2) for α^* follow readily from their respective counterparts for α . We can then also induce a structure α^{**} on X^{**} , and the restriction of α^{**} to X coincides with α . If α is ideal, then the analogue of (1.3) holds for weak*-continuous operators, i.e. we have

$$\|(S^*x_1^*,\ldots,S^*x_n^*)\|_{\alpha^*} \le C \, \|S\| \|\mathbf{x}^*\|_{\alpha^*}, \qquad \mathbf{x}^* \in (X^*)^n, \quad S \in \mathcal{L}(X).$$

In particular, α^* is ideal if α is ideal and X is reflexive.

If we prefer to express the dual Euclidean structure in terms of a norm on $\mathcal{F}(\ell^2, X)$, we can employ trace duality. If $T \in \mathcal{F}(X)$ and we have two representations of T, i.e.

$$T = \sum_{k=1}^{n} x_k^* \otimes x_k = \sum_{j=1}^{m} \bar{x}_j^* \otimes \bar{x}_j,$$

where $x_k, \bar{x}_j \in X$ and $x_k^*, \bar{x}_j^* \in X^*$, then $\sum_{k=1}^n \langle x_k, x_k^* \rangle = \sum_{j=1}^m \langle \bar{x}_j, \bar{x}_j^* \rangle$ (see **Jam87**, Proposition 1.3]). Therefore we can define the *trace* of *T* as

$$\operatorname{tr}(T) = \sum_{k=1}^{n} \langle x_k, x_k^* \rangle$$

for any finite representation of T. We define the norm α^* on $\mathcal{F}(\ell^2, X^*)$ as

$$||T||_{\alpha^*} := \sup\{|\operatorname{tr}(S^*T)| : S \in \mathcal{F}(\ell^2, X), \alpha(S) \le 1\}, \qquad T \in \mathcal{F}(\ell^2, X^*)$$

This definition coincides with the definition in terms of vectors in X^n . Indeed, for $\mathbf{x}^* \in (X^*)^n$ and $T \in \mathcal{F}(\ell^2, X^*)$ defined as $T = \sum_{k=1}^n e_k \otimes x_k^*$ for some orthonormal

sequence $(e_k)_{k=1}^n$ in ℓ^2 , we have that

$$\|\mathbf{x}^*\|_{\alpha^*} = \sup\left\{\sum_{k=1}^n |x_k^*(x_k)| : \mathbf{x} \in X^n, \|\mathbf{x}\|_{\alpha} \le 1\right\}$$

= $\sup\left\{\left|\sum_{k=1}^n \langle Se_k, Te_k \rangle\right| : S \in \mathcal{F}(\ell^2, X), \|S\|_{\alpha} \le 1\right\}$
= $\sup\left\{\left|\operatorname{tr}(S^*T)\right| : S \in \mathcal{F}(\ell^2, X), \|S\|_{\alpha}\right\} = \|T\|_{\alpha^*}.$

Note that if for two Euclidean structures α and β on X we have $\alpha \lesssim \beta$, then $\beta^* \leq \alpha^*$ on X^* . Part of the reason why the γ - and the ℓ^2 -structure work well in practice, is the fact that they are self-dual under certain assumptions on X. This is contained in the following proposition, along with a few other relations between dual Euclidean structures.

PROPOSITION 1.1.4. On X^* we have

- (i) $\operatorname{op}^* = \nu$ and $\nu^* = \operatorname{op}$.
- (ii) $\gamma^* \leq \gamma$. Moreover $\gamma \lesssim \gamma^*$ if and only if X has nontrivial type.
- (iii) If X is a Banach lattice, $(\ell^2)^* = \ell^2$

PROOF. Fix $\mathbf{x}^* \in (X^*)^n$. For (i) let $\mathbf{y}^* \in (X^*)^m$ be such that $\mathbf{x}^* = \mathbf{A}\mathbf{y}^*$ with $\mathbf{A} \in M_{n,m}(\mathbb{C})$ and $\sum_{k=1}^n |A_{kj}|^2 \leq 1$ for $1 \leq j \leq m$. Then we have

$$\begin{aligned} \|\mathbf{x}^*\|_{\mathrm{op}^*} &= \sup\left\{\sum_{k=1}^n |x_k^*(x_k)| : \mathbf{x} \in X^n, \left\|\sum_{k=1}^n b_k x_k\right\|_X \le 1, \sum_{k=1}^n |b_k|^2 \le 1\right\} \\ &\le \sup\left\{\sum_{j=1}^m \sum_{k=1}^n |y_j^*(A_{kj} x_k)| : \mathbf{x} \in X^n, \left\|\sum_{k=1}^n b_k x_k\right\|_X \le 1, \sum_{k=1}^n |b_k|^2 \le 1\right\} \\ &\le \sum_{j=1}^m \|y_j^*\|_X, \end{aligned}$$

so taking the infimum over all such \mathbf{y} shows $\|\mathbf{x}^*\|_{\mathrm{op}^*} = \|\mathbf{x}^*\|_{\nu}$. This also implies that $\|\mathbf{x}^{**}\|_{\mathrm{op}^*} = \|\mathbf{x}^{**}\|_{\nu}$ for all $\mathbf{x}^{**} \in (X^{**})^n$. Dualizing and restricting to X^* we obtain that $\nu^* = \text{op on } X^*$.

For (ii) we have for a Gaussian sequence $(\gamma_k)_{k=1}^n$ by Hölder's inequality that

$$\|\mathbf{x}^*\|_{\gamma^*} = \sup\left\{\left|\mathbb{E}\sum_{k=1}^n \langle \gamma_k x_k, \gamma_k x_k^* \rangle\right| : \mathbf{x} \in X^n, \|\mathbf{x}\|_{\gamma} \le 1\right\} \le \|\mathbf{x}^*\|_{\gamma}.$$

The converse estimate defines the notion of Gaussian K-convexity of X, which is equivalent to K-convexity of X by [HNVW17, Corollary 7.4.20]. It is a deep result of Pisier [Pis82] that K-convexity is equivalent to nontrivial type, see [HNVW17, Theorem 7.4.15].

For (iii) we note that since $X(\ell_n^2)^* = X^*(\ell_n^2)$ by [LT79, Section 1.d], we have

$$\left\| \left(\sum_{k=1}^{n} |x_k^*|^2 \right)^{1/2} \right\|_X = \sup \left\{ \sum_{k=1}^{n} |x_k^*(x_k)| : \mathbf{x} \in X^n, \left\| \left(\sum_{k=1}^{n} |x_k|^2 \right)^{1/2} \right\|_X \le 1 \right\},$$

determined $\ell^2 = (\ell^2)^*.$

so indeed $\ell^2 = (\ell^2)^*$.

Using a duality argument we can compare the $\ell_n^2(X)$ -norm and the α -norm of a vector in a finite dimensional subspace of X^n .

PROPOSITION 1.1.5. Let E be a finite dimensional subspace of X. Then for $\mathbf{x} \in E^n$ we have

$$(\dim(E))^{-1} \left(\sum_{k=1}^{n} \|x_k\|_X^2\right)^{1/2} \le \|\mathbf{x}\|_{\alpha} \le \dim(E) \left(\sum_{k=1}^{n} \|x_k\|_X^2\right)^{1/2}$$

PROOF. For $\mathbf{x} \in E^n$ we have by Proposition 1.1.2 that

$$\|\mathbf{x}\|_{\alpha} \le \|\mathbf{x}\|_{\nu} \le \dim(E) \|\mathbf{x}\|_{\text{op}} \le \dim(E) \left(\sum_{k=1}^{n} \|x_k\|_X^2\right)^{1/2}$$

Conversely take $\mathbf{x}^* \in (E^*)^n$ with $\|\mathbf{x}^*\|_{\alpha^*} \leq 1$ and $\|\mathbf{x}\|_{\alpha} = \sum_{k=1}^n x_k^*(x_k)$. Then

$$\|\mathbf{x}^*\|_{\alpha^*} \le \dim(E) \left(\sum_{k=1}^n \|x_k^*\|_{X^*}^2\right)^{1/2}$$

and therefore

$$\left(\sum_{k=1}^{n} \|x_k\|_X^2\right)^{1/2} \le \dim(E) \|\mathbf{x}\|_{\alpha}.$$

Unconditionally stable Euclidean structures. We end this section with an additional property of a Euclidean structure that will play an important role in Chapter 4-6. We will say that a Euclidean structure α on X is unconditionally stable if there is a C > 0 such that

(1.6)
$$\|\mathbf{x}\|_{\alpha} \leq C \sup_{|\epsilon_k|=1} \left\|\sum_{k=1}^{n} \epsilon_k x_k\right\|_X, \qquad \mathbf{x} \in X^n$$

(1.7)
$$\|\mathbf{x}^*\|_{\alpha^*} \le C \sup_{|\epsilon_k|=1} \left\|\sum_{k=1}^n \epsilon_k x_k^*\right\|_{X^*} \qquad \mathbf{x}^* \in (X^*)^n.$$

The next proposition gives some examples of unconditionally stable structures.

Proposition 1.1.6.

- (i) The ℓ^{g} -structure on X is unconditionally stable.
- (ii) If X has finite cotype, the γ -structure on X is unconditionally stable.
- (iii) If X is a Banach lattice, the ℓ^2 -structure on X is unconditionally stable.

PROOF. Fix $\mathbf{x} \in X^n$ and $\mathbf{x}^* \in (X^*)^n$. For (i) let $V: X \to L^1(S)$ be a norm-one operator. Then by Proposition 1.0.1 we have

$$\|(Vx_1,\ldots,Vx_n)\|_{\ell^2} \lesssim \mathbb{E} \|\sum_{k=1}^n \varepsilon_k Vx_k\|_{L^1(S)} \leq \sup_{|\epsilon_k|=1} \|\sum_{k=1}^n \epsilon_k x_k\|_X,$$

where $(\varepsilon_k)_{k\geq 1}$ is a Rademacher sequence. So taking the supremum over all such V yields (1.6). Now suppose that

$$\sup_{|\epsilon_k|=1} \left\| \sum_{k=1}^n \epsilon_k x_k^* \right\|_{X^*} = 1.$$

Define $V : X \to \ell_n^1$ by $Vx = (x_1^*(x), \dots, x_n^*(x))$, for which we have $||V|| \leq 1$. Suppose that $||\mathbf{x}||_{\ell^g} \leq 1$. Then $||(Vx_1, \dots, Vx_n)||_{\ell^2} \leq 1$, i.e.

$$\sum_{j=1}^{n} \left(\sum_{k=1}^{n} |x_j^*(x_k)|^2 \right)^{1/2} \le 1$$

and hence

$$\sum_{j=1}^{n} |x_j^*(x_j)| \le 1.$$

This means that $\|\mathbf{x}\|_{(\ell^{g})^*} \leq 1$, so (1.7) follows. For (ii), we have by Proposition 1.0.1 that

$$\|\mathbf{x}\| < \left(\mathbb{E}\|\sum_{n=0}^{n} \varepsilon_{n, T_{n}}\|^{2}\right)^{1/2} < \sup_{n=0}^{n} \|\sum_{n=0}^{n} \varepsilon_{n, T_{n}}\|^{2}$$

$$\|\mathbf{x}\|_{\gamma} \lesssim \left(\mathbb{E}\left\|\sum_{k=1}^{\infty} \varepsilon_{k} x_{k}\right\|_{X}^{2}\right) \leq \sup_{|\epsilon_{k}|=1} \left\|\sum_{k=1}^{\infty} \epsilon_{k} x_{k}\right\|_{X},$$

where $(\varepsilon_k)_{k\geq 1}$ is a Rademacher sequence. For (1.7) assume that $\|\mathbf{x}\|_{\gamma} \leq 1$. Then again by Proposition 1.0.1 we have

$$\begin{split} \left|\sum_{k=1}^{n} \langle x_k, x_k^* \rangle \right| &= \left| \mathbb{E} \langle \sum_{k=1}^{n} \varepsilon_k x_k, \sum_{k=1}^{n} \varepsilon_k x_k^* \rangle \right| \\ &\leq \left(\mathbb{E} \left\| \sum_{k=1}^{n} \varepsilon_k x_k \right\|_X^2 \right)^{1/2} \left(\mathbb{E} \left\| \sum_{k=1}^{n} \varepsilon_k x_k^* \right\|_{X^*}^2 \right)^{1/2} \\ &\lesssim \sup_{|\epsilon_k|=1} \left\| \sum_{k=1}^{n} \epsilon_k x_k^* \right\|_{X^*}. \end{split}$$

Finally (iii) follows from (i) and the equivalence of the ℓ^2 -structure and the ℓ^g -structure, see Proposition 1.1.3.

1.2. α -bounded operator families

Having introduced Euclidean structures in the preceding section, we will now connect Euclidean structures to operator theory.

DEFINITION 1.2.1. Let α be a Euclidean structure on X. A family of operators $\Gamma \subseteq \mathcal{L}(X)$ is called α -bounded if

$$\|\Gamma\|_{\alpha} := \sup \{ \|(T_1x_1, \dots, T_nx_n)\|_{\alpha} : T_k \in \Gamma, \mathbf{x} \in X^n, \|\mathbf{x}\|_{\alpha} \le 1 \}$$

is finite. If α is a global Euclidean structure, this definition can analogously be given for $\Gamma \subseteq \mathcal{L}(X, Y)$, where Y is another Banach space.

We allow repetitions of the operators in the definition of α -boundedness. In the case that $\alpha = \ell^2$ it is known that it is equivalent to test the definition only for distinct operators, see [**KVW16**, Lemma 4.3]. For $\alpha = \gamma$ this is an open problem.

Closely related to γ and ℓ^2 -boundedness is the notion of \mathcal{R} -boundedness. We say that $\Gamma \subseteq \mathcal{L}(X)$ is \mathcal{R} -bounded if there is a C > 0 such that for all $\mathbf{x} \in X^n$

$$\left(\mathbb{E}\left\|\sum_{k=1}^{n}\varepsilon_{k}T_{k}x_{k}\right\|^{2}\right)^{1/2} \leq C\left(\mathbb{E}\left\|\sum_{k=1}^{n}\varepsilon_{k}x_{k}\right\|^{2}\right)^{1/2}, \qquad T_{k} \in \Gamma,$$

where $(\varepsilon_k)_{k=1}^{\infty}$ is a Rademacher sequence. Note that the involved \mathcal{R} -norms do not form a Euclidean structure, as they do not satisfy (1.2). However, we have the following connections (see [**KVW16**]):

- \mathcal{R} -boundedness implies γ -boundedness. Moreover γ -boundedness and \mathcal{R} -boundedness are equivalent on X if and only if X has finite cotype.
- ℓ^2 -boundedness, γ -boundedness and \mathcal{R} -boundedness are equivalent on a Banach lattice X if and only if X has finite cotype.

Following the breakthrough papers [**CPSW00**, **Wei01b**], γ - and ℓ^2 - and \mathcal{R} -boundedness have played a major role in the development of vector-valued analysis over the past decades (see e.g. [**HNVW17**, Chapter 8]).

We call an operator $T \in \mathcal{L}(X)$ α -bounded if $\{T\}$ is α -bounded. It is not always the case that any $T \in \mathcal{L}(X)$ is α -bounded. In fact we have the following characterization:

PROPOSITION 1.2.2. Let α be a Euclidean structure on X. Then every $T \in \mathcal{L}(X)$ is α -bounded with $||\{T\}||_{\alpha} \leq C ||T||$ if and only if α is ideal with constant C.

PROOF. First assume that α is ideal with constant C. Then we have for all $T \in \mathcal{L}(X)$

$$\|\{T\}\|_{\alpha} = \sup\{\|(Tx_1, \dots Tx_n)\|_{\alpha} : \mathbf{x} \in X^n, \|\mathbf{x}\|_{\alpha} \le 1\} \le C \|T\|,\$$

where C is the ideal constant of α . Now suppose that for all $T \in \mathcal{L}(X)$ we have $\|\{T\}\|_{\alpha} \leq C \|T\|$. Then for $\mathbf{x} \in X^n$ and $T \in \mathcal{L}(X)$ we have

$$||(Tx_1, \dots Tx_n)||_{\alpha} \le ||\{T\}||_{\alpha} ||\mathbf{x}||_{\alpha} \le C ||T|| ||\mathbf{x}||_{\alpha},$$

so α is ideal with constant C.

Next we establish some basic properties of α -bounded families of operators.

PROPOSITION 1.2.3. Let α be a Euclidean structure on X and let $\Gamma, \Gamma' \subseteq \mathcal{L}(X)$ be α -bounded.

- (i) For $\Gamma'' = \{TT' : T \in \Gamma, T' \in \Gamma'\}$ we have $\|\Gamma''\|_{\alpha} \le \|\Gamma\|_{\alpha} \|\Gamma'\|_{\alpha}$.
- (ii) For $\Gamma^* = \{T^* : T \in \Gamma\}$ we have $\|\Gamma^*\|_{\alpha^*} = \|\Gamma\|_{\alpha}$.
- (iii) For α -bounded $\Gamma_k \subseteq \mathcal{L}(X)$ for $k \in \mathbb{N}$ we have $\|\bigcup_{k=1}^{\infty} \Gamma_k\|_{\alpha} \leq \sum_{k=1}^{\infty} \|\Gamma_k\|_{\alpha}$.
- (iv) For the absolutely convex hull $\tilde{\Gamma}$ of Γ we have $\|\Gamma\|_{\alpha} = \|\Gamma\|_{\alpha}$.
- (v) For the closure of Γ in the strong operator topology $\tilde{\Gamma}$ we have $\|\tilde{\Gamma}\|_{\alpha} = \|\Gamma\|_{\alpha}$.

PROOF. (i) is immediate from the definition, (ii) is a consequence of our definition of duality, (iii) follows from the triangle inequality and (v) is clear from the definition of an α -bounded family of operators.

For (iv) we first note that $\|\bigcup_{0\leq\theta\leq 2\pi}e^{i\theta}\Gamma\|_{\alpha} = \|\Gamma\|_{\alpha}$. It remains to check that $\|\operatorname{conv}(\Gamma)\|_{\alpha} = \|\Gamma\|_{\alpha}$. Suppose that for $1\leq j\leq m$ we have $S_j = \sum_{k=1}^n a_{jk}T_k$ where $T_1,\ldots,T_n\in\Gamma, a_{jk}\geq 0$ and $\sum_{k=1}^n a_{jk}=1$ for $1\leq j\leq m$. Let $(\xi_j)_{j=1}^m$ be a sequence of independent random variables with $\mathbb{P}(\xi_j=k)=a_{jk}$ for $1\leq k\leq n$. Then

$$\|(S_1x_1,\ldots,S_nx_n)\|_{\alpha} = \|\mathbb{E}(T_{\xi_1}x_1,\ldots,T_{\xi_n}x_n)\|_{\alpha}$$
$$\leq \mathbb{E}\|(T_{\xi_1}x_1,\ldots,T_{\xi_n}x_n)\|_{\alpha}$$
$$\leq \|\Gamma\|_{\alpha}$$

for all $\mathbf{x} \in X^n$ with $\|\mathbf{x}\|_{\alpha} \leq 1$, so $\|\operatorname{conv}(\Gamma)\|_{\alpha} = \|\Gamma\|_{\alpha}$.

As a corollary of Proposition 1.2.3(iv) and (v) we also have the α -boundedness of L^1 -integral means of α -bounded sets. Moreover from the triangle inequality for $\|\cdot\|_{\alpha}$ we obtain boundedness of L^{∞} -integral means. If $\alpha = \gamma$, there is a scale of results under type and cotype assumptions (see [**HV09**]).

COROLLARY 1.2.4. Let α be a Euclidean structure on a Banach space X, let (S, μ) be a measure space and let $f : S \to \Gamma$ be strongly measurable.

(i) If
$$\Gamma := \{f(s) : s \in S\}$$
 is α bounded, then the set

$$\Gamma_f^1 := \left\{ \int_S \varphi(s) f(s) \, \mathrm{d}s : \|\varphi\|_{L^1(S)} \le 1 \right\}$$

is α -bounded with $\|\Gamma_f^1\|_{\alpha} \leq \|\Gamma\|$.

(ii) If $\int_{S} ||f|| d\mu < \infty$ and α is ideal, then the set

$$\Gamma_f^\infty := \left\{ \int_S \varphi(s) f(s) \, \mathrm{d}s : \|\varphi\|_{L^\infty(S)} \le 1 \right\}$$

is α -bounded with $\|\Gamma_f^{\infty}\|_{\alpha} \lesssim \int_S \|f\| d\mu$.

A technical lemma. We end this section with a technical lemma that will be crucial in the representation theorems in this chapter, as well as in the more concrete factorization theorems in Chapter 2. The proof of this lemma (in the case $\Gamma = \emptyset$) is a variation of the proof of [AK16, Theorem 7.3.4], which is the key ingredient to prove the Maurey-Kwapień theorem on factorization of an operator $T: X \to Y$ through a Hilbert space (see [Kwa72, Mau74]). We will make the connection to the Maurey-Kwapień factorization theorem clear in Section 2.1, where we will prove a generalization of that theorem.

LEMMA 1.2.5. Let α be a Euclidean structure on X and let $Y \subseteq X$ be a subspace. Suppose that $F: X \to [0, \infty)$ and $G: Y \to [0, \infty)$ are two positive homogeneous functions such that

(1.8)
$$\left(\sum_{k=1}^{n} F(x_k)^2\right)^{\frac{1}{2}} \le \|\mathbf{x}\|_{\alpha}, \qquad \mathbf{x} \in X^n,$$

(1.9)
$$\|\mathbf{y}\|_{\alpha} \le \left(\sum_{k=1}^{n} G(y_k)^2\right)^{\frac{1}{2}}, \qquad \mathbf{y} \in Y^n$$

Let $\Gamma \subseteq \mathcal{L}(X)$ and be α -bounded family of operators. Then there exists a Γ -invariant subspace $Y \subseteq X_0 \subseteq X$ and a Hilbertian seminorm $\|\cdot\|_0$ on X_0 such that

(1.10) $||Tx||_0 \le 2 ||\Gamma||_{\alpha} ||x||_0 \qquad x \in X_0, T \in \Gamma,$

(1.11)
$$||x||_0 \ge F(x)$$
 $x \in X_0,$

(1.12)
$$||x||_0 \le 4G(x)$$
 $x \in Y.$

PROOF. Let X_0 be the smallest Γ -invariant subspace of X containing Y, i.e. set $Y_0 := Y$, define for $N \ge 1$

$$Y_N := \Big\{ Tx : T \in \Gamma, x \in Y_{N-1} \Big\}.$$

and take $X_0 := \bigcup_{N>0} Y_N$. We will prove the lemma in three steps.

Step 1: We will first show that G can be extended to a function G_0 on X_0 , such that $2G_0$ satisfies (1.9) for all $\mathbf{y} \in X_0^n$. For this pick a sequence of real numbers $(a_N)_{N=1}^{\infty}$ such that $a_N > 1$ and $\prod_{N=1}^{\infty} a_N = 2$ and define $b_M := \prod_{N=1}^M a_N$. For $y \in Y$ we set $G_0(y) = G(y)$ and proceed by induction. Suppose that G_0 is defined on $\bigcup_{N=0}^M Y_N$ for some $M \in \mathbb{N}$ with

(1.13)
$$\|\mathbf{y}\|_{\alpha} \le b_M \Big(\sum_{k=1}^n G_0(y_k)^2\Big)^{1/2}$$

for any $\mathbf{y} \in \left(\bigcup_{N=0}^{M} Y_N\right)^n$.

For $y \in Y_{M+1} \setminus \bigcup_{N=0}^{M} Y_N$ pick a $T \in \Gamma$ and an $x \in Y_M$ such that Tx = y and

define

$$G_0(y) := \frac{1}{a_{M+1} - 1} \|\Gamma\|_{\alpha} \cdot G_0(x).$$

For $\mathbf{y} \in \left(\bigcup_{N=0}^{M+1} Y_N\right)^n$ we let $\mathcal{I} = \{k : y_k \in \bigcup_{N=0}^M Y_N\}$. For $k \notin \mathcal{I}$ we let $T_k \in \Gamma$ and $x_k \in \bigcup_{N=0}^M Y_N$ be such that $T_k x_k = y_k$. Then by our definition of G_0 we have

$$\begin{aligned} \|\mathbf{y}\|_{\alpha} &\leq \left\| (\mathbf{1}_{k\in\mathcal{I}} y_{k})_{k=1}^{n} \right\|_{\alpha} + \left\| (\mathbf{1}_{k\notin\mathcal{I}} y_{k})_{k=1}^{n} \right\|_{\alpha} \\ &\leq b_{M} \left(\sum_{k\in\mathcal{I}} G_{0}(y_{k})^{2} \right)^{1/2} + b_{M} \|\Gamma\|_{\alpha} \left(\sum_{k\notin\mathcal{I}} G_{0}(x_{k})^{2} \right)^{1/2} \\ &\leq b_{M} \left(\sum_{k=1}^{n} G_{0}(y_{k})^{2} \right)^{1/2} + b_{M} (a_{M+1} - 1) \left(\sum_{k=1}^{n} G_{0}(y_{k})^{2} \right)^{1/2} \\ &= b_{M+1} \left(\sum_{k=1}^{n} G_{0}(y_{k})^{2} \right)^{1/2} \end{aligned}$$

So G_0 satisfies (1.13) for M + 1. Therefore, by induction we can define G_0 on X_0 , such that $2G_0$ satisfies (1.9) for all $\mathbf{y} \in X_0^n$.

Step 2: For $x \in X$ define the function $\phi_x : X^* \to \mathbb{R}_+$ by $\phi_x(x^*) := |x^*(x)|^2$. We will construct a sublinear functional on the space

$$\mathbb{V} := \operatorname{span}\{\phi_x : x \in X_0\}.$$

For this note that every $\psi \in \mathbb{V}$ has a representation of the form

(1.14)
$$\psi = \sum_{k=1}^{n_u} \phi_{u_k} - \sum_{k=1}^{n_v} \phi_{v_k} + \sum_{k=1}^{n_x} (\phi_{T_k x_k} - \phi_{2 \|\Gamma\|_{\alpha} x_k})$$

with $u_k \in X_0, v_k, x_k \in X$ and $T_k \in \Gamma$. Define $p : \mathbb{V} \to [-\infty, \infty)$ by

$$p(\psi) = \inf \left\{ 16 \sum_{k=1}^{n_u} G_0(u_k)^2 - \sum_{k=1}^{n_v} F(v_k)^2 \right\},\$$

where the infimum is taken over all representations of ψ in the form of (1.14). This functional clearly has the following properties

(1.15)
$$p(a\psi) = ap(\psi), \qquad \qquad \psi \in \mathbb{V}, a > 0,$$

(1.16)
$$p(\psi_1 + \psi_2) \le p(\psi_1) + p(\psi_2), \qquad \psi_1, \psi_2 \in \mathbb{V},$$

(1.17) $p(\phi_{Tx} - \phi_{2||\Gamma||_{-x}}) \le 0, \qquad x \in X_0, T \in \Gamma,$

(1.17)
$$p(\phi_{Tx} - \phi_{2\|\Gamma\|_{\alpha}x}) \le 0,$$

(1.18)
$$p(-\phi_x) \le -F(x)^2, \qquad x \in X_0,$$

(1.19)
$$p(\phi_x) \le 16G_0(x)^2, \qquad x \in X_0.$$

We will check that p(0) = 0. It is clear that $p(0) \le 0$. Let

$$0 = \sum_{k=1}^{n_u} \phi_{u_k} - \sum_{k=1}^{n_v} \phi_{v_k} + \sum_{k=1}^{n_x} (\phi_{T_k x_k} - \phi_{2 \|\Gamma\|_{\alpha} x_k})$$

be a representation of the form of (1.14). So for any $x^* \in X^*$ we have

(1.20)
$$\sum_{k=1}^{n_u} |x^*(u_k)|^2 + \sum_{k=1}^{n_x} |x^*(T_k x_k)|^2 = \sum_{k=1}^{n_v} |x^*(v_k)|^2 + \sum_{k=1}^{n_x} |x^*(2\|\Gamma\|_{\alpha} x_k)|^2.$$

Let

$$\mathbf{u} := (u_k)_{k=1}^{n_u} \in X_0^{n_u}, \qquad \mathbf{v} := (v_k)_{k=1}^{n_v} \in X^{n_v}, \\ \mathbf{x} := (x_k)_{k=1}^{n_x} \in X^{n_x}, \qquad \mathbf{y} := (T_k x_k)_{k=1}^{n_x} \in X^n$$

and define the vectors

$$\mathbf{\bar{u}} = \begin{pmatrix} \mathbf{u} \\ \mathbf{y} \end{pmatrix}, \quad \mathbf{\bar{v}} = \begin{pmatrix} \mathbf{v} \\ 2 \|\Gamma\|_{\alpha} \mathbf{x} \end{pmatrix}.$$

Note that (1.20) implies, by the Hahn-Banach theorem, that

$$v_1, \cdots, v_{n_v}, x_1, \cdots, x_{n_x} \in \operatorname{span}\{u_1, \cdots, u_{n_u}, T_1x_1, \cdots, T_{n_x}x_{n_x}\}.$$

Therefore there exists a matrix \mathbf{A} with $\|\mathbf{A}\| = 1$ such that $\mathbf{\bar{v}} = \mathbf{A}\mathbf{\bar{u}}$. Thus by property (1.2) of a Euclidean structure we get that $\|\mathbf{\bar{v}}\|_{\alpha} \leq \|\mathbf{\bar{u}}\|_{\alpha}$. Now we have, using the triangle inequality, that

$$\|\bar{\mathbf{v}}\|_{\alpha} \leq \|\bar{\mathbf{u}}\|_{\alpha} \leq \|\mathbf{u}\|_{\alpha} + \frac{1}{2} \|2\|\Gamma\|_{\alpha} \mathbf{x}\|_{\alpha} \leq \|\mathbf{u}\|_{\alpha} + \frac{1}{2} \|\bar{\mathbf{v}}\|_{\alpha},$$

which means that

$$\|\mathbf{v}\|_{\alpha} \le \|\mathbf{\bar{v}}\|_{\alpha} \le 2\|\mathbf{u}\|_{\alpha}$$

By assumption (1.8) on F and (1.9) on $2G_0$ we have

$$\sum_{k=1}^{n_v} F(v_k)^2 \le \|\mathbf{v}\|_{\alpha}^2 \le 4\|\mathbf{u}\|_{\alpha}^2 \le 16 \sum_{k=1}^{n_u} G_0(u_k)^2,$$

which means that $p(0) \ge 0$ and thus p(0) = 0. Now with property (1.16) of p we have

$$p(\psi) + p(-\psi) \ge p(0) = 0$$

so $p(\psi) > -\infty$ for all $\psi \in \mathbb{V}$. Combined with properties (1.15) and (1.16) this means that p is a sublinear functional.

Step 3. To complete the prove of the lemma, we construct a semi-inner product from our sublinear functional p using Hahn–Banach. Indeed, by applying the Hahn-Banach theorem [**Rud91**, Theorem 3.2], we obtain a linear function f on \mathbb{V} such that $f(\psi) \leq p(\psi)$ for all $\psi \in \mathbb{V}$. By property (1.18) we know that $p(-\phi_x) \leq 0$ and thus $f(\phi_x) \geq 0$ for all $x \in X_0$. We take the complexification of \mathbb{V}

$$\mathbb{V}^{\mathbb{C}} = \{ v_1 + iv_2 : v_1, v_2 \in \mathbb{V} \}$$

with addition and scalar multiplication defined as usual. We extend f to a complex linear functional on this space by $f(v_1 + iv_2) = f(v_1) + if(v_2)$ and define a pseudoinner product on X_0 by $\langle x, y \rangle = f(\rho_{x,y})$ with $\rho_{x,y} : X^* \to \mathbb{C}$ defined as $\rho_{x,y}(x^*) = x^*(x)\overline{x^*(y)}$ for all $x^* \in X^*$. This is well-defined since

$$\rho_{x,y} = \frac{1}{4}(\phi_{x+y} - \phi_{x-y} + i\phi_{x+iy} - i\phi_{x-iy}) \in \mathbb{V}^{\mathbb{C}}.$$

On X_0 we define $\|\cdot\|_0$ as the seminorm induced by this semi-inner product, i.e. $\|x\|_0 := \sqrt{\langle x, x \rangle} = \sqrt{f(\phi_x)}$. Then for $x \in X_0$ and $T \in \Gamma$ we have by property (1.17) of p

$$||Tx||_0^2 \le p(\phi_{Tx} - \phi_{2||\Gamma||_{\alpha}x}) + f(\phi_{2||\Gamma||_{\alpha}x}) \le 4||\Gamma||_{\alpha}^2 ||x||_0^2.$$

By property (1.18) of p we have

$$||x||_0^2 = f(\phi_x) \ge -p(-\phi_x) \ge F(x)^2, \qquad x \in X_0,$$

and by property (1.19) of p we have

$$||y||_0^2 = f(\phi_y) \le p(\phi_y) \le 16G_0(y)^2 = 16G(y)^2, \quad y \in Y.$$

So $\|\cdot\|_0$ satisfies (1.10)-(1.12).

1.3. The representation of α -bounded operator families on a Hilbert space

We will now represent an α -bounded family of operators Γ as a corresponding family of operators $\widetilde{\Gamma} \subseteq \mathcal{L}(H)$ for some Hilbert space H. As a preparation we record an important special case of Lemma 1.2.5.

LEMMA 1.3.1. Let α be a Euclidean structure on X and let $\Gamma \subseteq \mathcal{L}(X)$ be α bounded. Then for any $\eta = (y_0, y_1) \in X \times X$ there exists a Γ -invariant subspace $X_\eta \subseteq X$ with $y_0 \in X_\eta$ and a Hilbertian seminorm $\|\cdot\|_\eta$ on X_η such that

(1.21)
$$||Tx||_{\eta} \le 2||\Gamma||_{\alpha}||x||_{\eta} \qquad x \in X_{\eta}, T \in \Gamma,.$$

$$(1.22) ||y_0||_{\eta} \le 4||y_0||_X$$

 $(1.23) ||y_1||_{\eta} \ge ||y_1||_X$

J

PROOF. Define $F_{\eta}: X \to [0, \infty)$ as

$$F_{\eta}(x) = \begin{cases} \|x\|_X & \text{if } x \in \text{span} \{y_1\}, \\ 0 & \text{otherwise,} \end{cases}$$

if $y_1 \in X_n$

let $Y = \operatorname{span}\{y_0\}$ and define $G_\eta : Y \to [0, \infty)$ as $G_\eta(x) = ||x||_X$. Then F_η and G_η satisfy (1.8) and (1.9) by Proposition 1.1.5, so by Lemma 1.2.5 we can find a Γ -invariant subspace X_η of X containing y_0 and a seminorm $|| \cdot ||_\eta$ on X_η induced by a semi-inner product for which (1.10)-(1.12) hold, from which (1.21)-(1.23) directly follow.

With Lemma 1.3.1 we can now represent a α -bounded family of Banach space operators on a Hilbert space. Note that by Proposition 1.2.3 we know that without loss of generality we can restrict to families of operators that are absolutely convex and closed in the strong operator topology.

THEOREM 1.3.2. Let α be a Euclidean structure on X and let $\Gamma \subseteq \mathcal{L}(X)$ be absolutely convex, closed in the strong operator topology and α -bounded. Define $\|T\|_{\Gamma} = \inf \{\lambda > 0 : \lambda^{-1}T \in \Gamma\}$ on the linear span of Γ denoted by $\mathcal{L}_{\Gamma}(X)$. Then there is a Hilbert space H, a closed subalgebra \mathcal{B} of $\mathcal{L}(H)$, a bounded algebra homomorphism $\rho \colon \mathcal{B} \to \mathcal{L}(X)$ and a bounded linear operator $\tau \colon \mathcal{L}_{\Gamma}(X) \to \mathcal{B}$ such

that

$$\rho\tau(T) = T, \qquad T \in \mathcal{L}_{\Gamma}(X),$$
$$\|\rho\| \le 4,$$
$$\|\tau\| \le 2\|\Gamma\|_{\alpha}.$$

Furthermore, if \mathcal{A} is the algebra generated by Γ , τ extends to an algebra homomorphism of \mathcal{A} into \mathcal{B} such that $\rho\tau(S) = S$ for all $S \in \mathcal{A}$.

PROOF. Let \mathcal{A} be the algebra generated by Γ . For any $\eta \in X \times X$ we let $(X_{\eta}, \|\cdot\|_{\eta})$ be as in Lemma 1.3.1 and take $N_{\eta} = \{x \in X_{\eta} : \|x\|_{\eta} = 0\}$. Let H_{η} be the completion of the quotient space X_{η}/N_{η} , which is a Hilbert space. Let $\pi_{\eta} : \mathcal{A} \to \mathcal{L}(H_{\eta})$ be the algebra homomorphism mapping elements of \mathcal{A} to their representation on H_{η} , which is well-defined since X_{η} is \mathcal{A} invariant.

Define $E = \{(x, Sx) : x \in X, S \in \mathcal{A}\} \subseteq X \times X$. We define the Hilbert space H by the direct sum $H = \bigoplus_{\eta \in E} H_{\eta}$ with norm $\|\cdot\|_{H}$ given by

$$||h||_{H} = \left(\sum_{\eta \in E} ||h_{\eta}||_{\eta}^{2}\right)^{\frac{1}{2}}$$

for $h \in H$ with $h = (h_\eta)_{\eta \in E}$. Furthermore we define the algebra homomorphism $\tau : \mathcal{A} \to \mathcal{L}(H)$ by $\tau = \bigoplus_{\eta \in E} \pi_{\eta}$.

For all $T \in \mathcal{L}_{\Gamma}(X)$ we then have

$$\begin{aligned} |\tau(T)|| &= ||T||_{\Gamma} \sup \left\{ \left(\sum_{\eta \in E} \left\| \pi_{\eta} \left(\frac{T}{||T||_{\Gamma}} \right) (x_{\eta}) \right\|_{\eta}^{2} \right)^{\frac{1}{2}} : \left\| (x_{\eta})_{\eta \in E} \right\| \le 1 \right\} \\ &\le 2 ||\Gamma||_{\alpha} ||T||_{\Gamma}. \end{aligned}$$

Therefore the restriction $\tau|_{\Gamma} : \mathcal{L}_{\Gamma}(X) \to \mathcal{L}(H)$ is a bounded linear operator with $\|\tau|_{\Gamma}\| \leq 2\|\Gamma\|_{\alpha}$. Now for $S \in \mathcal{A}$ and $x \in X$ with $\|x\|_X \leq 1$ define $\zeta = (x, Sx)$. We have

$$\|\tau(S)\| \ge \sup_{\eta \in E} \|\pi_{\eta}(S)\| \ge \|\pi_{\zeta}(S)(x)\|_{\zeta} \|x\|_{\zeta}^{-1} \ge \|Sx\|_{X} \cdot (4\|x\|_{X})^{-1}$$

using (1.22) and (1.23). So $\|\tau(S)\| \ge \frac{1}{4}\|S\|$, which means that τ is injective. If we now define \mathcal{B} as the closure of $\tau(\mathcal{A})$ in $\mathcal{L}(H)$, we can extend $\rho = \tau^{-1}$ to an algebra homomorphism $\rho: \mathcal{B} \to \mathcal{L}(X)$ with $\|\rho\| \le 4$ since $\|\tau\| \ge \frac{1}{4}$. This proves the theorem. \Box

1.4. The equivalence of α -boundedness and C^* -boundedness

There is also a converse to Theorem 1.3.2, for which we will have to make a detour into operator theory. We will introduce matricial algebra norms in order to connect α -boundedness of a family of operators to the theory of completely bounded maps. For background on the theory developed in this section we refer to **[BL04, ER00, Pau02, Pis03]**.

Matricial algebra norms. Denote the space of $m \times n$ -matrices with entries in a complex algebra \mathcal{A} by $M_{m,n}(\mathcal{A})$. A matricial algebra norm on \mathcal{A} is a norm $\|\cdot\|_{\mathcal{A}}$ defined on each $M_{m,n}(\mathcal{A})$ such that

$$\|\mathbf{ST}\|_{\mathcal{A}} \leq \|\mathbf{S}\|_{\mathcal{A}} \|\mathbf{T}\|_{\mathcal{A}}, \qquad \mathbf{S} \in M_{m,k}(\mathcal{A}), \mathbf{T} \in M_{k,n}(\mathcal{A}) \\ \|\mathbf{ATB}\|_{\mathcal{A}} \leq \|\mathbf{A}\| \|\mathbf{T}\|_{\mathcal{A}} \|\mathbf{B}\|, \qquad \mathbf{A} \in M_{m,j}(\mathbb{C}), \mathbf{T} \in M_{j,k}(\mathcal{A}), \mathbf{B} \in M_{k,n}(\mathbb{C}).$$

The algebra \mathcal{A} with an associated matricial algebra norm will be called a *matricial* normed algebra. In the case that $\mathcal{A} \subseteq \mathcal{L}(X)$, we call a matricial algebra norm coherent if the norm of a 1×1 -matrix is the operator norm of its entry, i.e. if $\|(T)\|_{\mathcal{A}} = \|T\|$ for all $T \in \mathcal{A}$.

The following example shows that any Euclidean structure induces a matricial algebra norm on $\mathcal{L}(X)$.

EXAMPLE 1.4.1. Let α be a Euclidean structure on X. For $T \in M_{m,n}(\mathcal{L}(X))$ we define

$$\|\mathbf{T}\|_{\hat{\alpha}} = \sup\{\|\mathbf{T}\mathbf{x}\|_{\alpha} : \mathbf{x} \in X^n, \|\mathbf{x}\|_{\alpha} \le 1\}.$$

Then $\|\cdot\|_{\hat{\alpha}}$ is a coherent matricial algebra norm on $\mathcal{L}(X)$.

PROOF. Take $\mathbf{S} \in M_{m,k}(\mathcal{L}(X))$ and $\mathbf{T} \in M_{k,n}(\mathcal{L}(X))$. We have

$$\|\mathbf{ST}\|_{\hat{\alpha}} = \sup\left\{\frac{\|\mathbf{Sy}\|_{\alpha}}{\|\mathbf{y}\|_{\alpha}}\frac{\|\mathbf{y}\|_{\alpha}}{\|\mathbf{x}\|_{\alpha}} : \mathbf{x} \in X^n, \mathbf{y} = \mathbf{Tx}\right\} \le \|\mathbf{S}\|_{\hat{\alpha}}\|\mathbf{T}\|_{\hat{\alpha}}.$$

Moreover for any $\mathbf{A} \in M_{m,j}(\mathbb{C})$, $\mathbf{T} \in M_{j,k}(\mathcal{L}(X))$ and $\mathbf{B} \in M_{k,n}(\mathbb{C})$ we have by property (1.2) of the Euclidean structure that

$$\|\mathbf{ATB}\|_{\hat{\alpha}} \leq \sup\{\|\mathbf{A}\|\|\mathbf{TBx}\|_{\alpha} : \mathbf{x} \in X^n, \|\mathbf{Bx}\|_{\alpha} \leq \|\mathbf{B}\|\} \leq \|\mathbf{A}\|\|\mathbf{T}\|_{\hat{\alpha}}\|\mathbf{B}\|,$$

so $\|\cdot\|_{\hat{\alpha}}$ is a matricial algebra norm. Its coherence follows from

$$||(T)||_{\hat{\alpha}} = \sup\{||Tx|| : x \in X, ||x|| \le 1\} = ||T||$$

for $T \in \mathcal{L}(X)$, where we used property (1.1) of the Euclidean structure.

Using this induced matricial algebra norm we can reformulate α -boundedness. Indeed, for a family of operators $\Gamma \subseteq \mathcal{L}(X)$ we have

(1.24)
$$\|\Gamma\|_{\alpha} = \sup\{\|\mathbf{T}\|_{\hat{\alpha}} : \mathbf{T} = \operatorname{diag}(T_1, \dots, T_n), \quad T_1, \dots, T_n \in \Gamma\}.$$

This reformulation allows us to characterize those Banach spaces on which α boundedness is equivalent to uniform boundedness, using a result of Blecher, Ruan and Sinclair [**BRS90**].

PROPOSITION 1.4.2. Let α be a Euclidean structure on X such that for any family of operators $\Gamma \subseteq \mathcal{L}(X)$ we have

$$\|\Gamma\|_{\alpha} = \sup_{T \in \Gamma} \|T\|.$$

Then X is isomorphic to a Hilbert space.

PROOF. Take $T_1, \ldots, T_n \in \mathcal{L}(X)$ and let $\Gamma = \{T_k : 1 \le k \le n\}$. We have

$$\sup_{1 \le k \le n} \|T_k\| \le \|\operatorname{diag}(T_1, \dots, T_n)\|_{\hat{\alpha}} \le \|\Gamma\|_{\alpha} = \sup_{1 \le k \le n} \|T_k\|$$

which implies by $[\mathbf{BRS90}]$ that $\mathcal{L}(X)$ is isomorphic to an operator algebra and that therefore X is isomorphic to a Hilbert space by $[\mathbf{Eid40}]$.

 C^* -boundedness. Now let us turn to the converse of Theorem 1.3.2, for which we need to reformulate its conclusion. By an *operator algebra* \mathcal{A} we shall mean a closed unital subalgebra of a C^* -algebra. By the Gelfand-Naimark theorem we may assume without loss of generality that \mathcal{A} consists of bounded linear operators on a Hilbert space H. We say that $\Gamma \subseteq \mathcal{L}(X)$ is C^* -bounded if there exists a C > 0, an operator algebra \mathcal{A} and a bounded algebra homomorphism $\rho : \mathcal{A} \to \mathcal{L}(X)$ such that

$$\Gamma \subseteq \left\{ \rho(T) : T \in \mathcal{A}, \|T\| \le \frac{C}{\|\rho\|} \right\}.$$

The least admissible C is denoted by $\|\Gamma\|_{C^*}$.

From Theorem 1.3.2 we can directly deduce that any α -bounded family of operators is C^* -bounded. We will show that any C^* -bounded family of operators is α -bounded for some Euclidean structure α . As a first step we will prove a converse to Example 1.4.1, i.e. we will show that a matricial algebra norm on a subalgebra of $\mathcal{L}(X)$ gives rise to a Euclidean structure.

PROPOSITION 1.4.3. Let \mathcal{A} be a subalgebra of $\mathcal{L}(X)$ and let $\|\cdot\|_{\mathcal{A}}$ be a matricial algebra norm on \mathcal{A} such that $\|(T)\|_{\mathcal{A}} \geq \|T\|$ for all $T \in \mathcal{A}$. Then there is a Euclidean structure α on X such that $\|\mathbf{T}\|_{\hat{\alpha}} \leq \|\mathbf{T}\|_{\mathcal{A}}$ for all $\mathbf{T} \in M_{m,n}(\mathcal{A})$.

PROOF. Define the α -norm of a column vector $\mathbf{x} \in X^n$ by

$$\|\mathbf{x}\|_{\alpha} = \max\{\|\mathbf{x}\|_{\beta}, \|\mathbf{x}\|_{\mathrm{op}}\}\$$

with

$$\|\mathbf{x}\|_{\beta} = \sup\{\|\mathbf{S}\mathbf{x}\| : \mathbf{S} \in M_{1,n}(\mathcal{A}), \|\mathbf{S}\|_{\mathcal{A}} \le 1\}.$$

Then $\|\cdot\|_{\alpha}$ is a Euclidean structure, since we already know op is a Euclidean structure and for β we have

$$(x)\|_{\beta} \le \sup\{\|Sx\|_X : S \in \mathcal{A}, \|S\| \le 1\} = \|x\|_X$$

for any $x \in X$, so (1.1) holds. Moreover, if $\mathbf{A} \in M_{m,n}(\mathbb{C})$ and $\mathbf{x} \in X^n$, we have

$$\|\mathbf{A}\mathbf{x}\|_{\beta} = \sup\{\|\mathbf{S}\mathbf{A}\mathbf{x}\| : \mathbf{S} \in M_{1,m}(\mathcal{A}), \|\mathbf{S}\|_{\mathcal{A}} \le 1\} \le \|\mathbf{A}\| \|\mathbf{x}\|_{\beta}$$

so β satisfies (1.2).

Now suppose that $\mathbf{T} \in M_{m,n}(\mathcal{A}), \mathbf{x} \in X^n$ with $\|\mathbf{x}\|_{\alpha} \leq 1$ and $\mathbf{y} = \mathbf{T}\mathbf{x}$, then

$$\begin{aligned} \|\mathbf{y}\|_{\beta} &= \sup\{\|\mathbf{STx}\| : \mathbf{S} \in M_{1,m}(\mathcal{A}), \|\mathbf{S}\|_{\mathcal{A}} \leq 1\} \\ &\leq \sup\{\|\mathbf{Sx}\| : \mathbf{S} \in M_{1,n}(\mathcal{A}), \|\mathbf{S}\|_{\mathcal{A}} \leq \|\mathbf{T}\|_{\mathcal{A}}\} = \|\mathbf{T}\|_{\mathcal{A}} \|\mathbf{x}\|_{\beta} \end{aligned}$$

and

$$\begin{aligned} \|\mathbf{y}\|_{\mathrm{op}} &= \sup\{\|\mathbf{A}\mathbf{T}\mathbf{x}\| : \mathbf{A} \in M_{1,m}(\mathbb{C}), \|\mathbf{A}\| \le 1\} \\ &\leq \sup\{\|\mathbf{S}\mathbf{x}\| : \mathbf{S} \in M_{1,n}(\mathcal{A}), \|\mathbf{S}\|_{\mathcal{A}} \le \|\mathbf{T}\|_{\mathcal{A}}\} = \|\mathbf{T}\|_{\mathcal{A}} \|\mathbf{x}\|_{\beta}. \end{aligned}$$

From this we immediately get

$$\|\mathbf{T}\|_{\hat{\alpha}} = \sup\{\|\mathbf{y}\|_{\alpha} : \mathbf{y} = \mathbf{T}\mathbf{x}, \mathbf{x} \in X^{n}, \|\mathbf{x}\|_{\alpha} \le 1\} \le \|\mathbf{T}\|_{\mathcal{A}},$$

which proves the proposition.

If \mathcal{A} and \mathcal{B} are two matricial normed algebras, then an algebra homomorphism $\rho : \mathcal{A} \to \mathcal{B}$ naturally induces a map $\rho : M_{m,n}(\mathcal{A}) \to M_{m,n}(\mathcal{B})$ by setting $\rho(\mathbf{T}) = (\rho(T_{jk}))_{j,k=1}^{m,n}$ for $\mathbf{T} \in M_{m,n}(\mathcal{A})$. The algebra homomorphism ρ is called *completely bounded* if these maps are uniformly bounded for $m, n \in \mathbb{N}$.

We will use Proposition 1.4.3 to prove that the bounded algebra homomorphism ρ in the definition of C^* -boundedness can be used to construct a Euclidean structure on X such that ρ is completely bounded if we equip the operator algebra \mathcal{A} with its natural matricial algebra norm given by

$$\|\mathbf{T}\|_{\mathcal{A}} = \|\mathbf{T}\|_{\mathcal{L}(\ell_n^2(H), \ell_m^2(H))}, \qquad T \in M_{m,n}(\mathcal{A})$$

and we equip $\mathcal{L}(X)$ with the matricial algebra norm $\hat{\alpha}$ induced by α .

PROPOSITION 1.4.4. Let H be a Hilbert space and suppose that $\mathcal{A} \subseteq \mathcal{L}(H)$ is an operator algebra. Let $\rho \colon \mathcal{A} \to \mathcal{L}(X)$ be a bounded algebra homomorphism. Then there exists a Euclidean structure α on X such that

$$\|\rho(\mathbf{T})\|_{\hat{\alpha}} \leq \|\rho\| \|\mathbf{T}\|_{\mathcal{L}(\ell_n^2(H),\ell_m^2(H))}$$

for all $\mathbf{T} \in M_{m,n}(\mathcal{A})$, i.e. ρ is completely bounded.

PROOF. We induce a matricial algebra norm β on $\rho(\mathcal{A})$ by setting for $\mathbf{S} \in M_{m,n}(\rho(\mathcal{A}))$

$$\|\mathbf{S}\|_{\beta} = \|\rho\| \inf\{\|\mathbf{T}\|_{\mathcal{L}(\ell_{n}^{2}(H),\ell_{m}^{2}(H))} : \mathbf{T} \in M_{m,n}(\mathcal{A}), \rho(\mathbf{T}) = \mathbf{S}\}.$$

This is indeed a matricial algebra norm since for $\mathbf{S} \in M_{m,k}(\rho(\mathcal{A}))$ and $\mathbf{T} \in M_{k,n}(\rho(\mathcal{A}))$ we have that

$$\begin{aligned} \|\mathbf{ST}\|_{\beta} &= \|\rho\| \inf\{\|\mathbf{U}\| : \mathbf{U} \in M_{m,n}(\mathcal{A}), \rho(\mathbf{U}) = \mathbf{ST}\} \\ &\leq \|\rho\| \inf\{\|\mathbf{U}\| \|\mathbf{V}\| : \rho(\mathbf{U}) = \mathbf{S}, \rho(\mathbf{V}) = \mathbf{T}\} \\ &\leq \|\mathbf{S}\|_{\beta} \|\mathbf{T}\|_{\beta} \end{aligned}$$

as $\|\rho\| \ge 1$. Moreover for any $S \in \rho(\mathcal{A})$ we have

$$||(S)||_{\beta} = ||\rho|| \inf\{||T|| : T \in \mathcal{A}, \rho(T) = S\} \ge ||S||.$$

Hence, by Proposition 1.4.3, there exists a Euclidean structure α such that $\|\mathbf{S}\|_{\hat{\alpha}} \leq \|\mathbf{S}\|_{\beta}$ for all $\mathbf{S} \in M_{m,n}(\rho(\mathcal{A}))$. This means

$$\|\rho(\mathbf{T})\|_{\hat{\alpha}} \le \|\rho(\mathbf{T})\|_{\beta} \le \|\rho\|\|\mathbf{T}\|$$

for all $\mathbf{T} \in M_{m,n}(\mathcal{A})$, proving the proposition.

REMARK 1.4.5. If $\mathcal{A} = C(K)$ for K compact and X has Pisier's contraction property, then one can take $\alpha = \gamma$ in Proposition 1.4.4 (cf. [**PR07**, **KL10**]). For further results on γ -bounded representations of groups, see [**LM10**].

We now have all the necessary preparations to turn Theorem 1.3.2 into an 'if and only if' statement.

THEOREM 1.4.6. Let $\Gamma \subseteq \mathcal{L}(X)$. Then Γ is C^* -bounded if and only if there exists a Euclidean structure α on X such that Γ is α -bounded. Moreover $\|\Gamma\|_{\alpha} \simeq \|\Gamma\|_{C^*}$.

PROOF. First suppose that α is a Euclidean structure on X such that Γ is α -bounded. Let $\tilde{\Gamma}$ be the closure in the strong operator topology of the absolutely convex hull of $\Gamma \cup (\|\Gamma\|_{\alpha} \cdot I_X)$, where I_X is the identity operator on X. By Proposition 1.2.3 we know that $\tilde{\Gamma}$ is α -bounded with $\|\tilde{\Gamma}\|_{\alpha} \leq 2 \|\Gamma\|_{\alpha}$. Then, by Theorem

1.3.2, we can find a closed subalgebra \mathcal{A} of a C^* -algebra and a bounded algebra homomorphism $\rho : \mathcal{A} \to \mathcal{L}(X)$ such that

$$\Gamma \subseteq \tilde{\Gamma} \subseteq \{\rho\tau(T) : T \in \mathcal{L}_{\tilde{\Gamma}}(X), \|T\|_{\tilde{\Gamma}} \leq 1\}$$
$$\subseteq \Big\{\rho(T) : T \in \mathcal{A}, \|T\| \leq \frac{16 \|\Gamma\|_{\alpha}}{\|\rho\|}\Big\}.$$

So Γ is C^* -bounded with constant $\|\Gamma\|_{C^*} \leq 16 \|\Gamma\|_{\alpha}$.

Now assume that Γ is C^* -bounded. Let \mathcal{A} be an operator algebra over a Hilbert space H and let $\rho : \mathcal{A} \to \mathcal{L}(X)$ a bounded algebra homomorphism such that

$$\Gamma \subseteq \left\{ \rho(S) : S \in \mathcal{A}, \|S\| \le \frac{\|\Gamma\|_{C^*}}{\|\rho\|} \right\}.$$

By Proposition 1.4.4 there is a Euclidean structure α such that

$$\|
ho(\mathbf{T})\|_{\hat{lpha}} \le \|
ho\| \|\mathbf{T}\|_{\mathcal{L}(\ell^2_n(H))}$$

for all $\mathbf{T} \in M_{m,n}(\mathcal{A})$. Take $T_1, \ldots, T_n \in \Gamma$ and let $S_1, \ldots, S_n \in \mathcal{A}$ be such that $\rho(S_k) = T_k$ and $||S_k|| \leq \frac{||\Gamma||_{C^*}}{||\rho||}$ for $1 \leq k \leq n$. Then we have, using the Hilbert space structure of H, that

$$\|\operatorname{diag}(T_1,\ldots,T_n)\|_{\hat{\alpha}} = \|\rho(\operatorname{diag}(S_1,\ldots,S_n))\|_{\hat{\alpha}}$$
$$\leq \|\rho\|\|\operatorname{diag}(S_1,\ldots,S_n)\|_{\mathcal{L}(\ell_n^2(H))}$$
$$\leq \|\Gamma\|_{C^*}.$$

So by (1.24) we obtain that Γ is α -bounded with $\|\Gamma\|_{\alpha} \leq \|\Gamma\|_{C^*}$.
CHAPTER 2

Factorization of α -bounded operator families

In Chapter 1 we have seen that the α -boundedness of a family of operators is inherently tied up with a Hilbert space hiding in the background. However, we did not obtain information on the structure of this Hilbert space, nor on the form of the algebra homomorphism connecting the Hilbert and Banach space settings. In this chapter we will highlight some special cases in which the representation can be made more explicit.

The first special case that we will treat is the case where α is either the γ - or the π_2 -structure. In this case Lemma 1.2.5 implies a γ -bounded version of the Kwapień–Maurey factorization theorem. Moreover we will show that π_2 -boundedness can be characterized in terms of factorization through a Hilbert space, i.e. yielding control over the algebra homomorphism.

Afterwards we turn our attention to the case in which X is a Banach function space. In Section 2.2 we will show that, under a mild additional assumption on the Euclidean structure α , an α -bounded family of operators on a Banach function space is actually ℓ^2 -bounded. This implies that the ℓ^2 -structure is the canonical structure to consider on Banach function spaces.

In Section 2.3 we prove a version of Lemma 1.3.1 for Banach function spaces, in which the abstract Hilbert space is replaced by a weighted L^2 -space. This is remarkable, since this is gives us crucial information on the Hilbert space H. This formulation resembles the work of Maurey, Nikišin and Rubio de Francia on weighted versus vector-valued inequalities, but has the key advantage that no geometric properties of the Banach function space are used. In Section 2.4 we use this version of Lemma 1.3.1 to prove a vector-valued extension theorem with weaker assumptions than the one in the work of Rubio de Francia. This has applications in vector-valued harmonic analysis, of which we will give a few examples.

2.1. Factorization of γ - and π_2 -bounded operator families

In this section we will consider the special case where α is either the γ - or the π_2 -structure. For these Euclidean structures we will show that α -bounded families of operators can be factorized through a Hilbert space under certain geometric conditions on the underlying Banach spaces. All results in this section will be based on the following lemma, which is a special case of Lemma 1.2.5.

LEMMA 2.1.1. Let X and Y be Banach spaces. Let $\Gamma_1 \subseteq \mathcal{L}(X, Y)$ and suppose that there is a C > 0 such that for all $\mathbf{x} \in X^n$ and $S_1, \ldots, S_n \in \Gamma_1$ we have

$$\|(S_1x_1,\ldots,S_nx_n)\|_{\pi_2} \le C\left(\sum_{k=1}^n \|x_k\|^2\right)^{1/2}.$$

Let $\Gamma_2 \subseteq \mathcal{L}(Y)$ be a π_2 -bounded family of operators. Then there is a Hilbert space H, a contractive embedding $U: H \to Y$, a $\widetilde{S} \in \mathcal{L}(X, H)$ for every $S \in \Gamma_1$ and a $\widetilde{T} \in \mathcal{L}(H)$ for every $T \in \Gamma_2$ such that

$$\begin{split} \|S\| &\leq 4C, \qquad \qquad S \in \Gamma_1, \\ \|\widetilde{T}\| &\leq 2 \, \|\Gamma\|_{\pi_2}, \qquad \qquad T \in \Gamma_2. \end{split}$$

and the following diagram commutes:

$$\begin{array}{ccc} X & \xrightarrow{S} & Y & \xrightarrow{T} & Y \\ & \searrow \overset{\tilde{S}}{\searrow} & U & & U \\ & H & \xrightarrow{\tilde{T}} & H \end{array}$$

PROOF. Define $F : Y \to [0, \infty)$ as $F(y) = ||y||_Y$. Then we have, by the definition of the π_2 -structure, for any $\mathbf{y} \in Y^n$

$$\left(\sum_{k=1}^{n} F(y_k)^2\right)^{1/2} = \left(\sum_{k=1}^{n} \|y_k\|_X^2\right)^{1/2} \le \|\mathbf{y}\|_{\pi_2}.$$

Let

$$\widetilde{Y} = \{Sx : S \in \Gamma_1, x \in X\} \subseteq Y$$

and define $G \colon \widetilde{Y} \to [0,\infty)$ by

$$G(y) := C \cdot \inf \big\{ \|x\|_X : x \in X, \, Sx = y, \, S \in \Gamma_1 \big\}.$$

Fix $\mathbf{y} \in \widetilde{Y}$, then for any $S_1, \ldots, S_n \in \Gamma_1$ and $\mathbf{x} \in X^n$ such that $y_k = Sx_k$ we have

$$\|\mathbf{y}\|_{\pi_2} \le C \cdot \left(\sum_{k=1}^n \|x_k\|_X^2\right)^{1/2}.$$

Thus, taking the infimum over all such S_k and \mathbf{x} , we obtain

$$\|\mathbf{y}\|_{\pi_2} \le \left(\sum_{k=1}^n G(y_k)^2\right)^{1/2}.$$

Hence by Lemma 1.2.5 there is a Hilbertian seminorm $\|\cdot\|_0$ on a Γ_2 -invariant subspace Y_0 of Y which contains \widetilde{Y} and satisfies (1.10)-(1.12). In particular, for $y \in Y_0$ we have $\|y\|_Y = F(y) \le \|y\|_0$, so $\|\cdot\|_0$ is a norm.

Let H be the completion of $(Y_0, \|\cdot\|_0)$ and let $U \colon H \to Y$ be the inclusion mapping. For every $S \in \Gamma_1$ let $\tilde{S} \colon X \to H$ be the mapping $x \mapsto Sx \in \tilde{Y} \subseteq Y_0$. Then we have for any $x \in X$ that

$$||Sx||_0 \le 4 G(Sx) \le 4 C ||x||_X,$$

so $\|\widetilde{S}\| \leq 4C$. Moreover we have $S = U\widetilde{S}$. Finally let \widetilde{T} be the canonical extension of $T \in \Gamma_2$ to H. Then we have $\|\widetilde{T}\| \leq 2\|\Gamma_2\|_{\pi_2}$ and $TU = U\widetilde{T}$, which proves the lemma.

A γ -bounded Kwapień–Maurey factorization theorem. As a first application of Lemma 2.1.1 we prove a γ -bounded version of the Kwapień-Maurey factorization theorem (see [Kwa72, Mau74] and [AK16, Theorem 7.4.2]).

THEOREM 2.1.2. Let X be a Banach space with type 2 and Y a Banach space with cotype 2. Let $\Gamma_1 \subseteq \mathcal{L}(X, Y)$ and $\Gamma_2 \subseteq \mathcal{L}(Y)$ be γ -bounded families of operators. Then there is a Hilbert space H, a contractive embedding $U : H \to Y$, a $\widetilde{S} \in \mathcal{L}(X, H)$ for every $S \in \Gamma_1$ and a $\widetilde{T} \in \mathcal{L}(H)$ for every $T \in \Gamma_2$ such that

$$\begin{split} \|\widetilde{S}\| \lesssim \|\Gamma_1\|_{\gamma} & S \in \Gamma_1 \\ \|\widetilde{T}\| \lesssim \|\Gamma_2\|_{\gamma}, & T \in \Gamma_2. \end{split}$$

and the following diagram commutes:

$$\begin{array}{ccc} X & \stackrel{S}{\longrightarrow} Y & \stackrel{T}{\longrightarrow} Y \\ & \searrow \stackrel{\tilde{S}}{\searrow} \stackrel{U}{\uparrow} & \stackrel{U}{\uparrow} \\ & H & \stackrel{\tilde{T}}{\longrightarrow} H \end{array}$$

Note that the Kwapień-Maurey factorization theorem follows from Theorem 2.1.2 by taking $\Gamma_1 = \{S\}$ for some $S \in \mathcal{L}(X, Y)$ and taking $\Gamma_2 = \emptyset$. In particular the fact that any Banach space with type 2 and cotype 2 is isomorphic to a Hilbert space follows by taking X = Y, $\Gamma_1 = \{I_X\}$ and $\Gamma_2 = \emptyset$.

PROOF OF THEOREM 2.1.2. Note that γ -boundedness and π_2 -boundedness are equivalent on a space with cotype 2 by Proposition 1.1.3. Thus Γ_2 is π_2 -bounded on Y. Furthermore, using Proposition 1.1.3, the γ -boundedness of Γ_1 and Proposition 1.0.1, we have for $\mathbf{x} \in X^n$ and $T_1, \ldots, T_n \in \Gamma$

$$\begin{aligned} \|(T_1x_1, \dots, T_nx_n)\|_{\pi_2} &\lesssim \|(T_1x_1, \dots, T_nx_n)\|_{\gamma} \\ &\leq \|\Gamma\|_{\gamma}\|(x_1, \dots, x_n)\|_{\gamma} \\ &\lesssim \|\Gamma\|_{\gamma} \Big(\sum_{k=1}^n \|x_k\|^2\Big)^{1/2}. \end{aligned}$$

Therefore the theorem follows from Lemma 2.1.1.

Factorization of π_2 -bounded operator families through a Hilbert space. If we let X be a Hilbert space in Lemma 2.1.1, we can actually characterize the π_2 -boundedness of a family of operators on Y by a factorization property. In order to prove this will need the π_2 -summing norm for operators $T \in \mathcal{L}(Y, Z)$, where Y and Z are Banach spaces. It is defined as

$$||T||_{\pi_2} := \sup\left\{ \left(\sum_{k=1}^n ||Ty_k||_Z^2 \right)^{1/2} : \mathbf{y} \in Y^n, \sup_{||y^*||_{Y^*} \le 1} \left(\sum_{k=1}^n |\langle y_k, y^* \rangle|^2 \right)^{1/2} \le 1 \right\}$$

Clearly $||T|| \leq ||T||_{\pi_2}$ and T is called 2-summing if $||T||_{\pi_2} < \infty$. For a connection between p-summing operators and factorization through L^p we refer to [**Tom89**, **DJT95**] and the references therein. If $Y = \ell^2$ this definition coincides with the definition given in Section 1.1, which follows from the fact that $\mathcal{L}(\ell^2)$ is isometrically isomorphic to $\ell^2_{\text{weak}}(\ell^2)$, the space of all sequences $(y_n)_{n\geq 1}$ in ℓ^2 for which

$$\|(y_n)_{n\geq 1}\|_{\ell^2_{\text{weak}}(\ell^2)} := \sup_{\|y^*\|_{\ell^2} \leq 1} \left(\sum_{n=1}^{\infty} |\langle y_n, y^* \rangle|^2\right)^{1/2},$$

is finite, see e.g. [DJT95, Proposition 2.2].

THEOREM 2.1.3. Let Y be a Banach space and let $\Gamma \subseteq \mathcal{L}(Y)$. Then Γ is π_2 bounded if and only if there is a C > 0 such that for any Hilbert space X and $S \in \mathcal{L}(X,Y)$, there is a Hilbert space H, a contractive embedding $U : H \to Y$, a $\widetilde{S} \in \mathcal{L}(X,H)$ and a $\widetilde{T} \in \mathcal{L}(H)$ for every $T \in \Gamma$ such that

$$\begin{split} \|\widetilde{S}\| &\leq 4\|S\| \\ \|\widetilde{T}\| &\leq C, \end{split} \qquad \qquad T \in \Gamma \end{split}$$

and the following diagram commutes

$$\begin{array}{ccc} X & \xrightarrow{S} & Y & \xrightarrow{T} & Y \\ & \searrow \tilde{S} & U & \downarrow & U \\ & & & U & \downarrow \\ & & H & \xrightarrow{\tilde{T}} & H \end{array}$$

Moreover C > 0 can be chosen such that $\|\Gamma\|_{\pi_2} \simeq C$.

PROOF. For the 'only if' statement let X be a Hilbert space and $S \in \mathcal{L}(X, Y)$. Note that by the ideal property of the π_2 -structure and the coincidence of the π_2 -norm and the Hilbert-Schmidt norm on Hilbert spaces we have for all $\mathbf{x} \in X^n$

$$||(Sx_1,\ldots,Sx_n)||_{\pi_2} \le ||S|| ||(x_1,\ldots,x_n)||_{\pi_2} = \left(\sum_{k=1}^n ||x_k||^2\right)^{1/2}.$$

Therefore the 'only if' statement follows directly from Lemma 2.1.1 using $\Gamma_1 = \{S\}$.

For the 'if' statement let $T_1, \ldots, T_n \in \Gamma$. Let $\mathbf{y} \in Y^n$ with $\|\mathbf{y}\|_{\pi_2} \leq 1$ and let V be the finite rank operator associated to \mathbf{y} , i.e.

$$Vf := \sum_{k=1}^{n} \langle f, e_k \rangle y_k, \qquad f \in \ell^2$$

for some orthonormal sequence $(e_k)_{k=1}^n$ in ℓ^2 . We will combine the given Hilbert space factorization with Pietsch factorization theorem to factorize V and T_1, \ldots, T_n . In particular we will construct operators such that the following diagram commutes:

$$\begin{array}{cccc} \ell^2 & & V & & Y & \overset{T_k}{\longrightarrow} Y \\ \downarrow \widetilde{v} & & & & \downarrow \uparrow & & U \\ L^{\infty}(\Omega) & \overset{J}{\longrightarrow} L^2(\Omega) = X & \overset{\widetilde{S}}{\longrightarrow} H & \overset{\widetilde{T}_k}{\longrightarrow} H \end{array}$$

As $||V||_{\pi_2} \leq 1$, by the Pietsch factorization theorem [**DJT95**, p.48] there is a probability space (Ω, \mathbb{P}) and operators $\tilde{V} \colon \ell^2 \to L^{\infty}(\Omega)$ and $S \colon L^2(\Omega) \to Y$, such that $||\tilde{V}|| \leq 1$, $||S|| \leq 1$ and $V = SJ\tilde{V}$, where $J \colon L^{\infty}(\Omega) \to L^2(\Omega)$ is the canonical inclusion.

We now use the assumption with $X = L^2(\Omega)$ and S to construct H, U, \tilde{S} and \tilde{T}_k for $1 \leq k \leq n$ with the prescribed properties. Define $R \in \mathcal{F}(\ell^2, X)$ by $Re_k = T_k Ve_k$ and $\tilde{R} \in \mathcal{F}(\ell^2, L^2(\Omega))$ by $\tilde{R}e_k = \tilde{T}_k \tilde{S} J \tilde{V} e_k$ for $1 \leq k \leq n$. Then R = UR (see the diagram above) and therefore we have

$$\begin{aligned} \|R\|_{\pi_{2}} &\leq \|U\| \left\|\widetilde{R}\right\|_{\pi_{2}} = \left\|\widetilde{R}\right\|_{\mathrm{HS}} = \left(\sum_{k=1}^{n} \left\|\widetilde{T}_{k}\widetilde{S}J\widetilde{V}e_{k}\right\|^{2}\right)^{1/2} \\ &\leq 4C\left(\sum_{k=1}^{n} \left\|J\widetilde{V}e_{k}\right\|^{2}\right)^{1/2}. \end{aligned}$$

Since $||J||_{\pi_2} = 1$ by [**DJT95**, Example 2.9(d)], we have $||J\tilde{V}||_{\pi_2} \leq 1$ by the ideal property of the π_2 -summing norm. Moreover $||(T_1y_1, \ldots, T_ny_n)||_{\pi_2} = ||R||_{\pi_2}$, so we can conclude

$$\|(T_1y_1,\ldots,T_ny_n)\|_{\pi_2} \le 4C,$$

i.e. Γ is π_2 -bounded with $\|\Gamma\|_{\pi_2} \le 4C.$

In Theorem 2.1.3 it suffices to consider the case where S and \tilde{S} are injective, which allows us to restate the theorem in terms of Hilbert spaces embedded in Y.

COROLLARY 2.1.4. Let Y be a Banach space and let $\Gamma \subseteq \mathcal{L}(Y)$. Then Γ is π_2 -bounded if and only if there is a C > 0 such that:

For any Hilbert space X contractively embedded in Y there is a Hilbert space H (\star) with $X \subseteq H \subseteq Y$ such that H is contractively embedded in Y, the embedding

(*) $X \hookrightarrow H$ has norm at most 4, and such that T is an operator on H with $||T||_{\mathcal{L}(H)} \leq C$ for all $T \in \Gamma$.

Moreover C > 0 can be chosen such that $\|\Gamma\|_{\pi_2} \simeq C$.

PROOF. For the 'if' statement note that in the proof of Theorem 2.1.3 the orthonormal sequence can be chosen such that V is injective, and thus S can be made injective by restricting to $J\widetilde{V}(\ell^2) \subseteq L^2(\Omega)$. For the converse note that if S is injective in the proof of Lemma 2.1.1, then the constructed \widetilde{S} is as well.

Since the π_2 -structure is equivalent to the γ -structure if Y has cotype 2 by Proposition 1.1.3, we also have:

COROLLARY 2.1.5. Let Y be a Banach space with cotype 2 and let $\Gamma \subseteq \mathcal{L}(Y)$. Then Γ is γ -bounded if and only if there is a C > 0 such that (\star) of Corollary 2.1.4 holds. Moreover C > 0 can be chosen such that $\|\Gamma\|_{\gamma} \simeq C$.

Finally we note that we can dualize Theorem 2.1.3, Corollary 2.1.4 and 2.1.5. For example we have:

COROLLARY 2.1.6. Let Y be a Banach space with cotype 2 and let $\Gamma \subseteq \mathcal{L}(Y)$. Then Γ is γ -bounded if and only if there is a C > 0 such that:

For any Hilbert space X in which Y is contractively embedded there is a Hilbert space H with $Y \subseteq H \subseteq X$ such that Y is contractively embedded in H, the embedding $H \hookrightarrow X$ has norm at most 4, and such that T extends boundedly to

 $H \text{ with } \|T\|_{\mathcal{L}(H)} \leq C \text{ for all } T \in \Gamma.$

Moreover C > 0 can be chosen such that $\|\Gamma\|_{\gamma} \simeq C$.

PROOF. Note that since Y has type 2, Y^* has non-trivial type and cotype 2. Therefore by Proposition 1.1.4 the γ^* -structure is equivalent to the γ -structure on X^* . Moreover by Proposition 1.2.3 we know that Γ^* is γ^* -bounded on Y^* with $\|\Gamma^*\|_{\gamma^*} \leq C$. So the corollary follows by dualizing Corollary 2.1.5.

2.2. α -bounded operator families on Banach function spaces

For the remainder of this chapter, we will study Euclidean structures and factorization in the case that X is a Banach function space.

DEFINITION 2.2.1. Let (S, μ) be a σ -finite measure space. A subspace X of the space of measurable functions on S, denoted by $L^0(S)$, equipped with a norm $\|\cdot\|_X$ is called a *Banach function space* if it satisfies the following properties:

- (i) If $x \in L^0(S)$ and $y \in X$ with $|x| \le |y|$, then $x \in X$ and $||x||_X \le ||y||_X$.
- (ii) There is an $x \in X$ with x > 0 a.e.
- (iii) If $0 \le x_n \uparrow x$ for $(x_n)_{n=1}^{\infty}$ in $X, x \in L^0(S)$ and $\sup_{n \in \mathbb{N}} ||x_n||_X < \infty$, then $x \in X$ and $||x||_X = \sup_{n \in \mathbb{N}} ||x_n||_X$.

A Banach function space X is called *order-continuous* if additionally

(iv) If $0 \le x_n \uparrow x \in X$ with $(x_n)_{n=1}^{\infty}$ a sequence in X and $x \in X$, then $||x_n - x||_X \to 0$.

Order-continuity of a Banach function space X ensures that the dual X^* is a Banach function space (see [LT79, Section 1.b]) and that the Bochner space $L^p(S'; X)$ is a Banach function space on $(S \times S', \mu \times \mu')$ for any σ -finite measure space (S', μ') . As an example we note that any Banach function space which is reflexive or has finite cotype is order-continuous.

Since a Banach function space X is in particular a Banach lattice, it admits the ℓ^2 -structure. The main result of this section will be that the ℓ^2 -structure is actually the canonical structure to study on Banach function spaces. Indeed, we will show that, under mild assumptions on the Euclidean structure α , α -boundedness implies ℓ^2 -boundedness. We start by noting the following property of a Hilbertian seminorm on a space of functions.

LEMMA 2.2.2. Let (S, μ) be a measure space and let $X \subseteq L^0(S)$ be a vector space with a Hilbertian seminorm $\|\cdot\|_0$. Suppose that there is a C > 0 such that for $x \in L^0(S)$ and $y \in X$

$$|x| \leq |y| \Rightarrow x \in X \text{ and } ||x||_0 \leq C ||y||_0.$$

Then there exists a seminorm $\|\cdot\|_1$ on X such that

$$\begin{split} & \frac{1}{C} \|x\|_0 \le \|x\|_1 \le C \, \|x\|_0 \qquad \qquad x \in X, \\ \|x+y\|_1^2 = \|x\|_1^2 + \|y\|_1^2, \qquad \qquad x, y \in X : x \land y = 0. \end{split}$$

PROOF. Let Π be the collection of all finite measurable partitions of S, partially ordered by refinement. We define

$$\|x\|_{1} = \inf_{\pi \in \Pi} \sup_{\pi' \ge \pi} \left(\sum_{E \in \pi'} \|x \mathbf{1}_{E}\|_{0}^{2} \right)^{1/2}, \qquad x \in X,$$

which is clearly a seminorm. For a $\pi \in \Pi$, write $\pi = \{E_1, \dots, E_n\}$ and let $(\varepsilon_k)_{k=1}^n$ be a Rademacher sequence. Then we have for all $x \in X$ that

$$\sum_{k=1}^{n} \|x \mathbf{1}_{E_{k}}\|_{0}^{2} = \mathbb{E} \sum_{j=1}^{n} \sum_{k=1}^{n} \varepsilon_{j} \varepsilon_{k} \langle x \mathbf{1}_{E_{j}}, x \mathbf{1}_{E_{k}} \rangle = \mathbb{E} \left\| \sum_{k=1}^{n} \varepsilon_{k} \cdot x \mathbf{1}_{E_{k}} \right\|_{0}^{2} \le C^{2} \|x\|_{0}^{2}$$

and, since $\bigcup_{k=1}^{n} E_k = S$, we deduce in the same fashion

$$\|x\|_{0}^{2} \leq C^{2} \mathbb{E} \left\| \sum_{k=1}^{n} \varepsilon_{k} \cdot x \, \mathbf{1}_{E_{k}} \right\|_{0}^{2} = C^{2} \sum_{k=1}^{n} \|x \, \mathbf{1}_{E_{k}}\|_{0}^{2} = C^{2} \sum_{k=1}^{n} \|x \, \mathbf{1}_{E_{k}}\|_{0}^{2}.$$

Therefore we have $\frac{1}{C} ||x||_0 \le ||x||_1 \le C ||x||_0$ for all $x \in X_\eta$ and $\pi \in \Pi$. Furthermore if $x, y \in X$ with $x \land y = 0$, then for $\pi \ge \{ \operatorname{supp} x, S \setminus \operatorname{supp} x \}$ we have

$$\sum_{E \in \pi} \|(x+y) \mathbf{1}_E\|_0^2 = \sum_{E \in \pi} \|x \mathbf{1}_E\|_0^2 + \sum_{E \in \pi} \|y \mathbf{1}_E\|_0^2$$

So we also get $||x + y||_1^2 = ||x||_1^2 + ||y||_1^2$, which proves the lemma.

Let X be a Banach function space. For $m \in L^{\infty}(S)$ we define the pointwise multiplication operator $T_m \colon X \to X$ by $T_m x = m \cdot x$ and denote the collection of pointwise multiplication operators on X by

(2.1)
$$\mathcal{M} = \{T_m : m \in L^{\infty}(S), \|m\|_{L^{\infty}(S)} \le 1\} \subseteq \mathcal{L}(X).$$

It turns out that if α is a Euclidean structure on X such that \mathcal{M} is α -bounded, then α -boundedness implies ℓ^2 -boundedness. This will follow from the fact that an \mathcal{M} -invariant subspace of X satisfies the assumptions of Lemma 2.2.2.

THEOREM 2.2.3. Let X be a Banach function space on (S, μ) , let α be a Euclidean structure on X and assume that \mathcal{M} is α -bounded. If $\Gamma \subseteq \mathcal{L}(X)$ is α -bounded, then Γ is ℓ^2 -bounded with $\|\Gamma\|_{\ell^2} \lesssim \|\Gamma\|_{\alpha}$, where the implicit constant only depends on $\|\mathcal{M}\|_{\alpha}$.

PROOF. Let $\mathbf{x} \in X^n$ and $T_1, \ldots, T_n \in \Gamma$. We will first reduce the desired estimate to an estimate for simple functions. Define $z_0 := (\sum_{k=1}^n |x_k|^2)^{\frac{1}{2}}$. Then we have $x_k z_0^{-1} \in L^{\infty}(S)$, which means that we can find simple functions $\mathbf{u} \in X^n$ such that $||u_k - x_k z_0^{-1}||_{L^{\infty}(S)} \leq \frac{1}{n}$ for $1 \leq k \leq n$. Defining $y_0 := (\sum_{k=1}^n |u_k z_0|^2)^{1/2}$, we have

$$||z_0 - y_0||_X \le \sum_{k=1}^n ||x_k - u_k z_0||_X \le \sum_{k=1}^n ||z_0||_X ||u_k - x_k z_0^{-1}||_{L^{\infty}(S)} \le ||z_0||_X.$$

Define $z_1 := (\sum_{k=1}^n |T_k(u_k z_0)|^2)^{1/2}$. Then similarly, using $||T_k|| \le ||\Gamma||_{\alpha}$, we have

$$\left\| \left(\sum_{k=1}^{n} |T_k x_k|^2 \right)^{1/2} - z_1 \right\|_X \le \|z_0\|_X \|\Gamma\|_{\alpha}$$

For $1 \le k \le n$ we have $T(u_k z_0) z_1^{-1} \in L^{\infty}(S)$, which means that we can find simple functions $\mathbf{v} \in X^n$ such that $|v_k| \le |T(u_k z_0) z_1^{-1}|$ and

$$\|v_k - T(u_k z_0) z_1^{-1}\|_{L^{\infty}(S)} \le \|\Gamma\|_{\alpha} \frac{\|z_0\|_X}{\|z_1\|_X} \frac{1}{n}.$$

It follows that

(2.2)
$$|v_k z_1| \le |T(u_k z_0)|$$

and, defining $y_1 := (\sum_{k=1}^n |v_k z_1|^2)$, that

$$||z_1 - y_1||_X \le \sum_{k=1}^n ||T_k(u_k z_0) - v_k z_1||_X \le ||z_0||_X ||\Gamma||_{\alpha}.$$

Thus, combining the various estimates, we have

$$\left\| \left(\sum_{k=1}^{n} |T_k(x_k)|^2 \right)^{\frac{1}{2}} \right\|_X \le \|y_1\|_X + 2 \|\Gamma\|_{\alpha} \left\| \left(\sum_{k=1}^{n} |x_k|^2 \right)^{\frac{1}{2}} \right\|_X \\ \|y_0\| \le 2 \left\| \left(\sum_{k=1}^{n} |x_k|^2 \right)^{\frac{1}{2}} \right\|_X,$$

so it suffices to prove $||y_1||_X \leq ||\Gamma||_{\alpha} ||y_0||_X$, which is the announced reduction to simple functions.

Define $C_{\Gamma} := 4 \|\Gamma\|_{\alpha}, C_{\mathcal{M}} := 4 \|\mathcal{M}\|_{\alpha}$ and set

$$\Gamma_0 := \left(\frac{1}{2\|\Gamma\|_{\alpha}} \cdot \Gamma\right) \cup \left(\frac{1}{2\|\mathcal{M}\|_{\alpha}} \cdot \mathcal{M}\right).$$

Then Γ_0 is α -bounded with $\|\Gamma_0\|_{\alpha} \leq 1$ by Proposition 1.2.3. Thus, applying Lemma 1.3.1 to Γ_0 and $\eta = (y_0, y_1)$, we can find a Γ - and \mathcal{M} -invariant subspace $X_\eta \subseteq X$ with $y_0 \in X_\eta$ and a Hilbertian seminorm $\|\cdot\|_\eta$ on X_η such that (1.22) and (1.23) hold and

(2.3)
$$||Tx||_{\eta} \le C_{\Gamma} ||x||_{\eta}, \qquad x \in X_{\eta}, T \in \Gamma.$$

(2.4)
$$||Tx||_{\eta} \le C_{\mathcal{M}} ||x||_{\eta}, \qquad x \in X_{\eta}, T \in \mathcal{M}.$$

In particular, (2.4) implies if $x \in L^0(S)$, $\tilde{x} \in X_\eta$, and $|x| \leq |\tilde{x}|$, then $x \in X_\eta$ and

$$\|x\|_{\eta} \le C_{\mathcal{M}} \|\tilde{x}\|_{\eta}.$$

Therefore we deduce that $u_k y_0, T_k(u_k y_0), v_k z_1 \in X_\eta$ for $1 \le k \le n$ and $y_1, z_1 \in X_\eta$. Moreover, by Lemma 2.2.2 there is a seminorm $\|\cdot\|_{\nu}$ on X_η such that

(2.6)
$$\frac{1}{C_{\mathcal{M}}} \|x\|_{\eta} \le \|x\|_{\nu} \le C_{\mathcal{M}} \|x\|_{\eta} \qquad x \in X,$$

(2.7)
$$\|x_1 + x_2\|_{\nu}^2 = \|x_1\|_{\nu}^2 + \|x_2\|_{\nu}^2, \qquad x_1, x_2 \in X : x_1 \wedge x_2 = 0.$$

Let Σ' be a coarsest σ -algebra such that **u** and **v** are measurable and let $E_1, \ldots, E_m \in \Sigma'$ be the atoms of this finite σ -algebra. By applying (2.2)-(2.7), we get

$$\begin{split} \|y_1\|_{\eta}^2 &\leq C_{\mathcal{M}}^2 \sum_{j=1}^m \left\| \left(\sum_{k=1}^n |v_k|^2 \right)^{1/2} z_1 \, \mathbf{1}_{E_j} \right\|_{\nu}^2 \qquad \text{by } (2.6) + (2.7) \\ &= C_{\mathcal{M}}^2 \sum_{k=1}^n \sum_{j=1}^m \|v_k z_1 \, \mathbf{1}_{E_j}\|_{\nu}^2 \qquad \text{since } v_k \text{ is constant on } E_j \\ &\leq C_{\mathcal{M}}^4 \sum_{k=1}^n \|v_k z_1\|_{\eta}^2 \qquad \text{by } (2.6) + (2.7) \\ &\leq C_{\mathcal{M}}^6 \sum_{k=1}^n \|T_k(u_k z_0)\|_{\eta}^2 \qquad \text{by } (2.2) + (2.5) \\ &\leq C_{\mathcal{M}}^6 C_{\Gamma}^2 \sum_{k=1}^n \|u_k z_0\|_{\eta}^2 \qquad \text{by } (2.3) \\ &\leq C_{\mathcal{M}}^8 C_{\Gamma}^2 \sum_{k=1}^n \sum_{j=1}^m \|u_k z_0 \, \mathbf{1}_{E_j}\|_{\nu}^2 \qquad \text{by } (2.6) + (2.7) \end{split}$$

$$= C_{\mathcal{M}}^{8} C_{\Gamma}^{2} \sum_{j=1}^{m} \left\| \left(\sum_{k=1}^{n} |u_{k}|^{2} \right)^{1/2} z_{0} \mathbf{1}_{E_{j}} \right\|_{\nu}^{2} \qquad \text{since } u_{k} \text{ is constant on } E_{j}$$

$$\leq C_{\mathcal{M}}^{10} C_{\Gamma}^{2} \|y_{0}\|_{n}^{2} \qquad \qquad \text{by } (2.6) + (2.7)$$

Hence, by combining this estimate with (1.22) and (1.23), we get

$$||y_1||_X \le ||y_1||_\eta \le C^5_{\mathcal{M}} C_{\Gamma} ||y_0||_\eta \le 4 C^5_{\mathcal{M}} C_{\Gamma} ||y_0||_X,$$

which concludes the proof.

In view of Theorem 2.2.3 it would be interesting to investigate sufficient conditions on a general Banach space X such that α -boundedness of a family of operators on X implies e.g. γ -boundedness.

Remark 2.2.4.

- (i) By Theorem 1.4.6 one could replace the assumption on Γ and \mathcal{M} in Theorem 2.2.3 by the assumption that $\Gamma \cup \mathcal{M}$ is C^* -bounded.
- (ii) \mathcal{M} is γ -bounded if and only if X has finite cotype. Indeed, the 'if' statement follows from Proposition 1.1.3 and the fact that $\|\mathcal{M}\|_{\ell^2} = 1$. The 'only if' part follows from a variant of [**HNVW17**, Example 8.1.9] and the fact that if X does not have finite cotype, then ℓ_n^{∞} is $(1 + \varepsilon)$ -lattice finitely representable in X for any $n \in \mathbb{N}$ (see [**LT79**, Theorem 1.f.12]).
- (iii) The assumption that \mathcal{M} is α -bounded is not only sufficient, but also necessary in Theorem 2.2.3 if $\alpha = \gamma$. Indeed, for the γ -structure we know that γ -boundedness implies ℓ^2 -boundedness if and only if X has finite cotype, see [**KVW16**, Theorem 4.7]. Therefore if γ -boundedness implies ℓ^2 -boundedness on X, then X has finite cotype. This implies that \mathcal{M} is γ -bounded.

REMARK 2.2.5. On a Banach function space X one can also define for $q \in [1, \infty)$

$$\|\mathbf{x}\|_{\ell^q} := \left\| \left(\sum_{k=1}^n |x_k|^q \right)^{1/q} \right\|_X, \quad \mathbf{x} \in X^n.$$

and study the ℓ^q -boundedness of operators, which was initiated in [Wei01a] and done systematically in [KU14]. Our representation results of Chapter 1 rely heavily on the Hilbert structure of ℓ^2 and therefore a generalization of our representation results to an " ℓ^q -Euclidean structure" setting seems out of reach.

2.3. Factorization of ℓ^2 -bounded operator families through $L^2(S, w)$

As we have seen in the previous section, the ℓ^2 -structure is the canonical structure to consider on a Banach function space X. In this section we prove a version of Lemma 1.3.1 for the ℓ^2 -boundedness of a family of operators on a Banach function space, in which we have control over the the space X_η and the Hilbertian seminorm $\|\cdot\|_{\eta}$. Indeed, we will see that an ℓ^2 -bounded family of operators on a Banach function space X can be factorized through a weighted L^2 -space. In fact, this actually characterizes ℓ^2 -boundedness on X.

By a weight on a measure space (S, μ) we mean a measurable function $w : S \to [0, \infty)$. For $p \in [1, \infty)$ we let $L^p(S, w)$ be the space of all $f \in L^0(S)$ such that

$$||f||_{L^p(S,w)} := \left(\int_S |f|^p w \,\mathrm{d}\mu\right)^{1/p} < \infty.$$

Our main result is as follows. For the special case $X = L^p(S)$ this result can be found in the work of Le Merdy and Simard [LS02, Theorem 2.1]. See also Johnson and Jones [JJ78] and Simard [Sim99].

THEOREM 2.3.1. Let X be an order-continuous Banach function space on (S, Σ, μ) and let $\Gamma \subseteq \mathcal{L}(X)$. Γ is ℓ^2 -bounded if and only if there exists a constant C > 0 such that for all $y_0, y_1 \in X$ there exists a weight w such that $y_0, y_1 \in L^2(S, w)$ and

- (2.8) $||Tx||_{L^2(S,w)} \le C ||x||_{L^2(S,w)}, \quad x \in X \cap L^2(S,w), T \in \Gamma$
- (2.9) $||y_0||_{L^2(S,w)} \le c ||y_0||_X,$
- (2.10) $||y_1||_{L^2(S,w)} \ge \frac{1}{c} ||y_1||_X,$

where c is a numerical constant. Moreover C can be chosen such that $\|\Gamma\|_{\ell^2} \simeq C$.

PROOF. We will first prove the 'if' part. Let $\mathbf{x} \in X^n$ and $T_1, \ldots, T_n \in \Gamma$. Define $y_0 = (\sum_{k=1}^n |x_k|^2)^{\frac{1}{2}}$ and $y_1 = (\sum_{k=1}^n |T_k x_k|^2)^{\frac{1}{2}}$. Then we have by applying (2.8)-(2.10)

$$\|y_1\|_X^2 \le c^2 \sum_{k=1}^n \int_S |T_k x_k|^2 w \, \mathrm{d}\mu \le c^2 C^2 \sum_{k=1}^n \int_S |x_k|^2 w \, \mathrm{d}\mu \le c^4 C^2 \|y_0\|_X^2,$$

so $\|\Gamma\|_{\ell^2} \leq c^2 C$.

Now for the converse take $y_0, y_1 \in X$ arbitrary and let $\tilde{u} \in X$ with $\tilde{u} > 0$ a.e. Assume without loss of generality that $\|y_0\|_X = \|y_1\|_X = \|\tilde{u}\|_X = 1$ and define

$$u = \frac{1}{3} (|y_0| \vee |y_1| \vee \tilde{u}).$$

Then $||u||_X \le 1, u > 0$ a.e. and

(2.11)
$$||y_j^2 u^{-1}||_X \le ||y_j||_X ||y_j u^{-1}||_{L^{\infty}(S)} \le 3 ||y_j||_X, \quad j = 0, 1.$$

Let $Y = \{x \in X : x^2 u^{-1} \in X\}$ with norm $||x||_Y := ||x^2 u^{-1}||_X^{1/2}$. Then Y is an order-continuous Banach function space and for $\mathbf{v} \in Y^n$ we have

$$\begin{split} \left\| \left(\sum_{k=1}^{n} |v_k|^2 \right)^{1/2} \right\|_Y &= \left\| \sum_{k=1}^{n} |v_k|^2 u^{-1} \right\|_X^{1/2} \\ &\leq \left(\sum_{k=1}^{n} \left\| |v_k|^2 u^{-1} \right\|_X \right)^{1/2} = \left(\sum_{k=1}^{n} \left\| v_k \right\|_Y^2 \right)^{1/2}, \end{split}$$

i.e. Y is 2-convex. Moreover by Hölders inequality for Banach function spaces ([LT79, Proposition 1.d.2(i)]), we have

$$||x||_X \le ||x^2 u^{-1}||_X^{1/2} ||u||_X^{1/2} = ||x||_Y,$$

so Y is contractively embedded in X. By (2.11) we have $u, y_0, y_1 \in Y$. We will now apply Lemma 1.2.5. Define $F: X \to [0, \infty)$ by

$$F(x) = \begin{cases} \|x\|_X & \text{if } x \in \text{span}\{y_1\}\\ 0 & \text{otherwise} \end{cases}$$

and $G: Y \to [0, \infty)$ by $G(x) = ||x||_Y$. Then (1.8) holds by Proposition 1.1.5 and (1.9) follows from the 2-convexity of Y. Let \mathcal{M} be the pointwise multiplication operators as in (2.1) and define

$$\Gamma_0 := \left(\frac{1}{2\|\Gamma\|_{\ell^2}} \cdot \Gamma\right) \cup \left(\frac{1}{2} \cdot \mathcal{M}\right).$$

Then Γ_0 is α -bounded with $\|\Gamma_0\|_{\ell^2} \leq 1$ by Proposition 1.2.3. Applying Lemma 1.2.5 to Γ_0 , we can find a Γ - and \mathcal{M} -invariant subspace $Y \subseteq X_0 \subseteq X$ and a Hilbertian seminorm $\|\cdot\|_0$ such that

$$\begin{split} \|Tx\|_{0} &\leq 4 \, \|\Gamma\|_{\ell^{2}} \|x\|_{0}, & x \in X_{0}, \, T \in \Gamma, \\ \|Tx\|_{0} &\leq 4 \, \|x\|_{0}, & x \in X_{0}, \, T \in \mathcal{M}, \\ \|x\|_{0} &\leq 4 \, \|x\|_{Y}, & x \in Y, \\ \|y_{1}\|_{0} &\geq \|y_{1}\|_{X}. \end{split}$$

The second property implies that if $x \in L^0(S)$ and $\tilde{x} \in X_0$ with $|x| \leq |\tilde{x}|$, then $x \in X_0$ and $||x||_0 \leq 4 ||\tilde{x}||_0$. Thus we may, at the loss of a numerical constant, furthermore assume

(2.12)
$$||x_1 + x_2||_0^2 = ||x_1||_0^2 + ||x_2||_0^2, \qquad x_1, x_2 \in X : x_1 \land x_2 = 0$$

by Lemma 2.2.2.

Define a measure $\lambda(E) = ||u \mathbf{1}_E||_0^2$ for all $E \in \Sigma$. Using (2.12), the σ -additivity of this measure follows from

$$\lambda\left(\bigcup_{k=1}^{\infty} E_k\right) = \lim_{n \to \infty} \left\|\sum_{k=1}^n u \,\mathbf{1}_{E_k}\right\|_0^2 = \sum_{k=1}^{\infty} \lambda(E_k)$$

for $E_1, E_2, \ldots \in \Sigma$ pairwise disjoint, since $u \mathbf{1}_E \in Y$ for any $E \in \Sigma$ and Y is order-continuous. Moreover we have for any $E \in \Sigma$ with $\mu(E) = 0$ that

$$\lambda(E) = \|u \mathbf{1}_E\|_0^2 \lesssim \|u \mathbf{1}_E\|_Y^2 = \|\mathbf{1}_{\varnothing}\|_Y^2 = 0$$

so λ is absolutely continuous with respect to μ . Thus, by the Radon-Nikodym theorem, we can find an $f \in L^1(S)$ such that

$$\|u \mathbf{1}_E\|_0^2 = \lambda(E) = \int_E f \,\mathrm{d}\mu$$

for all $E \in \Sigma$. Define the weight $w := u^{-2} f$.

Take $x \in Y$ and let $(v_n)_{n=1}^{\infty}$ be a sequence functions of the form

$$v_n = u \sum_{j=1}^{m_n} a_j^n \mathbf{1}_{E_j^n}, \qquad a_j^n \in \mathbb{C}, \ E_j^n \in \Sigma,$$

such that $|v_n| \uparrow |x|$. Then $\lim_{n\to\infty} ||v_n - x||_0 = 0$ by the order-continuity of Y. So by (2.12) and the monotone convergence theorem

$$\|x\|_{0}^{2} = \lim_{n \to \infty} \sum_{j=1}^{m_{n}} |a_{j}^{n}|^{2} \|u \mathbf{1}_{E_{j}^{n}}\|_{0}^{2} = \lim_{n \to \infty} \sum_{j=1}^{m_{n}} \int_{E_{j}^{n}} |a_{j}^{n}|^{2} u^{2} w \,\mathrm{d}\mu = \int_{S} |x|^{2} w \,\mathrm{d}\mu.$$

In particular, $y_0, y_1 \in L^2(S, w)$ and, using (2.11), we have

$$\left(\int_{S} |y_{0}|^{2} w \, \mathrm{d}\mu\right)^{\frac{1}{2}} = \|y_{0}\|_{0} \le 16 \, \|y_{0}\|_{Y} \le 48 \, \|y_{0}\|_{X},$$
$$\left(\int_{S} |y_{1}|^{2} w \, \mathrm{d}\mu\right)^{\frac{1}{2}} = \|y_{1}\|_{0} \ge \frac{1}{4} \, \|y_{1}\|_{X},$$

so we can take c = 48. Take $T \in \Gamma$ and $x \in Y$ and define $m_n = \min(1, nu \cdot |Tx|^{-1})$ for $n \in \mathbb{N}$. Then $m_n \cdot Tx \in Y$ and $|m_n \cdot Tx| \uparrow |Tx|$. So by the monotone convergence theorem we have

$$\left(\int_{S} |Tx|^{2} w \, \mathrm{d}\mu\right)^{\frac{1}{2}} = \lim_{n \to \infty} \left(\int_{S} |m_{n} \cdot Tx|^{2} w \, \mathrm{d}\mu\right)^{\frac{1}{2}}$$
$$= \lim_{n \to \infty} ||m_{n} \cdot Tx||_{0}$$
$$\leq 4^{6} ||\Gamma||_{\ell^{2}} \left(\int_{S} |x|^{2} w \, \mathrm{d}\mu\right)^{\frac{1}{2}}.$$

To conclude, note that Y is dense in $X \cap L^2(S, w)$ by order-continuity. Therefore, since T is bounded on X as well, this estimate extends to all $x \in X \cap L^2(S, w)$. This means that (2.8)-(2.10) hold with $C \leq 4^6 \|\Gamma\|_{\ell^2}$.

REMARK 2.3.2. In the view of Theorem 1.4.6 and Theorem 2.2.3 we may replace the assumption that Γ is ℓ^2 -bounded by the assumption that $\Gamma \cup \mathcal{M}$ is C^* -bounded in Theorem 2.3.1.

The role of 2-convexity. If the Banach function space X is 2-convex, i.e. if

$$\left\| \left(\sum_{k=1}^{n} |x_k|^2 \right)^{1/2} \right\|_X \le \left(\sum_{k=1}^{n} \|x_k\|_X^2 \right)^{1/2}, \qquad \mathbf{x} \in X^n,$$

we do not have to construct a 2-convex Banach function space Y as we did in the proof of Theorem 2.3.1. Instead, we can just use X in place of Y, which yields more stringent conditions on the weight in Theorem 2.3.1.

THEOREM 2.3.3. Let X be an order-continuous, 2-convex Banach function space on (S, Σ, μ) and let $\Gamma \subseteq \mathcal{L}(X)$. Then Γ is ℓ^2 -bounded if and only if there exists a constant C > 0 such that for any weight w with

$$||x||_{L^2(S,w)} \le ||x||_X \qquad x \in X,$$

there exists a weight $v \ge w$ such that

$$\begin{aligned} \|Tx\|_{L^{2}(S,v)} &\leq C \, \|x\|_{L^{2}(S,v)} & x \in X, \, T \in \Gamma \\ \|x\|_{L^{2}(S,v)} &\leq c \, \|x\|_{X} & x \in X, \end{aligned}$$

where c is a numerical constant. Moreover C > 0 can be chosen such that $\|\Gamma\|_{\ell^2} \simeq C$.

PROOF. The proof is similar to, but simpler than, the proof of Theorem 2.3.1. The a priori given weight w allows us to define $F: X \to [0, \infty)$ as

$$F(x) = \left(\int_S |x|^2 w \,\mathrm{d}\mu\right)^{1/2}$$

and the 2-convexity allows us to use Y = X and define $G : X \to [0, \infty)$ as $G(x) = ||x||_X$. For more details, see [Lor16, Theorem 4.6.3]

REMARK 2.3.4. Theorem 2.3.3 is closely related to the work of Rubio de Francia, which was preceded by the factorization theory of Nikišin [Nik70] and Maurey [Mau73]. In his work Rubio de Francia proved Theorem 2.3.3 with all 2's replaced by any $q \in [1, \infty)$ for the following special cases:

• For $X = L^p(S)$ in [**Rub82**],

- - -

• For $\Gamma = \{T\}$ with $T \in \mathcal{L}(X)$ in [**Rub86**, III Lemma 1],

see also [GR85]. These results have been combined in [ALV19, Lemma 3.4], yielding Theorem 2.3.3 with all 2's replaced by any $q \in [1, \infty)$. These results are proven using different techniques and for $q \neq 2$, as discussed in Remark 2.2.5, seem out of reach using our approach.

2.4. Banach function space-valued extensions of operators

In this final section on the ℓ^2 -structure on Banach function spaces we will apply Theorem 2.3.1 to obtain an extension theorem in the spirit of Rubio de Francia's extension theorem for Banach function space-valued functions (see [**Rub86**, Theorem 5]). We will apply this theorem to deduce the following results related to the UMD property for Banach function spaces:

- We will provide a quantitative proof of the boundedness of the lattice Hardy-Littlewood maximal function on UMD Banach function spaces.
- We will show that the so-called dyadic UMD⁺ property is equivalent to the UMD property for Banach function spaces.
- We will show that the UMD property is necessary for the ℓ^2 -sectoriality of certain differentiation operators on $L^p(\mathbb{R}^d; X)$, where X is a Banach function space.

Tensor extensions and Muckenhoupt weights. Let us first introduce the notions we need to state the main theorem of this section. Let $p \in [1, \infty)$, w a weight on \mathbb{R}^d and suppose that T is a bounded linear operator on $L^p(\mathbb{R}^d, w)$. We may define a linear operator \widetilde{T} on the tensor product $L^p(\mathbb{R}^d, w) \otimes X$ by setting

$$\widetilde{T}(f \otimes x) := Tf \otimes x, \qquad f \in L^p(\mathbb{R}^d, w), \ x \in X,$$

and extending by linearity. For $p \in [1, \infty)$ the space $L^p(\mathbb{R}^d, w) \otimes X$ is dense in the Bochner space $L^p(\mathbb{R}^d, w; X)$ and it thus makes sense to ask whether \widetilde{T} extends to a bounded operator on $L^p(\mathbb{R}^d, w; X)$. If this is the case, we will denote this operator again by \widetilde{T} . For a family of operators $\Gamma \subseteq \mathcal{L}(L^p(\mathbb{R}^d, w))$ we write

$$\widetilde{\Gamma} := \{ \widetilde{T} : T \in \Gamma \}.$$

We denote the Lebesgue measure λ on \mathbb{R}^d of a measurable set $E \subseteq \mathbb{R}^d$ by |E|. For $p \in (1, \infty)$ we will say that a weight w on \mathbb{R}^d is in the *Muckenhoupt class* A_p and write $w \in A_p$ if the weight characteristic

$$[w]_{A_p} := \sup_Q \frac{1}{|Q|} \int_Q w \, \mathrm{d}\lambda \cdot \left(\frac{1}{|Q|} \int_Q w^{-\frac{1}{p-1}} \, \mathrm{d}\lambda\right)^{p-1}$$

is finite, where the supremum is taken over all cubes $Q\subseteq \mathbb{R}^d$ with sides parallel to the axes.

An abstract extension theorem. We can now state the main theorem of this section.

THEOREM 2.4.1. Let X be an order-continuous Banach function space on (S, Σ, μ) , let $p \in (1, \infty)$ and $w \in A_p$. Assume that there is a family of operators $\Gamma \subseteq \mathcal{L}(L^p(\mathbb{R}^d, w))$ and an increasing function $\phi \colon \mathbb{R}_+ \to \mathbb{R}_+$ such that

• For all weights $v \colon \mathbb{R}^d \to (0, \infty)$ we have

$$[v]_{A_2} \le \phi \Big(\sup_{T \in \Gamma} \|T\|_{\mathcal{L}(L^2(\mathbb{R}^d, v))} \Big).$$

• $\widetilde{\Gamma}$ is ℓ^2 -bounded on $L^p(\mathbb{R}^d, w; X)$.

Let $f, g \in L^p(\mathbb{R}^d, w; X)$ and suppose that there is an increasing function $\psi \colon \mathbb{R}_+ \to \mathbb{R}_+$ such that for all $v \in A_2$ we have

$$\|f(\cdot,s)\|_{L^2(\mathbb{R}^d,v)} \le \psi([v]_{A_2}) \|g(\cdot,s)\|_{L^2(\mathbb{R}^d,v)}, \qquad s \in S.$$

Then there is a numerical constant c > 0 such that

$$\|f\|_{L^p(\mathbb{R}^d,w;X)} \le c \cdot \psi \circ \phi(c \,\|\Gamma\|_{\ell^2}) \|g\|_{L^p(\mathbb{R}^d,w;X)}.$$

One needs to take care when considering $f(\cdot, s)$ for $f \in L^p(\mathbb{R}^d, w; X)$ and $s \in S$ in Theorem 2.4.1, as this is not necessarily a function in $L^2(\mathbb{R}^d, v)$. This technicality can in applications be circumvented by only using e.g. simple functions or smooth functions with compact support and a density argument.

PROOF OF THEOREM 2.4.1. Let $u \in L^p(\mathbb{R}^d, w)$ be such that there is a $c_K > 0$ with $u \ge c_K \mathbf{1}_K$ for every compact $K \subseteq \mathbb{R}^d$. Let $x \in X$ be such that x > 0 a.e. and

$$\|u \otimes x\|_{L^p(\mathbb{R}^d, w; X)} \le \|g\|_{L^p(\mathbb{R}^d, w; X)}.$$

Since X is order-continuous, $L^p(\mathbb{R}^d, w; X)$ is an order-continuous Banach function space over the measure space $(\mathbb{R}^d \times S, w \, d\lambda \, d\mu)$, so by Theorem 2.3.1 we can find a weight v on $\mathbb{R}^d \times S$ and a numerical constant c > 0 such that for all $T \in \widetilde{\Gamma}$ and $h \in L^p(\mathbb{R}^d, w; X) \cap L^2(\mathbb{R}^d \times S, v \cdot w)$

(2.13)
$$||Th||_{L^2(\mathbb{R}^d \times S, v \cdot w)} \le c ||\Gamma||_{\ell^2} ||h||_{L^2(\mathbb{R}^d \times S, v \cdot w)}$$

(2.14)
$$\||g| + u \otimes x\|_{L^2(\mathbb{R}^d \times S, v \cdot w)} \le c \||g| + u \otimes x\|_{L^p(\mathbb{R}^d, w; X)},$$

(2.15)
$$||f||_{L^2(\mathbb{R}^d \times S, v \cdot w)} \ge \frac{1}{c} ||f||_{L^p(\mathbb{R}^d, w; X)}.$$

Note that (2.14) and the definition of x imply

(2.16)
$$||g||_{L^2(\mathbb{R}^d \times S, v \cdot w)} \le 2c \, ||g||_{L^p(\mathbb{R}^d, w; X)}$$

and $u \in L^2(\mathbb{R}^d, v(\cdot, s) \cdot w)$ for μ -a.e. $s \in S$. Therefore, by the definition of u, we know that $v(\cdot, s) \cdot w$ is locally integrable on \mathbb{R}^d for μ -a.e. $s \in S$. Let \mathcal{A} be the \mathbb{Q} -linear span of indicator functions of rectangles with rational corners, which is a countable, dense subset of both $L^p(\mathbb{R}^d, w)$ and $L^2(\mathbb{R}^d, v(\cdot, s) \cdot w)$ for μ -a.e. $s \in S$. Define

$$\mathcal{B} = \{ \psi \otimes (x \mathbf{1}_E) : \psi \in \mathcal{A}, E \in \Sigma \}.$$

Then $\mathcal{B} \subseteq L^p(\mathbb{R}^d, w; X) \cap L^2(\mathbb{R}^d \times S, v \cdot w)$ since $u \otimes x \in L^2(\mathbb{R}^d \times S, v \cdot w)$. Testing (2.13) on all $h \in \mathcal{B}$ we find that for all $T \in \Gamma$ and $\psi \in \mathcal{A}$

$$\|T\psi\|_{L^2(\mathbb{R}^d, v(\cdot, s) \cdot w)} \le c \, \|\Gamma\|_{\ell^2} \, \|\psi\|_{L^2(\mathbb{R}^d, v(\cdot, s) \cdot w)}, \qquad s \in S.$$

Since \mathcal{A} is countable and dense in $L^2(\mathbb{R}^d, v(\cdot, s) \cdot w)$, we have by assumption that $v(\cdot, s) \cdot w \in A_2$ with $[v(\cdot, s) \cdot w]_{A_2} \leq \phi(c \|\widetilde{\Gamma}\|_{\ell^2})$ for μ -a.e. $s \in S$. Therefore, using Fubini's theorem, our assumption, (2.15) and (2.16), we obtain

$$\begin{split} \|f\|_{L^{p}(\mathbb{R}^{d},w;X)} &\leq c \left(\int_{S} \int_{\mathbb{R}^{d}} |f|^{2} v \cdot w \, \mathrm{d}\lambda \, \mathrm{d}\mu \right)^{1/2} \\ &\leq c \cdot \psi \circ \phi \big(c \, \|\widetilde{\Gamma}\|_{\ell^{2}} \big) \Big(\int_{S} \int_{\mathbb{R}^{d}} |g|^{2} v \cdot w \, \mathrm{d}\lambda \, \mathrm{d}\mu \Big)^{1/2} \\ &\leq 2c^{2} \cdot \psi \circ \phi \big(c \, \|\widetilde{\Gamma}\|_{\ell^{2}} \big) \|g\|_{L^{p}(\mathbb{R}^{d},w;X)}, \end{split}$$

proving the statement.

We say that a Banach space X has the UMD property if the martingale difference sequence of any finite martingale in $L^p(S; X)$ on a σ -finite measure space (S, μ) is unconditional for some (equivalently all) $p \in (1, \infty)$. That is, if for all finite martingales $(f_k)_{k=1}^n$ in $L^p(S; X)$ and scalars $|\epsilon_k| = 1$ we have

(2.17)
$$\left\|\sum_{k=1}^{n} \epsilon_k df_k\right\|_{L^p(S;X)} \le C \left\|\sum_{k=1}^{n} df_k\right\|_{L^p(S;X)}.$$

The least admissible constant C > 0 in (2.17) will be denoted by $\beta_{p,X}$. For a detailed account of the theory UMD Banach spaces we refer the reader to [**HNVW16**, Chapter 4] and [**Pis16**].

Let us point out some choices of $\Gamma \subseteq \mathcal{L}(L^p(\mathbb{R}^d, w))$ that satisfy the assumptions Theorem 2.4.1 when X has UMD property:

- $\Gamma = \{H\}$, where *H* is the Hilbert transform.
- $\Gamma = \{R_k : k = 1, \dots, d\}$ where R_k is the k-th Riesz projection.
- $\Gamma := \{T_Q : Q \text{ a cube in } \mathbb{R}^d\}$, where $T_Q : L^p(\mathbb{R}^d) \to L^p(\mathbb{R}^d)$ is the averaging operator

$$T_Q f(t) := \left(\frac{1}{|Q|} \int_Q f \, \mathrm{d}\lambda\right) \mathbf{1}_Q(t), \qquad t \in \mathbb{R}^d.$$

We will encounter these choices of Γ in the upcoming applications of Theorem 2.4.1. For these choices of Γ one obtains an extension theorem for UMD Banach function spaces in the spirit of [**Rub86**, Theorem 5].

COROLLARY 2.4.2. Let X be a UMD Banach function space and let T be a bounded linear operator on $L^{p_0}(\mathbb{R}^d, v)$ for some $p_0 \in (1, \infty)$ and all $v \in A_{p_0}$. Suppose that there is an increasing function $\phi \colon \mathbb{R}_+ \to \mathbb{R}_+$ such that

$$||T||_{L^{p_0}(\mathbb{R}^d, v) \to L^{p_0}(\mathbb{R}^d, v)} \le \phi([v]_{A_{p_0}}), \qquad v \in A_{p_0}.$$

Then \widetilde{T} extends uniquely to a bounded linear operator on $L^p(\mathbb{R}^d, w; X)$ for all $p \in (1, \infty)$ and $w \in A_p$.

PROOF. For $k = 1, \dots, d$ let R_k denote the k-th Riesz projection on $L^p(\mathbb{R}^d, w)$ and set $\Gamma = \{R_k : k = 1, \dots, d\}$. Then we have for any weight v on \mathbb{R}^d that

$$[v]_{A_2} \lesssim_d \left(\sup_{T \in \Gamma} \|T\|_{L^2(\mathbb{R}^d, v) \to L^2(\mathbb{R}^d, v)} \right)^4.$$

by [Gra14, Theorem 7.4.7]. Moreover by the triangle inequality, the ideal property of the ℓ^2 -structure, [HNVW16, Theorem 5.5.1] and [HH14, Corollary 2.11] we

have

$$\|\widetilde{\Gamma}\|_{\ell^{2}} \lesssim \sum_{k=1}^{d} \|R_{k}\|_{L^{p}(\mathbb{R}^{d}, w; X) \to L^{p}(\mathbb{R}^{d}, w; X)} \lesssim_{X, p, d} [w]_{A_{p}}^{\max\{\frac{1}{p-1}, 1\}}$$

Thus Γ satisfies the assumptions of Theorem 2.4.1. Let $f \in L^p(\mathbb{R}^d, w) \otimes X$. By Rubio de Francia extrapolation (see [**CMP12**, Theorem 3.9]) there is an increasing function $\psi \colon \mathbb{R}_+ \to \mathbb{R}_+$, depending on ϕ, p, p_0, d , such that for all $v \in A_2$ we have

$$\|Tf(\cdot,s)\|_{L^{2}(\mathbb{R}^{d},v)} \leq \psi([v]_{A_{2}})\|f(\cdot,s)\|_{L^{2}(\mathbb{R}^{d},v)}, \qquad s \in S.$$

Therefore by Theorem 2.4.1 we obtain

$$\|Tf\|_{L^{p}(\mathbb{R}^{d},w;X)} \leq c \cdot \psi \left(C_{X,p,d} \cdot [w]_{A_{p}}^{4 \cdot \max\{\frac{1}{p-1},1\}} \right) \|f\|_{L^{p}(\mathbb{R}^{d},w;X)}$$

which yields the desired result by density.

The advantages of Theorem 2.4.1 and Corollary 2.4.2 over [Rub86, Theorem 5] are as follows

- Theorem 2.4.1 yields a quantitative estimate of the involved constants, whereas this dependence is hard to track in [Rub86, Theorem 5].
- Theorem 2.4.1 and Corollary 2.4.2 allow weights in the conclusion, whereas [**Rub86**, Theorem 5] only yields an unweighted extension.
- [Rub86, Theorem 5] relies upon the boundedness of the lattice Hardy-Littlewood maximal operator on $L^p(\mathbb{R}^d; X)$, whereas this is not used in the proof of Theorem 2.4.1. Therefore, we can use Theorem 2.4.1 to give a quantitative proof of the boundedness of the lattice Hardy-Littlewood maximal operator on UMD Banach function spaces, see Theorem 2.4.4.
- Instead of assuming the UMD property of X, the assumptions of Theorem 2.4.1 are flexible enough to allow one to deduce the UMD property of X from ℓ^2 -boundedness of other operators, see Theorem 2.4.9.

REMARK 2.4.3. Rubio de Francia's extension theorem for UMD Banach function spaces has also been generalized in [ALV19, LN19, LN20]:

- In [ALV19, Corollary 3.6] a rescaled version of Corollary 2.4.2 has been obtained by adapting the original proof of Rubio de Francia.
- In [LN19] the proof of [Rub86, Theorem 5] has been generalized to allow for a multilinear limited range variant. The proof of Theorem 2.4.1 does not lend itself for such a generalization.
- Using the stronger assumption of sparse domination, the result in [LN19] has been made quantitative and has been extended to multilinear weight and UMD classes in [LN20].

The lattice Hardy–Littlewood maximal operator. As a first application of Theorem 2.4.1, we will show the boundedness of the lattice Hardy–Littlewood maximal operator on UMD Banach function spaces. Let X be an order-continuous Banach function space and $p \in (1, \infty)$. For $f \in L^p(\mathbb{R}^d; X)$ the lattice Hardy– Littlewood maximal operator is defined as

$$\widetilde{M}f(t) := \sup_{Q \ni t} \left(\frac{1}{|Q|} \int_{Q} |f| \, \mathrm{d}\lambda \right), \qquad t \in \mathbb{R}^{d},$$

where the supremum is taken in the lattice sense over cubes $Q \subseteq \mathbb{R}^d$ containing t (see [GMT93] or [HL19, Section 5] for the details). The boundedness of the

lattice Hardy-Littlewood maximal operator for UMD Banach function spaces X is a deep result shown by Bourgain [Bou84] and Rubio de Francia [Rub86]. Using this result, the following generalizations were subsequently shown on UMD Banach function spaces:

- García-Cuerva, Macías and Torrea showed in $[\mathbf{GMT93}]$ that \widetilde{M} is bounded on $L^p(\mathbb{R}^d, w; X)$ for all Muckenhoupt weights $w \in A_p$. Sharp dependence on the weight characteristic was obtained in $[\mathbf{HL19}]$ by Hänninen and the second author.
- Deleaval, Kemppainen and Kriegler showed in $[\mathbf{DKK18}]$ that \widetilde{M} is bounded on $L^p(S; X)$ for any space of homogeneous type S.
- Deleaval and Kriegler obtained dimension free estimates for a centered version of *M* on L^p(R^d; X) in [**DK19**].

With Theorem 2.4.1 we can reprove the result of Bourgain and Rubio de Francia and obtain an explicit estimate of the operator norm of \widetilde{M} in terms of the UMD constant of X. Tracking this dependence in the proof of Bourgain and Rubio de Francia would be hard, as it involves the weight characteristic dependence of the inequality [**Rub86**, (a.5)].

THEOREM 2.4.4. Let X be a UMD Banach function space with cotype $q \in (1, \infty)$ with constant $c_{q,X}$. The lattice Hardy-Littlewood maximal operator \widetilde{M} is bounded on $L^p(\mathbb{R}^d; X)$ for all $p \in (1, \infty)$ with

$$\|\widetilde{M}\|_{\mathcal{L}(L^p(\mathbb{R}^d;X))} \lesssim q(c_{q,X}\beta_{p,X})^2,$$

where the implicit constant only depends on p and d.

PROOF. Let $p \in (1,\infty)$ and $f \in L^p(\mathbb{R}^d)$. Define for any cube $Q \subseteq \mathbb{R}^d$ the averaging operator

$$T_Q f(t) := \left(\frac{1}{|Q|} \int_Q f \, \mathrm{d}\lambda\right) \mathbf{1}_Q(t), \qquad t \in \mathbb{R}^d$$

and set $\Gamma := \{T_Q : Q \text{ a cube in } \mathbb{R}^d\}$. Then we know that $\widetilde{\Gamma}$ is ℓ^2 -bounded on $L^p(\mathbb{R}^d; X)$ with

$$\|\widetilde{\Gamma}\|_{\ell^2} \lesssim \sqrt{q} c_{q,X} \beta_{p,X}$$

by [HNVW17, Theorem 7.2.13 and Proposition 8.1.13], where the implicit constant depends on p and d.

Let $w \colon \mathbb{R}^d \to (0, \infty)$ and set $C := \sup_{T \in \Gamma} ||T||_{\mathcal{L}(L^2(\mathbb{R}^d, w))}$. Fix a cube $Q \subseteq \mathbb{R}^d$. Applying T_Q to the function $(w + \varepsilon)^{-1} \mathbf{1}_Q$ for some $\varepsilon > 0$ we obtain

$$\int_{Q} \left(\frac{1}{|Q|} \int_{Q} (w(t) + \varepsilon)^{-1} \, \mathrm{d}t \right)^{2} w(s) \, \mathrm{d}s \le C^{2} \int_{Q} \frac{w(t)}{(w(t) + \varepsilon)^{2}} \, \mathrm{d}t$$

which implies

$$\left(\frac{1}{|Q|}\int_{Q}w(t)\,\mathrm{d}t\right)\left(\frac{1}{|Q|}\int_{Q}(w(t)+\varepsilon)^{-1}\,\mathrm{d}t\right)\leq C^{2}$$

So by letting $\varepsilon \to 0$ with the monotone convergence theorem, we obtain $w \in A_2$ with $[w]_{A_2} \leq C^2$. So Γ satisfies the assumptions of Theorem 2.4.1 with $\phi(t) = t^2$.

By Theorem 2.4.1, using the weighted boundedness of the scalar-valued Hardy-Littlewood maximal operator from [Gra14, Theorem 7.1.9], we know that for any simple function $f \colon \mathbb{R}^d \to X$

$$\left\|\widetilde{M}f\right\|_{L^{p}(\mathbb{R}^{d};X)} \lesssim q\left(c_{q,X}\beta_{p,X}\right)^{2} \|f\|_{L^{p}(\mathbb{R}^{d};X)}.$$

where the implicit constant depends on p and d. So, by the density of the simple functions in $L^p(\mathbb{R}^d; X)$, we obtain the desired result.

Remark 2.4.5.

- One could also use $\Gamma = \{H\}$ or $\Gamma = \{R_k : k = 1, \dots, d\}$ in the proof of Theorem 2.4.4, where H is the Hilbert transform and R_k is the k-th Riesz projection. Then the first assumption on Γ in Theorem 2.4.1 follows from [Gra14, Theorem 7.4.7] and the second from [HNVW16, Theorem 5.1.1 and 5.5.1] and the ideal property of the ℓ^2 -structure.
- In Theorem 2.4.4 the assumption that X has finite cotype may be omitted, since the UMD property implies that there exists a constant $C_p > 0$ such that X has cotype $C_p \beta_{p,X}$ with constant less than C_p (see [HLN16, Lemma 32]). This yields the bound $\|\widetilde{M}\|_{\mathcal{L}(L^p(\mathbb{R}^d;X))} \lesssim \beta_{p,X}^3$ in the conclusion of Theorem 2.4.4.
- One would be able to avoid the cotype constant in the conclusion of Theorem 2.4.4 if one can find a single operator T that both characterizes $v \in A_p$ with $\phi(t) = t^2$ and is bounded on $L^p(\mathbb{R}^d; X)$ with $||T||_{\mathcal{L}(L^p(\mathbb{R}^d; X))} \lesssim \beta_{p,X}$.

Randomized UMD **properties.** As a second application of Theorem 2.4.1 we will prove the equivalence of the UMD property and the dyadic UMD^+ property. Let us start by introducing the randomized UMD properties for a Banach space X.

We say that X has the UMD⁺ (respectively UMD⁻) property if for some (equivalently all) $p \in (1, \infty)$ there exists a constant $\beta^+ > 0$ (respectively $\beta^- > 0$) such that for all finite martingales $(f_k)_{k=1}^n$ in $L^p(S; X)$ on a σ -finite measure space (S, μ) we have

(2.18)
$$\frac{1}{\beta^{-}} \left\| \sum_{k=1}^{n} df_{k} \right\|_{L^{p}(S;X)} \leq \left\| \sum_{k=1}^{n} \varepsilon_{k} df_{k} \right\|_{L^{p}(S \times \Omega;X)} \leq \beta^{+} \left\| \sum_{k=1}^{n} df_{k} \right\|_{L^{p}(S;X)},$$

where $(\varepsilon_k)_{k=1}^n$ is a Rademacher sequence on (Ω, \mathbb{P}) . The least admissible constants in (2.18) will be denoted by $\beta_{p,X}^+$ and $\beta_{p,X}^-$. If (2.18) holds for all Paley-Walsh martingales on a probability space (S, μ) we say that X has the dyadic UMD⁺ or UMD⁻ property respectively and denote the least admissible constants by $\beta_{p,X}^{\Delta,+}$ and $\beta_{p,X}^{\Delta,-}$.

As for the UMD property, the UMD⁺ and UMD⁻ properties are independent of $p \in (1, \infty)$ (see [Gar90]). We trivially have $\beta_{p,X}^{\Delta,-} \leq \beta_{p,X}^{-}$ and $\beta_{p,X}^{\Delta,+} \leq \beta_{p,X}^{+}$. Furthermore we know that X has the UMD property if and only if it has the UMD⁺ and UMD⁻ properties with

$$\max\{\beta_{p,X}^-, \beta_{p,X}^+\} \le \beta_{p,X} \le \beta_{p,X}^- \beta_{p,X}^+,$$

see e.g. [HNVW16, Proposition 4.1.16]. The relation between the norm of the Hilbert transform on $L^p(\mathbb{T}; X)$ and $\beta_{p,X}^{\Delta,+}$ and $\beta_{p,X}^{\Delta,-}$ has recently been investigated in [OY19] and the UMD⁺ property was recently shown to be equivalent to a recoupling property for tangent martingales in [Yar20]. We refer to [HNVW16, Ver07] for further information on these randomized UMD properties.

Two natural questions regarding these randomized UMD properties are the following:

- Does either the UMD⁻ property or the UMD⁺ property imply the UMD property? For the UMD⁻ property it turns out that this is not the case, as any L^1 -space has it, see [Gar90]. For the UMD⁺ property this is an open problem. For general Banach spaces it is known that one cannot expect a better than quadratic bound relating $\beta_{p,X}$ and $\beta_{p,X}^+$ (see [Gei99, Corollary 5]).
- The dyadic UMD property implies its non-dyadic counterpart. Does the same hold for the dyadic UMD⁺ and UMD⁻ properties? For the UMD⁻ property it is known that the constants $\beta_{p,X}^{-}$ and $\beta_{p,X}^{\Delta,-}$ are not the same in general, as explained in [CV11].

Using Theorem 2.4.1, we will show that on Banach function spaces the dyadic UMD^+ property implies the UMD property (and thus also the UMD^+ property), with a quadratic estimate of the respective constants.

The equivalence of the UMD⁺ property and the UMD property on Banach function spaces has previously been shown in unpublished work of T.P. Hytönen, using Stein's inequality to deduce the ℓ^2 -boundedness of the Poisson semigroup on $L^p(\mathbb{R}^d; X)$, from which the boundedness of the Hilbert transform on $L^p(\mathbb{R}^d; X)$ was concluded using Theorem 2.3.1.

THEOREM 2.4.6. Let X be a Banach function space on (S, Σ, μ) . Assume that X has the dyadic UMD⁺ property and cotype $q \in (1, \infty)$ with constant $c_{q,X}$. Then X has the UMD property with

$$\beta_{p,X} \lesssim q \left(c_{q,X} \beta_{p,X}^{\Delta,+} \right)^2,$$

where the implicit constant only depends on $p \in (1, \infty)$.

PROOF. Denote the standard dyadic system on [0, 1) by \mathcal{D} , i.e.

$$\mathscr{D} := \bigcup_{k \in \mathbb{N}} \mathscr{D}_k, \qquad \mathscr{D}_k := \{2^{-k}([0,1)+j) : j = 0, \dots, 2^k - 1\}.$$

Then $(\mathscr{D}_k)_{k=1}^n$ is a Paley-Walsh filtration on [0,1) for all $n \in \mathbb{N}$. Let $p \in (1,\infty)$ and define

$$\Gamma := \big\{ \mathbb{E}(\cdot | \mathscr{D}_k) : k \in \mathbb{N} \big\}$$

on $L^p(0,1)$. By a dyadic version of Stein's inequality, which can be proven analogously to [**HNVW16**, Theorem 4.2.23], we have

$$\left\|\sum_{k=1}^{n}\varepsilon_{k}\mathbb{E}(f_{k}|\mathscr{D}_{k})\right\|_{L^{p}([0,1)\times\Omega;X)} \leq \beta_{p,X}^{\Delta,+}\left\|\sum_{k=1}^{n}\varepsilon_{k}f_{k}\right\|_{L^{p}([0,1)\times\Omega;X)}$$

where $(\varepsilon_k)_{k=1}^n$ is a Rademacher sequence. So by [**HNVW17**, Theorem 7.2.13] and the ideal property of the ℓ^2 -structure, we know that $\widetilde{\Gamma}$ is ℓ^2 -bounded with

(2.19)
$$\left\|\Gamma\right\|_{\ell^2} \le C \sqrt{q} c_{q,X} \beta_{p,X}^{\Delta,+},$$

where C > 0 only depends on p.

Define the dyadic weight class $A_2^{\mathscr{D}}$ as all weights w on [0, 1) such that

$$[w]_{A_2^{\mathscr{D}}} := \sup_{I \in \mathscr{D}} \frac{1}{|I|} \int_I w \, \mathrm{d}\lambda \cdot \frac{1}{|I|} \int_I w^{-1} \, \mathrm{d}\lambda < \infty.$$

Let $w: [0,1) \to (0,\infty)$ be a weight. Arguing as in Theorem 2.4.4, we know that

$$[w]_{A_2^{\mathscr{D}}} \le \left(\sup_{T \in \Gamma} \|T\|_{\mathcal{L}(L^2(w))} \right)^2.$$

Furthermore note that, with a completely analogous proof, Theorem 2.4.1 is also valid for the interval [0,1) instead of \mathbb{R}^d and using weights $v \in A_2^{\mathscr{D}}$ instead of weights $v \in A_2$. Therefore we know that if $f, g \in L^p([0,1); X)$ are such that for all $v \in A_2^{\mathscr{D}}$ we have

(2.20)
$$\|f(\cdot,s)\|_{L^2([0,1),v)} \le C \cdot [v]_{A_2^{\mathscr{D}}} \|g(\cdot,s)\|_{L^2([0,1),v)}, \qquad s \in S,$$

then it follows that

(2.21)
$$\|f\|_{L^p([0,1);X)} \le c \cdot C \cdot \|\widetilde{\Gamma}\|_{\ell^2}^2 \|g\|_{L^p([0,1);X)},$$

for some numerical constant c.

Define for every interval $I \in \mathscr{D}$ the Haar function h_I by

$$h_I := |I|^{\frac{1}{2}} (\mathbf{1}_{I_-} - \mathbf{1}_{I_+}),$$

where I_+ and I_- are the left and right halve of I. For $f \in L^p([0,1); X)$ define the Haar projection D_I by

$$D_I f(t) := h_I(t) \int_0^1 f(s) h_I(s) \, \mathrm{d}s$$

Let \mathcal{A} be the set of all simple functions $f \in L^p([0,1); X)$ such that $D_I f \neq 0$ for only finitely many $I \in \mathcal{D}$. Then for all $f \in \mathcal{A}$, $w \in A_2^{\mathcal{D}}$ and $\epsilon_I \in \{-1, 1\}$ we have

$$\left\|\sum_{I\in\mathscr{D}}\epsilon_I D_I f(\cdot,\omega)\right\|_{L^2([0,1),w)} \lesssim [w]_{A_2^{\mathscr{D}}} \|f(\cdot,s)\|_{L^2([0,1),w)}, \qquad s\in S$$

by [Wit00], so (2.20) is satisfied. Therefore, using (2.19) and (2.21), we obtain that

(2.22)
$$\left\| \sum_{I \in \mathscr{D}} \epsilon_I D_I f \right\|_{L^p([0,1);X)} \le C q \left(c_{q,X} \beta_{p,X}^{\Delta,+} \right)^2 \|f\|_{L^p([0,1);X)}$$

for all $f \in \mathcal{A}$ and $\epsilon_I \in \{-1, 1\}$. Note that \mathcal{A} is dense in $L^p([0, 1); X)$ by [**HNVW16**, Lemma 4.2.12] and we may take $\epsilon_I \in \mathbb{C}$ with $|\epsilon_I| = 1$ by the triangle inequality. So

$$\beta_{p,X} \le C q \left(c_{q,X} \beta_{p,X}^{\Delta,+} \right)^2$$

as (2.22) characterizes the UMD property of X by [HNVW16, Theorem 4.2.13]. \Box

Remark 2.4.7.

- As in Remark 2.4.5, the assumption that X has finite cotype may be omitted in Theorem 2.4.6. This would yield the bound $\beta_{p,X} \leq C_p \left(\beta_{p,X}^{\Delta,+}\right)^3$ for all $p \in (1,\infty)$ in the conclusion of Theorem 2.4.6.
- A similar argument as in the proof of Theorem 2.4.6 can be used to show Theorem 2.4.4 with the sharper estimate

$$\|\widetilde{M}\|_{\mathcal{L}(L^p(\mathbb{R}^d;X))} \lesssim q \left(c_{q,X}\beta_{p,X}^{+,\Delta}\right)^2.$$

 ℓ^2 -sectoriality and the UMD property. For the ℓ^2 -structure on a Banach function space X we say that a sectorial operator A on X is ℓ^2 -sectorial if the resolvent set

$$\{\lambda R(\lambda, A) : \lambda \neq 0, |\arg \lambda| > \sigma\}$$

is ℓ^2 -bounded for some $\sigma \in (0, \pi)$. We will introduce α -sectorial operators properly in Chapter 4.

It is well-known that both the differentiation operator Df := f' with domain $W^{1,p}(\mathbb{R}; X)$ and the Laplacian $-\Delta$ with domain $W^{2,p}(\mathbb{R}^d; X)$ are \mathcal{R} -sectorial, and thus ℓ^2 -sectorial, if X has the UMD property (see [**KW04**, Example 10.2] and [**HNVW17**, Theorem 10.3.4]). Using Theorem 2.4.1 we can turn this into an 'if and only if' statement for order-continuous Banach function spaces.

LEMMA 2.4.8. Let $0 \neq \varphi \in L^1(\mathbb{R}^d) \cap L^2(\mathbb{R}^d)$ be real-valued and let w be a weight on \mathbb{R}^d . Suppose that there is a C > 0 such that for all $f \in L^2(\mathbb{R}^d, w)$ and $\lambda \in \mathbb{R}$ we have

$$\|\varphi_{\lambda} * f\|_{L^{2}(\mathbb{R}^{d},w)} \leq C \|f\|_{L^{2}(\mathbb{R}^{d},w)}$$

where $\varphi_{\lambda}(t) := |\lambda|^{d} \varphi(\lambda t)$ for $t \in \mathbb{R}^{d}$. Then $w \in A_{2}$ and $[w]_{A_{2}} \leq C^{4}$, where the implicit constant depends on φ and d

PROOF. Let $\psi = \varphi_{-1} * \varphi$. Then $\psi(-t) = \psi(t)$ for all $t \in \mathbb{R}^d$ and $\psi(0) = \|\varphi\|_{L^2(\mathbb{R}^d)}^2 > 0$. Moreover

$$\|\psi\|_{L^{\infty}(\mathbb{R}^d)} \le \|\varphi\|_{L^2(\mathbb{R}^d)}^2,$$

so ψ is continuous by the density of $C_c(\mathbb{R}^d)$ in $L^2(\mathbb{R}^d)$. Therefore we can find a $\delta > 0$ such that $\psi(t) > \delta$ for all $|t| < \delta$. Define $\psi_{\lambda}(t) := \lambda^d \psi(\lambda t)$ for $\lambda > 0$. Then we have for all $f \in L^1(\mathbb{R}^d) \cap L^2(\mathbb{R}^d, w)$ that

$$\|\psi_{\lambda} * f\|_{L^{2}(\mathbb{R}^{d},w)} = \|\varphi_{-\lambda} * \varphi_{\lambda} * f\|_{L^{2}(\mathbb{R}^{d},w)} \le C^{2} \|f\|_{L^{2}(\mathbb{R}^{d},w)}$$

Now let Q be a cube in \mathbb{R}^d and let $f \in L^1(\mathbb{R}^d) \cap L^2(\mathbb{R}^d, w)$ be nonnegative and supported on Q. Take $\lambda = \frac{\delta}{\operatorname{diam}(Q)}$, then for $t \in Q$

$$\psi_{\lambda} * f(t) = \lambda^d \int_Q \psi(\lambda(t-s)) f(s) \, \mathrm{d}s \ge \frac{\delta^{d+1}}{|Q|} \int_Q f(s) \, \mathrm{d}s.$$

So by the same reasoning as in the proof of Theorem 2.4.4, we have $w \in A_2$ with $[w]_{A_2} \leq C^4$ with an implicit constant depending on φ, d .

Using Lemma 2.4.8 to check the weight condition of Theorem 2.4.1, the announced theorem follows readily.

THEOREM 2.4.9. Let X be an order-continuous Banach function space and let $p \in (1, \infty)$. The following are equivalent:

- (i) X has the UMD property.
- (ii) The differentiation operator D on $L^p(\mathbb{R}; X)$ is ℓ^2 -sectorial.
- (iii) The Laplacian $-\Delta$ on $L^p(\mathbb{R}^d; X)$ is ℓ^2 -sectorial.

PROOF. We have already discussed the implications (i) \Rightarrow (ii) and (i) \Rightarrow (iii). We will prove (iii) \Rightarrow (i), the proof of (ii) \Rightarrow (i) being similar. Take $\lambda \in \mathbb{R}$ and define the operators

$$T_{\lambda} := -\lambda^2 \Delta (1 - \lambda^2 \Delta)^{-2} = -\Delta R \left(-\frac{1}{\lambda^2}, -\Delta \right) \cdot \frac{1}{\lambda^2} R \left(-\frac{1}{\lambda^2}, -\Delta \right).$$

Since $-\Delta$ is ℓ^2 -sectorial on $L^p(\mathbb{R}^d; X)$, we know that the family of operators $\widetilde{\Gamma} = \{\widetilde{T}_{\lambda} : \lambda \in \mathbb{R}\}$ is ℓ^2 -bounded on $L^p(\mathbb{R}^d; X)$. Furthermore we have for $f \in L^2(\mathbb{R}^d)$ that $T_1 f = \varphi * f$ with $\varphi \in L^1(\mathbb{R}^d) \cap L^2(\mathbb{R}^d)$ such that

$$\hat{\varphi}(\xi) = \frac{(2\pi|\xi|)^2}{\left(1 + (2\pi|\xi|)^2\right)^2}, \qquad \xi \in \mathbb{R}^d$$

Moreover $T_{\lambda}f = \varphi_{\lambda} * f$ for $\varphi_{\lambda}(x) = \lambda^{d}\varphi(\lambda x)$ and $\lambda \in \mathbb{R}$. Using Lemma 2.4.8 this implies that the assumptions of Theorem 2.4.1 are satisfied.

Now by Theorem 2.4.1 and the boundedness of the Riesz projections on $L^2(\mathbb{R}^d, w)$ for all $w \in A_2$ (see [**Pet08**]), we find that for all $f \in C_c^{\infty}(\mathbb{R}^d) \otimes X$

$$||R_k f||_{L^p(\mathbb{R}^d;X)} \lesssim ||\Gamma||_{\ell^2}^4 ||f||_{L^p(\mathbb{R}^d;X)}, \qquad k = 1, \dots, d.$$

So, by the density of $C_c^{\infty}(\mathbb{R}^d) \otimes X$ in $L^p(\mathbb{R}^d; X)$, the Riesz projections are bounded on $L^p(\mathbb{R}^d; X)$, which means that X has the UMD property by [**HNVW16**, Theorem 5.5.1].

The proof scheme of Theorem 2.4.9 can be adapted to various other operators. We mention two examples:

- In [Lor19] it was shown that the UMD property is sufficient for the ℓ^2 boundedness of a quite broad class of convolution operators on $L^p(\mathbb{R}^d; X)$. Using a similar proof as the one presented in Theorem 2.4.9, one can show that the UMD property of the Banach function space X is necessary for the ℓ^2 -boundedness of these operators.
- On general Banach spaces X we know by a result of Coulhon and Lamberton [CL86] (quantified by Hytönen [Hyt15]), that the maximal L^p-regularity of (-Δ)^{1/2} implies that X has the UMD property. Maximal L^p-regularity implies the *R*-sectoriality of (-Δ)^{1/2} on L^p(ℝ^d; X) by a result of Clément and Prüss [CP01] and the converse holds if X has the UMD property by [Wei01b]. It is therefore a natural question to ask whether the *R*-sectoriality of (-Δ)^{1/2} on L^p(ℝ^d; X) also implies that X has the UMD property. By the equivalence of *R*-sectoriality and ℓ²-sectoriality on Banach lattices with finite cotype, we can show that this is indeed the case for Banach function spaces with finite cotype, using a similar proof as in the proof of Theorem 2.4.9. The question for general Banach spaces remains open. This is also the case for the question whether the *R*-sectoriality of -Δ on L^p(ℝ^d; X) implies that X has the UMD property, see [HNVW17, Problem 7].

CHAPTER 3

Vector-valued function spaces and interpolation

In Chapter 1 we treated Euclidean structures as a norm on the space of functions from $\{1, \ldots, n\}$ to X or as a norm on the space of operators from ℓ_n^2 to X for each $n \in \mathbb{N}$. In this chapter we will extend this norm to include functions from an arbitrary measure space (S, μ) to X and to operators from an arbitrary Hilbert space H to X. After introducing the relevant concepts, we will study the properties of the so-defined function spaces $\alpha(S; X)$ and operator spaces $\alpha(H, X)$. Their most important property is that every bounded operator on $L^2(S)$, e.g. the Fourier transform or a singular integral operator on $L^2(\mathbb{R}^d)$, extends automatically to a bounded operator on the X-valued function space $\alpha(S; X)$ for any Banach space X. This is in stark contrast to the situation for the Bochner spaces $L^2(S; X)$ and greatly simplifies analysis for vector-valued functions in these spaces.

In the second halve of this chapter we will develop an interpolation method based on these vector-valued function spaces. A charming feature of this α -interpolation method is that its formulations modelled after the real and the complex interpolation methods are equivalent. The α -interpolation method can therefore be seen as a way to keep strong interpolation properties of Hilbert spaces in a Banach space context.

As a standing assumption throughout this chapter and the subsequent chapters we suppose that α is a Euclidean structure on X.

3.1. The spaces $\alpha(H, X)$ and $\alpha(S; X)$

Our first step is to extend the definition of the α -norm to infinite vectors. For an infinite vector **x** with entries in X we define

$$\|\mathbf{x}\|_{\alpha} = \sup_{n \in \mathbb{N}} \|(x_1, \dots, x_n)\|_{\alpha}.$$

We then define $\alpha_+(\mathbb{N}; X)$ as the space of all infinite column vectors \mathbf{x} such that $\|\mathbf{x}\|_{\alpha} < \infty$ and let $\alpha(\mathbb{N}; X)$ be the subspace of $\alpha_+(\mathbb{N}; X)$ consisting of all $\mathbf{x} \in \alpha_+(\mathbb{N}; X)$ such that

$$\lim_{n \to \infty} \| (0, \dots, 0, x_{n+1}, x_{n+2}, \dots) \|_{\alpha} = 0.$$

Proposition 1.1.5 shows that if $\mathbf{x} \in \alpha_+(\mathbb{N}; X)$ has finite dimensional range, then $\mathbf{x} \in \alpha(\mathbb{N}; X)$. This leads to following characterization of $\alpha(\mathbb{N}; X)$.

PROPOSITION 3.1.1. Let $\mathbf{x} \in \alpha_+(\mathbb{N}; X)$. Then $\mathbf{x} \in \alpha(\mathbb{N}; X)$ if and only if there exists an sequence $(\mathbf{x}^k)_{k=1}^{\infty}$ with finite dimensional range such that $\lim_{k\to\infty} ||\mathbf{x} - \mathbf{x}^k||_{\alpha} = 0$.

From Proposition 3.1.1 and Property (1.2) of a Euclidean structure we obtain directly the important fact that every bounded operator on ℓ^2 extends to a bounded operator on $\alpha(\mathbb{N}; X)$ and $\alpha_+(\mathbb{N}; X)$. PROPOSITION 3.1.2. If $\mathbf{x} \in \alpha_+(\mathbb{N}; X)$ and \mathbf{A} is an infinite matrix representing a bounded operator on ℓ^2 , then $\mathbf{A}\mathbf{x} \in \alpha_+(\mathbb{N}; X)$ with

$$\|\mathbf{A}\mathbf{x}\|_{lpha} \leq \|\mathbf{A}\|\|\mathbf{x}\|_{lpha}$$

If either $\mathbf{x} \in \alpha(\mathbb{N}; X)$ or \mathbf{A} represents a compact operator on ℓ^2 , then $\mathbf{A}\mathbf{x} \in \alpha(\mathbb{N}; X)$.

The space $\alpha(H, X)$. As announced we wish to extend the definition of the α -norms to functions on a measure space different from \mathbb{N} and to operators from a Hilbert space H to X for H different from ℓ^2 .

DEFINITION 3.1.3. Let H be a Hilbert space. We let $\alpha(H, X)$ (resp. $\alpha_+(H, X)$) be the space of all $T \in \mathcal{L}(H, X)$ such that $(Te_k)_{k=1}^{\infty} \in \alpha(\mathbb{N}; X)$ (resp. $(Te_k)_{k=1}^{\infty} \in \alpha_+(\mathbb{N}; X)$) for all orthonormal systems $(e_k)_{k=1}^{\infty}$ in H. We then set

$$||T||_{\alpha(H;X)} = ||T||_{\alpha_+(H;X)} := \sup ||(Te_k)_{k=1}^{\infty}||_{\alpha_+}$$

where the supremum is taken over all orthonormal systems $(e_k)_{k=1}^{\infty}$ in H.

If *H* is separable, then it suffices to compute $||(Te_k)_{k=1}^{\infty}||_{\alpha}$ for a fixed orthonormal basis $(e_k)_{k=1}^{\infty}$ of *H* by Proposition 3.1.2.

For $\alpha = \gamma$ the spaces $\gamma_+(H, X)$ and $\gamma(H, X)$ are already well-studied in literature (see for example [**KW16a**], [**HNVW17**, Chapter 9] and the references therein). Since many of the basic properties of $\alpha(H, X)$ have proofs similar to the ones for $\gamma(H, X)$, we can be brief here and refer to [**HNVW17**, Chapter 9] for inspiration. In particular:

- Both $\alpha_+(H, X)$ and $\alpha(H, X)$ are Banach spaces.
- $\alpha(H, X)^*$ can be canonically identified with $\alpha^*_+(H^*, X^*)$ through trace duality. Note that in this duality one should not identify H with its Hilbert space dual, see [**HNVW17**, Section 9.1.b] for a discussion.
- In many cases $\alpha(H, X)$ and $\alpha_+(H, X)$ coincide. For the Gaussian structure this is the case if and only if X does not contain a closed subspace isomorphic to c_0 .

It follows readily from Proposition 3.1.1 that $\alpha(H, X)$ is the closure of the finite rank operators in $\alpha_+(H, X)$. This can be used to show that every $T \in \alpha(H, X)$ is supported on a separable closed subspace of H:

PROPOSITION 3.1.4. Let H be a Hilbert space and $T \in \alpha(H, X)$. Then there is a separable closed subspace H_0 of H such that $T\varphi = 0$ for all $\varphi \in H_0^{\perp}$.

PROOF. Let $T = \lim_{k \to \infty} T_k$ in $\alpha(H, X)$ where each T_k is of the form

$$T_k\varphi = \sum_{j=1}^{m_k} \langle \varphi, \psi_{jk} \rangle x_{jk}$$

with $\psi_{jk} \in H^*$, $x_{jk} \in X$ for $1 \leq j \leq m_k$ and $k \in \mathbb{N}$. Let H_0 be the closure of the linear span of $\{\psi_{jk} : 1 \leq j \leq m_k, k \in \mathbb{N}\}$ in H. Then H_0 is separable and $T\varphi = 0$ for all $\varphi \in H_0^{\perp}$.

As we already noted, $\alpha(H, X)^*$ can be identified with $\alpha^*_+(H^*, X^*)$ through trace duality. In the converse direction we have the following proposition.

PROPOSITION 3.1.5. Let H be a Hilbert space, let $Y \subseteq X^*$ be norming for X and let $T \in \mathcal{L}(H, X)$. If there is a C > 0 such that for all finite rank operators $S: H^* \to Y$ we have

$$|\operatorname{tr}(S^*T)| \le C ||S||_{\alpha^*(H^*,X^*)}$$

Then $T \in \alpha_+(H, X)$ with $||T||_{\alpha_+(H, X)} \leq C$.

PROOF. Let $(e_k)_{k=1}^n$ be an orthonormal sequence in H and $\varepsilon > 0$. Define $x_k = Te_k$ and let $(x_k^*)_{k=1}^n$ be a sequence in Y with $||(x_k^*)_{k=1}^n||_{\alpha^*} \leq 1$ and

$$||(x_k)_{k=1}^n||_{\alpha} \le \sum_{k=1}^n |x_k^*(x_k)| + \varepsilon.$$

Then, for the finite rank operator $S = \sum_{k=1}^{n} e_k \otimes x_k^*$, we have

$$\|(T_k e_k)_{k=1}^n\|_{\alpha} \le \sum_{k=1}^n |x_k^*(x_k)| + \varepsilon = |\operatorname{tr}(S^*T)| + \varepsilon \le C + \varepsilon.$$

Taking the supremum over all orthonormal sequences in H finishes the proof. \Box

The space $\alpha(S; X)$. We will mostly be using $H = L^2(S)$ for a measure space (S, μ) . We abbreviate

$$\alpha(S;X) := \alpha(L^2(S),X)$$
$$\alpha_+(S;X) := \alpha_+(L^2(S),X)$$

For an operator $T \in \mathcal{L}(L^2(S), X)$ we say that T is *representable* if there exists a strongly measurable $f: S \to X$ with $x^* \circ f \in L^2(S)$ for all $x^* \in X$ such that

(3.1)
$$T\varphi = \int_{S} \varphi f \, \mathrm{d}\mu, \qquad \varphi \in L^{2}(S).$$

Here the integral is well defined by Pettis' theorem [HNVW16, Theorem 1.2.37]. Equivalently T is representable if there exists a strongly measurable $f: S \to X$ such that for all $x^* \in X^*$ we have

(3.2)
$$x^* \circ f = T^*(x^*).$$

Conversely, if we start from a strongly measurable function $f: S \to X$ with $x^* \circ f \in L^2(S)$ for all $x^* \in X$, we can define the operator $T_f: L^2(S) \to X$ as in (3.1), which is again well defined by Pettis' theorem. If $T_f \in \alpha(S; X)$ (resp. $\alpha_+(S; X)$) we can identify f and T_f , since f is the unique representation of T_f . In this case we write $f \in \alpha(S; X)$ (resp. $f \in \alpha_+(S; X)$) and assign to f the α -norm

$$\|f\|_{\alpha(S;X)} := \|T_f\|_{\alpha(S;X)}, \\ \|f\|_{\alpha_+(S;X)} := \|T_f\|_{\alpha_+(S;X)}.$$

In Proposition 3.1.9, we will see that

 $\alpha_{\bullet}(S;X) := \{T \in \alpha(S;X) : T \text{ is representable by a function } f \colon S \to X\}$

is usually not all of $\alpha(S, X)$. However it is often useful to think of the space $(\alpha_{\bullet}(S; X), \|\cdot\|_{\alpha(S;X)})$ as a normed function space and of $\alpha(S; X)$ as its completion, where the elements of $\alpha(S; X) \setminus \alpha_{\bullet}(S; X)$ are interpreted as operators $T : L^2(S) \to X$. If $S = \mathbb{R}^d$ we have $C_0^{\infty}(\mathbb{R}^d) \subseteq L^2(\mathbb{R}^d) \xrightarrow{T} X$ and we may also think of $\alpha(S; X)$ as a space of X-valued distributions. Then (3.1) conforms with the usual interpretation of a locally integrable function f as a distribution T.

The following proposition tells us that $\alpha(S; X)$ is indeed the completion of $\alpha_{\bullet}(S; X)$.

PROPOSITION 3.1.6. Let (S, μ) be a measure space and let \mathcal{A} be a dense subset of $L^2(S)$. Then

$$\operatorname{span}{f \otimes x : f \in \mathcal{A}, x \in X}$$

is dense in $\alpha(S; X)$.

PROOF. Since the finite rank operators are dense in $\alpha(S; X)$, it suffices to show that every rank one operator $T = g \otimes x$ with $g \in L^2(S)$ and $x \in X$ can be approximated by operators T_{f_n} with $f_n \in \text{span}\{h \otimes x : h \in \mathcal{A}\}$. For this let $(h_n)_{k=1}^{\infty}$ be such that $h_n \to g$ in $L^2(S)$ and define $f_n = h_n \otimes x$. Then we have, using Proposition 3.1.2,

$$||T - T_{f_n}||_{\alpha(S;X)} = ||(g - h_n) \otimes x||_{\alpha(S;X)} = ||g - h_n||_{L^2(S)} ||x||_X \square$$

Proposition 3.1.6 allows us to work with work with functions rather than operators in $\alpha(S; X)$. The following lemma sometimes allows us to reduce considerations even further to bounded functions on sets of finite measure.

LEMMA 3.1.7. Let (S, μ) be a measure space and let $f: S \to X$ be strongly measurable. Then there exists a partition $\Pi = \{E_n\}_{n=1}^{\infty}$ of S such that E_n has positive finite measure and f is bounded on E_n for all $n \in \mathbb{N}$. Furthermore, there exists a sequence of such partitions $\Pi_m = \{E_{nm}\}_{n=1}^{\infty}$ such that for the associated averaging projections

$$P_m f(s) := \sum_{n=1}^{\infty} \mathbf{1}_{E_{nm}}(s) \frac{1}{\mu(E_{nm})} \int_{E_{nm}} f \, \mathrm{d}\mu, \qquad s \in S, \ m \in N,$$

we have $P_m f(s) \to f(s)$ for μ -a.e. $s \in S$.

PROOF. By [HNVW16, Proposition 1.1.15] we know that f vanishes off a σ -finite subset of S, so without loss of generality we may assume that (S, μ) is σ -finite. Let $(S_n)_{n=1}^{\infty}$ be a sequence of disjoint measurable sets of finite measure such that $S = \bigcup_{n=1}^{\infty} S_n$. For $n, k \in \mathbb{N}$ set

$$A_{n,k} := \{ s \in S_n : k - 1 \le \| f(s) \|_X < k \}.$$

The sets $(A_{n,k})_{n,k=1}^{\infty}$ are pairwise disjoint, have finite measure and f is bounded on $A_{n,k}$ for all $n, k \in \mathbb{N}$. Relabelling and leaving out all sets with measure zero proves the first part of the lemma.

For the second part note that by the first part we may assume that S has finite measure and f is bounded. By [**HNVW16**, Lemma 1.2.19] there exists a sequence of simple functions $(f_m)_{m=1}^{\infty}$ and a sequence of measurable sets $(B_m)_{m=1}^{\infty}$ such that

$$\sup_{s \in B_m} \|f_m(s) - f(s)\|_X < \frac{1}{m} \quad \text{and} \quad \mu(S \setminus B_m) < 2^{-m-1}.$$

Upon replacing B_m by $\bigcap_{j\geq m} B_j$ the sequence $(B_m)_{m=1}^{\infty}$ can be taken to be increasing with $\mu(S \setminus B_m) \to 0$ for $m \to \infty$. For each $m \in \mathbb{N}$ let $\Pi_m = \{E_{nm}\}_{n=1}^{M_m}$ be the partition of S consisting of the atoms of the finite σ -algebra generated by B_1, \ldots, B_m and the simple functions f_1, \ldots, f_m .

Take $s \in S$, fix $\varepsilon > 0$ and let $N \in \mathbb{N}$ such that $N > \varepsilon/2$ and $s \in B_N$, which is possible for a.e. $s \in S$ since $\mu(\bigcup_{m=1}^{\infty} B_m) = \mu(S)$. Then we have for all $m \ge N$ that $||f_m - f||_X < 1/m$ on B_m , and thus in particular on E_{jm} for $j \in N$ such that $s \in E_{jm}$. Therefore

$$||P_m f(s) - P_m f_m(s)||_X < \frac{1}{m}$$

and since $P_m f_m = f_m$ we conclude

$$||P_m f(s) - f(s)||_X \le ||P_m f(s) - P_m f_m(s)||_X + ||f_m(s) - f(s)||_X < \frac{2}{m} < \epsilon.$$

Therefore $P_m f(s) \to f(s)$ for μ -a.e. $s \in S$, which concludes the proof.

Representability of operators in $\alpha(S; X)$. We will now study the representability of elements of $\alpha(S; X)$ with the aim of characterizing when all elements of $\alpha(S; X)$ are representable by a function $f: S \to X$. If (S, μ) is atomic, then it is clear that every element of $\alpha(S; X)$ is representable by a function. All elements of $\alpha(S; X)$ are also representable by a function if $\alpha = \pi_2$ and X is a Hilbert space, since the Hilbert-Schmidt norm coincides with the π_2 -norm in this case and we have the following, well-known lemma.

LEMMA 3.1.8. Let H be a Hilbert space, (S, μ) a measure space and suppose that $T: L^2(S) \to H$ is a Hilbert-Schmidt operator. Then there is a strongly measurable $f: S \to H$ such that $T\varphi = \int_S \varphi f \, d\mu$ for all $\varphi \in L^2(S)$.

PROOF. We can represent T in the form

$$T\varphi = \sum_{k=1}^{\infty} a_k \langle \varphi, e_k \rangle h_k, \qquad \varphi \in L^2(S),$$

where $(a_k)_{k=1}^{\infty} \in \ell^2$, $(e_k)_{k=1}^{\infty}$ is an orthonormal sequence in $L^2(S)$ and $(h_k)_{k=1}^{\infty}$ is an orthonormal sequence in H. Let

$$f(s) := \sum_{k=1}^{\infty} a_k e_k(s) h_k, \qquad \mu\text{-a.e. } s \in S.$$

This defines a strongly measurable map $f: S \to H$ since

$$\int_S \sum_{k=1}^\infty |a_k|^2 |e_k|^2 \,\mathrm{d}\mu < \infty.$$

Moreover $T\varphi = \int_{S} \varphi f \, d\mu$ for all $\varphi \in L^2(S)$.

It turns out that the two discussed cases, i.e. (S, μ) atomic or X a Hilbert space and $\alpha = \pi_2$, are in a certain sense the only occasions in which all elements of $\alpha(S; X)$ are representable by a function. This will be a consequence of the following proposition.

PROPOSITION 3.1.9. Let (S, Σ, μ) be a non-atomic measure space. Every operator in $\alpha(S, X)$ is representable if and only if $\pi_2 \leq \alpha$.

PROOF. Let us first show that if $T \in \alpha(S; X) \subseteq \pi_2(S; X)$, then T is representable. Note that T is 2-summing, so by the Pietsch factorization theorem **[DJT95**, p.48], we know that T has a factorization T = UJV:

$$\begin{array}{ccc} L^2(S) & & \xrightarrow{T} & X \\ & \downarrow_V & & U \\ L^{\infty}(S') & & \xrightarrow{J} & L^2(S') \end{array}$$

where (S', μ') is a finite measure space and J is the inclusion map. Since J is 2-summing by Grothendiek's theorem (see e.g. [**DJT95**, Theorem 3.7]), VJ is also 2-summing and thus a Hilbert-Schmidt operator. Therefore T is representable by Lemma 3.1.8.

Conversely suppose that every $T \in \alpha(S; X)$ is representable. By restricting to a subset of S we may assume $\mu(S) < \infty$ and then by rescaling we may assume $\mu(S) = 1$. We define a map

$$J: \alpha(S; X) \to L^0(S; X)$$

such that JT is a representing function for $T \in \alpha(S; X)$. This map is well-defined since the representing function is unique up to μ -a.e. equality. Let us consider the topology of convergence in measure on $L^0(S; X)$. If $T_n \to T$ in $\alpha(S; X)$ and $f_n := JT_n \to f$ in $L^0(S; X)$, then it is clear by the dominated convergence theorem that

$$T\varphi = \int_{S} \varphi f \,\mathrm{d}\mu, \qquad \varphi \in \mathcal{A},$$

where

$$\mathcal{A} := \{ \varphi \in L^2(S) : \int_S |\varphi| \sup_{n \in \mathbb{N}} ||f_n||_X \, \mathrm{d}\mu < \infty \}.$$

Since \mathcal{A} is dense in $L^2(S)$ by Lemma 3.1.7, this shows that f = JT. Hence J has a closed graph and is therefore continuous. In particular it follows that there is a constant C > 0 so that if $||JT(s)||_X \ge 1$ for μ -a.e. $s \in S$, then $||T||_{\alpha(S;X)} \ge C^{-1}$. Now take $\mathbf{x} \in X^n$ such that $\sum_{k=1}^n ||x_k||^2 = 1$ and partition S into sets E_1, \ldots, E_n with measure $||x_1||^2, \ldots, ||x_n||^2$, which is possible since (S, μ) is non-atomic. Define

$$e_k = \mathbf{1}_{E_k} ||x_k||^{-1},$$

 $f(s) = \sum_{k=1}^n x_k e_k(s), \qquad s \in S$

for $1 \le k \le n$. Then $(e_k)_{k=1}^n$ is an orthonormal sequence in $L^2(S)$ and $||f(s)||_X = 1$ for $s \in S$, so

$$\|\mathbf{x}\|_{\alpha} = \left\|\sum_{k=1}^{n} e_k \otimes x_k\right\|_{\alpha(S;X)} \ge C^{-1}.$$

This implies that $\left(\sum_{k=1}^{n} \|x_k\|^2\right)^{1/2} \leq C \|\mathbf{x}\|_{\alpha}$ for all $\mathbf{x} \in X^n$. Thus for any $\mathbf{A} \in M_{m,n}(\mathbb{C})$ with $\|\mathbf{A}\| \leq 1$ we have

$$\left(\sum_{j=1}^{m} \|\mathbf{A}\mathbf{x}\|^{2}\right)^{1/2} \leq C \, \|\mathbf{A}\mathbf{x}\|_{\alpha} \leq C \, \|\mathbf{x}\|_{\alpha},$$

which shows that $\pi_2 \lesssim \alpha$.

COROLLARY 3.1.10. Let (S, μ) be a non-atomic measure space. All $S \in \alpha(S; X)$ and $T \in \alpha^*(S; X^*)$ are representable if and only if α and α^* are equivalent to the π_2 -structure and X is isomorphic to a Hilbert space.

PROOF. The 'if' part follows directly from Lemma 3.1.8. For the 'only if' part note that, by Proposition 3.1.9, Proposition 1.1.3 and Proposition 1.1.4, we have on X^*

(3.3)
$$\pi_2 \lesssim \alpha^* \lesssim \pi_2^* \le \gamma^* \le \gamma \le \pi_2.$$

This implies that α^* is equivalent to the π_2 -structure on X^* . A similar argument on X^{**} implies that α^{**} is equivalent to the π_2 -structure on X^{**} , so α is equivalent to the π_2 -structure on X. By (3.3) and Propositions 1.1.3 and 1.1.4, we also have that X^* has nontrivial type and cotype 2. Therefore by [HNVW17, Proposition 7.4.10] we know that X^{**} , and thus X, has type 2. A similar chain of inequalities on X^{**} shows that X^{**} , and thus X, has cotype 2. So by Theorem 2.1.2 we know that X is isomorphic to a Hilbert space.

We end this section with a representation result for the ℓ^2 -structure on a Banach function space X or a $C_0(K)$ space. Note that by $\ell^2(S;X)$ we mean the space $\alpha(S;X)$ where α is the ℓ^2 -structure, not the sequence space ℓ^2 indexed by S with values in X.

PROPOSITION 3.1.11. Let (S, μ) be a measure space and suppose that X is either an order-continuous Banach function space or $C_0(K)$ for some locally compact K. Then for any strongly measurable $f: S \to X$ we have $f \in \ell^2(S; X)$ if and only if $\left(\int_S |f|^2 d\mu\right)^{1/2} \in X$ with

$$||f||_{\ell^2(S;X)} = \left\| \left(\int_S |f|^2 \, \mathrm{d}\mu \right)^{1/2} \right\|_X.$$

PROOF. We will prove the 'only if' statement, the 'if' statement being similar, but simpler. Let $f: S \to X$ be strongly measurable. By [**HNVW16**, Proposition 1.1.15] we may assume that S is σ -finite and by Proposition 3.1.4 we may assume that $L^2(S)$ is separable. Suppose that $(e_k)_{k=1}^{\infty}$ is an orthonormal basis of $L^2(S)$ such that $\int_S |e_k| ||f||_X < \infty$ for all $k \in \mathbb{N}$. Such a basis can for example be constructed by partitioning S into sets of finite measure where f is bounded as in Lemma 3.1.7. Let $x_k := \int_S e_k f \, d\mu$ and $f_n := \sum_{k=1}^n \overline{e}_k \otimes x_k$. Then $f \in \ell^2(S; X)$ if and only if $(x_k)_{k=1}^{\infty} \in \ell^2(\mathbb{N}; X)$. This occurs if and only if

$$\lim_{n \to \infty} \left\| \left(\sum_{k=n+1}^{\infty} |x_k|^2 \right)^{1/2} \right\|_X = 0$$

By order-continuity or Dini's theorem respectively, this occurs if and only if we have $\left(\sum_{k=1}^{\infty} |x_k|^2\right)^{1/2} \in X$. Since

$$\left(\sum_{k=1}^{\infty} |x_k|^2\right)^{1/2} = \left(\int_S |f|^2 \,\mathrm{d}\mu\right)^{1/2}$$

the result follows.

If for example $X = L^p(\mathbb{R})$, then a measurable $f : \mathbb{R} \to L^p(\mathbb{R})$ belongs to $\ell^2(\mathbb{R}; X)$ if and only if

$$\|f\|_{\ell^2(\mathbb{R};X)} = \left(\int_{\mathbb{R}} \left(\int_{\mathbb{R}} |f(t,s)|^2 \,\mathrm{d}t\right)^{p/2} \,\mathrm{d}s\right)^{1/p} < \infty.$$

For a Banach function space with finite cotype we also have that

$$||f||_{\gamma(S;X)} \simeq ||f||_{\ell^2(S;X)} = \left\| \left(\int_S |f|^2 \, \mathrm{d}\mu \right)^{1/2} \right\|_X$$

which follows from Proposition 1.1.3 (see also [**HNVW17**, Theorem 9.3.8]). This equation suggests to think of the norms $\|\cdot\|_{\gamma(S;X)}$ and $\|\cdot\|_{\alpha(S;X)}$ as generalizations of the classical square functions in L^p -spaces to the Banach space setting. We will

support this heuristic in the next section by showing that α -norms have properties quite similar to the usual function space properties of $L^p(S'; L^2(S))$. In Chapter 5 we will use this heuristic to generalize the classical L^p -square functions for sectorial operators to arbitrary Banach spaces.

3.2. Function space properties of $\alpha(S; X)$

We will now take a closer look at the space $\alpha(S; X)$ as the completion of a function space over the measure space (S, μ) . We start with some embedding between these spaces and the more classical Bochner spaces $L^2(S; X)$. If E is a finite-dimensional subspace of X and $f: S \to E$ is strongly measurable, then $f \in \alpha(S, X)$ if and only if $f \in L^2(S; X)$. In fact, by Proposition 1.1.5, we have

(3.4)
$$(\dim(E))^{-1} \|f\|_{L^2(S;X)} \le \|f\|_{\alpha(S;X)} \le \dim(E) \|f\|_{L^2(S;X)}.$$

Moreover if $\dim(L^2(S)) = \infty$, it is known that for the γ -structure we have

(3.5)
$$||f||_{\gamma(S;X)} \lesssim ||f||_{L^2(S;X)}, \qquad f \in L^2(S;X),$$

if and only if X has type 2 and

(3.6)
$$||f||_{L^2(S;X)} \lesssim ||f||_{\gamma(S;X)}, \qquad f \in \gamma(S;X)$$

if and only if X has cotype 2, see [HNVW17, Section 9.2.b]. Further embeddings under smoothness conditions can be found in [HNVW17, Section 9.7]. We leave the generalization of these embeddings to a general Euclidean structure α to the interested reader.

Extension of bounded operators on $L^2(S)$. One of the main advantages the spaces $\alpha(S; X)$ have over the Bochner spaces $L^p(S; X)$ is the fact that any operator $T \in \mathcal{L}(L^2(S_1), L^2(S_2))$ can be extended to a bounded operator $\widetilde{T}: \alpha(S_1; X) \rightarrow \alpha(S_2; X)$. Indeed, putting $\widetilde{T}U := U \circ T^*$ for $U \in \alpha(S_1; X)$, we have that \widetilde{T} is bounded by Proposition 3.1.2. For functions this read as follows:

PROPOSITION 3.2.1. Let (S_1, μ_1) and (S_2, μ_2) be measure spaces and let $f: S_1 \to X$ be a strongly measurable function in $\alpha(S_1; X)$. Take $T \in \mathcal{L}(L^2(S_1), L^2(S_2))$ and suppose that there exists a strongly measurable $g: S_2 \to X$ such that for every $x^* \in X^*$ we have

$$x^* \circ g = T(x^* \circ f)$$

or equivalently $x^* \circ g \in L^2(S)$ and

$$\int_{S_2} \varphi g \,\mathrm{d}\mu_2 = \int_{S_1} (T^* \varphi) f \,\mathrm{d}\mu_1, \qquad \varphi \in L^2(S_2).$$

Then $g \in \alpha(S_2; X)$ and

$$||g||_{\alpha(S_2;X)} \le ||T|| ||f||_{\alpha(S_1;X)}.$$

In the setting of Proposition 3.2.1 we write Tf = g. As typical examples, we note that multiplication by an L^{∞} -function is a bounded operation on $\alpha(\mathbb{R}; X)$ and we show that the Fourier transform can be extended from an isometry on $L^2(\mathbb{R})$ to an isometry on $\alpha(\mathbb{R}; X)$. Combining these examples we would obtain a Fourier multiplier theorem, which we will treat more generally in Corollary 3.2.9. EXAMPLE 3.2.2. Let (S, μ) be a measure space and suppose that $f \in \alpha(S; X)$. For any $m \in L^{\infty}(S)$ we have $mf \in \alpha(S; X)$ with

$$|mf||_{\alpha(S;X)} \le ||m||_{L^{\infty}(S)} ||f||_{\alpha(S;X)}.$$

EXAMPLE 3.2.3. Suppose that $f \in L^1(\mathbb{R}; X)$ with $f \in \alpha(\mathbb{R}; X)$. Define

$$\mathcal{F}f(\xi) := \widehat{f}(\xi) := \int_{\mathbb{R}} f(t) e^{-2\pi i t\xi} dt, \qquad \xi \in \mathbb{R},$$
$$\mathcal{F}^{-1}f(\xi) := \check{f}(\xi) := \int f(t) e^{2\pi i t\xi} dt \qquad \xi \in \mathbb{R}$$

$$\mathcal{F}^{-1}f(\xi) := \check{f}(\xi) := \int_{\mathbb{R}} f(t) \mathrm{e}^{2\pi i t\xi} \,\mathrm{d}t \qquad \xi \in \mathbb{R}$$

Then $\hat{f}, \check{f} \in \alpha(\mathbb{R}; X)$ with $\|\hat{f}\|_{\alpha(\mathbb{R}; X)} = \|\check{f}\|_{\alpha(\mathbb{R}; X)} = \|f\|_{\alpha(\mathbb{R}; X)}$

The α -Hölder inequality. Next we will prove Hölder's inequality for α -spaces, which is a realisation of the duality pairing between $\alpha_+(S; X)$ and $\alpha_+^*(S; X^*)$ for representable elements. Conversely, we will show that the representable elements of a subspace of $\alpha_+^*(S; X^*)$ are norming for $\alpha_+(S; X)$ using Proposition 3.1.5.

PROPOSITION 3.2.4. Let (S, μ) be a measure space.

(i) Suppose that $f: S \to X$ and $g: S \to X^*$ are in $\alpha_+(S; X)$ and $\alpha^*_+(S; X^*)$ respectively. Then $\langle f, g \rangle \in L^1(S)$ and

$$\int_{S} |\langle f, g \rangle| \, \mathrm{d}\mu \le \|f\|_{\alpha_{+}(S;X)} \|g\|_{\alpha_{+}^{*}(S;X^{*})}$$

(ii) Let $Y \subseteq X^*$ be norming for X and let $f: S \to X$ be strongly measurable. If there is a C > 0 such that for all $g \in L^2(S) \otimes Y$ we have

$$\int_{S} |\langle f,g \rangle| \, \mathrm{d}\mu \leq C \, \|g\|_{\alpha^{*}(S;X^{*})}$$

then $f \in \alpha_{+}(S;X)$ with $\|f\|_{\alpha_{+}(S;X)} \leq C$.

PROOF. We will only prove (i), as (ii) follows directly from of Proposition 3.1.5 and the fact that any finite rank operator is representable as an element of $L^2(S) \otimes Y$. Let $\Pi_m = \{E_{km}\}_{k=1}^{\infty}$ be a sequence of partitions of S as in Lemma 3.1.7 and let P_m be the associated averaging projections. Then

$$\int_{S} |\langle f, g \rangle| d\mu \le \sup_{m \in \mathbb{N}} \int_{S} |\langle P_m f, g \rangle| \, \mathrm{d}\mu.$$

For each $m \in \mathbb{N}$ we define a measurable function h_m with $|h_m| = 1$ μ -a.e. and $h_m \langle P_m f, g \rangle = |\langle P_m f, g \rangle|$. Define $Q_{nm}, R_{nm} : L^2(S) \to L^2(S)$ by

$$\begin{aligned} Q_{nm}\varphi &:= \mathbf{1}_{\bigcup_{k=1}^{n} E_{km}} P_{m}\varphi, & \varphi \in L^{2}(S), \\ R_{nm}\varphi &:= h_{m} \mathbf{1}_{\bigcup_{k=1}^{n} E_{km}} P_{m}\varphi, & \varphi \in L^{2}(S). \end{aligned}$$

and extend these operators to bounded operators on $\alpha_+(S; X)$ and $\alpha_+(S; X^*)$ using Proposition 3.2.1. Define

$$x_{km} \coloneqq \left(\mu(E_{km})\right)^{-1/2} \int_{E_{km}} f \,\mathrm{d}\mu,$$
$$x_{km}^* \coloneqq \left(\mu(E_{km})\right)^{-1/2} \int_{E_{km}} h_m g \,\mathrm{d}\mu$$

and note that

$$Q_{nm}f = \sum_{k=1}^{n} (\mu(E_{km}))^{-1/2} x_{km} \mathbf{1}_{E_{km}},$$
$$R_{nm}g = \sum_{k=1}^{n} (\mu(E_{km}))^{-1/2} x_{km}^* \mathbf{1}_{E_{km}}.$$

It follows that

$$\int_{\bigcup_{k=1}^{n} E_{km}} |\langle P_m f, g \rangle| = \sum_{k=1}^{n} \langle x_{km}, x_{km}^* \rangle \le \|(x_{km})_{k=1}^{n}\|_{\alpha} \|(x_{km}^*)_{k=1}^{n}\|_{\alpha^*}.$$

Since we have

$$\begin{aligned} \|(x_{km})_{k=1}^n\|_{\alpha} &= \|Q_{nm}f\|_{\alpha_+(S;X)} \le \|f\|_{\alpha_+(S;X)},\\ \|(x_{km}^*)_{k=1}^n\|_{\alpha} &= \|R_{nm}g\|_{\alpha_+^*(S;X^*)} \le \|g\|_{\alpha_+^*(S;X^*)}, \end{aligned}$$

the result follows by first letting $n \to \infty$ and then $m \to \infty$ using Fatou's lemma. \Box

Convergence properties. In the function spaces $L^p(S; X)$ we have convergence theorems like Fatou's lemma and the dominated convergence theorem. In the next proposition we summarize some convergence properties of the α -norms. For example (i) can be seen as an α -version of Fatou's lemma. It is important to note that even if all f_n 's are in $\alpha(S; X)$, we can only deduce that f is in $\alpha_+(S; X)$.

PROPOSITION 3.2.5. Let $f: S \to X$ be a strongly measurable function.

(i) Suppose that $f_n: S \to X$ are functions in $\alpha_+(S; X)$ such that

$$\sup_{n\in\mathbb{N}}\|f_n\|_{\alpha_+(S;X)}<\infty.$$

If $f_n(s)$ converges weakly to f(s) μ -a.e, then $f \in \alpha_+(S;X)$ with

$$\|f\|_{\alpha_+(S;X)} \le \liminf_{n \to \infty} \|f_n\|_{\alpha_+(S;X)}$$

Now suppose that $f \in \alpha(S; X)$.

- (ii) Let $(g_n)_{n=1}^{\infty}$ be a sequence in $L^{\infty}(S)$ with $|g_n| \leq 1$ and $g_n(s) \to 0$ μ -a.e. Then $\lim_{n\to\infty} ||g_n \cdot f||_{\alpha(S;X)} = 0.$
- (iii) If α is ideal and $T_n, T \in \mathcal{L}(X)$ with $\lim_{n \to \infty} T_n x = Tx$ for $x \in X$, then $\lim_{n \to \infty} T_n \circ f \to T \circ f$ in $\alpha(S; X)$.

PROOF. For (i) note that for all $x^* \in X^*$ we have

$$\sup_{n \in \mathbb{N}} \|x^* \circ f_n\|_{L^2(S)} \le \sup_{n \in \mathbb{N}} \|f_n\|_{\alpha_+(S;X)} \|x^*\|_{X^*} < \infty.$$

Let $(e_m)_{m=1}^{\infty}$ be an orthonormal sequence in $L^2(S)$, set $x_{nm} = \int_S e_m f_n d\mu$ and $x_m = \int_S e_m f d\mu$. Then by the dominated convergence theorem we have for all $x^* \in X^*$

$$\lim_{n \to \infty} \langle x_{nm}, x^* \rangle = \lim_{n \to \infty} \int_S e_m \langle f_n, x^* \rangle \, \mathrm{d}\mu = \int_S e_m \langle f, x^* \rangle \, \mathrm{d}\mu = \langle x_m, x^* \rangle.$$

Thus by α -duality we have for each $m \in \mathbb{N}$

$$\|(x_1,\ldots,x_m)\|_{\alpha} \leq \liminf_{n\to\infty} \|(x_{n1},\ldots,x_{nm})\|_{\alpha} \leq \liminf_{n\to\infty} \|f_n\|_{\alpha_+(S;X)},$$

so (i) follows by taking the supremum over all orthonormal sequences in $L^2(S)$. For (ii) let $\varepsilon > 0$. By Proposition 3.1.6 we can find a finite dimensional subspace

 $E \subseteq X$ and an $h \in L^2(S; E)$ such that $||f - g||_{\alpha(S;X)} < \varepsilon$. Then by (3.4) and the dominated convergence theorem we have

$$\lim_{n \to \infty} \|g_n \cdot f\|_{\alpha(S;X)} \le \dim(E) \lim_{n \to \infty} \|g_n \cdot h\|_{L^2(S;X)} + \varepsilon = \varepsilon.$$

The proof of (iii) is similar.

The α -multiplier theorem. We now come to one of the main theorems of this section, which characterize α -boundedness of a family of operators in terms of the boundedness of a pointwise multiplier on $\alpha(S; X)$. This will be very useful later. We say that a function $T: S \to \mathcal{L}(X)$ is strongly measurable in the strong operator topology if $Tx: S \to X$ is strongly measurable for all $x \in X$. For $f: S \to X$ we define $Tf: S \to X$ by

$$Tf(s) := T(s)f(s), \qquad s \in S$$

THEOREM 3.2.6. Let (S, μ) be a measure space, let $T : S \to \mathcal{L}(X)$ be strongly measurable in the strong operator topology and set $\Gamma = \{T(s) : s \in S\}$. If Γ is α -bounded, then $Tf \in \alpha_+(S; X)$ with

$$||Tf||_{\alpha_+(S;X)} \le ||\Gamma||_{\alpha} ||f||_{\alpha(S;X)}$$

for all $f \in \alpha(S; X)$.

PROOF. Let $\Pi = \{E_n\}_{n=1}^{\infty}$ be a partition of S with associated averaging projection P as in Lemma 3.1.7 for f and let $\Pi' = \{E'_n\}_{n=1}^{\infty}$ be a partition of S with associated averaging projection P' as in Lemma 3.1.7 for TPf. Then

$$P'TPf = \sum_{n=1}^{\infty} S_n x_n \, \mathbf{1}_{E_n}$$

where

$$x_n = \frac{1}{\mu(E_n)} \int_{E_n} f \, \mathrm{d}\mu,$$

$$S_n x = \frac{1}{\mu(E'_n)} \int_{E'_n} Tx \, \mathrm{d}\mu, \qquad x \in X.$$

So we obtain

$$\|Pf\|_{\alpha(S;X)} = \left\|\sum_{n=1}^{\infty} x_n \, \mathbf{1}_{E_n}\right\|_{\alpha(S;X)} = \left\|\left(x_n \mu(E_n)^{1/2}\right)_{n=1}^{\infty}\right\|_{\alpha},\\ \|P'TPf\|_{\alpha(S;X)} = \left\|\sum_{n=1}^{\infty} S_n x_n \, \mathbf{1}_{E_n}\right\|_{\alpha(S;X)} = \left\|\left(S_n x_n \mu(E_n)^{1/2}\right)_{n=1}^{\infty}\right\|_{\alpha}$$

Since S_n belongs to the strong operator topology closure of the convex hull of Γ , it follows from Proposition 1.2.3 and Proposition 3.2.1 that

$$\|P'TPf\|_{\alpha(S;X)} \le \|\Gamma\|_{\alpha}\|Pf\|_{\alpha(S;X)} \le \|\Gamma\|_{\alpha}\|f\|_{\alpha(S;X)}.$$

Now let P_m be a sequence of such averaging projections for f as in Lemma 3.1.7 and for every $m \in \mathbb{N}$ let $P'_{m'}$ be a sequence of such averaging projections for $TP_m f$ as in Lemma 3.1.7. Then we have

$$\lim_{m\to\infty}\lim_{m'\to\infty}P'_{m'}TP_mf(s)=Tf(s),\qquad s\in S,$$

so the conclusion follows by applying Proposition 3.2.5(i) twice.

REMARK 3.2.7. Since we use Proposition 3.2.5(i) in the proof of Theorem 3.2.6, we do not know whether $Tf \in \alpha(S; X)$. We refer to [**HNVW17**, Section 9.5] for a discussion on sufficient conditions such that one can conclude $Tf \in \alpha(S; X)$ in the case $\alpha = \gamma$.

We also have a converse of Theorem 3.2.6, for which we need to assume that the measure space (S, μ) has more structure. A *metric measure space* (S, d, μ) is a complete separable metric space (S, d) with a locally finite Borel measure μ . We denote by $\text{supp}(\mu)$ the smallest closed set with the property that its complement has measure zero.

THEOREM 3.2.8. Let (S, d, μ) be a metric measure space, let $T : S \to \mathcal{L}(X)$ be continuous in the strong operator topology and set

$$\Gamma = \{T(s) : s \in \operatorname{supp}(\mu)\}.$$

If we have $Tf \in \alpha_+(S; X)$ for all $f \in \alpha(S; X)$ with

$$|Tf||_{\alpha_+(S;X)} \le C ||f||_{\alpha(S;X)},$$

then Γ is α -bounded with $\|\Gamma\|_{\alpha} \leq C$.

PROOF. Take $T_1, \ldots, T_n \in \Gamma$ and $\mathbf{x} \in X^n$. Let $s_1, \ldots, s_n \in \operatorname{supp}(\mu)$ be such that $T_k = T(s_k)$ for $1 \leq k \leq n$ and let $\varepsilon > 0$. For $1 \leq k \leq n$, using the continuity of T and the fact that $s_k \in \operatorname{supp}(\mu)$, we can select an open ball $O_k \subseteq \operatorname{supp}(\mu)$ with finite positive measure such that $s_k \in O_k$ and

(3.7)
$$||T(s)x_k - T(s_k)x_k|| \le n^{-1}\varepsilon, \qquad s \in O_k.$$

If $O_{k_1} \cap O_{k_2} \neq \emptyset$ for $1 \leq k_1 \neq k_2 \leq n$, then $\mu(O_{k_1} \cap O_{k_2}) > 0$. Since μ is non-atomic, there are disjoint E_1, E_2 with positive measure such that $O_{k_1} \cap O_{k_2} = E_1 \cup E_2$. Iteratively replacing O_{k_1} by $O_{k_1} \setminus E_1$ and O_{k_2} by $O_{k_2} \setminus E_2$ for all pairs $1 \leq k_1 \neq k_2 \leq n$, we obtain pairwise disjoint sets O_1, \ldots, O_n of positive finite measure such that (3.7) holds.

Let P be the averaging projection associated to O_1, \ldots, O_n and define $f = \sum_{k=1}^n \mu(O_k)^{-1/2} x_k \mathbf{1}_{O_k}$. Then

$$PTf = \sum_{k=1}^{n} \mu(O_k)^{-1/2} y_k \, \mathbf{1}_{O_k}$$

for

$$y_k = \frac{1}{\mu(O_k)} \int_{O_k} Tx_k \,\mathrm{d}\mu, \qquad 1 \le k \le n.$$

Note that $||y_k - T_k x_k|| \leq n^{-1}\varepsilon$, so we have by Proposition 3.2.1, the fact that $(\mu(O_k)^{-1/2} \mathbf{1}_{O_k})_{k=1}^n$ is an orthonormal system in $L^2(S)$ and our assumption, that

$$\|\mathbf{y}\|_{\alpha} = \|PTf\|_{\alpha(S;X)} \le C \|f\|_{\alpha(S;X)} = C \|\mathbf{x}\|_{\alpha}.$$

Therefore $||(T_1x_1, \ldots, T_nx_n)||_{\alpha} \leq C ||\mathbf{x}||_{\alpha} + \varepsilon$, which proves the theorem. \Box

We conclude this section by combining Theorem 3.2.6 and Example 3.2.3 into the following Fourier multiplier theorem.

COROLLARY 3.2.9. Suppose that $m : \mathbb{R} \to \mathcal{L}(X)$ is strongly measurable in the strong operator topology and $\{m(s) : s \in \mathbb{R}\}$ is α -bounded. For $f \in L^1(\mathbb{R}; X)$ such that $\widehat{f} \in L^1(\mathbb{R}; X)$ we define

$$T_m f(s) = \mathcal{F}^{-1}(m(s)\widehat{f}(s)), \qquad s \in S.$$

If
$$f \in \alpha(\mathbb{R}; X)$$
, then $Tf \in \alpha_+(\mathbb{R}; X)$ with

$$||T_m f||_{\alpha_+(\mathbb{R};X)} \le ||\{m(s) : s \in \mathbb{R}\}||_{\alpha} ||f||_{\alpha(\mathbb{R};X)}.$$

3.3. The α -interpolation method

In this section we will develop a theory of interpolation using Euclidean structures. This method seems especially well-adapted to the study of sectorial operators and semigroups, which we will explore further in Chapter 5. Although we develop this interpolation method in more generality, the most important example is the Gaussian structure, which gives rise to the Gaussian method of interpolation. A discrete version of the Gaussian method was already considered in [**KKW06**], where it is used to the study the H^{∞} -calculus of various differential operators. The continuous version of the Gaussian method was studied in [**SW06**, **SW09**], where Gaussian interpolation of Bochner spaces $L^p(S; X)$ and square function spaces $\gamma(S; X)$, as well as a Gaussian version of abstract Stein interpolation, was treated. Furthermore, for Banach function spaces, an ℓ^q -version of this interpolation method was developed in [**Kun15**]. An abstract framework covering these interpolation methods, as well as the real and complex interpolation methods, is developed in [**LL21**].

The results in [**KKW06**, **SW06**, **SW09**] were based on a draft version of this memoir, which explains why some of these papers omit various proofs with a reference to this memoir, see e.g. [**KKW06**, Proposition 7.3] and [**SW06**, Section 2].

Throughout this section we let α be a global Euclidean structure, (X_0, X_1) a compatible pair of Banach spaces and $\theta \in (0, 1)$. We will define interpolation spaces $(X_0, X_1)^{\alpha}_{\theta}$ and $(X_0, X_1)^{\alpha+}_{\theta}$ and refer to these methods of interpolation as the α -method and the α_+ -method. Note that we will only use the Euclidean structures α_0 on X_0 and α_1 on X_1 for our construction, so the assumption that α is a global Euclidean structure is only for notational convenience.

Let us consider the space

$$L^{2}(\mathbb{R}) + L^{2}(\mathbb{R}, e^{-2t}dt) = L^{2}(\mathbb{R}, \min\{1, e^{-2t}\}dt).$$

We call an operator

$$T: L^2(\mathbb{R}) + L^2(\mathbb{R}, \mathrm{e}^{-2t} \mathrm{d}t) \to X_0 + X_1.$$

admissible and write $T \in \mathcal{A}$ (respectively $T \in \mathcal{A}_+$) if $T \in \alpha(\mathbb{R}, e^{-2jt} dt; X_j)$ (respectively $T \in \alpha_+(\mathbb{R}, e^{-2jt} dt; X_j)$) for j = 0, 1. We define

$$||T||_{\mathcal{A}} := \max_{j=0,1} ||T_j||_{\alpha(\mathbb{R}, e^{-2jt} dt; X_j)},$$

$$||T||_{\mathcal{A}_+} := \max_{j=0,1} ||T_j||_{\alpha_+(\mathbb{R}, e^{-2jt} dt; X_j)},$$

where T_j denotes the operator T from $L^2(\mathbb{R}, e^{-2jt}dt)$ into X_j . Both \mathcal{A} and \mathcal{A}_+ are complete with respect to their norm.

Denote by e_{θ} the function $t \mapsto e^{\theta t}$. We define $(X_0, X_1)^{\alpha}_{\theta}$ as the space of all $x \in X_0 + X_1$ such that

$$\|x\|_{(X_0,X_1)^{\alpha}_{\theta}} := \inf\{\|T\|_{\mathcal{A}} : T \in \mathcal{A}, T(\mathbf{e}_{\theta}) = x\} < \infty.$$

The space $(X_0, X_1)^{\alpha}_{\theta+}$ is defined similarly as the space of all $x \in X_0 + X_1$ such that

$$\|x\|_{(X_0,X_1)_{\theta}^{\alpha_+}} := \inf\{\|T\|_{\mathcal{A}_+} : \ T \in \mathcal{A}_+, \ T(\mathbf{e}_{\theta}) = x\} < \infty.$$

Then $(X_0, X_1)^{\alpha}_{\theta}$ and $(X_0, X_1)^{\alpha_+}_{\theta}$ are quotient spaces of \mathcal{A} and \mathcal{A}_+ respectively and thus Banach spaces. For brevity we will sometimes write $X_{\theta} := (X_0, X_1)^{\alpha}_{\theta}$ and $X_{\theta,+} := (X_0, X_1)^{\alpha_+}_{\theta}$.

PROPOSITION 3.3.1 (α -Interpolation of operators). Suppose that (X_0, X_1) and (Y_0, Y_1) are compatible pairs of Banach spaces and α is ideal. Assume that $S: X_0 + X_1 \rightarrow Y_0 + Y_1$ is a bounded operator such that $S(X_0) \subset Y_0$ and $S(X_1) \subset Y_1$. Then $S: X_{\theta} \rightarrow Y_{\theta}$ is bounded with

$$||S||_{X_{\theta} \to Y_{\theta}} \le ||S||_{X_{0} \to Y_{0}}^{1-\theta} ||S||_{X_{1} \to Y_{1}}^{\theta}.$$

A similar statement holds for $S_+: X_{\theta,+} \to Y_{\theta,+}$.

PROOF. Suppose $T \in \mathcal{A}$. Fix τ so that $||S||_{X_1 \to Y_1} = e^{\tau} ||S||_{X_0 \to Y_0}$ and let U_{τ} be the shift operator given by $U\varphi = \varphi(\cdot - \tau)$, which satisfies

(3.8)
$$||U_{\tau}||_{\mathcal{L}(L^2(\mathbb{R}, e^{-2jt}dt))} \le e^{-j\tau}, \quad j = 0, 1.$$

The ideal property of α means that STU_{τ} is admissible and

$$\|STU_{\tau}\|_{\mathcal{A}} \le \max_{j=0,1} \{\|S\|_{X_{j}\to Y_{j}}\|T\|_{\alpha(\mathbb{R}, e^{-2jt}dt; X_{j})} e^{-j\tau} \} \le \|S\|_{X_{0}\to Y_{0}}\|T\|_{\mathcal{A}}.$$

Now if $T(\mathbf{e}_{\theta}) = x$, then $\mathbf{e}^{\theta \tau} \cdot STU_{\tau}(\mathbf{e}_{\theta}) = Sx$ and therefore

$$\|S\|_{X_{\theta} \to Y_{\theta}} \le e^{\theta \tau} \|S\|_{X_{0} \to Y_{0}} = \|S\|_{X_{0} \to Y_{0}}^{1-\theta} \|S\|_{X_{1} \to Y_{1}}^{\theta}.$$

In interpolation theory it is often useful to know that $X_0 \cap X_1$ is dense in the intermediate spaces, which is the content of the next lemma.

PROPOSITION 3.3.2. The set of finite rank operators $T \in \mathcal{A}$ is dense in \mathcal{A} . In particular, $X_0 \cap X_1$ is dense in X_{θ} .

PROOF. If $T \in \mathcal{A}$, we consider the operators $S_{\lambda,n}$ given by

$$S_{\lambda,n}\varphi(t) \coloneqq \sum_{|k| \le n} \left(\frac{1}{\lambda} \int_{k\lambda}^{(k+1)\lambda} \varphi(s) \mathrm{d}s\right) \mathbf{1}_{[k\lambda,(k+1)\lambda)}(t), \qquad t \in \mathbb{R}$$

for $\varphi \in L^2(\mathbb{R}) + L^2(\mathbb{R}, e^{-2t} dt)$. As $T_j S_{\lambda,n}$ has finite rank, it suffices to show that for j = 0, 1

(3.9)
$$\lim_{\lambda \to 0} \lim_{n \to \infty} \|T_j - T_j S_{\lambda,n}\|_{\alpha(\mathbb{R}, \mathrm{e}^{-2jt} \mathrm{d}t; X_j)} = 0.$$

Note that for a finite rank operator $U \in \alpha(\mathbb{R}, e^{-2jt} dt; X_j)$ and j = 0, 1

$$\lim_{\lambda \to 0} \lim_{n \to \infty} \|U - US_{\lambda,n}\|_{\alpha(\mathbb{R}, \mathrm{e}^{-2jt} \mathrm{d}t; X_j)} = 0$$

by the Lebesgue differentiation theorem. Moreover we have

$$\|S_{\lambda,n}\|_{\mathcal{L}(L^2(\mathbb{R}))} = 1,$$

$$\|S_{\lambda,n}\|_{\mathcal{L}(L^2(\mathbb{R}, e^{-2t} dt))} = \frac{\sinh \lambda}{\lambda},$$

so by density we obtain (3.9) for j = 0, 1. To conclude note that if $T \in \mathcal{A}$ has finite rank, then necessarily

$$T(L^{2}(\mathbb{R}) + L^{2}(\mathbb{R}, \mathrm{e}^{-2t} \mathrm{d}t)) \subseteq X_{0} \cap X_{1},$$
since $T \in \alpha(\mathbb{R}, e^{-2jt} dt; X_j)$ for j = 0, 1. Thus $X_0 \cap X_1$ is dense in X_{θ} .

Duality. If $X_0 \cap X_1$ is dense in both X_0 and X_1 , then the pair (X_0^*, X_1^*) is also compatible. We can then define the classes $\mathcal{A}^*, \mathcal{A}^*_+$ for the pair (X_0^*, X_1^*) with the global Euclidean structure α^* and define the interpolation spaces $(X_0^*, X_1^*)_{\theta}^{\alpha^*}$ and $(X_0^*, X_1^*)_{\theta}^{\alpha^*_+}$, which we write as X_{θ}^* and $X_{\theta,+}^*$ for brevity.

If $T \in \mathcal{A}^*_+$ we can view T^* as the operator from $X_0 \cap X_1$ to $L^2(\mathbb{R}) \cap L^2(\mathbb{R}, e^{-2t} dt)$ so that for $x \in X_0 \cap X_1$

$$\langle T^*x,\varphi\rangle=\langle x,T\varphi\rangle,\qquad \varphi\in L^2(\mathbb{R})+L^2(\mathbb{R},\mathrm{e}^{-2t}\mathrm{d} t),$$

using the densely defined bilinear form $\langle \cdot, \cdot \rangle$ on $L^2(\mathbb{R}) + L^2(\mathbb{R}, e^{-2t}dt)$ given by

(3.10)
$$\langle \varphi_1, \varphi_2 \rangle = \int_{\mathbb{R}} \varphi_1(t) \varphi_2(-t) \, \mathrm{d}t$$

for all φ_1 and φ_2 such that $\varphi_1(\cdot)\varphi_2(-\cdot) \in L^1(\mathbb{R})$, which holds in particular if

$$\varphi_1 \in L^2(\mathbb{R}) \cap L^2(\mathbb{R}, \mathrm{e}^{-2t} \mathrm{d}t),$$

$$\varphi_2 \in L^2(\mathbb{R}) + L^2(\mathbb{R}, \mathrm{e}^{-2t} \mathrm{d}t).$$

Then T^* extends to the adjoints $T_j^* : X_j \to L^2(\mathbb{R}, e^{-2jt} dt)$.

LEMMA 3.3.3. Suppose that $X_0 \cap X_1$ is dense in X_0 and X_1 . If $S \in \mathcal{A}$ and $T \in \mathcal{A}_+^*$, then

$$\operatorname{tr}(T_0^*S_0) = \operatorname{tr}(T_1^*S_1).$$

PROOF. Let us fix $T \in \mathcal{A}_+^*$. The equality is trivial if S has finite rank and thus range contained in $X_0 \cap X_1$, since T^*S then has finite rank and range contained in $L^2(\mathbb{R}) \cap L^2(\mathbb{R}, e^{-2t}dt)$. Since the functionals $S \mapsto \operatorname{tr}(T_0^*S_0)$ and $S \mapsto \operatorname{tr}(T_1^*S_1)$ are continuous, the result follows from Proposition 3.3.2

By Lemma 3.3.3 we can now define the pairing

$$\langle S,T\rangle := \operatorname{tr}(T_0^*S_0) = \operatorname{tr}(T_1^*S_1), \qquad S \in \mathcal{A}, \ T \in \mathcal{A}_+^*$$

and note that

$$(3.11) \qquad |\langle S,T\rangle| \le \min_{j=0,1} ||S_j||_{\alpha(\mathbb{R},\mathrm{e}^{-2jt}\mathrm{d}t;X_j)} ||T_j||_{\alpha^*_+(\mathbb{R},\mathrm{e}^{-2jt}\mathrm{d}t;X_j)} \le ||S||_{\mathcal{A}} ||T||_{\mathcal{A}^*_+}$$
for $C \in \mathcal{A}$ and $T \in \mathcal{A}^*$

for $S \in \mathcal{A}$ and $T \in \mathcal{A}_+^*$.

THEOREM 3.3.4. Suppose that $X_0 \cap X_1$ is dense in X_0 and X_1 . Then we have $(X_{\theta})^* = X_{\theta,+}^*$ isomorphically.

PROOF. Let $x^* \in X^*_{\theta,+}$ and take $T \in \mathcal{A}^*_+$ with $T(\mathbf{e}_{\theta}) = x^*$. Fix $x \in X_0 \cap X_1$ and take an $S \in \mathcal{A}$ with finite rank and $S(\mathbf{e}_{\theta}) = x$. For $\tau \in \mathbb{R}$ let U_{τ} be the shift operator given by $U_{\tau}\varphi = \varphi(\cdot - \tau)$. For $\varphi \in L^2(\mathbb{R}) \cap L^2(\mathbb{R}, \mathbf{e}^{2jt} \mathrm{d}t)$ we note that

$$\int_{\mathbb{R}} e^{\theta \tau} U_{\tau} \varphi \, \mathrm{d}\tau = \int_{\mathbb{R}} e^{\theta(\tau + \cdot)} \varphi(-\tau) \, \mathrm{d}\tau = \langle \mathrm{e}_{\theta}, \varphi \rangle \mathrm{e}_{\theta}$$

as Bochner integral in $L^2(\mathbb{R}) + L^2(\mathbb{R}, e^{2jt}dt)$. Thus, since the range of T^*SU_{τ} is contained in a fixed finite-dimensional subspace of $L^2(\mathbb{R}) \cap L^2(\mathbb{R}, e^{2jt}dt)$ for all $\tau \in \mathbb{R}$, we have

$$\int_{\mathbb{R}} e^{\theta \tau} \langle SU_{\tau}, T \rangle d\tau = \langle S(e_{\theta}), T(e_{\theta}) \rangle.$$

Now by (3.8) and (3.11) we have

$$e^{\theta\tau} \langle SU_{\tau}, T \rangle \leq \begin{cases} e^{(\theta-1)\tau} \|S_1\|_{\alpha(\mathbb{R}, e^{-2jt} dt; X_j)} \|T_1\|_{\alpha^*_+(\mathbb{R}, e^{-2jt} dt; X_j)}, & \tau \ge 0, \\ e^{\theta\tau} \|S_0\|_{\alpha(\mathbb{R}, e^{-2jt} dt; X_j)} \|T_0\|_{\alpha^*_+(\mathbb{R}, e^{-2jt} dt; X_j)}, & \tau < 0, \end{cases}$$

from which it follows that

$$|\langle x, x^* \rangle| = |\langle S(\mathbf{e}_{\theta}), T(\mathbf{e}_{\theta}) \rangle| \le (\theta(1-\theta))^{-1} ||S||_{\mathcal{A}} ||T||_{\mathcal{A}^*_+}$$

Hence, taking the infimum over all such S and T and using Proposition 3.3.2, we have

$$|\langle x, x^* \rangle| \le (\theta(1-\theta))^{-1} ||x||_{X_{\theta}} ||x^*||_{X_{\theta+1}^*}.$$

By the density of $X_0 \cap X_1$ in X_{θ} this implies that $X_{\theta,+}^*$ embeds continuously into $(X_{\theta})^*$.

We now turn to the other embedding. Given $x^* \in (X_{\theta})^*$ we must show $x^* \in X_{\theta,+}^*$ with $||x^*||_{X_{\theta,+}^*} \leq C ||x^*||_{(X_{\theta})^*}$. First note that x^* induces a linear functional ψ on \mathcal{A} by $\psi(S) = x^*(S(e_{\theta}))$ for $S \in \mathcal{A}$. Moreover there is a natural isometric embedding of \mathcal{A} into

$$\alpha(\mathbb{R}; X_0) \oplus_{\infty} \alpha(\mathbb{R}, \mathrm{e}^{-2t} \mathrm{d}t; X_1)$$

via the map $S \mapsto (S_0, S_1)$. Hence by the Hahn-Banach theorem we can extend x^* to a functional on this larger space, i.e. there is a

$$T = (T_0, T_1) \in \alpha_+^*(\mathbb{R}, X_0^*) \oplus_1 \alpha_+^*(\mathbb{R}, e^{-2t} dt, X_1^*)$$

such that $||T|| = ||x^*||_{(X_{\theta})^*}$ and

$$\operatorname{tr}(T_0^*S_0) + \operatorname{tr}(T_1^*S_1) = x^*(S(\mathbf{e}_\theta)), \qquad S \in \mathcal{A}$$

Let us apply this to the rank one operator $S = \varphi \otimes x$ for some $\varphi \in L^2(\mathbb{R}) \cap L^2(\mathbb{R}, e^{-2t} dt)$ and $x \in X_0 \cap X_1$. Then

$$\langle x, T_0(\varphi) \rangle + \langle x, T_1(\varphi) \rangle = x^*(x) \langle \mathbf{e}_{\theta}, \varphi \rangle,$$

so we have, by the density of $X_0 \cap X_1$, that

(3.12)
$$T_0(\varphi) + T_1(\varphi) = \langle \mathbf{e}_{\theta}, \varphi \rangle x^*, \qquad \varphi \in L^2(\mathbb{R}) \cap L^2(\mathbb{R}, \mathbf{e}^{-2t} \mathrm{d}t)$$

as functionals on X_{θ} . Let $U = e^{\theta}U_1 - I$, where U_1 is the shift operator given by $U_1\varphi = \varphi(\cdot - 1)$. Then we have

(3.13)
$$T_0(U\varphi) + T_1(U\varphi) = \left(e^{\theta} \langle e_{\theta}, U_1 \varphi \rangle - \langle e_{\theta}, \varphi \rangle\right) x^* = 0.$$

Note that

$$\begin{aligned} \|T_0 U\|_{\alpha^*_+(L^2(\mathbb{R}), X^*_0)} &\leq (\mathrm{e}^{\theta} + 1) \|x^*\|_{(X_{\theta})^*}, \\ \|T_1 U\|_{\alpha^*_+(L^2(\mathbb{R}, \mathrm{e}^{-2t} \mathrm{d}t), X^*_1)} &\leq (\mathrm{e}^{\theta - 1} + 1) \|x^*\|_{(X_{\theta})^*} \end{aligned}$$

So it follows from (3.13) that $V: L^2(\mathbb{R}) + L^2(\mathbb{R}, e^{-2t}dt) \to X_0 + X_1$ given by

$$V\varphi = \begin{cases} T_0 U\varphi, & \varphi \in L^2(\mathbb{R}) \\ -T_1 U\varphi, & \varphi \in L^2(\mathbb{R}, e^{-2t} dt) \end{cases}$$

is a well-defined element of \mathcal{A}^*_+ and $\|V\|_{\mathcal{A}^*_+} \leq (e^{\theta} + 1)\|x^*\|_{(X_{\theta})^*}$. Let us compute $V(e_{\theta})$. We have, using (3.12), that

$$V(\mathbf{e}_{\theta}) = T_0 U(\mathbf{e}_{\theta} \mathbf{1}_{(-\infty,0)}) - T_1 U(\mathbf{e}_{\theta} \mathbf{1}_{(0,\infty)})$$

= $T_0(\mathbf{e}_{\theta} \mathbf{1}_{(0,1)}) + T_1(\mathbf{e}_{\theta} \mathbf{1}_{(0,1)})$
= $\langle \mathbf{e}_{\theta}, \mathbf{e}_{\theta} \mathbf{1}_{(0,1)} \rangle x^* = x^*$

Thus we have $x^* \in X^*_{\theta,+}$ with $\|x^*\|_{X^*_{\theta,+}} \leq \|V\|_{\mathcal{A}^*_+} \leq (e^{\theta}+1)\|x^*\|_{(X_{\theta})^*}$ and the proof is complete. \Box

3.4. A comparison with real and complex interpolation

We will now compare the α -interpolation method with the more well-known real and complex interpolation methods. We will only consider the α -interpolation method in this section and leave the adaptations necessary to treat the α_+ -interpolation method to the interested reader. As in the previous section, throughout this section α is a global Euclidean structure, (X_0, X_1) is a compatible pair of Banach spaces and $0 < \theta < 1$.

Real interpolation. We will start with a formulation of the α -interpolation method in the spirit of the real interpolation method. More precisely, we will give a formulation of the α -interpolation method analogous to the Lions-Peetre mean method, which is equivalent to the real interpolation method in terms of the K-functional (see [LP64]). Let \mathcal{A}_{\bullet} be the set of all strongly measurable functions $f : \mathbb{R}_+ \to X_0 \cap X_1$ such that $t \mapsto t^j f(t) \in \alpha(\mathbb{R}_+, \frac{dt}{t}; X_j)$) for j = 0, 1. Define for $f \in \mathcal{A}_{\bullet}$

$$||f||_{\mathcal{A}_{\bullet}} := \max_{j=0,1} ||t \mapsto t^j f(t)||_{\alpha(\mathbb{R}_+, \frac{dt}{t}; X_j)}.$$

PROPOSITION 3.4.1. For $x \in X_{\theta}$ we have

$$\|x\|_{X_{\theta}} = \inf \left\{ \|f\|_{\mathcal{A}_{\bullet}} : f \in \mathcal{A}_{\bullet} \text{ with } \int_{0}^{\infty} t^{\theta} f(t) \frac{\mathrm{d}t}{t} = x \right\}$$

where the integral converges in the Bochner sense in $X_0 + X_1$.

PROOF. Note that for $f \in \mathcal{A}_{\bullet}$ we have $t \mapsto f(e^t) \in \alpha(\mathbb{R}, e^{-2jt}, X_j))$ for j = 0, 1. Therefore, using the transformation $t \mapsto e^t$, we may identify \mathcal{A}_{\bullet} with a subset of \mathcal{A} . So the inequality " \leq " is immediate.

To obtain the converse inequality note that it suffices to prove the inequality for $x \in X_0 \cap X_1 \setminus$ by Proposition 3.3.2. Let $\varepsilon > 0$ and $T \in \mathcal{A}$ with $T(\mathbf{e}_{\theta}) = x$ and $\|T\|_{\mathcal{A}} < (1 + \varepsilon) \|x\|_{X_{\theta}}$. For $\lambda > 0$ we consider the convolution operator

$$K_{\lambda}\varphi = \frac{1}{2\lambda} \int_{-\lambda}^{\lambda} \varphi(\cdot - t) \mathrm{e}^{\theta t} \, \mathrm{d}t, \qquad \varphi \in L^{2}(\mathbb{R}) + L^{2}(\mathbb{R}, \mathrm{e}^{-2t} \, \mathrm{d}t).$$

Then $K_{\lambda}(\mathbf{e}_{\theta}) = \mathbf{e}_{\theta}$, hence $TK_{\lambda}(\mathbf{e}_{\theta}) = x$. Note that for j = 0, 1

$$\|K_{\lambda}\|_{\mathcal{L}(L^{2}(\mathbb{R}, \mathrm{e}^{-2jt}\mathrm{d}t))} \leq \frac{1}{2\lambda} \int_{-\lambda}^{\lambda} \mathrm{e}^{(\theta-j)t} \,\mathrm{d}t \leq \begin{cases} \frac{\sinh(\theta\lambda)}{\theta\lambda} & j=0, \\ \frac{\sinh((1-\theta)\lambda)}{(1-\theta)\lambda} & j=1. \end{cases}$$

Hence for small enough $\lambda > 0$

(3.14)
$$||TK_{\lambda}||_{\mathcal{A}} < (1+\varepsilon)||x||_{X_{\theta}}.$$

Now we show that TK_{λ} is representable by a function. Let

$$F(t) = \begin{cases} T(\mathbf{1}_{(0,t)} \mathbf{e}_{\theta}) & t > 0, \\ -T(\mathbf{1}_{(t,0]} \mathbf{e}_{\theta}) & t \le 0, \end{cases}$$

then we have

$$K_{\lambda}\varphi = \frac{1}{2\lambda} \int_{\mathbb{R}} \varphi(t) \,\mathbf{1}_{(t-\lambda,t+\lambda)} \,\mathrm{e}^{\theta(\cdot-t)} \,\mathrm{d}t, \qquad \varphi \in L^{2}(\mathbb{R}) + L^{2}(\mathbb{R},\mathrm{e}^{-2t} \,\mathrm{d}t)$$

as a Bochner integral in $L^2(\mathbb{R}) + L^2(\mathbb{R}, e^{-2t} dt)$. Hence

(3.15)
$$TK_{\lambda}\varphi = \frac{1}{2\lambda} \int_{\mathbb{R}} \varphi(t) e^{-\theta t} \left(F(t+\lambda) - F(t-\lambda) \right) dt,$$

so we can take

$$g(t) = \frac{\mathrm{e}^{-\theta t}}{2\lambda} \big(F(t+\lambda) - F(t-\lambda) \big), \qquad t \in \mathbb{R}.$$

Then, for $f(t) = g(\ln(t))$, we have by (3.14) and (3.15)

$$\max_{j=0,1} \|t \mapsto t^j f(t)\|_{\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X_j)} = \|g\|_{\mathcal{A}} = \|TK_\lambda\|_{\mathcal{A}} \le (1+\varepsilon)\|x\|_{X_\theta},$$

which proves the inequality " \geq ".

The Lions-Peetre mean method also admits a discretized version. Using Proposition 3.4.1 we can also give a discretized version of the α -interpolation method in the same spirit. On a Banach space with finite cotype this will show that the γ -interpolation method is equivalent with the Rademacher interpolation method introduced in [**KKW06**, Section 7]. Moreover, it connects the α -interpolation method to the abstract interpolation framework developed in [LL21].

Let $\mathcal{A}_{\#}$ be the set of all infinite sequences $(x_k)_{k\in\mathbb{Z}}$ in $X_0 \cap X_1$ such that $(x_k)_{k\in\mathbb{Z}}\in\alpha(\mathbb{Z};X_0)$ and $(2^kx_k)_{k\in\mathbb{Z}}\in\alpha(\mathbb{Z};X_1)$, equipped with the norm

$$\|(x_k)_{k\in\mathbb{Z}}\|_{\mathcal{A}_{\#}} := \max\{\|(x_k)_{k\in\mathbb{Z}}\|_{\alpha(\mathbb{Z};X_0)}, \|(2^k x_k)_{k\in\mathbb{Z}}\|_{\alpha(\mathbb{Z};X_1)}\}.$$

PROPOSITION 3.4.2. For $x \in X_{\theta}$ we have

$$\|x\|_{X_{\theta}} \simeq \inf \big\{ \|\mathbf{y}\|_{\mathcal{A}_{\#}} : \mathbf{y} \in \mathcal{A}_{\#}, \sum_{k \in \mathbb{Z}} 2^{k\theta} y_k = x \big\},\$$

where the series converges in $X_0 + X_1$.

PROOF. Fix $x \in X_{\theta}$. By Proposition 3.4.1 it suffices to prove

(3.16)

$$\inf \{ \|\mathbf{y}\|_{\mathcal{A}_{\#}} : \mathbf{y} \in \mathcal{A}_{\#} \text{ with } \sum_{k \in \mathbb{Z}} 2^{k\theta} y_k = x \} \\
\simeq \inf \{ \|f\|_{\mathcal{A}_{\bullet}} : f \in \mathcal{A}_{\bullet} \text{ with } \int_0^\infty t^{\theta} f(t) \frac{\mathrm{d}t}{t} = x \}$$

First let $f \in \mathcal{A}_{\bullet}$ be such that $\int_{0}^{\infty} t^{\theta} f(t) \frac{\mathrm{d}t}{t} = x$. Define $g(t) = f(2^{t})$, then we have $\ln(2) \cdot \int_{\mathbb{R}} 2^{t\theta} g(t) dt = x$ and for j = 0, 1

(3.17)
$$\|t \mapsto 2^{jt}g(t)\|_{\alpha(\mathbb{R};X_j)} \lesssim \|t \mapsto t^j f(t)\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X_j)}$$

by the boundedness of the map $h \mapsto (t \mapsto h(2^t))$ from $L^2(\mathbb{R}_+, \frac{\mathrm{d}t}{t})$ to $L^2(\mathbb{R})$. For $k \in \mathbb{Z}$ define

$$y_k = \ln(2) \int_k^{k+1} 2^{(t-k)\theta} g(t) \, \mathrm{d}t \in X_0 \cap X_1.$$

For j = 0, 1 we have, since the functions

$$\varphi_k(t) := 2^{(t-k)(\theta-j)} \mathbf{1}_{[k,k+1)}, \qquad t \in \mathbb{R}$$

are orthogonal and uniformly bounded in $L^2(\mathbb{R})$, that

$$\|(2^{jk}y_k)_{k\in\mathbb{Z}}\|_{\alpha(\mathbb{Z};X_j)} \leq \sup_{k\in\mathbb{Z}} \|\varphi_k\|_{L^2(\mathbb{R})} \|t\mapsto 2^{jt}g(t)\|_{\alpha(\mathbb{R};X_j)}.$$

Combined with (3.17) this yields $\mathbf{y} \in \mathcal{A}_{\#}$ with

$$\|\mathbf{y}\|_{\mathcal{A}_{\#}} \lesssim \|f\|_{\mathcal{A}_{\bullet}}.$$

Since $\sum_{k \in \mathbb{Z}} 2^{k\theta} y_k = x$ this proves " \leq " of (3.16). Conversely take $\mathbf{y} \in \mathcal{A}_{\#}$ such that $\sum_{k \in \mathbb{Z}} 2^{k\theta} y_k = x$ and define

$$f(t) := \sum_{k \in \mathbb{Z}} y_k 2^{(k-t)\theta} \mathbf{1}_{[k,k+1)}(t), \qquad t \in \mathbb{R}.$$

Then $\int_{\mathbb{R}} 2^{t\theta} f(t) dt = x$ and note that $f = \sum_{k \in \mathbb{Z}} \varphi_k \otimes y_k$ with

$$\varphi_k(t) = 2^{(k-t)\theta} \mathbf{1}_{[k,k+1)}(t), \qquad t \in \mathbb{R}.$$

Since the φ_k 's are orthogonal and since we can compute the $\alpha(\mathbb{R}; X_0)$ -norm of f using a fixed orthonormal basis of $L^2(\mathbb{R})$, this implies that

$$\|f\|_{\alpha(\mathbb{R};X_0)} \leq \sup_{k \in \mathbb{N}} \|\varphi_k\|_{L^2(\mathbb{R})} \|\mathbf{y}\|_{\alpha(\mathbb{Z};X_0)} \lesssim \|\mathbf{y}\|_{\alpha(\mathbb{Z};X_0)}.$$

Combined with a similar computation for the $\alpha_1(\mathbb{R}; X_1)$ -norm of $t \mapsto 2^t f(t)$, this yields for

$$g(t) = \frac{f\left(\ln(t)/\ln(2)\right)}{\ln(2)}$$

that we have

$$\|g\|_{\mathcal{A}_{\bullet}} = \ln(2)^{-1/2} \max_{j=0,1} \|t \mapsto 2^t f(t)\|_{\alpha(\mathbb{R};X_j)} \lesssim \|\mathbf{y}\|_{\mathcal{A}_{\#}}.$$

Since $\int_0^\infty t^\theta g(t) \frac{dt}{t} = x$, this proves " \gtrsim " of (3.16).

Complex interpolation. Next we will give a formulation of the α -method in the spirit of the complex interpolation method. Denote by the strip

$$\mathbb{S} = \{ z \in \mathbb{C} : 0 < \operatorname{Re}(z) < 1 \}.$$

Let $\mathcal{H}(\overline{\mathbb{S}})$ be the space of all bounded continuous functions $f:\overline{\mathbb{S}}\to X_0+X_1$ such that

• f is a holomorphic $(X_0 + X_1)$ -valued function on \mathbb{S} .

• $f_j(t) := f(j+it)$ is a bounded, continuous, X_j -valued function for j = 0, 1.

We let $\mathcal{A}_{\mathbb{S}}$ be the subspace of all $f \in \mathcal{H}(\overline{\mathbb{S}})$ such that $f_j \in \alpha(\mathbb{R}; X_j)$ and we define

$$||f||_{\mathcal{A}_{\mathbb{S}}} := \max_{i=0,1} ||f_j||_{\alpha(\mathbb{R};X_j)}.$$

PROPOSITION 3.4.3. For $x \in X_{\theta}$ we have

$$\|x\|_{X_{\theta}} = (2\pi)^{-1/2} \inf\{\|f\|_{\mathcal{A}_{\mathbb{S}}} : f \in \mathcal{A}_{\mathbb{S}}, f(\theta) = x\}.$$

71

PROOF. Let $h_k \in C_c^{\infty}(\mathbb{R})$ and $x_k \in X_0 \cap X_1$ for $k = 1, \ldots, n$ and define

(3.18)
$$T = \sum_{k=1}^{n} h_k \otimes x_k \in \mathcal{A}.$$

Set $e_z(t) = e^{tz}$ for $z \in \mathbb{C}$. Then we have for $f(z) := T(e_z)$ and j = 0, 1 that

$$f(j - 2\pi i t) = \sum_{k=1}^{n} \int_{\mathbb{R}} h_k(\xi) e^{(j - 2\pi i t)\xi} d\xi \cdot x_k$$
$$= \mathcal{F}\left(\sum_{k=1}^{n} \xi \mapsto h_k(\xi) e^{j\xi} \cdot x_k\right).$$

Therefore, by Example 3.2.3, we have

$$\|f_j\|_{\alpha(\mathbb{R};X_j)} = (2\pi)^{-1/2} \left\| \xi \mapsto \sum_{k=1}^n h_k(\xi) e^{j\xi} \cdot x_k \right\|_{\alpha(\mathbb{R};X_j)}$$
$$= (2\pi)^{-1/2} \|T\|_{\alpha(\mathbb{R},e^{-2jt}dt;X_j)},$$

so we have $f \in \mathcal{A}_{\mathbb{S}}$ with $(2\pi)^{-1/2} ||f||_{\mathcal{A}_{\mathbb{S}}} = ||T||_{\mathcal{A}}$. Since $C_c^{\infty}(\mathbb{R})$ is dense in $L^2(\mathbb{R}) \cap L^2(\mathbb{R}, e^{-2t}dt)$, the collection of all T as in (3.18) is dense in \mathcal{A} by Proposition 3.3.2. So the inequality " \geq " follows.

For the converse let $f \in \mathcal{A}_{\mathbb{S}}$. Take $\varphi \in C_c^{\infty}(\mathbb{R})$ and let $\tilde{\varphi}(z) := \int_{\mathbb{R}} e^{-zt} \varphi(t) dt$ be its Laplace transform. Then $\tilde{\varphi}$ is entire and for any $s_1 < s_2$ we have an estimate

$$|\tilde{\varphi}(z)| \le C(1+|z|)^{-2}, \qquad s_1 \le \operatorname{Re} z \le s_2.$$

Therefore we can define

$$T\varphi := \frac{1}{2\pi} \int_{\mathbb{R}} \tilde{\varphi}(\frac{1}{2} + it) f(\frac{1}{2} + it) dt$$

as a Bochner integral in $X_0 + X_1$. An application of Cauchy's theorem shows that

$$T\varphi = \frac{1}{2\pi} \int_{\mathbb{R}} \tilde{\varphi}(s+it) f(s+it) \, \mathrm{d}t$$

for 0 < s < 1. By the dominated convergence theorem, using that f is bounded and $t \mapsto \varphi(j + it) \in L^1(\mathbb{R})$, we get for j = 0, 1

$$T\varphi = \frac{1}{2\pi} \int_{\mathbb{R}} \tilde{\varphi}(j+it) f_j(t) \, \mathrm{d}t$$

as Bochner integrals in X_j . Since we have

$$\int_{\mathbb{R}} |\tilde{\varphi}(j+it)|^2 \, \mathrm{d}t = 2\pi \int_{\mathbb{R}} |\varphi(t)|^2 \mathrm{e}^{-2jt} \mathrm{d}t < \infty, \qquad j = 0, 1,$$

and $f_j \in \alpha(\mathbb{R}, X_j)$, it follows that T extends to bounded operators

$$T_j: L^2(\mathbb{R}, \mathrm{e}^{-2jt} \mathrm{d}t) \to X_j, \qquad j = 0, 1.$$

Therefore T can be extended to be in \mathcal{A} and in particular we have

$$||T||_{\mathcal{A}} = \max_{j=0,1} ||T_j||_{\alpha(\mathbb{R}, e^{-2jt} dt; X_j)} = \max_{j=0,1} (2\pi)^{-1/2} ||f_j||_{\alpha(\mathbb{R}; X_j)} = (2\pi)^{-1/2} ||f||_{\mathcal{A}_{\mathbb{S}}}.$$

To conclude the proof of the inequality " \leq " we show that $T(e_{\theta}) = f(\theta)$. For this note that for any $\varphi \in C_c^{\infty}(\mathbb{R})$ we have

$$T(\varphi \cdot \mathbf{e}_{\theta}) = \frac{1}{2\pi} \int_{\mathbb{R}} \int_{\mathbb{R}} e^{-(\theta + it)s} \varphi(s) e^{\theta s} f(\theta + it) \, \mathrm{d}s \, \mathrm{d}t$$
$$= \frac{1}{2\pi} \int_{\mathbb{R}} \tilde{\varphi}(it) f(\theta + it) \, \mathrm{d}t.$$

Now fix φ such that $\varphi(0) = 1$ and for $n \in \mathbb{N}$ set

$$\varphi_n(t) = \varphi(nt), \qquad t \in \mathbb{R}$$

Then $\varphi_n \cdot \mathbf{e}_{\theta} \to \mathbf{e}_{\theta}$ in $L^2(\mathbb{R}) + L^2(\mathbb{R}, \mathbf{e}^{-2t} \, \mathrm{d}t)$ and therefore

$$T(\mathbf{e}_{\theta}) = \lim_{n \to \infty} T(\varphi_n \cdot \mathbf{e}_{\theta}) = \lim_{n \to \infty} \frac{1}{2\pi} \int_{\mathbb{R}} \frac{1}{n} \tilde{\varphi}(it/n) f(\theta + it) \, \mathrm{d}t = f(\theta),$$

where the last step follows from

$$\left\| t \mapsto \tilde{\varphi}(it) \right\|_{L^1(\mathbb{R})} = \left\| t \mapsto \hat{\varphi}(t/2\pi) \right\|_{L^1(\mathbb{R})} = 2\pi \cdot \varphi(0) = 2\pi$$

and [HNVW16, Theorem 2.3.8]. This concludes the proof.

A comparison of α -interpolation with real and complex interpolation. We conclude this section by comparing the α -interpolation method with the actual real and complex interpolation methods. Recall that if X_i has Fourier type $p_i \in$ [1,2] for j = 0,1, i.e. if the Fourier transform is bounded from $L^{p_j}(\mathbb{R};X_j)$ to $L^{p'_j}(\mathbb{R}; X_j)$, then by a result of Peetre [**Pee69**] we know that we have continuous embeddings

$$(3.19) (X_0, X_1)_{\theta, p} \hookrightarrow [X_0, X_1]_{\theta} \hookrightarrow (X_0, X_1)_{\theta, p'}$$

where $\frac{1}{p} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1}$. In particular the real method $(X_0, X_1)_{\theta,2}$ and the complex method $[X_0, X_1]_{\theta}$ are equivalent on Hilbert spaces. Using Proposition 3.4.2 we can prove a similar statement for the real and Gaussian interpolation method under type and cotype assumptions. Note that Fourier type p implies type p and cotype p', but the converse only holds on Banach lattices (see [**GKT96**]).

Theorem 3.4.4.

(i) If X_0 and X_1 have type $p_0, p_1 \in [1, 2]$ and cotype $q_0, q_1 \in [2, \infty]$ respectively, then we have continuous embeddings

$$(X_0, X_1)_{\theta, p} \hookrightarrow (X_0, X_1)_{\theta}^{\gamma} \hookrightarrow (X_0, X_1)_{\theta, q}$$

where $\frac{1}{p} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1}$ and $\frac{1}{q} = \frac{1-\theta}{q_0} + \frac{\theta}{q_1}$. (ii) If X_0 and X_1 have type 2, then we have the continuous embedding

$$[X_0, X_1]_{\theta} \hookrightarrow (X_0, X_1)_{\theta}^{\gamma}$$

If X_0 and X_1 have cotype 2, then we have the continuous embedding

$$(X_0, X_1)^{\gamma}_{\theta} \hookrightarrow [X_0, X_1]_{\theta}$$

(iii) If X_0 and X_1 are order-continuous Banach function spaces, then

$$(X_0, X_1)_{\theta}^{\ell^2} = [X_0, X_1]_{\theta}$$

isomorphically.

(iv) If X_0 and X_1 are Banach lattices with finite cotype, then

$$(X_0, X_1)^{\gamma}_{\theta} = (X_0, X_1)^{\ell^2}_{\theta}$$

isomorphically.

PROOF. For (i) we note that we have, by the discrete version of the Lions– Peetre mean method (see [LP64, Chapitre 2]), that

$$\|x\|_{(X_0,X_1)_{\theta,p}} \simeq \inf \Big\{ \max_{j=0,1} \|(2^{jk}y_k)_{k\in\mathbb{Z}}\|_{\ell^{p_j}(\mathbb{Z};X_j)} : \sum_{k\in\mathbb{Z}} 2^{k\theta}y_k = x \Big\},\$$

where the infimum is taken over all sequences $(y_k)_{k\in\mathbb{Z}}$ in $X_0 \cap X_1$ such that the involved norms are finite. For a finitely non-zero sequence $(y_k)_{k\in\mathbb{Z}}$ in $X_0 \cap X_1$ we have, using type p_j of X_j and Proposition 1.0.1

$$\|(2^{jk}y_k)_{k\in\mathbb{Z}}\|_{\gamma(\mathbb{Z};X_j)} \lesssim \|(2^{jk}y_k)_{k\in\mathbb{Z}}\|_{\ell^{p_j}(\mathbb{Z};X_j)}, \qquad j = 0, 1.$$

By Proposition 3.1.1 this inequality extends to any sequence in $X_0 \cap X_1$ such that the right hand-side is finite. Therefore the first embedding in (i) follows from Proposition 3.4.2. The proof of the second embedding in (i) is similar.

For (ii) let $f \in \mathcal{H}(\overline{\mathbb{S}})$. Then $g(z) := e^{z^2 - \theta^2} f(z)$ has the property that $g(\theta) = f(\theta)$ and thus by Proposition 3.4.3 and (3.5)

$$\begin{split} \|f(\theta)\|_{(X_0,X_1)_{\theta}^{\gamma}} &\leq \max_{j=0,1} \|t \mapsto g(j+it)\|_{\gamma(\mathbb{R};X_j)} \\ &\lesssim \max_{j=0,1} \|t \mapsto g(j+it)\|_{L^2(\mathbb{R};X)} \\ &\leq \max_{j=0,1} \|t \mapsto e^{(j+it)^2 - \theta^2}\|_{L^2(\mathbb{R})} \sup_{t \in \mathbb{R}} \|f(j+it)\|_X, \\ &\lesssim \sup_{t \in \mathbb{R}} \|f(j+it)\|_X, \end{split}$$

from which the first embedding follows by the definition of the complex interpolation method. For the second embedding let $f \in \mathcal{A}_{\mathbb{S}}$. Then we have by [HNVW16, Corollary C.2.11] and (3.6)

$$\begin{aligned} \|f(\theta)\|_{[X_0,X_1]_{\theta}} &\lesssim \max_{j=0,1} \|t \mapsto f(j+it)\|_{L^2(\mathbb{R};X_j)} \\ &\lesssim \max_{j=0,1} \|t \mapsto f(j+it)\|_{\gamma(\mathbb{R};X_j)}, \end{aligned}$$

from which the second embedding follows.

For (iii) denote the measure space over which X is defined by (S, μ) . Note that $[X_0, X_1]_{\theta}$ is given by the Calderón-Lozanovskii space $X_0^{1-\theta}X_1^{\theta}$, which consists of all $x \in L^0(S)$ such that $|x| = |x_0|^{1-\theta} |x_1|^{\theta}$ with $x_j \in X_j$ for j = 0, 1. The norm is given by

$$||x||_{X_0^{1-\theta}X_1^{\theta}} = \inf \{ \max_{j=0,1} ||x_j||_{X_j} : |x| = |x_0|^{1-\theta} |x_1|^{\theta}, x_0 \in X_0, x_1 \in X_1 \},\$$

see [Cal64, Loz69].

First suppose that $0 \le x \in X_0^{1-\theta} X_1^{\theta}$ factors in the form $x = |x_0|^{1-\theta} |x_1|^{\theta}$ with $x_j \in X_j$ for j = 0, 1 and $\max_{j=0,1} ||x_j||_{X_j} \le 2 ||x||_{[X_0,X_1]_{\theta}}$. We define

$$f(z) := e^{z^2 - \theta^2} |x_0|^{1-z} |x_1|^z, \qquad z \in \overline{\mathbb{S}}.$$

Then since, for j = 0, 1, we have

$$\left(\int_{\mathbb{R}} |f(j+it)(s)|^2 \, \mathrm{d}t\right)^{\frac{1}{2}} = \left(\int_{\mathbb{R}} \mathrm{e}^{2(j^2 - t^2 - \theta^2)} \, \mathrm{d}t\right)^{\frac{1}{2}} |x_j(s)|, \qquad s \in S,$$

we have by Proposition 3.1.11 that $f_j \in \ell^2(\mathbb{R}; X_j)$ and therefore $f \in \mathcal{A}_{\mathbb{S}}$. By Proposition 3.4.3 this shows that

$$\|x\|_{(X_0,X_1)_{\theta}^{\ell^2}} \le \max_{j=0,1} \|f_j\|_{\ell^2(\mathbb{R};X)} \lesssim \max_{j=0,1} \|x_j\|_{X_j} \le 2 \|x\|_{[X_0,X_1]_{\theta}}.$$

For the converse direction take $f \in \mathcal{A}_{\mathbb{S}}$. By [**HNVW16**, Lemma C.2.10(2)] with $X_0 = X_1 = \mathbb{C}$ and Hölder's inequality, we have for a.e. $s \in S$

$$|f(\theta)(s)| \lesssim \left(\left(\int_{\mathbb{R}} |f(it)(s)|^2 \, \mathrm{d}t \right)^{(1-\theta)/2} \cdot \left(\int_{\mathbb{R}} |f(1+it)(s)|^2 \, \mathrm{d}t \right)^{\theta/2} \right).$$

Therefore, by Proposition 3.1.11, we have

$$\|f(\theta)\|_{[X_0,X_1]_{\theta}} = \|f(\theta)\|_{X_0^{1-\theta}X_1^{\theta}} \lesssim \max_{j=0,1} \left\| \left(\int_{\mathbb{R}} |f(j+it)|^2 \, \mathrm{d}t \right)^{1/2} \right\|_{X_j}$$
$$= \max_{j=0,1} \|f_j\|_{\ell^2(\mathbb{R};X_j)},$$

which implies the result by Proposition 3.4.3. Finally (iv) follows directly from Proposition 1.1.3. $\hfill \Box$

CHAPTER 4

Sectorial operators and H^{∞} -calculus

On a Hilbert space H, a sectorial operator A has a bounded H^{∞} -calculus if and only if it has BIP. In this case A has even a bounded H^{∞} -calculus for operatorvalued analytic functions which commute with the resolvent of A. If A and B are resolvent commuting sectorial operators with a bounded H^{∞} -calculus, then (A, B)has a joint H^{∞} -calculus. Moreover if only one of the commuting operators has a bounded H^{∞} -calculus, then still the "sum of operators" theorem holds, i.e.

$$||Ax||_{H} + ||Bx||_{H} \le ||Ax + Bx||_{H}, \qquad x \in D(A) \cap D(B).$$

These theorems are very useful in regularity theory of partial differential operators and in particular in the theory of evolution equations. However, none of these important theorems hold in general Banach spaces without additional assumptions.

In this chapter we show that the missing "ingredient" in general Banach spaces is an α -boundedness assumption, which allows one to reduce the problem via the representation in Theorem 1.3.2 and its converse in Theorem 1.4.6 to the Hilbert space case. Indeed, rather than designing an α -bounded version of the Hilbertian proof for each of the aforementioned results, we will prove a fairly general "transference principle" (Theorem 4.4.1) adapted to this task. Our analysis will in particular shed new light on the connection between the γ -structure and sectorial operators, which has been extensively studied (see [**HNVW17**, Chapter 10] and the references therein).

In the upcoming sections we will introduce the notions of (almost) α -sectoriality, (α)-bounded H^{∞} -calculus and (α)-BIP for a sectorial operator A. We will prove the following relations between these concepts:



Implications (1), (3), (5), (6) and (9) are trivial. The 'if and only if' statement in (2) is proven in Theorem 4.3.2, implication (4) is one of our main results and is proven Theorem 4.5.6, implication (7) follows from Theorem 4.5.4, and implication (8) is

contained in Proposition 4.5.3 under the assumption that α is ideal. In the case that either $\alpha = \ell^2$ or $\alpha = \gamma$ and X has Pisier's contraction property, implications (1), (3) and (4) are 'if and only if' statements (see Theorem 4.3.5). Moreover, if X has the so-called triangular contraction property, then a bounded H^{∞} -calculus implies γ -sectoriality (see [KW01] or [HNVW17, Theorem 10.3.4]).

Besides these connections between the α -versions of the boundedness of the H^{∞} -calculus, BIP and sectoriality, we will study operator-valued and joint H^{∞} -calculus using Euclidean structures in Section 4.4. In particular, we will use our transference principle to deduce the boundedness of these calculi from α -boundedness of the H^{∞} -calculus. Moreover we will prove a sums of operators theorem.

Throughout this chapter we will keep the standing assumption that α is a Euclidean structure on X.

4.1. The Dunford calculus

In this preparatory section we will recall the definition and some well-known properties of the so-called Dunford calculus. For a detailed treatment and proofs of the statements in this section we refer the reader to [HNVW17, Chapter 10] (see also [Haa06a, Chapter 2] and [KW04, Section 9]).

If $0 < \sigma < \pi$ we denote by Σ_{σ} the sector in the complex plane given by

$$\Sigma_{\sigma} = \{ z \in \mathbb{C} : z \neq 0, |\arg z| < \sigma \}.$$

We let Γ_{σ} be the boundary of Σ_{σ} , i.e. $\Gamma_{\sigma} = \{|t|e^{i\sigma \operatorname{sgn}(t)} : t \in \mathbb{R}\}$, which we orientate counterclockwise. A closed injective operator A with dense domain D(A) and dense range R(A) is called *sectorial* if there exists a $0 < \sigma < \pi$ so that the spectrum of A, denoted by $\sigma(A)$, is contained in $\overline{\Sigma}_{\sigma}$ and the resolvent $R(\lambda, A) := (\lambda - A)^{-1}$ for $\lambda \in \mathbb{C} \setminus \sigma(A) =: \rho(A)$ satisfies

$$\sup\{\|\lambda R(\lambda, A)\| : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma}\} \le C_{\sigma}.$$

We denote by $\omega(A)$ the infimum of all σ so that this inequality holds. The definition of sectoriality varies in the literature. In particular, one could omit the dense domain, dense range and injectivity assumptions on A. However, these assumptions are not very restrictive, as one can always restrict to the part of A in $\overline{D(A) \cap R(A)}$, which has dense domain and range and is injective. Moreover if X is reflexive, then A automatically has dense domain and we have a direct sum decomposition

$$X = N(A) \oplus \overline{R(A)}.$$

For $p \in [1, \infty]$ we define the Hardy space $H^p(\Sigma_{\sigma})$ as the space of all holomorphic $f \colon \Sigma_{\sigma} \to \mathbb{C}$ such that

$$||f||_{H^p(\Sigma_{\sigma})} := \sup_{|\theta| < \sigma} ||t \mapsto f(\mathrm{e}^{i\theta} t)||_{L^p(\mathbb{R}_+, \frac{\mathrm{d}t}{t})}$$

is finite. We will mostly work with the spaces $H^1(\Sigma_{\sigma})$ and $H^{\infty}(\Sigma_{\sigma})$. For $0 < \sigma' < \sigma$ we have the continuous inclusion

(4.1)
$$H^p(\Sigma_{\sigma}) \hookrightarrow H^{\infty}(\Sigma_{\sigma'}).$$

The Dunford calculus. Let A be a sectorial operator on X, suppose $\omega(A) < \nu < \sigma < \pi$ and let $f \in H^1(\Sigma_{\sigma})$. Then we can define $f(A) \in \mathcal{L}(X)$ by the Bochner integral

(4.2)
$$f(A) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} f(z) R(z, A) \, \mathrm{d}z.$$

with norm estimate

$$||f(A)|| \le C_{\nu} ||f||_{H^1(\Sigma_{\sigma})} < \infty.$$

This is called the Dunford calculus of A. Let us note a few key properties of this calculus

- The definition is independent of ν by Cauchy's integral theorem.
- The calculus is multiplicative and thus commutative, i.e. if $f, g \in H_1(\Sigma_{\sigma})$, then f(A)g(A) = (fg)(A).
- For sectorial operators we have

$$D(A) \cap R(A) = R(A(I+A)^{-2})$$

and using the Dunford calculus for $\varphi(z) = z(1+z)^{-2}$ we have $\varphi(A) = A(I+A)^{-2}$. Thus for this φ we have $\varphi(A) \colon X \to D(A) \cap R(A)$.

Let $f \in H^1(\Sigma_{\sigma})$, $x \in D(A) \cap R(A)$ and fix $y \in X$ such that $x = \varphi(A)y$ with $\varphi(z) = z(1+z)^{-2}$. Using the multiplicativity of the calculus and Fubini's theorem we have

$$\begin{split} \int_0^\infty &\|f(tA)x\|_X \frac{\mathrm{d}t}{t} \le C_\nu \int_0^\infty \int_{\Gamma_\nu} |f(tz)| |\varphi(z)| \|y\|_X \frac{|\mathrm{d}z|}{|z|} \frac{\mathrm{d}t}{t} \\ &\le C_\nu \|f\|_{H^1(\Sigma_\sigma)} \|\varphi\|_{H^1(\Sigma_\sigma)} \|y\|_X < \infty, \end{split}$$

so $t \mapsto f(tA)x$ is Bochner integrable. Since

$$\int_0^\infty f(tz)\varphi(z)\frac{\mathrm{d}t}{t} = c\varphi(z), \qquad z \in \Sigma_\sigma$$

with $c \mathrel{\mathop:}= \int_0^\infty f(t) \frac{\mathrm{d} t}{t}$ by analytic continuation, we have the useful identity

(4.3)
$$\int_0^\infty f(tA)x \frac{\mathrm{d}t}{t} = c x, \qquad x \in D(A) \cap R(A).$$

The extended Dunford calculus. We extend will now extend the Dunford calculus to include functions like e^{-wz} and z^s , for details we refer to [Haa06a, Chapter 3] and [KW04, Section 15]. Define for $n \in \mathbb{N}$ the functions

(4.4)
$$\varphi_n(z) := \frac{n}{z+n} - \frac{1}{nz+1}, \qquad z \in \mathbb{C} \setminus (0, \infty).$$

These φ_n 's have the following properties:

- (i) $\varphi_n \in H^1(\Sigma_{\sigma})$ for all $0 < \sigma < \pi$ and $n \in \mathbb{N}$.
- (ii) By Cauchy's integral formula we have for $n \in \mathbb{N}$

$$\varphi_n(A) = -nR(-n, A) + n^{-1}R(-n^{-1}, A)$$

and thus by the sectoriality of A we have $\sup_{n \in \mathbb{N}} \|\varphi_n(A)\| < \infty$

- (iii) The range of each $\varphi_n(A)$ is $D(A) \cap R(A)$.
- (iv) For all $x \in X$ we have $\varphi_n(A)x \to x$ as $n \to \infty$.

Now suppose that f is a holomorphic function on Σ_{σ} satisfying an estimate

$$|f(z)| \le C|z|^{-\delta} (1+|z|)^{2\delta}, \qquad z \in \Sigma,$$

for some $\delta > 0$ and C > 0. For $x \in D(A^m) \cap R(A^m)$ with $m > \delta$ let $y \in X$ be such that $\varphi^m(A)y = x$ with $\varphi(z) = z(1+z)^{-2}$. Then we can define

$$f(A)x := f\varphi^m(A)y,$$

which is independent of $m > \delta$. For $x \in D(A^m) \cap R(A^m)$ we have, by the multiplicativity of the Dunford calculus, that

$$f(A)x = \lim_{n \to \infty} (f\varphi_n^m)(A)x.$$

We extend this definition to the set the set D(f(A)) of all $x \in X$ for which this limit exists. It can be shown that this defines f(A) as a closed operator with dense domain for which $D(A^m) \cap R(A^m)$ is a core. Let us note a few examples of functions that are allows in the extended Dunford calculus.

- If $\omega(A) < \pi/2$ we can take $f(z) = e^{-wz}$ for $w \in \Sigma_{\pi/2-\sigma}$. This leads to the bounded analytic semigroup $(e^{-wA})_{w \in \Sigma_{\pi/2-\sigma}}$.
- Taking $f(z) = z^w$ we obtain the fractional powers A^w for $w \in \mathbb{C}$. For $z, w \in \mathbb{C}$ we have

$$A^{z+w}x = A^z A^w x, \qquad x \in D(A^z A^w) = D(A^{z+w}) \cap D(A^w)$$

and
$$A^{z+w} = A^z A^w$$
 if $\operatorname{Re} z \cdot \operatorname{Re} w > 0$.

The fractional powers A^s for $s \in \mathbb{R}$ with $|s| < \frac{\pi}{\omega(A)}$ are sectorial operators with $\omega(A^s) = |s| \omega(A)$. For such $s \in \mathbb{R}$ we have $\varphi_s(A) = \varphi(A^s)$ with

$$\varphi_s(z) = z^s (1+z^s)^{-2}, \qquad \varphi(z) = z(1+z)^{-2}$$

by the composition rule and therefore

(4.5)
$$R(\varphi_s(A)) = R(A^s(I+A^s)^{-2}) = D(A^s) \cap R(A^s).$$

Related to these fractional powers we have for 0 < s < 1 and $f \in H^1(\Sigma_{\sigma})$ the representation formula

(4.6)
$$f(A) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} f(z) z^{-s} A^{s} R(z, A) \, \mathrm{d}z,$$

This is sometimes a useful alternative to (4.2), since $A^s R(z, A) = \varphi_z(A)$ with $\varphi_z(w) = \frac{w^s}{z-w}$ and φ_z is a $H^1(\Sigma_{\mu})$ -function for $\omega(A) < \mu < |\arg(z)|$.

4.2. (Almost) α -sectorial operators

After the preparations in the previous section, we start our investigation by studying the boundedness of the resolvent of a sectorial operator A on X. We say that A is α -sectorial if there exists a $\omega(A) < \sigma < \pi$ such that

$$\{\lambda R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma}\}$$

is α -bounded and we let $\omega_{\alpha}(A)$ be the infimum of all such σ . α -sectoriality has already been studied in the following special cases:

• \mathcal{R} -sectoriality, which is equivalent to maximal L^p -regularity (see [CP01, Wei01a]), has been studied thoroughly over the past decades (see e.g. [DHP03, KKW06, KW01, KW04]). γ -sectoriality is equivalent to \mathcal{R} -sectoriality if X has finite cotype by Proposition 1.0.1.

\$\ell^2\$-sectoriality, or more generally \$\ell^q\$-sectoriality, has previously been studied in [KU14]. We already used \$\ell^2\$-sectoriality in Subsection 2.4.

We will also study a slightly weaker notion, analogous to the notion of almost \mathcal{R} -sectoriality and almost γ -sectoriality introduced in [**KKW06**, **KW16a**]. We will say that A is almost α -sectorial if there exists a $\omega(A) < \sigma < \pi$ such that the family $\{\lambda AR(\lambda, A)^2 : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma}\}$ is α -bounded and we let $\tilde{\omega}_{\alpha}(A)$ be the infimum of all such σ . This notion will play an important role in Section 4.5 and Chapter 5.

 α -sectoriality implies almost α -sectoriality by Proposition 1.2.3. The converse is not true, as we will show in Section 6.3. If an operator is α -sectorial, then we do have equality of the angle of α -sectoriality and almost α -sectoriality.

PROPOSITION 4.2.1. Let A be an almost α -sectorial operator on X. If

$$\{tR(-t, A) : t > 0\}$$

is α -bounded, then A is α -sectorial with $\omega_{\alpha}(A) = \tilde{\omega}_{\alpha}(A)$. In particular, if A is α -sectorial, then $\omega_{\alpha}(A) = \tilde{\omega}_{\alpha}(A)$.

PROOF. Take $\tilde{\omega}_{\alpha}(A) < \sigma < \pi$ and take $\lambda = t e^{i\theta}$ for some t > 0 and $\sigma \leq |\theta| < \pi$. Suppose that $\sigma \leq \theta < \pi$, then we have

$$\lambda R(\lambda, A) + tR(-t, A) = i \int_{\theta}^{\pi} t e^{is} AR(t e^{is}, A)^2 \, \mathrm{d}s.$$

A similar formula holds if $\sigma \leq -\theta < \pi$. Now since $\{tR(-t, A) : t > 0\}$ is α -bounded and $\sigma > \tilde{\omega}_{\alpha}(A)$ we know by Proposition 1.2.3 and Corollary 1.2.4 that

$$\left\{\int_{\theta}^{\pi} t \mathrm{e}^{is} AR(t \mathrm{e}^{is}, A)^2 \, \mathrm{d}s : \sigma \le |\theta| < \pi\right\}$$

is α -bounded. Therefore $\{\lambda R(\lambda, A) : |\arg(\lambda)| \ge \sigma\}$ is α -bounded, which means that $\omega_{\alpha}(A) \le \sigma$. Combined with the trivial estimate $\tilde{\omega}_{\alpha}(A) \le \omega_{\alpha}(A)$, the proposition follows.

We can characterize almost α -sectoriality nicely using the Dunford calculus of A, for which we will need the following consequence of the maximum modulus principle.

LEMMA 4.2.2. Let $0 < \sigma < \pi$ and let Σ be an open sector in \mathbb{C} bounded by Γ_{σ} . Suppose that $f: \Sigma \cup \Gamma_{\sigma} \to \mathcal{L}(X)$ is bounded, continuous, and holomorphic on Σ . If $\{f(z): z \in \Gamma_{\sigma}\}$ is α -bounded, then $\{f(z): z \in \Sigma\}$ is α -bounded.

PROOF. Suppose that $\mathbf{x} \in X^n$ and $z \in \Sigma$, then by the maximum modulus principle we have

$$\|(x_1,\ldots,f(z)x_k,\ldots,x_n)\|_{\alpha} \leq \sup_{w\in\Gamma_{\sigma}} \|(x_1,\ldots,f(w)x_k,\ldots,x_n)\|_{\alpha}.$$

By iteration we have for $z_1, \ldots, z_n \in \Sigma$ that

$$\|(f(z_1)x_1,\ldots,f(z_n)x_n)\|_{\alpha} \le \sup_{w_1,\ldots,w_n\in\Gamma_{\sigma}} \|(f(w_1)x_1,\ldots,f(w_n)x_n)\|_{\alpha},$$

which proves the lemma.

PROPOSITION 4.2.3. Let A be a sectorial operator on X and take $\omega(A) < \sigma < \pi$. The following conditions are equivalent:

(i) A is almost α -sectorial with $\tilde{\omega}_{\alpha}(A) < \sigma$.

(ii) There is a $0 < \sigma' < \sigma$ such that for some (all) 0 < s < 1 the set $\{\lambda^s A^{1-s} R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma'}\}$

is α -bounded.

(iii) There is a $0 < \sigma' < \sigma$ such that the set

$$\{f(tA): t > 0, f \in H^1(\Sigma_{\sigma'}), \|f\|_{H^1(\Sigma_{\sigma'})} \le 1\}$$

is α -bounded.

(iv) There is a $0 < \sigma' < \sigma$ such that for all $f \in H^1(\Sigma_{\sigma'})$ the set

$$\{f(tA): t > 0\}$$

is α -bounded.

PROOF. We start by proving the implication (i) \Rightarrow (ii). Fix $\tilde{\omega}_{\alpha}(A) < \mu < \sigma$ and 0 < s < 1. For $\mu < |\theta| < \sigma$ define

$$f(z) := (e^{-i\theta}z)^{1-s}(1 - e^{-i\theta}z)^{-1}, \qquad z \in \Sigma_{\mu}$$

and set

$$F(z) := \int_0^{|z|} \frac{f(t e^{i \arg z})}{t e^{i \arg z}} \, \mathrm{d}t, \qquad z \in \Sigma_\mu$$

Let $c := \int_0^\infty \frac{f(t)}{t} \, \mathrm{d}t$ and define

$$G(z) := F(z) - c \frac{z}{1+z}, \qquad z \in \Sigma_{\mu}$$

Since there is a C > 0 such that

$$|f(z)| \le C |z|^{1-s} (1+|z|)^{-1}, \qquad z \in \Sigma_{\mu},$$

one can show that $G \in H^1(\Sigma_{\mu})$. Clearly $G'(z) = f(z)/z - c(1+z)^{-2}$, from which we can see that $zG'(z) \in H^1(\Sigma_{\mu})$ as well. Since we have for $\tilde{\omega}_{\alpha}(A) < \nu < \mu$

$$G(tA) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} G(z) R(z, tA) \,\mathrm{d}z, \qquad t > 0$$

as a Bochner integral, we may differentiate under the integral sign by the dominated convergence theorem and obtain for t>0

$$\begin{split} tAG'(tA) &= \frac{1}{2\pi i} \int_{\Gamma_{\nu}} tzG'(tz)R(z,A)\mathrm{d}z \\ &= t\frac{\mathrm{d}}{\mathrm{d}t}G(tA) \\ &= \frac{1}{2\pi i} \int_{\Gamma_{\nu}} G(z)ztAR(z,tA)^2\frac{\mathrm{d}z}{z}. \end{split}$$

Since $G \in H^1(\Sigma_{\mu})$ and tA is almost α -sectorial, it follows from Corollary 1.2.4 that the set

$$\{ (te^{i\theta})^s A^{1-s} R(te^{i\theta}, A) : t > 0 \} = \{ f(tA) : t > 0 \}$$

= $\{ tAG'(tA) + ctA(1+tA)^{-2} \}$

is α -bounded. Therefore by Lemma 1.2.3(iii) and Lemma 4.2.2 we deduce that

$$\{\lambda^s A^{1-s} R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{|\theta|}\}\$$

is α -bounded.

Next we show that (ii) \Rightarrow (iii). Fix $\sigma' < \nu < \sigma'' < \sigma$. By (4.6) we have the following representation for $f \in H^1(\Sigma_{\sigma''})$

$$f(tA) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} f(tz) z^{s} A^{1-s} R(z, A) \frac{\mathrm{d}z}{z}, \qquad t > 0.$$

Since $f(t) \in H^1(\Sigma_{\sigma''})$ independent of t > 0, it follows by Corollary 1.2.4 that

$$\{f(tA): t > 0, f \in H^1(\Sigma_{\sigma''}), \|f\|_{H^1(\Sigma_{\sigma''})} \le 1\}.$$

is α -bounded. The implication (iii) \Rightarrow (iv) is trivial.

For (iv) \Rightarrow (i) take $f(z) = e^{-i\theta} z (1 - e^{-i\theta} z)^{-2}$ with $\sigma' < |\theta| < \sigma$. Then $f \in H^1(\Sigma_{\sigma'})$, so the set

$$\{te^{i\theta}AR(te^{i\theta},A)^2:t>0\}$$

is α -bounded. Therefore by Lemma 1.2.3(iii) and Lemma 4.2.2 we deduce that

$$\{\lambda A(1+\lambda A)^{-2}:\lambda\in\mathbb{C}\setminus\overline{\Sigma}_{|\theta|}\}$$

is α -bounded and thus $\tilde{\omega}_{\alpha}(A) \leq |\theta| < \sigma$.

When $\omega(A) < \frac{\pi}{2}$, the sectorial operator A generates an analytic semigroup. In the next proposition we connect the (almost) α -sectoriality of A to α -boundedness of the associated semigroup.

PROPOSITION 4.2.4. Let A be a sectorial operator on X with $\omega(A) < \pi/2$ and take $\omega(A) < \sigma < \pi/2$. Then

(i) A is α -sectorial with $\omega_{\alpha}(A) \leq \sigma$ if and only if

 $\{e^{-zA}: z \in \Sigma_{\nu}\}$

is α -bounded for all $0 < \nu < \pi/2 - \sigma$. (ii) A is almost α -sectorial with $\tilde{\omega}_{\alpha}(A) \leq \sigma$ if and only if

 $\{zAe^{-zA}: z \in \Sigma_{\nu}\}$

is α -bounded for all $0 < \nu < \pi/2 - \sigma$.

PROOF. For the 'if' statement of (i) take $\sigma < \nu' < \nu < \pi/2$. The α -boundedness of $\{te^{\pm i\nu}R(te^{\pm i\nu}, A) : t > 0\}$ follows from the Laplace transform representation of $R(te^{\pm i\nu}, A)$ in terms of the semigroups generated by $-e^{\pm i(\pi/2-\nu')}A$ (see [HNVW17, Proposition G.4.1]) and Corollary 1.2.4. The α -boundedness of

$$\{\lambda R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu}\}$$

then follows from Lemma 4.2.2.

For the only if take $0 < \nu < \pi/2 - \sigma$ and note that by [HNVW17, Proposition 10.2.7]

$$e^{-zA} = z^{-1}R(z^{-1}, A) + f_z(A), \qquad z \in \Sigma_{\nu},$$

where $f_z(w) = e^{-zw} - (1+zw)^{-1}$. Since $f_z \in H^1(\Sigma_{\sigma})$, the α -boundedness of e^{-zA} on the boundary of Σ_{ν} follows from Proposition 4.2.3 and the α -boundedness in the interior of Σ_{ν} then follows from Lemma 4.2.2.

The proof of (ii) is similar. For the 'if' statement one uses an appropriate Laplace transform representation of $R(te^{\pm i\nu}, A)^2$ and the 'only if' statement is simpler as zwe^{-zw} is an H^1 -function.

As noted in Section 4.1, the operator A^s is sectorial as long as $|s| < \frac{\pi}{\omega(A)}$ and in this case $\omega(A^s) = |s| \omega(A)$. We end this section with a similar result for (almost) α -sectoriality.

PROPOSITION 4.2.5. Let A be a sectorial operator on X

- (i) Suppose that A is almost α -sectorial and $0 < |s| < \pi/\tilde{\omega}_{\alpha}(A)$. Then A^s is almost α -sectorial with $\tilde{\omega}_{\alpha}(A^s) = |s| \tilde{\omega}_{\alpha}(A)$.
- (ii) Suppose that A is α -sectorial and $0 < |s| < \pi/\omega_{\alpha}(A)$. Then A^s is α -sectorial with $\omega_{\alpha}(A^s) = |s| \omega_{\alpha}(A)$.

PROOF. Since A is (almost) α -sectorial if and only if A^{-1} is (almost) α -sectorial with equal angles by the resolvent identity, it suffices to consider the case s > 0. (i) follows from Proposition 4.2.3 and the fact that for $0 < s < \pi/\tilde{\omega}_{\alpha}(A)$ we have $f \in H^1(\Sigma_{\sigma})$ if and only if $g \in H^1(\Sigma_{\sigma s})$, where $g(z) = f(z^s)$.

For (ii) suppose that A is α -sectorial and fix $0 < s < \pi/\omega_{\alpha}(A)$. Define

$$\psi(z) = \frac{z - z^s}{(1 + z^s)(1 - z)}, \qquad z \in \Sigma_{\sigma}$$

and note that $\psi \in H_1(\Sigma_{\sigma})$ for $\sigma < \pi/s$. By [KW04, Lemma 15.17] we have

$$-tR(-t, A^s) = -t^{1/s}R(-t^{1/s}, A) + \psi(t^{-1/s}A), \qquad t > 0.$$

Therefore $\{-tR(-t, A^s) : t > 0\}$ is α -bounded by the α -sectoriality of A, Proposition 4.2.3 and Proposition 1.2.3. Therefore A^s is α -sectorial and by (i) and Proposition 4.2.1 we have

$$\omega_{\alpha}(A^{s}) = \tilde{\omega}_{\alpha}(A^{s}) = s \,\tilde{\omega}_{\alpha}(A) = s \,\omega_{\alpha}(A)$$

which finishes the proof.

4.3. α -bounded H^{∞} -calculus

We now turn to the study of the H^{∞} -calculus of a sectorial operator A on X, which for Hilbert spaces dates back to the ground breaking paper of McIntosh [McI86]. For Banach spaces, in particular L^p -spaces, the central paper is by Cowling, Doust, McIntosh and Yagi [CDMY96]. For examples of operators with or without a bounded H^{∞} -calculus important in the theory of evolution equations, see e.g. [Haa06a, Chapter 8], [HNVW17, Section 10.8], [KW04, Section 14] and the references therein.

We will focus on situations where the H^{∞} -calculus is α -bounded. This has already been thoroughly studied for the γ -structure, through the notion of \mathcal{R} boundedness, in [**KW01**]. For a general Euclidean structure we will first use Theorem 1.4.6 to obtain an abstract result, which we afterwards make more specific under specific assumptions on X and α .

We will briefly recall the definition of the H^{∞} -calculus and refer to [Haa06a, Chapter 2], [HNVW17, Chapter 10] or [KW04, Section 9] for a proper introduction. Note that some of these references take a slightly different, but equivalent approach to the H^{∞} -calculus.

The H^{∞} -calculus for A is an extension of the Dunford calculus to all functions in $H^{\infty}(\Sigma_{\sigma})$ for some $\omega(A) < \sigma < \pi$. Recall that for $\varphi(z) = z(1+z)^{-2}$ we have

 $R(\varphi(A)) = D(A) \cap R(A)$ and we can thus define for $f \in H^{\infty}(\Sigma_{\sigma})$ and $x \in D(A) \cap R(A)$ the map

$$f(A)x := (f\varphi)(A)y$$

where $y \in X$ is such that $x = \varphi(A)y$. This definition coincides with the extended Dunford calculus and for $f \in H^1(\Sigma_{\sigma})$ it coincides with the Dunford calculus. Moreover it it is easy to check that y = 0 implies x = 0, so f(A)x is well-defined.

By the properties of the φ_n 's as in (4.4) we have $||f(A)x||_X \leq C||x||_X$ for all $x \in D(A) \cap R(A)$ if and only if $\sup_{n \in \mathbb{N}} ||(f\varphi_n)(A)|| < \infty$. If one of these equivalent conditions hold we can extend f(A) to a bounded operator on X by density, for which we have

(4.7)
$$f(A)x = \lim_{n \to \infty} (f\varphi_n)(A)x, \qquad x \in X.$$

We say that A has a bounded H^{∞} -calculus if there is a $\omega(A) < \sigma < \pi$ such that f(A) extends to a bounded operator on X for all $f \in H^{\infty}(\Sigma_{\sigma})$ and we denote the infimum of all such σ by $\omega_{H^{\infty}}(A)$. Just like the Dunford calculus, the H^{∞} -calculus is multiplicative. We say that A has an α -bounded $H^{\infty}(\Sigma_{\sigma})$ -calculus if the set

$$\left\{f(A): f \in H^{\infty}(\Sigma_{\sigma}), \|f\|_{H^{\infty}(\Sigma_{\sigma})} \le 1\right\}$$

is α -bounded for some $\omega_{H^{\infty}}(A) < \sigma < \pi$. We denote the infimum of all such σ by $\omega_{\alpha-\mathrm{H}^{\infty}}(A)$.

We note that the $(\alpha$ -)bounded H^{∞} -calculus of A implies the $(\alpha$ -)bounded H^{∞} calculus of A^s . This follows directly from the composition rule $f(A) = g(A^s)$ for $f \in H^{\infty}(\Sigma_{\sigma})$ and $g \in H^{\infty}(\Sigma_{s\sigma})$ with $f(z) = g(z^s)$ (see e.g. [Haa06a, Theorem 2.4.2]).

PROPOSITION 4.3.1. Let A be a sectorial operator on X.

- (i) Suppose that A has a bounded H^{∞} -calculus and $0 < |s| < \pi/\omega_{H^{\infty}}(A)$. Then A^s has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A^s) = |s| \omega_{H^{\infty}}(A)$.
- (ii) Suppose that A has an α -bounded H^{∞} -calculus and $0 < |s| < \pi/\omega_{\alpha-H^{\infty}}(A)$. Then A^s has an α -bounded H^{∞} -calculus with $\omega_{\alpha-H^{\infty}}(A^s) = |s| \omega_{\alpha-H^{\infty}}(A)$.

Our first major result with respect to an α -bounded H^{∞} -calculus follows almost immediately from the transference results in Chapter 1. Indeed, using Theorem 1.4.6 we can show that one can always upgrade a bounded H^{∞} -calculus to an α -bounded H^{∞} -calculus.

THEOREM 4.3.2. Let A be a sectorial operator with a bounded H^{∞} -calculus. For every $\omega_{H^{\infty}}(A) < \sigma < \pi$ there exists a Euclidean structure α on X such that A has an α -bounded H^{∞} -calculus with $\omega_{\alpha-H^{\infty}}(A) < \sigma$.

PROOF. Fix $\omega_{H^{\infty}}(A) < \nu < \sigma$. Note that $H^{\infty}(\Sigma_{\nu})$ is a closed unital subalgebra of the C^* -algebra of bounded continuous functions on Σ_{ν} and that the algebra homomorphism $\rho \colon H^{\infty}(\Sigma_{\nu}) \to \mathcal{L}(X)$ given by $f \mapsto f(A)$ is bounded since A has a bounded $H^{\infty}(\Sigma_{\nu})$ -calculus. Therefore the set

$$\{f(A) : f \in H^{\infty}(\Sigma_{\nu}), \|f\|_{H^{\infty}(\Sigma_{\nu})} \le 1\}$$

is C^* -bounded. So by Theorem 1.4.6 we know that there is a Euclidean structure α such that A has a bounded H^{∞} -calculus with $\omega_{\alpha-\mathrm{H}^{\infty}}(A) \leq \nu$.

Control over the Euclidean structure. In general we have no control over the choice of the Euclidean structure α in Theorem 4.3.2, as we will see in Example 4.4.5. However, under certain geometric assumptions we can actually indicate a specific Euclidean structure such that A has an α -bounded H^{∞} -calculus. The following proposition will play a key role in this.

PROPOSITION 4.3.3. Let α be a global ideal Euclidean structure and assume that

$$\alpha(\mathbb{N} \times \mathbb{N}; X) = \alpha(\mathbb{N}; \alpha(\mathbb{N}; X))$$

isomorphically with constant C_{α} . Let $(U_k)_{k\geq 1}$ and $(V_k)_{k\geq 1}$ be sequences of operators in $\mathcal{L}(X)$, which for all $n \in \mathbb{N}$ satisfy

$$\| (U_1 x, \dots, U_n x) \|_{\alpha} \le M_U \| x \|_X, \qquad x \in X \| (V_1^* x^*, \dots, V_n^* x^*) \|_{\alpha^*} \le M_V \| x^* \|_{X^*}, \qquad x^* \in X^*$$

for some constants $M_U, M_V > 0$. If Γ is an α -bounded family of operators, then the family

$$\left\{\sum_{k=1}^{n} V_k T_k U_k : T_1, \dots, T_n \in \Gamma, n \in \mathbb{N}\right\}$$

is also α -bounded with bound at most $C^2_{\alpha} M_U M_V \|\Gamma\|_{\alpha}$.

PROOF. Fix $n, m \in \mathbb{N}$ and define $U: X \to \alpha(\ell_n^2; X)$ by

$$Ux = (U_1x, \dots, U_nx), \qquad x \in X$$

By assumption we have $||U|| \leq M_U$. Take $\mathbf{x} \in X^m$. Using the global ideal property of α and the isomorphism between $\alpha(\ell_{mn}^2; X)$ and $\alpha(\ell_m^2; \alpha(\ell_n^2; X))$, we have

$$\left\| (U_k x_j)_{j,k=1}^{m,n} \right\|_{\alpha(\ell_{mn}^2;X)} \le C_\alpha \left\| (Ux_j)_{j=1}^m \right\|_{\alpha(\ell_m^2;\alpha(\ell_n^2;X))} \le C_\alpha M_U \|\mathbf{x}\|_\alpha.$$

Analogously we have for any $\mathbf{x}^* \in (X^*)^m$ that

$$\left\| (V_k^* x_j^*)_{j,k=1}^{m,n} \right\|_{\alpha^*(\ell_{mn}^2;X^*)} \le C_\alpha M_V \|\mathbf{x}^*\|_{\alpha^*}.$$

Now let $S_j = \sum_{k=1}^n V_k T_{jk} U_k$ for $1 \le j \le m$ with $T_{jk} \in \Gamma \cup \{0\}$. By the duality $\alpha(\ell_m^2; X)^* = \alpha^*(\ell_m^2; X^*)$, we can pick $\mathbf{x}^* \in (X^*)^m$ such that $\|\mathbf{x}^*\|_{\alpha^*} = 1$ and

$$\left\| (S_1 x_1, \dots, S_m x_m) \right\|_{\alpha} = \sum_{j=1}^m \langle S_j x_j, x_j^* \rangle.$$

Using the α -boundedness of Γ we obtain

$$\begin{aligned} \left\| (S_{j}x_{j})_{j=1}^{m} \right\|_{\alpha} &= \sum_{j=1}^{m} \sum_{k=1}^{n} \langle T_{jk}U_{k}x_{k}, V_{k}^{*}x_{k}^{*} \rangle \\ &\leq \left\| (T_{k}U_{k}x_{j})_{j,k=1}^{m,n} \right\|_{\alpha(\ell_{mn}^{2};X)} \left\| (V_{k}^{*}x_{j}^{*})_{j,k=1}^{m,n} \right\|_{\alpha^{*}(\ell_{mn}^{2};X^{*})} \\ &\leq \left\| \Gamma \right\|_{\alpha} \left\| (U_{k}x_{j})_{j,k=1}^{m,n} \right\|_{\alpha(\ell_{mn}^{2};X)} \left\| (V_{k}^{*}x_{j}^{*})_{j,k=1}^{m,n} \right\|_{\alpha^{*}(\ell_{mn}^{2};X^{*})} \\ &\leq C_{\alpha}^{2} M_{U} M_{V} \left\| \Gamma \right\|_{\alpha} \left\| \mathbf{x} \right\|_{\alpha}. \end{aligned}$$

The theorem now follows by taking suitable T_{jk} .

Using the fact that the γ -structure is unconditionally stably, as shown in Proposition 1.1.6, we notice that Proposition 4.3.3 is a generalization of a similar statement for \mathcal{R} -boundedness in [**KW01**, Theorem 3.3].

We will also need a special case of the following lemma, which is a generalization of [HNVW17, Proposition H.2.3]. We will use the full power of this generalization in Chapter 6.

LEMMA 4.3.4. Fix $0 < \nu < \sigma < \pi$ and let $(\lambda_k)_{k=1}^{\infty}$ be a sequence in Σ_{ν} . Suppose that there is a c > 1 such that $|\lambda_{k+1}| \ge c |\lambda_k|$ for all $k \in \mathbb{N}$. For

$$g(z) := \sum_{k=1}^{\infty} a_k f(\lambda_k z), \qquad f \in H^1(\Sigma_{\sigma}), \ \boldsymbol{a} \in \ell^{\infty}$$

we have $g \in H^{\infty}(\Sigma_{\sigma-\nu})$ with $\|g\|_{H^{\infty}(\Sigma_{\sigma-\nu})} \lesssim \|a\|_{\ell^{\infty}} \|f\|_{H^{1}(\Sigma_{\sigma})}$.

PROOF. We will prove the claim on the strip

$$\mathbb{S}_{\sigma} := \{ z \in \mathbb{C} : |\mathrm{Im}(z)| < \sigma \}.$$

Define $\bar{f}: \mathbb{S}_{\sigma} \to \mathbb{C}$ by $\bar{f}(z) = f(e^z), \ \bar{g}: \mathbb{S}_{\sigma-\nu} \to \mathbb{C}$ by $\bar{g}(z) = g(e^z), \ \text{fix} \ z \in \Sigma_{\sigma-\nu}$ and set

$$\bar{\lambda}_k := \log(\lambda_k), \qquad \bar{z} := \log(z), \qquad \bar{c} := \log(c)$$

Then $|\bar{\lambda}_j - \bar{\lambda}_k| > \bar{c} > 0$, thus the disks

$$D_k := \left\{ w \in \mathbb{C} : |w - (\bar{\lambda}_k + \bar{z})| < \frac{\bar{c}}{2} \wedge \sigma \right\}, \qquad k \in \mathbb{N}$$

are pairwise disjoint and contained in \mathbb{S}_{σ} . Therefore we have, by the mean value property, that

$$\begin{split} g(z)| &\leq \|\boldsymbol{a}\|_{\ell^{\infty}} \sum_{k=1}^{\infty} \left| \bar{f}(\bar{\lambda}_{k} + \bar{z}) \right| \\ &= \|\boldsymbol{a}\|_{\ell^{\infty}} \sum_{k=1}^{\infty} \left| \frac{1}{|D_{k}|} \int_{D_{k}} \bar{f}(x + iy) \, \mathrm{d}x \, \mathrm{d}y \right| \\ &\leq \|\boldsymbol{a}\|_{\ell^{\infty}} \frac{1}{\pi(\frac{\bar{c}}{2} \wedge \sigma)^{2}} \int_{\mathbb{S}_{\frac{\bar{c}}{2} \wedge \sigma}} |\bar{f}(x + iy)| \, \mathrm{d}x \, \mathrm{d}y \\ &\leq \|\boldsymbol{a}\|_{\ell^{\infty}} \frac{1}{\pi(\frac{\bar{c}}{2} \wedge \sigma)} \sup_{|y| < \sigma} \int_{\mathbb{R}} |\bar{f}(x + iy)| \, \mathrm{d}x \\ &\lesssim \|\boldsymbol{a}\|_{\ell^{\infty}} \|f\|_{H^{1}(\Sigma_{\sigma})}. \end{split}$$

This proves the norm estimate, from which $g \in H^{\infty}(\Sigma_{\sigma-\nu})$ follows directly. \Box

We are now ready to prove some special cases in which we can indicate a Euclidean structure such that A has an α -bounded H^{∞} -calculus.

THEOREM 4.3.5. Let A be a sectorial operator on X with a bounded H^{∞} -calculus.

- (i) If X has Pisier's contraction property, then A has a γ -bounded H^{∞} calculus with $\omega_{\gamma-H^{\infty}}(A) = \omega_{H^{\infty}}(A)$.
- (ii) If X is a Banach lattice, then A has a ℓ^2 -bounded H^{∞} -calculus with $\omega_{\ell^2-H^{\infty}}(A) = \omega_{H^{\infty}}(A).$

We refer to [**Pis78**] and [**HNVW17**, Section 7.5] for the definition of Pisier's contraction property. Theorem 4.3.5(i) was already proven in [**KW01**, Theorem 5.3]. Here we will prove (i) and (ii) of Theorem 4.3.5 in a unified manner using Euclidean structures.

PROOF OF THEOREM 4.3.5. Take $\alpha = \gamma$ in case (i) and $\alpha = \ell^g$ (which is equivalent to $\alpha = \ell^2$ by Proposition 1.1.3) in case (ii). By Proposition 1.2.3 and (4.7) it suffices to show that the family of operators

$$\left\{f(A): f \in H^1(\Sigma_{\sigma}) \cap H^{\infty}(\Sigma_{\sigma}), \|f\|_{H^{\infty}(\Sigma_{\sigma})} \le 1\right\}$$

is α -bounded. For $f \in H^1(\Sigma_{\sigma}) \cap H^{\infty}(\Sigma_{\sigma})$ we compute, using the representation formula (4.6),

(4.8)
$$f(A) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} z^{-1/2} f(z) A^{\frac{1}{2}} R(z, A) dz$$
$$= \sum_{k \in \mathbb{Z}} \sum_{\epsilon = \pm 1} \frac{-\epsilon}{2\pi i} e^{\epsilon i\nu/2} \int_{1}^{2} f(e^{\epsilon i\nu} 2^{k} t) \varphi_{e^{\epsilon i\nu}}(t^{-1} 2^{-k} A) \frac{dt}{t}$$

with $\varphi_z(w) := w^{1/2}/(z-w)$.

Now fix $\omega_{H^{\infty}}(A) < \mu < \nu, 1 \leq t \leq 2$ and $\epsilon = \pm 1$. Set $\psi := \varphi_{e^{\epsilon i\nu}}^{1/2}$ and note that $\psi \in H^1(\Sigma_{\mu})$ since $e^{\epsilon i\nu} \notin \Sigma_{\mu}$. By Lemma 4.3.4 and the boundedness of the H^{∞} -calculus of A, this means that there is a $C_0 > 0$ such that for any $n \in \mathbb{N}$

$$\sup_{|\epsilon_k|=1} \left\| \sum_{k=-n}^n \epsilon_k \psi(t^{-1} 2^k A) \right\| \le C_0.$$

Note that α is unconditionally stable on X by Proposition 1.1.6. Moreover the family of multiplication operators $\{x \mapsto ax : |a| \leq 1\}$ on X is α -bounded by the right ideal property of α . Furthermore we have

$$\alpha(\mathbb{N} \times \mathbb{N}; X) = \alpha(\mathbb{N}; \alpha(\mathbb{N}; X)),$$

isomorphically, either by Pisier's contraction property if $\alpha = \gamma$ (see [HNVW17, Corollary 7.5.19]) or since α is equivalent to the ℓ^2 -structure on Banach lattices if $\alpha = \ell^g$. Therefore by Proposition 4.3.3 the family of operators

$$\Gamma_{t,\epsilon} := \left\{ \sum_{k=-n}^{n} a_k \psi(t^{-1} 2^k A)^2 : |a_{-n}|, \dots, |a_n| \le 1, n \in \mathbb{N} \right\}$$

is α -bounded and there is a constant $C_1 > 0$, independent of t and ϵ , such that $\|\Gamma_{t,\epsilon}\|_{\alpha} \leq C_1$.

Let $f_1, \ldots, f_m \in f \in H^1(\Sigma_{\sigma}) \cap H^{\infty}(\Sigma_{\sigma})$ with $||f_j||_{H^{\infty}(\Sigma_{\sigma})} \leq 1$ and take $\mathbf{x} \in X^n$. Then we have, using (4.8) in the first step, that

$$\begin{split} \left\| \left(f_{j}(A)x_{j} \right)_{j=1}^{m} \right\|_{\alpha} &\leq \frac{1}{\pi} \sup_{\epsilon = \pm 1} \left\| \left(\sum_{k \in \mathbb{Z}} \int_{1}^{2} f_{j}(\mathrm{e}^{\epsilon i\nu} 2^{k}t) \psi(t^{-1}2^{-k}A)^{2}x_{j} \frac{\mathrm{d}t}{t} \right)_{j=1}^{m} \right\|_{\alpha} \\ &\leq \sup_{\epsilon = \pm 1} \sup_{1 \leq t \leq 2} \sup_{n \in \mathbb{N}} \left\| \left(\sum_{k=-n}^{n} f_{j}(\mathrm{e}^{\epsilon i\nu} 2^{k}t) \psi(t^{-1}2^{-k}A)^{2}x_{j} \right)_{j=1}^{m} \right\|_{\alpha} \\ &\leq \sup_{\epsilon = \pm 1} \sup_{1 \leq t \leq 2} \left\| \Gamma_{t,\epsilon} \right\|_{\alpha} \left\| \mathbf{x} \right\|_{\alpha} \leq C_{1} \| \mathbf{x} \|_{\alpha}. \end{split}$$

Hence we see that A has an α -bounded H^{∞} -calculus with $\omega_{\alpha-H^{\infty}}(A) \leq \sigma$.

4.4. Operator-valued and joint H^{∞} -calculus

In this section we will study of the operator-valued and joint functional calculus for sectorial operators by reducing the problem to the Hilbert space case via Euclidean structures and the general representation theorem (Theorem 1.3.2). We will also deduce a theorem on the closedness of the sum of two commuting sectorial operators.

The idea of an operator-valued H^{∞} -calculus goes back to Albrecht, Franks and McIntosh [**AFM98**] in Hilbert spaces. For the construction we take $0 < \sigma < \pi$, $\Gamma \subseteq \mathcal{L}(X)$ and for $p \in [1, \infty]$ let $H^p(\Sigma_{\sigma}; \Gamma)$ be the set of all holomorphic functions $f: \Sigma_{\sigma} \to \Gamma$ such that

$$\|f\|_{H^p(\Sigma_{\sigma};\Gamma)} := \sup_{|\theta| < \sigma} \|t \mapsto f(e^{i\theta}t)\|_{L^p(\mathbb{R}_+,\frac{dt}{t};\mathcal{L}(X))}$$

is finite.

Take $\omega(A) < \nu < \sigma < \pi$ and let Γ be a family of bounded operators on X which commute with the resolvent of A. Then we define f(A) for $f \in H^1(\Sigma_{\sigma}; \Gamma)$ by the contour integral

(4.9)
$$f(A) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} f(z) R(z, A) \, \mathrm{d}z.$$

We define for $f \in H^{\infty}(\Sigma_{\sigma}; \Gamma)$ and $x \in D(A) \cap R(A)$

 $f(A)x := (f\varphi)(A)y$

where $y \in X$ is such that $x = \varphi(A)y$ with $\varphi(z) = z(1+z)^{-2}$. As for the H^{∞} calculus, this coincides with (4.9) for $f \in H^1(\Sigma_{\sigma}; \Gamma)$.

If $||f(A)x||_X \leq C ||x||_X$ for all $x \in D(A) \cap R(A)$ or equivalently if for φ_n as in (4.4) we have $\sup_{n \in \mathbb{N}} ||(f\varphi_n)(A)|| < \infty$, we can extend f(A) to a bounded operator on X by density. We can then approximate f(A)x as in (4.7) for the H^{∞} -calculus.

If there is a $\omega(A) < \sigma < \pi$ such that f(A) extends to a bounded operator on X for all $f \in H^{\infty}(\Sigma_{\sigma}; \Gamma)$ we say that A has a bounded $H^{\infty}(\Gamma)$ -calculus and we denote the infimum of all such σ by $\omega_{H^{\infty}(\Gamma)}(A)$. If the set

$$\left\{f(A): f \in H^{\infty}(\Sigma_{\sigma}), \|f\|_{H^{\infty}(\Sigma_{\sigma};\Gamma)} \leq 1\right\}$$

is α -bounded for some $\omega_{H^{\infty}(\Gamma)}(A) < \sigma < \pi$ we say that A has a α -bounded $H^{\infty}(\Gamma)$ calculus and we denote the infimum of all such σ by $\omega_{\alpha-\mathrm{H}^{\infty}(\Gamma)}(A)$.

We also want to study the joint functional calculus, first introduced in [Alb94] by Albrecht. For this let (A, B) be a pair of sectorial operators which commute in the sense that $R(\lambda, A)$ and $R(\mu, B)$ commute for all $\lambda \in \rho(A)$ and $\mu \in \rho(B)$. Under these hypotheses

$$DR(A,B) := D(A) \cap D(B) \cap R(A) \cap R(B)$$

is dense in X. Indeed, DR(A, B) is the range of $\varphi_n(A)\varphi_n(B)$ for each $n \in \mathbb{N}$ and $x = \lim_{n \to \infty} \varphi_n(A)\varphi_n(B)x$.

Suppose that $\omega(A) < \nu_A < \sigma_A < \pi$ and $\omega(B) < \nu_B < \sigma_B < \pi$ and $p \in [1, \infty]$. W let $H^p(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ be the set of all holomorphic $f \colon \Sigma_{\sigma_A} \times \Sigma_{\sigma_B} \to \mathbb{C}$ such that

$$\|f\|_{H^p(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})} := \sup_{|\theta_A| < \sigma_A, |\theta_B| < \sigma_B} \|(s,t) \mapsto f(\mathrm{e}^{i\theta_A}s, \mathrm{e}^{i\theta_B}t)\|_{L^p(\mathbb{R}_+ \times \mathbb{R}_+, \frac{\mathrm{d}s}{s} \frac{\mathrm{d}t}{t})}$$

is finite. Then for $f \in H^1(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ we can define the operator f(A, B) by

(4.10)
$$f(A,B) = -\frac{1}{4\pi^2} \int_{\Gamma_{\nu_A}} \int_{\Gamma_{\nu_B}} f(z,w) R(z,A) R(w,B) \, \mathrm{d}w \, \mathrm{d}z.$$

If $f \in H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ and $x \in DR(A, B)$ we define f(A, B)x by

 $f(A, B)x := (f\varphi)(A, B)y,$

where $y \in X$ is such that $x = \varphi(A, B)y$ with

$$\varphi(z,w) = z(1+z)^{-2}w(1+w)^{-2}, \qquad (z,w) \in \Sigma_{\sigma_A} \times \Sigma_{\sigma_B}.$$

Again this calculus is well-defined and it coincides with (4.10) if $f \in H^1(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$.

As before for $f \in H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ we have that f(A, B) extends to a bounded operator on X if $||f(A, B)x||_X \leq C||x||_X$ for all $x \in DR(A, B)$ or equivalently if

$$\sup_{n\in\mathbb{N}}\|(f\psi_n)(A,B)\|<\infty$$

where $\psi_n(z, w) = \varphi_n(z)\varphi_n(w)$. We say that (A, B) has a bounded joint H^{∞} calculus if there are $\omega(A) < \nu_A < \sigma_A < \pi$ and $\omega(B) < \nu_B < \sigma_B < \pi$ such that f(A, B) extends to a bounded operator for all $f \in H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ and denote the infimum over all such (σ_A, σ_B) by $\omega_{H^{\infty}}(A, B)$.

A general transference principle. Our main results in this and the next section will be based on the following transference principle, which basically tells us that the α -bounded versions of the introduced properties of sectorial operators may be studied in the Hilbert space setting.

THEOREM 4.4.1. Suppose that A and B are resolvent commuting α -sectorial operators on X. Take $\omega_{\alpha}(A) < \sigma_A < \pi$ and $\omega_{\alpha}(B) < \sigma_B < \pi$.

- Let Ξ_A and Ξ_B be subsets of $H^{\infty}(\Sigma_{\sigma_A})$ and $H^{\infty}(\Sigma_{\sigma_B})$ such that $\{f(A) : f \in \Xi_A\}$ and $\{f(B) : f \in \Xi_B\}$ are α -bounded.
- Let Γ_A be an α -bounded subset of $\mathcal{L}(X)$, which commutes with the resolvent of A.

Then there is a Hilbert space H and resolvent commuting operators \widetilde{A} and \widetilde{B} on H with $\omega(\widetilde{A}) < \sigma_A$ and $\omega(\widetilde{B}) < \sigma_B$, so that:

(i) There is a C > 0 such that

$$\sup\{\|f(A)\|_{\mathcal{L}(H)} : f \in \Xi_A\} \le C,$$
$$\sup\{\|f(\widetilde{B})\|_{\mathcal{L}(H)} : f \in \Xi_B\} \le C.$$

(ii) There is a C > 0 such that for all $f \in H^1(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ we have

$$\|f(A,B)\|_{\mathcal{L}(X)} \le C \|f(A,B)\|_{\mathcal{L}(H)}.$$

(iii) There is a C > 0 and a bounded subset $\widetilde{\Gamma}_A$ of $\mathcal{L}(H)$ commuting with the resolvent of \widetilde{A} such that all $f \in H^1(\Sigma_{\sigma_A}; \Gamma_A)$ there is a $\widetilde{f} \in H^1(\Sigma_{\sigma_A}; \widetilde{\Gamma}_A)$ with

$$\|f(A)\|_{\mathcal{L}(X)} \le C \|f(A)\|_{\mathcal{L}(H)}$$

PROOF. Fix $\omega_{\alpha}(A) < \nu_A < \sigma_A$ and $\omega_{\alpha}(B) < \nu_B < \sigma_B$. Define

$$\Gamma_{0} = \{\lambda R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_{A}}\} \cup \{\lambda R(\lambda, B) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_{B}}\} \\ \cup \{f(A) : f \in \Xi_{A}\} \cup \{f(B) : f \in \Xi_{B}\} \cup \Gamma_{A}$$

Let Γ be the closure in the strong operator topology of the absolutely convex hull of

$$\{T_1T_2T_3: T_1, T_2, T_3 \in \Gamma_0 \cup \{I\}\},\$$

where I denotes the identity operator on X. Then Γ is α -bounded by Proposition 1.2.3. Denote by $\mathcal{L}_{\Gamma}(X)$ the linear span of Γ normed by the Minkowski functional

$$||T||_{\Gamma} = \inf\{\lambda > 0 : \lambda^{-1}T \in \Gamma\}.$$

Then the map $z \mapsto R(z, A)$ is continuous from $\mathbb{C} \setminus \overline{\Sigma}_{\nu_A}$ to $\mathcal{L}_{\Gamma}(X)$. This follows directly from the fact that for $z, w \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_A}$ we have

$$R(z,A) - R(w,A) = (z^{-1} - w^{-1})zwR(z,A)R(w,A) \in (z^{-1} - w^{-1})\Gamma.$$

The same holds for the map $z \mapsto R(z, B)$ from $\mathbb{C} \setminus \overline{\Sigma}_{\nu_B}$ to $\mathcal{L}_{\Gamma}(X)$. Analogously, the map $(z, w) \mapsto R(z, A)R(w, B)$ is continuous from $(\mathbb{C} \setminus \overline{\Sigma}_{\nu_A}) \times (\mathbb{C} \setminus \overline{\Sigma}_{\nu_B})$ to $\mathcal{L}_{\Gamma}(X)$.

By Theorem 1.3.2 there is a closed subalgebra \mathcal{B} of $\mathcal{L}(H_0)$ for some Hilbert space H_0 , a bounded algebra homomorphism $\rho : \mathcal{B} \to \mathcal{L}(X)$ and a bounded linear operator $\tau : \mathcal{L}_{\Gamma}(X) \to \mathcal{B}$ so that $\rho\tau(T) = T$ for all $T \in \mathcal{L}_{\Gamma}(X)$. Furthermore, τ extends to an algebra homomorphism on the algebra \mathcal{A} generated by Γ .

Set $R_A(z) = \tau(R(z, A))$ for $z \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_A}$ and $R_B(z) = \tau(R(z, B))$ for $z \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_B}$. Then, since τ is an algebra homomorphism on \mathcal{A} , we know that R_A and R_B are commuting functions which obey the resolvent equations

$$R_A(z) - R_A(w) = (w - z)R_A(z)R_A(w), \qquad z, w \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_A},$$

$$R_B(z) - R_B(w) = (w - z)R_B(z)R_B(w), \qquad z, w \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_B}.$$

Furthermore we have

(4.11)
$$\sup_{\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_A}} \|\lambda R_A(\lambda)\| \le \|\tau\|, \qquad \sup_{\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_B}} \|\lambda R_B(\lambda)\| \le \|\tau\|.$$

Finally we note that, since $z \to R(z, A)$ is continuous from $\mathbb{C} \setminus \overline{\Sigma}_{\nu_A}$ into $\mathcal{L}_{\Gamma}(X)$, the map R_A is also continuous. A similar statement holds for R_B . Therefore it follows from the resolvent equation that both R_A and R_B are holomorphic.

Now let H be the subspace of H_0 of all $\xi \in H_0$ such that

(4.12)
$$\lim_{t \to \infty} \xi + tR_A(-t)\xi = \lim_{t \to 0} tR_A(-t)\xi = 0, \\ \lim_{t \to \infty} \xi + tR_B(-t)\xi = \lim_{t \to 0} tR_B(-t)\xi = 0.$$

As the operators $tR_A(-t)$ and $tR_B(-t)$ are uniformly bounded for t > 0, H is closed. Moreover, since R_A and R_B commute, $R_A(z)(H) \subseteq H$ for $z \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_A}$ and $R_B(z)(H) \subseteq H$ for $z \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_B}$.

For φ_n as in (4.4) we have $\varphi_n(A), \varphi_n(B) \in \mathcal{L}_{\Gamma}(X)$ with

(4.13)
$$\sup_{n \in \mathbb{N}} \|\varphi_n(A)\|_{\Gamma} \le C, \qquad \sup_{n \in \mathbb{N}} \|\varphi_n(B)\|_{\Gamma} \le C.$$

Moreover we claim that for all $n \in \mathbb{N}$ we have

(4.14)
$$\tau(\varphi_n(A)\varphi_n(B))(H_0) \subseteq H.$$

To prove this claim it suffices to show that if $\xi = \tau(\varphi_n(A))\eta$ for some $\eta \in H_0$, then

(4.15)
$$\lim_{t \to \infty} \xi + tR_A(-t)\xi = \lim_{t \to 0} tR_A(-t)\xi = 0$$

and an identical statement for B. We have

$$F(\varphi_n(A)) = n^{-1}R_A(-n^{-1}) - nR_A(-n)$$

and therefore if $t \neq n, n^{-1}$ we have

$$tR_A(-t)\tau(\varphi_n(A)) = tn^{-1}(t-n^{-1})^{-1} (R_A(-t) - R_A(-n^{-1})) - tn(t-n)^{-1} (R_A(-t) - R_A(-n)).$$

Combined with the uniform boundedness of $tR_A(-t)$ one can deduce (4.15) by taking the limits $t \to 0$ and $t \to \infty$ on each of the terms in this expression.

We can now define the sectorial operator A on H using R_A . For $\xi \in H$ we have by the resolvent equation that if $R_A(z)\xi = 0$ for some $z \in \mathbb{C} \setminus \overline{\Sigma}_{\nu_A}$ we have $tR_A(-t)\xi = 0$ for all t > 0. Hence $R_A(z)|_H$ is injective by (4.12). As domain we take the range of $R_A(-1)$ and define

$$\widetilde{A}(R_A(-1)\xi) := -\xi - R_A(-1)\xi, \qquad \xi \in H$$

Then \widetilde{A} is injective and has dense domain and range by (4.12) (See [EN00, Section II.4.a] and [HNVW17, Proposition 10.1.7(3)] for the details). Moreover by the resolvent equation we have $R(\cdot, \widetilde{A}) = R_A|_H$ and thus \widetilde{A} is sectorial on H with $\omega(\widetilde{A}) \leq \nu_A < \sigma_A$ by (4.11). We make a similar definition for \widetilde{B} .

Finally, we turn to the inequalities in (i)-(iii). For (i) take $f \in \Xi_A$ and let $\omega(\tilde{A}) < \mu_A < \sigma_A$. For any $n \in \mathbb{N}$ we have

$$(f\varphi_n)(A) = \frac{1}{2\pi i} \int_{\Gamma_{\mu_A}} f(z)\varphi_n(z)R(z,A) \,\mathrm{d}z$$

and this integral converges as a Bochner integral in $\mathcal{L}_{\Gamma}(X)$. Therefore, using the boundedness of τ , we have

$$(f\varphi_n)(\widetilde{A})\xi = \tau((f\varphi_n)(A))\xi, \quad \xi \in H.$$

By the multiplicativity of the H^{∞} -calculus, the boundedness of τ and (4.13) we obtain that there is a C > 0 such that for any $n \in \mathbb{N}$

$$\|(f\varphi_n)(A)\|_{\mathcal{L}(H)} \le \|\tau\|\|(f\varphi_n)(A)\|_{\Gamma} \le C.$$

We can prove an analogous estimate for $f(\tilde{B})$ for any $f \in \Xi_B$ and thus (i) follows. For (ii) take $f \in H^1(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$. We can express f(A, B) as a Bochner integral

in $\mathcal{L}_{\Gamma}(X)$ using (4.10). By the boundedness of τ we conclude that

$$\tau(f(A,B))\xi = f(A,B)\xi, \qquad \xi \in H$$

Fix $n \in \mathbb{N}$. Using the fact that τ extends to an algebra homomorphism on the algebra generated by Γ and (4.14), we have for $n \in \mathbb{N}$

$$\tau \big(f(A, B) \varphi_n(A) \varphi_n(B) \big) \eta = f(A, B) \tau \big(\varphi_n(A) \varphi_n(B) \big) \eta, \qquad \eta \in H_0$$

This means, by the boundedness of τ and (4.13), that

$$\|\tau(f(A,B)\varphi_n(A)\varphi_n(B))\|_{\mathcal{L}(H_0)} \le C_0 \|f(A,B)\|_{\mathcal{L}(H)}$$

with $C_0 > 0$ independent of f and n. Since ρ is also bounded this implies by a limiting argument that

$$\left\|f(A,B)\right\|_{\mathcal{L}(X)} \le C_1 \left\|f(\widetilde{A},\widetilde{B})\right\|_{\mathcal{L}(H)}$$

with $C_1 > 0$ again independent of f, proving (ii).

Finally for (iii) take $f \in H^1(\Sigma_{\sigma_A}; \Gamma_A)$. We can express f(A) as a Bochner integral in $\mathcal{L}_{\Gamma}(X)$ using (4.9). Define $\widetilde{\Gamma}_A := \{\tau(T) : T \in \Gamma_A\}$ and $\widetilde{f}(z) := \tau(f(z))$. By the boundedness of τ we have

$$\tau(f(A))\xi = \widetilde{f}(\widetilde{A})\xi, \qquad \xi \in H.$$

Arguing analogously to the proof of (ii) we can now deduce that

$$\|f(A)\|_{\mathcal{L}(X)} \le C \|f(A)\|_{\mathcal{L}(H)},$$

proving (iii).

The operator-valued H^{∞} -calculus. On a Hilbert space, any sectorial operator with a bounded H^{∞} -calculus has a bounded operator-valued H^{∞} -calculus, a result that is implicit in [LM96] (see also [LLL98, Remark 6.5] and [AFM98]). As a first application of the transference principle of Theorem 4.4.1 we obtain an analog of this statement in Banach spaces under additional α -boundedness assumptions. Similar results using \mathcal{R} -boundedness techniques are contained in [KW01, LLL98].

THEOREM 4.4.2. Suppose that A is a sectorial operator on X with an α -bounded H^{∞} -calculus. Let Γ be an α -bounded subset of $\mathcal{L}(X)$ which commutes with the resolvent of A. Then A has a bounded $H^{\infty}(\Gamma)$ -calculus with $\omega_{H^{\infty}(\Gamma)}(A) \leq \omega_{\alpha-H^{\infty}}(A)$.

PROOF. Fix $\omega_{\alpha-\mathrm{H}^{\infty}}(A) < \sigma < \pi$. We apply the transference principle of Theorem 4.4.1 to the sectorial operator A with $\Xi_A = H^{\infty}(\Sigma_{\sigma})$ and $\Gamma_A = \Gamma$. Then there is a sectorial operator \widetilde{A} on a Hilbert space H and a uniformly bounded family of operators $\widetilde{\Gamma}$ on H such that for all $f \in H^1(\Sigma_{\sigma}; \Gamma)$ there is a $\widetilde{f} \in H^1(\Sigma_{\sigma}; \widetilde{\Gamma})$ with

$$\|f(A)\|_{\mathcal{L}(X)} \le C \|f(A)\|_{\mathcal{L}(H)}.$$

As stated before the theorem, any sectorial operator on a Hilbert space with a bounded H^{∞} -calculus has a bounded operator-valued H^{∞} -calculus. So for any $f \in H^{\infty}(\Sigma_{\sigma}; \Gamma)$ we have

$$\sup_{n \in \mathbb{N}} \|(f\varphi_n)(A)\| \le C \sup \left\{ \|\widetilde{f}(\widetilde{A})\| : \widetilde{f} \in H^1(\Sigma_{\sigma}; \widetilde{\Gamma}) \right\} \le C,$$

which shows that A has a bounded $H^{\infty}(\Gamma)$ -calculus with $\omega_{H^{\infty}(\Gamma)}(A) \leq \sigma$.

In Theorem 4.4.2 we cannot avoid the α -boundedness assumptions. In [LLL98] it is shown that if the conclusion of Theorem 4.4.2 holds for all sectorial operators with a bounded H^{∞} -calculus and for all bounded and resolvent commuting families $\Gamma \subseteq \mathcal{L}(X)$, then X is isomorphic to a Hilbert space.

We can combine Theorem 4.3.5 and Theorem 4.4.2 to improve Theorem 4.4.2 in case the Euclidean structure α is either the γ - or the ℓ^2 -structure. A similar result using \mathcal{R} -boundedness can be found in [**KW01**, Theorem 4.4].

COROLLARY 4.4.3. Let A be a sectorial operator on X with a bounded H^{∞} calculus and let Γ be a subset of $\mathcal{L}(X)$ which commutes with the resolvent of A.

(i) If X has Pisier's contraction property and Γ is γ -bounded, then A has a γ -bounded $H^{\infty}(\Gamma)$ -calculus with $\omega_{\gamma-\mathrm{H}^{\infty}(\Gamma)}(A) \leq \omega_{H^{\infty}}(A)$.

(ii) If X is a Banach lattice and Γ is ℓ^2 -bounded, then A has an ℓ^2 -bounded $H^{\infty}(\Gamma)$ -calculus with $\omega_{\ell^2-\mathrm{H}^{\infty}(\Gamma)}(A) \leq \omega_{H^{\infty}}(A)$.

PROOF. Either take $\alpha = \gamma$ or $\alpha = \ell^2$. By Theorem 4.3.5 we know that A has an α -bounded H^{∞} -calculus with $\omega_{\alpha-H^{\infty}}(A) = \omega_{H^{\infty}}(A)$. Then by Theorem 4.4.2 we know that A has a bounded $H^{\infty}(\Gamma)$ -calculus with $\omega_{H^{\infty}(\Gamma)}(A) \leq \omega_{H^{\infty}}(A)$. Finally, by a repetition of the proof of Theorem 4.3.5 using the operator family

$$\Gamma_{t,\epsilon} := \left\{ \sum_{k=-n}^{n} T_k \psi(t^{-1} 2^k A)^2 : T_{-n}, \dots, T_n \in \Gamma, n \in \mathbb{N} \right\}$$

we can prove that A has a α -bounded $H^{\infty}(\Gamma)$ -calculus with $\omega_{\alpha-\mathrm{H}^{\infty}(\Gamma)}(A) \leq \omega_{H^{\infty}}(A)$.

The joint H^{∞} -calculus. On a Hilbert space any pair of resolvent commuting sectorial operators with bounded H^{∞} -calculi has a bounded joint H^{∞} -calculus (see [AFM98, Corollary 4.2]). Moreover the converse of this statement is trivial. Again using the transference principle of Theorem 4.4.1 we obtain a characterization of the boundedness of the joint H^{∞} -calculus of a pair of commuting sectorial operators (A, B) on a Banach space X in terms of the α -boundedness of the H^{∞} -calculi of A and B.

THEOREM 4.4.4. Suppose that A and B are resolvent commuting sectorial operators on X.

(i) If A and B have an α-bounded H[∞]-calculus, then (A, B) has a bounded joint H[∞]-calculus with

 $\omega_{H^{\infty}}(A, B) \leq (\omega_{\alpha \cdot H^{\infty}}(A), \omega_{\alpha \cdot H^{\infty}}(B)).$

(ii) If (A, B) has a bounded joint H^{∞} -calculus, then for any

 $\omega_{H^{\infty}}(A,B) < (\sigma_A,\sigma_B) < (\pi,\pi)$

there is a Euclidean structure α such that A and B have α -bounded H^{∞} -calculi with $\omega_{\alpha-H^{\infty}}(A) \leq \sigma_A$ and $\omega_{\alpha-H^{\infty}}(B) \leq \sigma_B$.

PROOF. The first part is a typical application of Theorem 4.4.1. Let $\omega_{\alpha-\mathrm{H}^{\infty}}(A) < \sigma_A < \pi$ and $\omega_{\alpha-\mathrm{H}^{\infty}}(B) < \sigma_B < \pi$. Using Theorem 4.4.1 we can find a pair of resolvent commuting sectorial operators \widetilde{A} and \widetilde{B} on a Hilbert space H such that $\omega_{H^{\infty}}(\widetilde{A}) < \sigma_A, \omega_{H^{\infty}}(\widetilde{B}) < \sigma_B$ and such that

$$\|f(A,B)\|_{\mathcal{L}(X)} \le C \|f(A,B)\|_{\mathcal{L}(H)}$$

for all $f \in H^1(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$. On a Hilbert space any pair of sectorial operators with a bounded H^{∞} -calculus has a bounded joint H^{∞} -calculus (see [**AFM98**, Corollary 4.2]), so by approximation this proves the first part.

For the second part note that $H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ is a closed unital subalgebra of the C^* -algebra of bounded continuous functions on $\Sigma_{\sigma_A} \times \Sigma_{\sigma_B}$ and that the algebra homomorphism $\rho \colon H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B}) \to \mathcal{L}(X)$ given by $f \mapsto f(A, B)$ is bounded since (A, B) has a bounded H^{∞} -calculus. Therefore the set

$$\{f(A,B): f \in H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B}), \|f\|_{H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})} \le 1\}$$

is C^* -bounded, from which the claim follows by Theorem 1.4.6 and restricting to functions $f: \Sigma_{\sigma_A} \times \Sigma_{\sigma_B} \to \mathbb{C}$ that are constant in one of the variables. \Box

As in the operator-valued H^{∞} -calculus case, in Theorem 4.4.4 we cannot omit the assumption of an α -bounded H^{∞} -calculus. We illustrate this with an example, see also [**KW01**, **LLL98**].

EXAMPLE 4.4.5. Consider the Schatten class S^p for $p \in (1, \infty)$. We represent a member $x \in S^P$ by an infinite matrix, i.e. $x = (x_{jk})_{j,k=1}^{\infty}$, and define $Ax = (2^j x_{jk})_{j,k=1}^{\infty}$ with as domain the set of all $x \in S^p$ such that $Ax \in S^p$. Analogously we define $Bx = (2^k x_{jk})_{j,k=1}^{\infty}$. Then A and B are both sectorial operators with bounded H^{∞} -calculus and $\omega_{H^{\infty}}(A) = \omega_{H^{\infty}}(B) = 0$. However (A, B) do not have a bounded joint H^{∞} -calculus for any choice of angles, unless p = 2 (see [LLL98, Theorem 3.9]).

REMARK 4.4.6. In particular Example 4.4.5 shows that the Euclidean structure α given by Theorem 4.3.2 for A must fail the ideal property. Indeed, let α be an ideal Euclidean structure such that A has an α -bounded H^{∞} -calculus. Then for any $f \in H^{\infty}(\Sigma_{\sigma})$ we have f(B) = Tf(A)T, where T is the transpose operator on S^p . So B has an α -bounded H^{∞} -calculus as well by the ideal property of α and therefore Theorem 4.4.4 would imply that (A, B) has a bounded joint H^{∞} -calculus, a contradiction with Example 4.4.5.

We can combine Theorem 4.3.5 and Theorem 4.4.4 to recover the following result of Lancien, Lancien and Le Merdy [LLL98] (see also [AFM98, FM98, KW01]).

COROLLARY 4.4.7. Suppose that X has Pisier's contraction property or is a Banach lattice. Let A and B be resolvent commuting sectorial operators on X with a bounded H^{∞} -calculus. Then (A, B) has a bounded joint H^{∞} -calculus with $\omega_{H^{\infty}}(A, B) = (\omega_{H^{\infty}}(A), \omega_{H^{\infty}}(B)).$

The sum of closed operators. We end this section with a sum of closed operators theorem. It is well known that an operator-valued H^{∞} -calculus implies theorems on the closedness of the sum of commuting operators, see e.g. [AFM98, KW01, LLL98] and [KW04, Theorem 12.13]. However, here we prefer to employ the transference principle of Theorem 4.4.1 once more.

THEOREM 4.4.8. Let A and B be resolvent commuting sectorial operators on X. Suppose that A has an α -bounded H^{∞} -calculus and B is α -sectorial with $\omega_{\alpha \cdot H^{\infty}}(A) + \omega_{\alpha}(B) < \pi$. Then A + B is closed on $D(A) \cap D(B)$ and

$$||Ax||_X + ||Bx||_X \leq ||Ax + Bx||_X, \quad x \in D(A) \cap D(B).$$

Moreover A + B is sectorial with $\omega(A + B) \leq \max\{\omega_{\alpha - H^{\infty}}(A), \omega_{\alpha}(B)\}.$

PROOF. Take $\sigma_A > \omega_{H^{\infty}}(A)$ and $\sigma_B > \omega_{\alpha}(B)$ with $\sigma_A + \sigma_B < \pi$. Choose $\Xi_A = H^{\infty}(\Sigma_{\sigma_A})$ and apply Theorem 4.4.1 to find a Hilbert space H and resolvent commuting sectorial operators \widetilde{A} , \widetilde{B} on H with $\omega_{H^{\infty}}(\widetilde{A}) < \sigma_A$ and $\omega(\widetilde{B}) < \sigma_B$. By the sum of operators theorem on Hilbert spaces due to Dore and Venni [**DV87**, Remark 2.11] (see also [**AFM98**]) we deduce that $\widetilde{A} + \widetilde{B}$ is a sectorial operator on $D(\widetilde{A}) \cap D(\widetilde{B})$ with

(4.16)
$$\|\widetilde{A}\xi\|_{H} + \|\widetilde{B}\xi\|_{H} \lesssim \|\widetilde{A}\xi + \widetilde{B}\xi\|_{H}, \quad \xi \in D(\widetilde{A}) \cap D(\widetilde{B}).$$

Using the joint functional calculus we wish to transfer this inequality to A and B. For this note that the function $f(z, w) = z(z+w)^{-1}$ belongs to $H^{\infty}(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ since $\sigma_A + \sigma_B < \pi$. Set $g_n(z) = (z+w)\varphi_n(z)^2\varphi_n(w)^2$ with φ_n as in (4.4). Then $g \in H^1(\Sigma_{\sigma_A} \times \Sigma_{\sigma_B})$ and by the resolvent identity we have

$$g_n(\widetilde{A},\widetilde{B}) = (\widetilde{A} + \widetilde{B})\varphi_n(\widetilde{A})^2\varphi_n(\widetilde{B})^2.$$

Therefore by the multiplicativity of the joint H^{∞} -calculus and (4.16) we have for $\eta \in R(\widetilde{A} + \widetilde{B})$ and $\xi \in D(\widetilde{A}) \cap D(\widetilde{B})$ with $\eta = \widetilde{A}\xi + \widetilde{B}\xi$

$$\begin{split} \|f(\widetilde{A},\widetilde{B})\varphi_{n}(\widetilde{A})^{2}\varphi_{n}(\widetilde{B})^{2}\eta\|_{H} &= \|f(\widetilde{A},\widetilde{B})(\widetilde{A}\xi + \widetilde{B}\xi)\varphi_{n}(\widetilde{A})^{2}\varphi_{n}(\widetilde{B})^{2}\xi\|_{H} \\ &= \|f(\widetilde{A},\widetilde{B})g_{n}(\widetilde{A},\widetilde{B})\xi\|_{H} \\ &= \|\widetilde{A}\varphi_{n}(\widetilde{A})^{2}\varphi_{n}(\widetilde{B})^{2}\xi\|_{H} \\ &\lesssim \|\varphi_{n}(\widetilde{A})^{2}\varphi_{n}(\widetilde{B})^{2}\eta\|_{H} \end{split}$$

Taking the limit $n \to \infty$ and using the density of $R(\widetilde{A} + \widetilde{B})$ in H, we see that $f(\widetilde{A}, \widetilde{B})$ is bounded on H. By part (ii) of Theorem 4.4.1 we therefore obtain

$$\sup_{n \in \mathbb{N}} \| (f\psi_n)(A, B)x \| \le \sup_{n \in \mathbb{N}} \| (f\psi_n)(\widetilde{A}, \widetilde{B}) \| \lesssim \| f(\widetilde{A}, \widetilde{B}) \|$$

with $\psi_n(z, w) = \varphi_n(z)\varphi_n(w)$. It follows that f(A, B) extends to a bounded operator on X. Therefore we have for all $x \in D(A) \cap D(B)$

$$\|A\varphi_n(A)^2\varphi_n(B)^2x\|_X = \|f(A,B)g_n(A,B)x\|_X$$
$$\lesssim \|(A+B)\varphi_n(A)^2\varphi_n(B)^2x\|_X$$

and taking the limit $n \to \infty$ this implies

$$||Ax||_X + ||Bx||_X \le 2||Ax||_X + ||Ax + Bx||_X \lesssim ||Ax + Bx||_X.$$

The closedness of A+B now follows from the closedness of A and B. The sectoriality of A+B is proven for example in [AFM98, Theorem 3.1].

As we have seen before in this section, Theorem 4.4.8 can be strengthened if the Euclidean structure α is either the γ - or the ℓ^2 -structure. For a similar statement using \mathcal{R} -sectoriality we refer to [**KW01**].

COROLLARY 4.4.9. Let A and B be resolvent commuting sectorial operators on X. Suppose that A has a bounded H^{∞} -calculus and B is α -sectorial with $\omega_{H^{\infty}}(A) + \omega_{\alpha}(B) < \pi$. Assume one of the following conditions:

- (i) X has Pisier's contraction property and $\alpha = \gamma$.
- (ii) X is a Banach lattice and $\alpha = \ell^2$.

Then A + B is closed on the domain $D(A) \cap D(B)$ and

$$||Ax||_X + ||Bx||_X \le C \, ||Ax + Bx||_X, \qquad x \in D(A) \cap D(B).$$

Moreover A + B is α -sectorial with $\omega_{\alpha}(A + B) \leq \max\{\omega_{H^{\infty}}(A), \omega_{\alpha}(B)\}.$

PROOF. The first part of the statement follows directly from Theorem 4.3.5 and Theorem 4.4.8. It remains to prove the α -sectoriality of A + B. Fix $\omega_{\alpha-H^{\infty}}(A) < \sigma_A < \pi$ and $\omega_{\alpha}(B) < \sigma_B < \pi$ such that $\sigma_A + \sigma_B < \pi$ and take $\max\{\sigma_A, \sigma_B\} < \nu < \pi$. Let $\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu}$ and define

$$g_{\lambda}(z) := \frac{\lambda}{\lambda - z} ((\lambda - z)R(\lambda - z, B)), \qquad z \in \Sigma_{\sigma_A}$$

Then $g_{\lambda} \in H^{\infty}(\Sigma_{\sigma_A}; \Gamma)$ with

$$\Gamma := \{ \lambda R(\lambda, B) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma_B} \}.$$

Note that $\frac{\lambda}{\lambda-z}$ is uniformly bounded for $\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\nu}$ and $z \in \Sigma_{\sigma_A}$. Therefore since Γ is α -bounded it follows from Corollary 4.4.3 that the family

$$\{g_{\lambda}(A): \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\mu}\}$$

is α -bounded. By an approximation argument similar to the one presented in Theorem 4.4.8 we have $g_{\lambda}(A) = R(\lambda, A + B)$. Therefore it follows that A + B is α -sectorial of angle ν .

4.5. α -bounded imaginary powers

Before the development of the H^{∞} -calculus for a sectorial operator A, the notion of bounded imaginary powers, i.e. A^{is} for $s \in \mathbb{R}$, played an important role in the study of sectorial operators. We refer to [**Bd92**, **DV87**, **Mon97**, **PS90**] for a few breakthrough results in using bounded imaginary powers.

Defined by the extended Dunford calculus, A^{is} for $s \in \mathbb{R}$ is a possibly unbounded operator whose domain includes $D(A) \cap R(A)$. A is said to have bounded imaginary powers, denoted by BIP, if A^{is} is bounded for all $s \in \mathbb{R}$. In this case $(A^{is})_{s \in \mathbb{R}}$ is a C_0 -group and by semigroup theory we then know that there are $C, \theta > 0$ such that $||A^{is}|| \leq C e^{\theta |s|}$ for $s \in \mathbb{R}$. Thus we can define

$$\omega_{\mathrm{BIP}}(A) := \inf\{\theta : \|A^{is}\| \le C\mathrm{e}^{\theta|s|}, s \in \mathbb{R}\}.$$

It is a celebrated result of Prüss and Sohr [**PS90**] that $\omega_{\text{BIP}}(A) \ge \omega(A)$ and it is possible to have $\omega_{\text{BIP}}(A) \ge \pi$, see [**Haa03**, Corollary 5.3]. If A has a bounded H^{∞} -calculus, then A has BIP and since

(4.17)
$$\sup_{z \in \Sigma_{\sigma}} z^{it} \le e^{\sigma t}, \quad t \in \mathbb{R}$$

we have $\omega_{\text{BIP}}(A) \leq \omega_{H^{\infty}}(A) < \pi$. Furthermore Cowling, Doust, McIntosh and Yagi [**CDMY96**] showed that in this case $\omega_{\text{BIP}}(A) = \omega_{H^{\infty}}(A)$. Conversely if X is a Hilbert space and A has BIP with $\omega_{\text{BIP}}(A) < \pi$, then A has a bounded H^{∞} calculus. However, the example given in [**CDMY96**] shows that even for $X = L^p$ with $p \neq 2$ this result fails, i.e. it is possible for a sectorial operator A on X without a bounded H^{∞} -calculus to have $\omega_{\text{BIP}}(A) < \pi$.

We will try to understand this from the point of view of Euclidean structures. For this we say that a sectorial operator A has α -BIP if the family $\{e^{-\theta|s|}A^{is} : s \in \mathbb{R}\}$ is α -bounded for some $\theta \geq 0$. In this case we set

$$\omega_{\alpha-\mathrm{BIP}}(A) = \inf \{ \theta : (\mathrm{e}^{-\theta|s|} A^{is})_{s \in \mathbb{R}} \text{ is } \alpha\text{-bounded} \}.$$

Since $(A^s)^{it} = A^{ist}$ for $|s| \le \pi/\omega(A)$ and $t \in \mathbb{R}$ (see [KW04, Theorem 15.16]), we know that A^s has $(\alpha$ -)BIP if A has $(\alpha$ -)BIP with

$$\omega_{\text{BIP}}(A^s) = |s| \,\omega_{\text{BIP}}(A)$$
$$\omega_{\alpha\text{-BIP}}(A^s) = |s| \,\omega_{\alpha\text{-BIP}}(A).$$

Moreover α -BIP implies BIP with $\omega_{\text{BIP}}(A) \leq \omega_{\alpha\text{-BIP}}(A)$. If α is ideal, we have equality of angles.

PROPOSITION 4.5.1. Let α be an ideal Euclidean structure and let A be a sectorial operator on X. Suppose that A has α -BIP, then $\omega_{\text{BIP}}(A) = \omega_{\alpha\text{-BIP}}(A)$.

PROOF. Since α is ideal, we have the estimate $||A^{in}||_{\alpha} \leq C ||A^{in}||$ for $n \in \mathbb{Z}$. Take $\theta > \omega_{\text{BIP}}(A)$, then by Proposition 1.2.3(iii) we know that

$$\{\mathrm{e}^{\theta|n|}A^{in}:n\in\mathbb{Z}\}$$

is α -bounded. Combined with the fact that $\{A^{is} : s \in [-1,1]\}$ is α -bounded we obtain by Proposition 1.2.3(i) that $\omega_{\alpha-\text{BIP}}(A) < \theta$

The connection between (α)-BIP and (almost) α -sectoriality. We have an integral representation of $\lambda^s A^s (1 + \lambda A)^{-1}$ in terms of the imaginary powers of A, which will allow us to connect BIP to almost α -sectoriality. The representation is based on the Mellin transform.

LEMMA 4.5.2. Let A be a sectorial operator with BIP with $\omega_{BIP}(A) < \pi$. Then we have for $\lambda \in \mathbb{C}$ with $|\arg(\lambda)| + \omega_{BIP}(A) < \pi$ that

$$\lambda^s A^s (1+\lambda A)^{-1} = \frac{1}{2} \int_{\mathbb{R}} \frac{1}{\sin(\pi(s-it))} \lambda^{it} A^{it} \, \mathrm{d}t, \qquad 0 < s < 1.$$

PROOF. Recall the following Mellin transform (see e.g. [Tit86])

(4.18)
$$\int_0^\infty \frac{z^{s-1}}{1+z} \, \mathrm{d}z = \frac{\pi}{\sin(\pi s)}, \qquad 0 < \operatorname{Re}(s) < 1.$$

Using the substitution $z = e^{2\pi\xi}$ this becomes a Fourier transform:

$$2\int_{\mathbb{R}} \frac{e^{2\pi s\xi}}{1 + e^{2\pi\xi}} e^{-2\pi i t\xi} d\xi = \frac{1}{\sin(\pi(s - it))}, \qquad 0 < s < 1, \ t \in \mathbb{R}.$$

Thus by the Fourier inversion theorem we have

$$\int_{\mathbb{R}} \frac{e^{2\pi i t\xi}}{\sin(\pi(s - it))} \, dt = \frac{2e^{2\pi s\xi}}{1 + e^{2\pi\xi}}, \qquad 0 < s < 1, \, \xi \in \mathbb{R}.$$

Therefore using the substitution $z = e^{2\pi\xi}$ we have

(4.19)
$$\int_{\mathbb{R}} \frac{z^{it}}{\sin(\pi(s-it))} \, \mathrm{d}t = \frac{2z^s}{1+z}, \qquad 0 < s < 1, \ z \in \mathbb{R}_+$$

for $z \in \mathbb{R}_+$, which extends by analytic continuation to all $z \in \mathbb{C}$ with $-\pi < \arg(z) < \pi$.

Take $\omega(A) < \nu < \pi - |\arg(\lambda)|$ and let $x \in D(A) \cap R(A)$. Then $A^{it}x$ is given by the Bochner integral

$$A^{it}x = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} z^{it}\varphi(z)R(z,A)y\,\mathrm{d}z,$$

where $\varphi(z) = z(1+z)^{-2}$ and $y \in X$ is such that $x = \varphi(A)y$. Thus, by Fubini's theorem, (4.19) and $|\arg(\lambda)| + \nu < \pi$, we have for 0 < s < 1

$$\frac{1}{2} \int_{\mathbb{R}} \frac{1}{\sin(\pi(s-it))} \lambda^{it} A^{it} x \, \mathrm{d}t = \frac{1}{4\pi i} \int_{\Gamma_{\nu}} \int_{\mathbb{R}} \frac{\lambda^{it} z^{it}}{\sin(\pi(s-it))} \, \mathrm{d}t \, \varphi(z) R(z,A) y \, \mathrm{d}z$$
$$= \lambda^s A^s (1+\lambda A)^{-1} x.$$

As λA has BIP with $\omega_{\text{BIP}}(\lambda A) < \pi$, the lemma now follows by a density argument. \Box

As announced this lemma allows us to connect BIP to almost α -sectoriality. PROPOSITION 4.5.3. Let A be a sectorial operator on X.

- (i) If A has α -BIP with ω_{α -BIP < π , then A is almost α -sectorial with $\tilde{\omega}_{\alpha}(A) \leq \omega_{\alpha}$ -BIP(A).
- (ii) If A has BIP with $\omega_{\text{BIP}} < \pi$ and α is ideal, then A is almost α -sectorial with $\tilde{\omega}_{\alpha}(A) \leq \omega_{\text{BIP}}(A)$.

PROOF. Either fix $\omega_{\alpha-\text{BIP}}(A) < \theta < \pi$ for (i) or fix $\omega_{\text{BIP}}(A) < \theta < \pi$ for (ii). Suppose that $\lambda_1, \ldots, \lambda_n \in \mathbb{C}$ satisfy $|\arg(\lambda_k)| \leq \pi - \theta$. Then for 0 < s < 1 and $\mathbf{x} \in X^n$ we have by Lemma 4.5.2

$$\begin{aligned} \left\| \left(\lambda_k^s A^s R(-\lambda_k, A) x_k\right)_{k=1}^n \right\|_{\alpha} &\leq \frac{1}{2} \int_{\mathbb{R}} \frac{1}{\left|\sin(\pi(s-it))\right|} \left\| \left(\lambda_k^{it} A^{it} x_k\right)_{k=1}^n \right\|_{\alpha} \mathrm{d}t \\ &\leq \frac{1}{2} \int_{\mathbb{R}} \frac{1}{\left|\sin(\pi(s-it))\right|} \mathrm{e}^{(\pi-\theta)|t|} \|A^{it}\|_{\alpha} \mathrm{d}t \, \|\mathbf{x}\|_{\alpha} \\ &\leq C \, \|\mathbf{x}\|_{\alpha}, \end{aligned}$$

where we used that there is a $\theta_0 < \theta$ such that

$$e^{-\theta_0|t|} \|A^{it}\|_{\alpha} \leq C$$

with C > 0 independent of $t \in \mathbb{R}$ in the last step.

With some additional effort we can self-improve Proposition 4.5.3(i) to conclude that A is actually α -sectorial rather than almost α -sectorial. They key ingredient will be the α -multiplier theorem (Theorems 3.2.6 and 3.2.8).

THEOREM 4.5.4. Let A be a sectorial operator on X. If A has α -BIP with $\omega_{\alpha-\text{BIP}} < \pi$, then A is α -sectorial with $\omega_{\alpha}(A) \leq \omega_{\alpha-\text{BIP}}(A)$.

PROOF. We will show that for $0 < s < \frac{1}{2}$ the families of operators

$$\Gamma_s := \{ t^s A^s (1 + tA)^{-1} : t > 0 \}$$

are α -bounded uniformly in s. Since we have for $x \in D(A) \cap R(A)$

$$\lim_{s \to 0} t^{1-s} A^s (t+A)^{-1} x = -tR(-t,A)x$$

by the dominated convergence theorem, we obtain for $x_1, \ldots, x_n \in D(A) \cap R(A)$ and $t_1, \ldots, t_n > 0$ that

$$\left\| \left(-t_k R(-t_k, A) x_k \right)_{k=1}^n \right\|_{\alpha} \le \liminf_{s \to 0} \left\| \left(t_k^{1-s} A^s (t_k + A)^{-1} x_k \right)_{k=1}^n \right\|_{\alpha}.$$

This implies that A is α -sectorial with

$$\omega_{\alpha}(A) = \tilde{\omega}_{\alpha}(A) \le \omega_{\alpha-\mathrm{BIP}}(A).$$

by Proposition 4.2.1 and Proposition 4.5.3.

We claim that it suffices to prove for f in the Schwartz class $\mathcal{S}(\mathbb{R}; X)$ that

(4.20)
$$\left\| t \mapsto \int_{\mathbb{R}} k_s(t-u) A^{i(t-u)} f(u) \, \mathrm{d}u \right\|_{\alpha(\mathbb{R};X)} \le C \, \|f\|_{\alpha(\mathbb{R};X)},$$

where C > 0 is independent of $0 < s < \frac{1}{2}$ and

$$k_s(t) := \frac{1}{2\sin(\pi(s-it))}, \qquad t \in \mathbb{R}.$$

Indeed, assuming this claim for the moment, we know by Fubini's theorem and Lemma 4.5.2

$$\int_{\mathbb{R}} \int_{\mathbb{R}} k_s(t-u) A^{i(t-u)} f(u) \, \mathrm{d}u \, \mathrm{e}^{-2\pi i t\xi} \, \mathrm{d}t = \int_{\mathbb{R}} k_s(t) A^{it} \mathrm{e}^{-2\pi i t\xi} \, \mathrm{d}t \int_{\mathbb{R}} f(u) \mathrm{e}^{-2\pi i u\xi} \, \mathrm{d}u = \mathrm{e}^{-2\pi\xi s} A^s (1 + \mathrm{e}^{-2\pi\xi} A)^{-1} \hat{f}(\xi)$$

for any $\xi \in \mathbb{R}$. Thus since the Fourier transform is an isometry on $\alpha(\mathbb{R}; X)$ by Example 3.2.3, we deduce that for any $g \in \mathcal{S}(\mathbb{R}; X)$

$$\|\xi \mapsto e^{-2\pi s\xi} A^s (1 + e^{-2\pi \xi} A)^{-1} g(\xi)\|_{\alpha(\mathbb{R};X)} \le C \|g\|_{\alpha(\mathbb{R};X)},$$

which extends to all strongly measurable $g: S \to X$ in $\alpha(S; X)$ by density, see Proposition 3.1.6. Then the converse of the α -multiplier theorem (Theorem 3.2.8) implies that Γ_s is α -bounded, which completes the proof.

To prove the claim fix $0 < s < \frac{1}{2}$ and set $I_m = [2m - 1, 2m + 1)$ for $m \in \mathbb{Z}$. For $n \in \mathbb{Z}$ we define the kernel

$$K_n(t,u) := \sum_{j \in \mathbb{Z}} k_s(t-u) \, \mathbf{1}_{I_j}(t) \, \mathbf{1}_{I_{j+n}}(u), \qquad t, u \in \mathbb{R}$$

where the sum consists of only one element for any point (t, u). Since $k_s \in L^1(\mathbb{R})$, the operator $T_n : L^2(\mathbb{R}) \to L^2(\mathbb{R})$ given by

$$T_n \varphi(t) := \int_{\mathbb{R}} K_n(t, u) \varphi(u) \, \mathrm{d}u, \qquad t \in \mathbb{R},$$

is well-defined. By the Mellin transform as in (4.19) we know

$$\hat{k}_s(\xi) = \frac{\mathrm{e}^{-s\xi}}{1 + \mathrm{e}^{-\xi}} \le 1, \qquad \xi \in \mathbb{R},$$

so by Plancherel's theorem we have for $\varphi \in L^2(\mathbb{R})$

$$\|T_n\varphi\|_{L^2(\mathbb{R})}^2 = \sum_{j\in\mathbb{Z}} \int_{\mathbb{R}} \mathbf{1}_{I_j}(t) \left| \int_{\mathbb{R}} k_s(t-u)\varphi(u) \,\mathbf{1}_{I_{j+n}}(u) \,\mathrm{d}u \right|^2 \mathrm{d}t$$
$$\leq \sum_{j\in\mathbb{Z}} \int_{I_{j+n}} |\varphi(t)|^2 \,\mathrm{d}t = \|\varphi\|_{L^2(\mathbb{R})}^2.$$

Moreover since $|K_n(t, u)| \leq |k_s(t-u)| \mathbf{1}_{|t-u| \geq 2(|n|-1)}$ for $t, u \in \mathbb{R}$ and

$$|k_s(t)| \le \frac{1}{2|\sinh(\pi t)|} \le e^{-\pi|t|}, \qquad |t| \ge 1,$$

we have by Young's inequality

$$||T_n||_{L^2(\mathbb{R})\to L^2(\mathbb{R})} \le C_0 e^{-2\pi|n|}, \qquad |n|\ge 2$$

for some constant $C_0 > 0$. We conclude that T_n extends to a bounded operator on $\alpha(\mathbb{R}; X)$ for all $n \in \mathbb{Z}$ with

$$||T_n||_{\alpha(\mathbb{R}:X)\to\alpha(\mathbb{R};X)} \le C_0 e^{-2\pi|n|}$$

For $t \in \mathbb{R}$ define p(t) = 2j with $j \in \mathbb{Z}$ such that $t \in I_j$. Then $|p(t) - t| \leq 1$ for all $t \in \mathbb{R}$. Take $\omega_{\alpha \text{-BIP}}(A) < \theta < \pi$ and let $C_1, C_2 > 0$ be such that

$$\begin{split} \big\| \{A^{is} : s \in [-1,1]\} \big\|_{\alpha} &\leq C_1, \\ \big\| A^{is} \big\|_{\alpha} &\leq C_2 \, \mathrm{e}^{\theta|s|}, \qquad s \in \mathbb{R}. \end{split}$$

Now take a Schwartz function $f \in \mathcal{S}(\mathbb{R}; X)$ and fix $n \in \mathbb{Z}$. Noting that p(t) = p(u) - 2n on the support of K_n , we estimate

$$\begin{aligned} \left\| t \mapsto \int_{\mathbb{R}} K_n(t, u) A^{i(t-u)} f(u) \, \mathrm{d}u \right\|_{\alpha(\mathbb{R}; X)} \\ &= \left\| t \mapsto \int_{\mathbb{R}} K_n(t, u) A^{i(t-p(t)+p(u)-u-2n)} f(u) \, \mathrm{d}u \right\|_{\alpha(\mathbb{R}; X)} \\ &\leq C_1 \left\| t \mapsto \int_{\mathbb{R}} K_n(t, u) A^{i(p(u)-u-2n)} f(u) \, \mathrm{d}u \right\|_{\alpha(\mathbb{R}; X)} \\ &\leq C_0 C_1 \, \mathrm{e}^{-2\pi |n|} \| t \mapsto A^{i(p(t)-t-2n)} f(t) \|_{\alpha(\mathbb{R}; X)} \\ &\leq C_0 C_1^2 C_2 \, \mathrm{e}^{-2(\pi-\theta)|n|} \| f \|_{\alpha(\mathbb{R}; X)} \end{aligned}$$

using Theorem 3.2.6 in the second and last step. Since

$$k_s(t-u) = \sum_{n \in \mathbb{Z}} K_n(t, u), \qquad t, u \in \mathbb{R},$$

the claim in (4.20) now follows from the triangle inequality.

REMARK 4.5.5. If the X has the UMD property and A is a sectorial operator with BIP, then it was shown in [**CP01**, Theorem 4] that A is γ -sectorial. The proof of that result can be generalized to a Euclidean structure α under the assumption that $\alpha(\mathbb{R}; X)$ has the UMD property, which in case of the γ -structure is equivalent to the assumption that X has the UMD property. Note that the proofs of Theorem 4.5.4 and [**CP01**, Theorem 4] are of a similar flavour. The key difference being the point at which one gets rid of the singular integral operators, employing their boundedness on $\alpha(\mathbb{R}; X)$ and $L^p(\mathbb{R}; X)$ respectively.

The characterization of H^{∞} -calculus in terms of α -BIP. With Theorem 4.5.4 at our disposal we turn to the main result of this section, which characterizes when A has a bounded H^{∞} -calculus in terms of α -BIP. For this we will combine the Mellin transform arguments from 4.5.4 with the self-improvement of a bounded H^{∞} -calculus in Theorem 4.3.2 and the transference principle in Theorem 4.4.1.

THEOREM 4.5.6. Let A be a sectorial operator on X. The following conditions are equivalent:

- (i) A has BIP with $\omega_{\text{BIP}}(A) < \pi$ and α -BIP for some Euclidean structure α on X.
- (ii) A has a bounded H^{∞} -calculus.

If one of these equivalent conditions holds, we have

 $\omega_{H^{\infty}}(A) = \omega_{BIP}(A) = \inf\{\omega_{\alpha-BIP}(A) : \alpha \text{ is a Euclidean structure on } X\}$

PROOF. Suppose that A has a bounded H^{∞} -calculus and let $\omega_{H^{\infty}}(A) < \sigma < \pi$. Then, by Theorem 4.3.2, there is a Euclidean structure α on X so that A has a α -bounded $H^{\infty}(\Sigma_{\sigma})$ -calculus. By (4.17) this implies that A has α -BIP with ω_{α -BIP $\leq \sigma$ and therefore

(4.21) $\inf \{ \omega_{\alpha-\text{BIP}}(A) : \alpha \text{ is a Euclidean structure on } X \} \le \omega_{H^{\infty}}(A)$

For the converse direction pick s > 0 so that $\omega_{\alpha-\text{BIP}}(A^s) < \pi$. Then A^s is α -sectorial by Theorem 4.5.4 with $\omega_{\alpha}(A^s) \leq \omega_{\alpha-\text{BIP}}(A^s)$. Take $\omega_{\alpha-\text{BIP}}(A^s) < \sigma < \pi$,

then by Theorem 4.4.1 we can find a sectorial operator \widetilde{A} on a Hilbert space H with $\omega(\widetilde{A}) = \omega_{\text{BIP}}(\widetilde{A}) < \sigma$ and such that

$$||f(A)||_{\mathcal{L}(X)} \lesssim ||f(\widetilde{A})||_{\mathcal{L}(H)}, \qquad f \in H^{\infty}(\Sigma_{\sigma}).$$

Since BIP implies a bounded H^{∞} -calculus on a Hilbert space by [McI86], \tilde{A} has a bounded $H^{\infty}(\Sigma_{\sigma})$ -calculus. Therefore A^s has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A^s) < \pi$. So since the BIP and H^{∞} -calculus angles are equal for sectorial operators with a bounded H^{∞} -calculus, it follows that

(4.22)
$$\omega_{H^{\infty}}(A^s) = \omega_{\text{BIP}}(A^s) = s \,\omega_{\text{BIP}}(A) < s \,\pi.$$

Thus A has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A) = s^{-1}\omega_{H^{\infty}}(A^s) = \omega_{\text{BIP}}(A)$ by Proposition 4.3.1. The claimed angle equalities follow by combining (4.21) and (4.22).

Combining Theorem 4.5.6 with Theorem 4.3.5 and Proposition 4.5.1 we obtain the following corollary, of which the first part recovers [KW16a, Corollary 7.5]

COROLLARY 4.5.7. Let A be a sectorial operator on X.

(i) If X has Pisier's contraction property, then A has a bounded H^{∞} -calculus if and only if A has γ -BIP with ω_{γ -BIP}(A) < π . In this case

$$\omega_{H^{\infty}}(A) = \omega_{\gamma-\mathrm{H}^{\infty}}(A) = \omega_{\mathrm{BIP}}(A) = \omega_{\gamma-\mathrm{BIP}}(A)$$

(ii) If X is a Banach lattice, then A has a bounded H^{∞} -calculus if and only if A has ℓ^2 -BIP with $\omega_{\ell^2-\text{BIP}}(A) < \pi$. In this case

 $\omega_{H^{\infty}}(A) = \omega_{\ell^2 - \mathrm{H}^{\infty}}(A) = \omega_{\mathrm{BIP}}(A) = \omega_{\ell^2 - \mathrm{BIP}}(A)$
CHAPTER 5

Sectorial operators and generalized square functions

Continuing our analysis of the connection between the H^{∞} -calculus of sectorial operators and Euclidean structures, we will characterize whether a sectorial operator A has a bounded H^{∞} -calculus in terms of generalized square function estimates and in terms of the existence of a dilation to a group of isometries in this chapter. Furthermore, for a given Euclidean structure α we will introduce certain spaces close to X on which A always admits a bounded H^{∞} -calculus. In order to do so we will need the full power of the vector-valued function spaces introduced in Chapter 3, in particular the α -multiplier theorem.

Our inspiration stems from [CDMY96], where Cowling, Doust, McIntosh and Yagi describe a general construction of some spaces associated to a given sectorial operator A on L^p for $p \in (1, \infty)$. They consider norms of the form

$$\left\| \left(\int_0^\infty |\psi(tA)x|^2 \frac{\mathrm{d}t}{t} \right)^{1/2} \right\|_{L^p}, \qquad x \in D(A) \cap R(A),$$

where $\psi \in H^1(\Sigma_{\sigma})$ for some $\omega(A) < \sigma < \pi$. They characterize the boundedness of the H^{∞} -calculus of A on X in terms of the equivalence of such expressions with $\|x\|_{L^p}$. Further developments in this direction can for example be found in [AMN97, FM98, KU14, KW16b, LL05, LM04, LM12].

In the language of this memoir the norms from [CDMY96] can be interpreted as

$$\left\| \left(\int_0^\infty |\psi(tA)x|^2 \frac{\mathrm{d}t}{t} \right)^{1/2} \right\|_{L^p} = \|t \mapsto \psi(tA)x\|_{\ell^2(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)},$$

which suggests to extend these results to the framework of Euclidean structures by replacing the ℓ^2 -structure with a general Euclidean structure α . Therefore, for a sectorial operator A on a general Banach space X equipped with a Euclidean structure α we will introduce the generalized square function norms $||t \mapsto \psi(tA)x||_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)}$ along with a discrete variant and study their connection with the H^{∞} -calculus of A in Section 5.1. In particular, we will characterize the boundedness of the H^{∞} -calculus of A in terms of a norm equivalence between these generalized square function norms and the usual norm on X. For the γ -structure, which is equivalent to the ℓ^2 -structure on L^p , this was already done in [KW16a] (see also [HNVW17, Section 10.4]). In Section 5.2 we will use these generalized square function norms to construct dilations of sectorial operators on the spaces $\alpha(\mathbb{R}; X)$, which characterize the boundedness of the H^{∞} -calculus of A.

Afterwards we introduce a scale of spaces $H^{\alpha}_{\theta,A}$ for $\theta \in \mathbb{R}$ in Section 5.3, which are endowed with such a generalized square function norm. These spaces are very close to the homogeneous fractional domain spaces, but behave better in many respects. In particular we will show that A induces a sectorial operator on these spaces which always has a bounded H^{∞} -calculus. Moreover we will show that these generalized square function spaces form an interpolation scale for the complex method and that when one applies the α -interpolation method as introduced in Section 3.3 to the fractional domain spaces of A, one obtains these generalized square function spaces. We will end this chapter with an investigation of the generalized square function spaces for sectorial operators that are not necessarily almost α -bounded in Section 5.4. This will allow us to construct some interesting counterexamples on the angle of the H^{∞} -calculus in Section 6.4.

As in the previous two chapters, we keep the standing assumption that α is a Euclidean structure on X throughout this chapter.

5.1. Generalized square function estimates

Let A be a sectorial operator on X. As announced in the introduction of this chapter, we start by studying the generalized square function norm

$$||t \mapsto \psi(tA)x||_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)},$$

and its discrete analog

$$\sup_{t\in[1,2]} \left\| (\psi(2^n tA)x)_{n\in\mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)}$$

for appropriate $x \in X$. For $\alpha = \gamma$ these norms were already studied in **[KW16a**] (see also **[HNVW17**, Section 10.4]).

We would like to work with x such that $t \mapsto \psi(tA)x$ defines an element of $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$, rather than just being an element of the larger space $\alpha_+(\mathbb{R}_+, \frac{dt}{t}; X)$. Our main tool, the α -multiplier theorem (Theorem 3.2.6), asserts that α -bounded pointwise multipliers act boundedly from $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$ to $\alpha_+(\mathbb{R}_+, \frac{dt}{t}; X)$. We will frequently use the following lemma to ensure that such a multiplier actually maps to $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$ for certain $x \in X$.

LEMMA 5.1.1. Let A be a sectorial operator on X and take $\omega(A) < \sigma < \pi$. Let $x \in R(\varphi(A))$ for some $\varphi \in H^1(\Sigma_{\sigma})$, e.g. take $x \in D(A) \cap R(A)$. Then for $f \in H^{\infty}(\Sigma_{\sigma})$ and $\psi \in H^1(\Sigma_{\sigma})$ we have

$$(t \mapsto f(A)\psi(tA)x) \in \alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X), (n \mapsto f(A)\psi(2^n tA)x) \in \alpha(\mathbb{Z}; X), \qquad t \in [1, 2].$$

PROOF. We will only show the first statement, the second being proven analogously. Take $\omega(A) < \nu < \sigma$ and let $y \in X$ be such that $x = \varphi(A)y$. By the multiplicativity of the H^{∞} -calculus we have

(5.1)
$$f(A)\psi(tA)x = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} f(z)\psi(tz)\varphi(z)R(z,A)y\,\mathrm{d}z.$$

For all $z \in \Gamma_{\nu}$ the function $\psi(\cdot z) \otimes f(z)\varphi(z)R(z,A)y$ belongs to $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$, with norm

 $\|\psi(\cdot z)\|_{L^2(\mathbb{R}_+,\frac{\mathrm{d}t}{4})}\|f(z)\varphi(z)R(z,A)y\|_X.$

By (4.1) we know that for any $\xi \in H^1(\Sigma_{\sigma})$ we have $\xi \in H^2(\Sigma_{\sigma'})$ for $\nu < \sigma' < \sigma$, so

$$\sup_{z\in\Gamma_{\nu}}\|\psi(\cdot z)\|_{L^{2}(\mathbb{R}_{+},\frac{\mathrm{d}t}{t})}<\infty$$

We can therefore interpret the integral (5.1) as a Bochner integral in $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$, which yields that $f(A)\psi(\cdot A)x$ defines an element of $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$.

The equivalence of discrete and continuous generalized square function norms. Next we will show that it does not matter whether one studies the discrete or the continuous generalized square functions, as these norms are equivalent. Because of this equivalence we will only state results for the continuous generalized square function norms in the remainder of this section. The statements for discrete generalized square function norms are left to the interested reader, see also [HNVW17, Section 10.4.a]. Situations in which one can take $\delta = 0$ in the following proposition will be discussed in Corollary 5.1.5 and Proposition 5.4.5.

PROPOSITION 5.1.2. Let A be a sectorial operator on X, take $\omega(A) < \sigma < \pi$ and let $\psi \in H^1(\Sigma_{\sigma})$. For all $0 < \delta < \sigma - \omega(A)$ there is a C > 0 such that for $x \in D(A) \cap R(A)$

$$\sup_{t\in[1,2]} \left\| (\psi(2^n tA)x)_{n\in\mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)} \le C \max_{\epsilon=\pm\delta} \|\psi_{\epsilon}(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)},$$

and

$$\|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \le C \sup_{|\epsilon|<\delta} \sup_{t\in[1,2]} \left\| (\psi_{\epsilon}(2^n tA)x)_{n\in\mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)}$$

where $\psi_{\epsilon}(z) = \psi(e^{i\epsilon}z)$.

PROOF. For $x \in D(A) \cap R(A)$ and $|\epsilon| \leq \delta$ we know that

$$(t \mapsto \psi_{\epsilon}(tA)x) \in \alpha(\mathbb{R}_{+}, \frac{\mathrm{d}t}{t}; X),$$
$$(n \mapsto \psi_{\epsilon}(2^{n}tA)x) \in \alpha(\mathbb{Z}; X), \qquad t \in [1, 2]$$

by Lemma 5.1.1. Therefore if $\alpha = \gamma$ the first inequality follows from [**HNVW17**, Proposition 9.7.10] with a = 0, $b = \delta/\log(2)$, $\alpha = \sigma/\log(2)$ and the observation that $z \mapsto 2^z$ maps the strip

$$\mathbb{S}_{\sigma/\log(2)} := \{ z \in \mathbb{C} : |\mathrm{Im}(z)| < \sigma/\log(2) \}$$

onto the sector Σ_{σ} . Analogously the second inequality for $\alpha = \gamma$ follows from [**HNVW17**, Proposition 9.7.20] with $\alpha = \delta/\log(2)$. The proofs carry over to an arbitrary Euclidean structure, as the only properties of the γ -structure used in the proof of [**HNVW17**, Proposition 9.7.10 and Proposition 9.7.20] are (1.1) and the right ideal property in (1.2) in the form of Proposition 3.2.1.

The equivalence of the generalized square function norms for different ψ . The first major result of this section will be the equivalence of the continuous generalized square function norms for different choices of $\psi \in H^1(\Sigma_{\sigma})$. As a preparation let us note the following easy corollary of Jensen's inequality.

LEMMA 5.1.3. Let $h \in L^1(\mathbb{R}_+, \frac{dt}{t})$. The operator S_h on $L^2(\mathbb{R}_+, \frac{dt}{t})$ given by

$$S_h u(s) := \int_0^\infty h(st)u(t) \frac{\mathrm{d}t}{t}, \qquad s \in \mathbb{R}_+$$

is bounded with $||S_h|| \leq ||h||_{L^1(\mathbb{R}_+, \frac{\mathrm{d}t}{t})}$.

PROOF. Let $c := \|h\|_{L^1(\mathbb{R}_+, \frac{dt}{t})}$. By Jensen's inequality and Fubini's theorem we have

$$\int_0^\infty \left| \int_0^\infty h(st) u(t) \frac{\mathrm{d}t}{ct} \right|^2 \frac{\mathrm{d}s}{s} \le \int_0^\infty \int_0^\infty |h(st)| |u(t)|^2 \frac{\mathrm{d}t}{ct} \frac{\mathrm{d}s}{s} = \|u\|_{L^2(\mathbb{R}_+, \frac{\mathrm{d}t}{t})}^2,$$

which yields $\|S_h u\|_{L^2(\mathbb{R}_+, \frac{\mathrm{d}t}{t})} \le \|h\|_{L^1(\mathbb{R}_+, \frac{\mathrm{d}t}{t})} \|u\|_{L^2(\mathbb{R}_+, \frac{\mathrm{d}t}{t})}.$

We are now ready to prove the announced equivalence, which relies upon the α multiplier theorem. For the ℓ^2 - and the γ -structure this recovers the corresponding results from [LM04] and [KW16a] respectively.

PROPOSITION 5.1.4. Let A be an almost α -sectorial operator on X. Take $\tilde{\omega}_{\alpha}(A) < \sigma < \pi$ and fix arbitrary non-zero $\psi, \varphi \in H^1(\Sigma_{\sigma})$. For all $f \in H^{\infty}(\Sigma_{\sigma})$ and $x \in D(A) \cap R(A)$ we have

$$\left\|f(A)\psi(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \lesssim \|f\|_{H^{\infty}(\Sigma_{\sigma})} \left\|\varphi(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}.$$

In particular, for $f \equiv 1$, we have

$$\left\|\psi(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}\simeq \left\|\varphi(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}$$

PROOF. First fix $f \in H^1(\Sigma_{\sigma}) \cap H^{\infty}(\Sigma_{\sigma})$, let $x \in D(A) \cap R(A)$ and note that both $f(A)\psi(\cdot A)x$ and $\varphi(\cdot A)x$ are in $\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)$ by Lemma 5.1.1. Let $\xi, \eta \in H^1(\Sigma_{\sigma})$ be non-zero such that

$$\int_0^\infty \xi(t)\eta(t)\varphi(t)\frac{\mathrm{d}t}{t} = 1$$

Then

$$\int_0^\infty \xi(tz)\eta(tz)\varphi(tz)\frac{\mathrm{d}t}{t} = 1, \qquad z \in \Sigma_\sigma,$$

which is clear for $z \in \mathbb{R}_+$ and then in general by analytic continuation. Take $\tilde{\omega}_{\alpha}(A) < \nu < \sigma$. We use the properties of the Dunford calculus of A and Fubini's theorem to calculate

$$f(A) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} \left(\int_{0}^{\infty} \xi(tz) \eta(tz) \varphi(tz) \frac{\mathrm{d}t}{t} \right) f(z) R(z, A) \,\mathrm{d}z$$
$$= \int_{0}^{\infty} \left(\frac{1}{2\pi i} \int_{\Gamma_{\nu}} \xi(tz) \eta(tz) \varphi(tz) f(z) R(z, A) \,\mathrm{d}z \right) \frac{\mathrm{d}t}{t}$$
$$= \int_{0}^{\infty} \xi(tA) \eta(tA) f(A) \varphi(tA) \frac{\mathrm{d}t}{t}$$

With this identity, Fubini's theorem and (4.6) we obtain for $x \in X$ and $s \in \mathbb{R}_+$

$$\begin{split} f(A)\psi(sA)x &= \int_0^\infty \psi(sA)\xi(tA)\eta(tA)f(A)\varphi(tA)x\frac{\mathrm{d}t}{t} \\ &= \frac{1}{2\pi i}\int_{\Gamma_\nu}\psi(sz)z^{\frac{1}{2}}A^{\frac{1}{2}}R(z,A)\int_0^\infty\xi(tz)\eta(tA)f(A)\varphi(tA)x\frac{\mathrm{d}t}{t}\frac{\mathrm{d}z}{z} \\ &= \frac{1}{2\pi i}\sum_{\epsilon=\pm 1}\int_0^\infty\psi(s\lambda\mathrm{e}^{i\epsilon\nu})M(\lambda\mathrm{e}^{i\epsilon\nu})\int_0^\infty\xi(t\lambda\mathrm{e}^{i\epsilon\nu})N(t)\varphi(tA)x\frac{\mathrm{d}t}{t}\frac{\mathrm{d}\lambda}{\lambda} \\ &= \frac{1}{2\pi i}\sum_{\epsilon=\pm 1}S_\psi\Big(\lambda\mapsto M(\lambda\mathrm{e}^{i\epsilon\nu})\cdot S_\xi\big(t\mapsto N(t)\varphi(tA)x\big)(\lambda\mathrm{e}^{i\epsilon\nu})\Big)(s), \end{split}$$

where

$$\begin{split} M(z) &:= z^{\frac{1}{2}} A^{\frac{1}{2}} R(z, A), \qquad \qquad z \in \Sigma_{\sigma} \\ N(t) &:= \eta(tA) f(A), \qquad \qquad t \in \mathbb{R}_+ \end{split}$$

and S_h for $h \in L^1(\mathbb{R}_+, \frac{dt}{t})$ is as in Lemma 5.1.3, which extends to a bounded operator on functions in $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$ by Proposition 3.2.1.

By Proposition 4.2.3 we have that $\{M(z) : z \in \Sigma_{\sigma}\}$ is α -bounded. For N we recall the representation of (4.6)

$$N(t) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} \eta(tz) f(z) z^{\frac{1}{2}} A^{\frac{1}{2}} R(z, A) \frac{\mathrm{d}z}{z}.$$

Thus since

$$\int_{\Gamma_{\nu}} |\eta(tz)| |f(z)| \left| \frac{\mathrm{d}z}{z} \right| \le \|f\|_{H^{\infty}(\Sigma_{\sigma})} \|\eta\|_{H^{1}(\Sigma_{\sigma})}$$

we have by the almost α -sectoriality of A and Corollary 1.2.4 that

$$\|\{N(t): t \in \mathbb{R}_+\}\|_{\alpha} \lesssim \|f\|_{H^{\infty}(\Sigma_{\sigma})} \|\eta\|_{H^1(\Sigma_{\sigma})}$$

Moreover by Lemma 5.1.1 we have

$$N(\cdot)\varphi(\cdot A)x = \eta(\cdot A)f(A)\varphi(\cdot A)x \in \alpha\left(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X\right)$$

where implicit constant depends on A.

Applying the boundedness of S_{ξ} and S_{ψ} and the α -multiplier theorem (Theorem 3.2.6) on M and N we obtain

$$\begin{split} \|f(A)\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} &\leq \|S_{\psi}\| \|S_{\xi}\| \left\| \{M(z): z \in \Sigma_{\sigma}\} \right\|_{\alpha} \\ &\quad \cdot \|\{N(t): t \in \mathbb{R}_+\}\|_{\alpha} \|\varphi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \\ &\qquad \qquad \leq \|f\|_{H^{\infty}(\Sigma_{\sigma})} \|\varphi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}, \end{split}$$

where the implicit constant depends on ψ , ξ , η and A.

The same estimate for a general $f \in H^{\infty}(\Sigma_{\sigma})$ follows by approximating f by $f_n := f\varphi_n \in H^1(\Sigma_{\sigma}) \cap H^{\infty}(\Sigma_{\sigma})$ with φ_n as in (4.4), noting that $f_n(A)\psi(tA)x \to f(A)\psi(tA)x$ for all $t \in \mathbb{R}_+$ and appealing to Proposition 3.2.5(i).

From Proposition 5.1.4 we can see that the square function norms corresponding to ψ and ψ_{ϵ} as in Proposition 5.1.2 are equivalent when A is almost α -sectorial. Thus we can take $\delta = 0$ in Proposition 5.1.4.

COROLLARY 5.1.5. Let A be an almost α -sectorial operator on X, take $\tilde{\omega}_{\alpha}(A) < \sigma < \pi$ and let $\psi \in H^1(\Sigma_{\sigma})$. For all $x \in D(A) \cap R(A)$ we have

$$\|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \simeq \sup_{t\in[1,2]} \left\| (\psi(2^n tA)x)_{n\in\mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)}$$

If the generalized square function norms are equivalent with $\|\cdot\|_X$, it follows immediately from Proposition 5.1.4 that A has a bounded H^{∞} -calculus, which is the content of the next theorem.

THEOREM 5.1.6. Let A be an almost α -sectorial operator on X and take $\tilde{\omega}_{\alpha}(A) < \sigma < \pi$. If there are non-zero $\psi, \varphi \in H^1(\Sigma_{\sigma})$ such that for all $x \in D(A) \cap R(A)$

$$\left\|\psi(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \lesssim \|x\|_X \lesssim \left\|\varphi(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)},$$

then A has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A) \leq \tilde{\omega}_{\alpha}(A)$.

PROOF. Our claim follows directly from Proposition 5.1.4. Indeed, for $f \in H^{\infty}(\Sigma_{\nu})$ for any $\tilde{\omega}_{\alpha}(A) < \nu \leq \sigma$ and $x \in D(A) \cap R(A)$ we have

$$\begin{split} \|f(A)x\| &\lesssim \left\|f(A)\varphi(tA)x\right\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)} \\ &\lesssim \left\|\psi(tA)x\right\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)} \lesssim \|x\|_X \end{split}$$

which extends by density to all $x \in X$.

The equivalence of the generalized square function norms with the norm of X. In the second halve of this section we will turn to the converse of Theorem 5.1.6, i.e. we will study when the boundedness of the H^{∞} -calculus of A implies the equivalence of the generalized square function norms with $\|\cdot\|_X$. In order to prove our results, we will need to use the adjoint of A. Recall that the adjoint of a sectorial operator is a closed operator, which may not be a sectorial operator as it may only have dense domain and dense range in the weak*-topology. To remedy this we introduce the so-called moon dual, see e.g. [FW06, KW04]. Define X^{\sharp} as $\overline{D(A^*)} \cap \overline{R(A^*)}$, where the closures are taken in the norm topology of X^* . The moon-dual operator A^{\sharp} of A is the part of A^* in X^{\sharp} , i.e.

$$A^{\sharp}x = A^*x^*, \qquad x \in D(A^{\sharp}) = \left\{ x^* \in D(A^*) \cap \overline{R(A^*)} : A^*x^* \in \overline{D(A^*)} \right\}.$$

Then the following properties hold:

- A^{\sharp} is a sectorial operator on X^{\sharp} with spectrum $\rho(A^{\sharp}) = \rho(A^{*}) = \rho(A)$.
- $X^{\sharp} \subseteq X^*$ is norming for X.
- $R(z, A^{\sharp})$ is the restriction of $R(z, A)^*$ to X^{\sharp} .
- $\varphi(A)^* x = \varphi(A^{\sharp}) x$ for $\varphi \in H^1(\Sigma_{\sigma})$ and $x^* \in X^{\sharp}$.
- If X is reflexive, then $X^{\sharp} = X^*$ and $A^{\sharp} = A^*$.
- If A has a bounded H^{∞} -calculus, then A^{\sharp} has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A^{\sharp}) = \omega_{H^{\infty}}(A)$.

We will start by showing that, up to a smoothing factor $\varphi(A)$ for $\varphi \in H^1(\Sigma_{\sigma})$, we always have the equivalence of the generalized square function norms with $\|\cdot\|_X$. Note that similar estimates hold for the adjoint A^* on X^* equipped with the Euclidean structure α^* , by applying the following proposition to A^{\sharp} on X^{\sharp} equipped with the Euclidean structure induced by α^* .

PROPOSITION 5.1.7. Let A be a sectorial operator on X, let $\omega(A) < \sigma < \pi$ and take non-zero $\psi, \varphi \in H^1(\Sigma_{\sigma})$. Then we have

$$\|\psi(\cdot A)\varphi(A)x\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)} \lesssim \|x\|_X, \qquad x \in X$$

and

108

$$\|\varphi(A)x\|_X \lesssim \|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)}, \qquad x \in D(A) \cap R(A).$$

PROOF. For the first inequality fix $x \in X$. Then we have $\psi(\cdot A)\varphi(A)x \in \alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)$ by Lemma 5.1.1. Furthermore by (4.6) we have for $\omega(A) < \nu < \sigma$ and t > 0

$$\begin{split} \psi(tA) &= \frac{1}{2\pi i} \int_{\Gamma_{\nu}} \psi(tz) (z^{-1}A)^{1/2} R(1, z^{-1}A) \frac{\mathrm{d}z}{z} \\ &= \sum_{\epsilon = \pm 1} \frac{-\epsilon}{2\pi i} \int_{0}^{\infty} \psi(ste^{\epsilon i\nu}) (\mathrm{e}^{-\epsilon i\nu}s^{-1}A)^{1/2} (1 - \mathrm{e}^{-\epsilon i\nu}s^{-1}A) \frac{\mathrm{d}s}{s} \\ &= \sum_{\epsilon = \pm 1} \frac{-\epsilon}{2\pi i} \int_{0}^{\infty} \psi(s^{-1}te^{\epsilon i\nu}) f_{\epsilon}(sA) \frac{\mathrm{d}s}{s} \end{split}$$

with $f_{\epsilon}(z) := (e^{-\epsilon i\nu} z)^{1/2} (1 - e^{-\epsilon i\nu} z)^{-1}$. As $f_{\epsilon} \in H^1(\Sigma_{\sigma'})$ for $\omega(A) < \sigma' < \nu$, we have by Fubini's theorem and the multiplicativity of the Dunford calculus

$$\int_0^\infty \|f_\epsilon(sA)\varphi(A)\| \frac{\mathrm{d}s}{s} \lesssim \|f\|_{H^1(\Sigma_{\sigma'})} \|\varphi\|_{H^1(\Sigma_{\sigma})}.$$

Therefore, by property (1.1) of a Euclidean structure and (4.1), we have

$$\begin{split} \|\psi(\cdot A)\varphi(A)x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} &\leq \sum_{\epsilon=\pm 1} \left\| t \mapsto \int_{0}^{\infty} \psi(s^{-1}t\mathrm{e}^{\epsilon i\nu})f_{\epsilon}(sA)\varphi(A)x\frac{\mathrm{d}s}{s} \right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} \\ &\leq \sum_{\epsilon=\pm 1} \int_{0}^{\infty} \left\| t \mapsto \psi(s^{-1}t\mathrm{e}^{\epsilon i\nu}) \right\|_{L^{2}(\mathbb{R}_{+},\frac{\mathrm{d}t}{t})} \|f_{\epsilon}(sA)\varphi(A)x\|_{X}\frac{\mathrm{d}s}{s} \\ &= \sum_{\epsilon=\pm 1} \left\| t \mapsto \psi(t\mathrm{e}^{\epsilon i\nu}) \right\|_{L^{2}(\mathbb{R}_{+},\frac{\mathrm{d}t}{t})} \int_{0}^{\infty} \|f_{\epsilon}(sA)\varphi(A)x\|_{X}\frac{\mathrm{d}s}{s} \lesssim \|x\|_{X}, \end{split}$$

which proves the first inequality. Applying this result to A^{\sharp} on X^{\sharp} equipped with the Euclidean structure induced by α^* yields

(5.2)
$$\|\psi(\cdot A)^*\varphi(A)^*x^*\|_{\alpha^*(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X^*)} \lesssim \|x^*\|_{X^*}, \quad x^* \in X^{\sharp}.$$

For the second inequality take $x \in D(A) \cap R(A)$. Then by Lemma 5.1.1 we have $\psi(\cdot A)x \in \alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)$. Thus, since $\psi \in H^2(\Sigma_{\sigma'})$ for $\omega(A) < \sigma' < \sigma$ by (4.1), we have by (4.3) and the multiplicativity of the Dunford calculus

$$c x = \int_0^\infty \psi(tA) \psi^*(tA) x \frac{\mathrm{d}t}{t}$$

where $\psi^*(z) := \overline{\psi(\overline{z})}$ and $c = \int_0^\infty |\psi(t)|^2 \frac{dt}{t} > 0$. Applying Proposition 3.2.4 and (5.2) we deduce for any $x^* \in X^{\sharp}$

$$\begin{aligned} |\langle \varphi(A)x, x^* \rangle| &\leq c^{-1} \int_0^\infty |\langle \psi(tA)x, \psi^*(tA)^*\varphi(A)^*x^* \rangle| \frac{\mathrm{d}t}{t} \\ &\leq c^{-1} \|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)} \|\psi^*(\cdot A)^*\varphi(A)^*x\|_{\alpha^*(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)} \\ &\lesssim \|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)} \|x^*\|_{X^*}, \end{aligned}$$

so taking the supremum over all $x^* \in X^{\sharp}$ yields the second inequality.

If we assume the Euclidean structure α to be unconditionally stable and A to have a bounded H^{∞} -calculus, we can get rid of the $\varphi(A)$ -terms in Proposition 5.1.7. For the Euclidean structures ℓ^2 and γ , this recovers results from [CDMY96, KW16a]

THEOREM 5.1.8. Let A be a sectorial operator on X with a bounded H^{∞} calculus and assume that α is unconditionally stable. Take $\omega_{H^{\infty}}(A) < \sigma < \pi$ and let $\psi \in H^1(\Sigma_{\sigma})$ be non-zero. Then for all $x \in D(A) \cap R(A)$ we have

$$\|x\|_X \simeq \left\|t \mapsto \psi(tA)x\right\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)} \simeq \sup_{t \in [1,2]} \left\|(\psi(2^n tA)x)_{n \in \mathbb{Z}}\right\|_{\alpha(\mathbb{Z};X)}$$

PROOF. Let $\omega_{H^{\infty}}(A) < \sigma' < \sigma$ and $\varphi \in H^1(\Sigma_{\sigma'})$. Note that $(\varphi(2^n t A) x)_{n \in \mathbb{Z}}$ is an element of $\alpha(\mathbb{Z}; X)$ by Lemma 5.1.1 and the functions

$$f(z) = \sum_{k=-n}^{n} \epsilon_k \varphi(2^k t z)$$

are uniformly bounded in $H^{\infty}(\Sigma_{\sigma})$ for $t \in [1, 2]$, $|\epsilon_k| = 1$ and $n \in \mathbb{N}$ by Lemma 4.3.4. Therefore, since α is unconditionally stable and A admits a bounded H^{∞} -calculus, we have for all $x \in D(A) \cap R(A)$

110

$$\sup_{t\in[1,2]} \left\| (\varphi(2^n tA)x)_{n\in\mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)} \lesssim \sup_{t\in[1,2]} \sup_{n\in\mathbb{N}} \sup_{|\epsilon_k|=1} \left\| \sum_{k=-n}^n \epsilon_k \varphi(2^k tA)x \right\|_X$$
$$\lesssim \|x\|_X.$$

Taking $\varphi = \psi$ in this inequality yields the first halve of the equivalence between the generalized discrete square function norms and $\|\cdot\|_X$. Furthermore, using this inequality with $\varphi = \psi_{\epsilon} = \psi(e^{i\epsilon} \cdot)$ with $\epsilon < \sigma - \sigma'$, we have by Proposition 5.1.2

$$\|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \lesssim \sup_{|\epsilon| < \sigma - \sigma'} \sup_{t \in [1,2]} \left\| (\psi_{\epsilon}(2^n tA)x)_{n \in \mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)} \lesssim \|x\|.$$

For the converse inequality we apply this result to the moon dual A^{\sharp} on X^{\sharp} equipped with the Euclidean structure induced by α^* to obtain

$$\|\psi(\cdot A)^* x^*\|_{\alpha^*(\mathbb{R}_+,\frac{dt}{t};X^*)} \lesssim \|x^*\|_{X^*}, \qquad x^* \in D(A^*) \cap R(A^*).$$

Since $\psi \in H^2(\Sigma_{\sigma'})$ for $\omega(A) < \sigma' < \sigma$ by (4.1), we have by the Calderón reproducing formula (4.3)

$$c\,x = \int_0^\infty \psi(tA)\psi^*(tA)x\frac{\mathrm{d}t}{t}$$

where $\psi^*(z) := \overline{\psi(\overline{z})}$ and $c = \int_0^\infty |\psi(t)|^2 \frac{dt}{t} > 0$. Applying Proposition 3.2.4, we deduce for any $x^* \in D(A^*) \cap R(A^*)$

$$\begin{aligned} |\langle x, x^* \rangle| &\leq c^{-1} \int_0^\infty |\langle \psi(tA)x, \psi^*(tA)^*x^* \rangle| \frac{\mathrm{d}t}{t} \\ &\leq c^{-1} \|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)} \|\psi^*(\cdot A)^*x\|_{\alpha^*(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)} \\ &\lesssim \|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)} \|x^*\|_{X^*}. \end{aligned}$$

So since $D(A^*) \cap R(A^*)$ is norming for X, this yields

$$\|x\|_X \lesssim \|\psi(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}.$$

Another application of Proposition 5.1.2 yields the same inequality for the discrete generalized square function norm, finishing the proof. $\hfill \Box$

The equality of the angles of almost α -sectoriality and H^{∞} -calculus. To conclude this section, we note that, by combining Theorem 5.1.6 and Theorem 5.1.8, we are now able to show the equality of the almost α -sectoriality angle and the H^{∞} -calculus angle of a sectorial operator A. Using the global, ideal, unconditionally stable Euclidean structure ℓ^g this in particular reproves the equality of the BIP and bounded H^{∞} -calculus angles, originally shown in [CDMY96, Theorem 5.4]. Furthermore if A is α -sectorial this implies $\omega_{H^{\infty}}(A) = \omega_{\alpha}(A)$, which for the γ -structure was shown in [KW01].

COROLLARY 5.1.9. Let A be an α -sectorial operator on X with a bounded H^{∞} -calculus and assume that α is unconditionally stable.

- (i) If A almost α -sectorial, then $\omega_{H^{\infty}}(A) \leq \tilde{\omega}_{\alpha}(A)$.
- (ii) If α is ideal, then A is almost α -bounded with

$$\omega_{H^{\infty}}(A) = \omega_{BIP}(A) = \tilde{\omega}_{\alpha}(A)$$

PROOF. For (i) we know by Theorem 5.1.8 that for $\omega_{H^{\infty}} < \sigma < \pi$ and a non-zero $\psi \in H^1(\Sigma_{\sigma})$

$$\|x\| \simeq \|t \mapsto \psi(tA)x\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)}, \qquad x \in D(A) \cap R(A).$$

Thus, by Theorem 5.1.6, we know that $\omega_{H^{\infty}}(A) \leq \tilde{\omega}_{\alpha}(A)$. (ii) follows from (4.17) and Proposition 4.5.3.

5.2. Dilations of sectorial operators

Extending a dilation result of Sz-Nagy [SN47], Le Merdy showed in [LM96, LM98] that a sectorial operator A on a Hilbert space H with $\omega(A) < \frac{\pi}{2}$ has a bounded H^{∞} -calculus if and only if the associated semigroup $(e^{-tA})_{t\geq 0}$ has a dilation to a unitary group $(U(t))_{t\in\mathbb{R}}$ on a larger Hilbert space \widetilde{H} , i.e. A has a dilation to a normal operator \widetilde{A} on \widetilde{H} . By the spectral theorem for normal operators (see e.g. [Con90, Theorem X.4.19]) we can think of \widetilde{A} as a multiplication operator.

In this section we will use the generalized square functions to characterize the boundedness of the H^{∞} -calculus of a sectorial operator A on a general Banach space X in terms of dilations. We say that a semigroup $(U(t))_{t\geq 0}$ on a Banach space \widetilde{X} is a dilation of $(e^{-tA})_{t\geq 0}$ if there is an isomorphic embedding $J: X \to \widetilde{X}$ and a bounded operator $Q: \widetilde{X} \to X$ such that

$$e^{-tA} = QU(t)J, \qquad t \ge 0.$$

A sectorial operator \widetilde{A} on \widetilde{X} is called a dilation of A if there are such J and Q with

$$R(\lambda, A) = QR(\lambda, \widetilde{A})J, \qquad \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\max(\omega(A), \omega(\widetilde{A}))}.$$

This can be expressed in terms of the commutation of the following diagrams

$\widetilde{X} \xrightarrow{U(t)} \widetilde{X}$	$\widetilde{X} \xrightarrow{R(\lambda,A)} \widetilde{X}$
J Q	J Q
$X \xrightarrow{\mathrm{e}^{-tA}} X$	$X \xrightarrow{R(\lambda,A)} X$

Taking t = 0 in the semigroup case we see that QJ = I and JQ is a bounded projection of \widetilde{X} onto R(J). The same conclusion can be drawn in the sectorial operator case by

$$x = \lim_{\lambda \to \infty} \lambda (\lambda + A)^{-1} x = \lim_{\lambda \to \infty} \lambda Q (\lambda + \widetilde{A})^{-1} J x = Q J x, \qquad x \in X,$$

We will choose $\widetilde{X} = \alpha(\mathbb{R}; X)$ for an unconditionally stable Euclidean structure α on X and for s > 0 consider the multiplication operator \mathcal{M}_s given by

$$\mathcal{M}_s g(t) := (it)^{\frac{2}{\pi}s} g(t), \qquad t \in \mathbb{R}$$

for strongly measurable $g: \mathbb{R} \to X$ such that $g, \mathcal{M}_s g \in \alpha(\mathbb{R}; X)$. Note that the spectrum of \mathcal{M}_s is given by

$$\sigma(\mathcal{M}_s) = \partial \Sigma_s$$

and that for a bounded measurable function $f: \Sigma_{\sigma} \to \mathbb{C}$ with $s < \sigma < \pi$ the operator $f(\mathcal{M}_s)$ defined by

$$f(\mathcal{M}_s)g(t) = f((it)^{\frac{2}{\pi}s})g(s), \qquad t \in \mathbb{R}$$

extends to a bounded operator on $\alpha(\mathbb{R}; X)$ by Example 3.2.2. Hence \mathcal{M}_s has a bounded Borel functional calculus and is therefore a worthy replacement for normal operators on a Hilbert space.

If \mathcal{M}_s on $\alpha(\mathbb{R}; X)$ is a dilation of a sectorial operator A on X for $\omega(A) < s < \pi$, we have for $f \in H^1(\Sigma_{\sigma}) \cap H^{\infty}(\Sigma_{\sigma})$ with $s < \nu < \sigma < \pi$ that

$$f(A) = \frac{1}{2\pi i} \int_{\Gamma_{\nu}} f(z) QR(z, \mathcal{M}_s) J \, \mathrm{d}z, = Qf(\mathcal{M}_s) J,$$

where $f(\mathcal{M}_s)$ can either be interpreted in the Borel functional calculus sense or the Dunford calculus sense. Therefore, the fact that \mathcal{M}_s is a dilation of A implies that A has a bounded H^{∞} -calculus and we have for $f \in H^{\infty}(\Sigma_{\sigma})$

(5.3)
$$f(A) = Qf(\mathcal{M}_s)J$$

The converse of this statement is the main result in this section, which characterizes the boundedness of the H^{∞} -calculus of A in terms of dilations.

THEOREM 5.2.1. Let A be a sectorial operator on X and $\omega(A) < s < \pi$. Consider the following statements:

- (i) A has a bounded $H^{\infty}(\Sigma_{\sigma})$ -calculus for some $\omega(A) < \sigma < s$.
- (ii) The operator \mathcal{M}_s on $\alpha(\mathbb{R}; X)$ is a dilation of A for all unconditionally stable Euclidean structures α on X.
- (iii) The operator \mathcal{M}_s on $\alpha(\mathbb{R}; X)$ is a dilation of A for some Euclidean structure α on X.
- (iv) A has a bounded $H^{\infty}(\Sigma_{\sigma})$ -calculus for all $s < \sigma < \pi$.

Then $(i) \implies (ii) \implies (iv)$. Moreover, if A is almost α -sectorial with $\tilde{\omega}_{\alpha}(A) < s$ for some unconditionally stable Euclidean structure α , then $(iv) \implies (i)$.

Since $\gamma(\mathbb{R}; H) = L^2(\mathbb{R}; H)$ and $\omega(A) = \tilde{\omega}_{\gamma}(A)$ for a sectorial operator A on a Hilbert space H, Theorem 5.2.1 extends the classical theorem on Hilbert spaces by Le Merdy [LM96]. If X has finite cotype, the γ -structure is unconditionally stable by Proposition 1.1.6, so we also recover the main result from Fröhlich and the third author [FW06, Theorem 5.1]. For further results on dilations in UMD Banach spaces and L^p -spaces we refer to [FW06] and [AFL17].

PROOF OF THEOREM 5.2.1. For (i) \implies (ii) we may assume without loss of generality that $s = \frac{\pi}{2}$, as we can always rescale by defining a sectorial operator $B := A^{\frac{\pi}{2s}}$ and using Proposition 4.3.1 and the observation that $(\mathcal{M}_s)^{\frac{\pi}{2s}} = \mathcal{M}_{\frac{\pi}{2}}$.

Define for $x \in D(A) \cap R(A)$

$$Jx(t) := A^{1/2}R(it, A)x \qquad t \in \mathbb{R}$$

Setting $\psi_{\pm}(z) = \frac{z^{1/2}}{\pm i - z}$, we have

$$Jx(t) = t^{-1/2}\psi_{+}(t^{-1}A)x, \qquad t \in \mathbb{R}_{+},$$

$$Jx(-t) = t^{-1/2}\psi_{-}(t^{-1}A)x, \qquad t \in \mathbb{R}_{+}.$$

Therefore $Jx \in \alpha(\mathbb{R}; X)$ by Lemma 5.1.1 and using Proposition 3.2.1 we obtain

$$\|Jx\|_{\alpha(\mathbb{R};X)} \simeq \|\psi_+(tA)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} + \|\psi_-(tA)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}.$$

Now by Theorem 5.1.8 the bounded $H^{\infty}(\Sigma_{\sigma})$ -calculus of A implies that

$$||Jx||_{\alpha(\mathbb{R};X)} \simeq ||x||_{\alpha(\mathbb{R};X)}$$

so by density J extends to an isomorphic embedding $J: X \to \alpha(\mathbb{R}; X)$. Next take $g \in \alpha(\mathbb{R}; X)$ such that $||g(t)||_X \leq (1 + |t|)^{-1}$ and define the operator

$$Qg := \frac{1}{\pi} \int_{\mathbb{R}} A^{1/2} R(-it, A) g(t) \, \mathrm{d}t \in X,$$

where the integral converges in the Bochner sense in X, since

$$||A^{1/2}R(-it,A)|| \lesssim \frac{1}{|t|^{1/2}}, \quad t \in \mathbb{R}.$$

By the α -Hölder inequality (Proposition 3.2.4) and Theorem 5.1.8 applied to the moon dual A^{\sharp} on X^{\sharp} equipped with the Euclidean structure induced by α^* , we have for $x^* \in D(A^*) \cap R(A^*)$ that

$$\begin{aligned} |\langle Qg, x^* \rangle| &\leq \frac{1}{\pi} \int_0^\infty \left| \left\langle g(t), t^{-1/2} \left(\psi_+(t^{-1}A)^* + \psi_-(t^{-1}A)^* \right) x^* \right\rangle \right| \mathrm{d}t \\ &\lesssim \|g\|_{\alpha(\mathbb{R};X)} \left(\|\psi_+(tA)^* x^*\|_{\alpha^*(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X^*)} + \|\psi_-(tA)^* x^*\|_{\alpha^*(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X^*)} \right) \\ &\lesssim \|g\|_{\alpha(\mathbb{R};X)} \|x^*\|. \end{aligned}$$

Since $D(A^*) \cap R(A^*)$ is norming for X and using Proposition 3.1.6, it follows that Q extends to a bounded operator $Q: \alpha(\mathbb{R}; X) \to X$. To show that $\mathcal{M}_{\frac{\pi}{2}}$ on $\alpha(\mathbb{R}; X)$ is a dilation of A we will show that

(5.4)
$$R(\lambda, A) = QR(\lambda, \mathcal{M}_{\frac{\pi}{2}})J, \qquad \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\frac{\pi}{2}}.$$

First note that for $t \in \mathbb{R}$ we have by the resolvent identity

$$AR(it, A)R(-it, A) = -\frac{1}{2it}(AR(it, A) - AR(-it, A))$$

= $-\frac{1}{2}(R(it, A) + R(-it, A)).$

So since

$$\sup_{t \in \mathbb{R}} \|A^{1/2} R(it, A) x\|_X < \infty, \qquad x \in R(A^{1/2})$$

by the resolvent equation, we have for $x \in D(A) \cap R(A)$ and $\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\frac{\pi}{2}}$ that

$$\left\|\frac{1}{\lambda - it}A^{1/2}R(it, A)x\right\|_X \lesssim (1 + |t|)^{-1}, \qquad t \in \mathbb{R}$$

and therefore

$$QR(\lambda, \mathcal{M}_{\frac{\pi}{2}})Jx = \frac{1}{\pi} \int_{\mathbb{R}} A^{1/2}R(-it, A) \frac{1}{\lambda - it} A^{1/2}R(it, A)x \, dt$$

$$= -\frac{1}{2\pi} \int_{\mathbb{R}} \frac{1}{\lambda - it} R(it, A)x \, dt - \frac{1}{2\pi} \int_{\mathbb{R}} \frac{1}{\lambda - it} R(-it, A)x \, dt$$

$$= \frac{1}{2\pi i} \int_{\Gamma_{\frac{\pi}{2}}} \frac{1}{\lambda - z} R(z, A)x \, dz + \frac{1}{2\pi i} \int_{\Gamma_{\frac{\pi}{2}}} \frac{1}{\lambda + z} R(z, A)x \, dz$$

$$= R(\lambda, A)x,$$

where the last step follows from [HNVW17, Example 10.2.9] and Cauchy's theorem. This proves (5.4) by density.

The implication (ii) \implies (iii) follows directly from the fact that the global lattice structure ℓ^g is unconditionally stable on any Banach space X by Proposition 1.1.6. Implication (iii) \implies (iv) is a direct consequence of (5.3). Finally, if A is almost

 α -sectorial with $\tilde{\omega}_{\alpha}(A) < s$ for some unconditionally stable Euclidean structure α , (iv) \Longrightarrow (i) is a consequence of Corollary 5.1.9.

As a direct corollary we obtain a dilation result for the semigroup $(e^{-tA})_{t\geq 0}$. Note that we could use any \mathcal{M}_s with $\tilde{\omega}_{\alpha}(A) < s < \pi$, but only $\mathcal{M}_{\frac{\pi}{2}}$ yields a group of isometries.

COROLLARY 5.2.2. Let A be an almost α -sectorial operator on X with $\tilde{\omega}_{\alpha}(A) < \frac{\pi}{2}$ and assume that α is unconditionally stable. Then the following are equivalent

- (i) A has a bounded H^{∞} -calculus.
- (ii) The group of isometries $(U(t))_{t \in \mathbb{R}}$ on $\alpha(\mathbb{R}; X)$ given by $U(t) = e^{-t\mathcal{M}\frac{\pi}{2}}$ is a dilation of the semigroup $(e^{-tA})_{t \geq 0}$.

PROOF. The implication (i) \Rightarrow (ii) follows directly from Theorem 5.2.1 and (5.3) for $f_t(z) = e^{-tz}$ with $t \ge 0$. For the implication (ii) \Rightarrow (i) we note that from the Laplace transform (see [**HNVW17**, Proposition G.4.1])

$$R(\lambda, A)x = -\int_0^\infty e^{\lambda t} e^{-tA} x \, dt, \qquad \operatorname{Re} \lambda < 0, \, x \in X,$$

and a similar equation for $\mathcal{M}_{\frac{\pi}{2}}$ we obtain that $\mathcal{M}_{\frac{\pi}{2}}$ on $\alpha(\mathbb{R}; X)$ is a dilation of A, which implies the statement by Theorem 5.2.1.

To conclude this section, we note that for Banach lattices we can actually construct a dilation of $(e^{-tA})_{t\geq 0}$ consisting of *positive* isometries. This provides a partial converse to the result of the third author in [Wei01b, Remark 4.c] that the negative generator of any bounded analytic semigroup of positive contractions on L^p has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A) < \frac{\pi}{2}$. For more elaborate results in this direction and a full L^p -counterpart to the Hilbert space result from [LM98] we refer to [AFL17, Fac14b]

COROLLARY 5.2.3. Let A be a sectorial operator on an order-continuous Banach function space X and suppose that A has a bounded H^{∞} -calculus with $\omega_{H^{\infty}}(A) < \frac{\pi}{2}$. Then the semigroup $(e^{-tA})_{t\geq 0}$ has a dilation to a positive C_0 -group of isometries $(U(t))_{t\in\mathbb{R}}$ on $\ell^2(\mathbb{R}; X)$.

PROOF. Let J and Q be the embedding and projection operator of the dilation in Theorem 5.2.1(ii) with $\alpha = \ell^2$. Let \mathcal{F} denote the Fourier transform on $\ell^2(\mathbb{R}; X)$ and define

$$R(\lambda, A) = QR(\lambda, \mathcal{M}_{\frac{\pi}{2}})J = Q_{\mathcal{N}}R(\lambda, \mathcal{N})J_{\mathcal{N}}, \qquad \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\frac{\pi}{2}},$$

where

$$\mathcal{N} := \mathcal{F}^{-1} \mathcal{M}_{\frac{\pi}{2}} \mathcal{F} = \frac{1}{2\pi} \frac{\mathrm{d}}{\mathrm{d}t}$$

on $\ell^2(\mathbb{R}; X)$, $J_{\mathcal{N}} := \mathcal{F}^{-1}J$ and $Q_{\mathcal{N}} := Q\mathcal{F}$. Since the Fourier transform is bounded on $\ell^2(\mathbb{R}; X)$ by Example 3.2.3, we obtain that $(e^{-t\mathcal{N}})_{t\in\mathbb{R}}$ is a dilation of $(e^{-tA})_{t\geq 0}$ by (5.3) for $f_t(z) = e^{-tz}$ with $t \geq 0$. Now the corollary follows from the fact that $(e^{-t\mathcal{N}})_{t\in\mathbb{R}}$ is the translation group on $\ell^2(\mathbb{R}; X)$, which is a positive C_0 -group of isometries by the order-continuity of X, the dominated convergence theorem and Proposition 3.1.11.

5.3. A scale of generalized square function spaces

For a sectorial operator A on the Banach space X the scale of homogeneous fractional domain spaces $\dot{X}_{\theta,A}$ reflects many properties of X and is very useful in spectral theory. However, the operators on $\dot{X}_{\theta,A}$ induced by A may not have a bounded H^{∞} -calculus or BIP, the scale $\dot{X}_{\theta,A}$ may not be an interpolation scale and even for a differential operator A they may not be easy to identify as function spaces. Therefore one also considers e.g. the real interpolation spaces $(X, D(A))_{\theta,q}$ for $q \in [1, \infty]$, on which the restriction of an invertible sectorial operator A always has a bounded H^{∞} -calculus (see [**Dor99**]), and which, in the case of $A = -\Delta$ on $L^p(\mathbb{R}^d)$, equal the Besov spaces $B_{p,q}^{2\theta}(\mathbb{R}^d)$. However, these spaces almost never equal the fractional domain scale $\dot{X}_{\theta,A}$ (see [**KW05**]).

In this section we will introduce a scale of intermediate spaces $H^{\alpha}_{\theta,A}$ which are defined in terms of the generalized square functions of Section 5.1. These spaces have, under reasonable assumptions on A and the Euclidean structure α , the following advantages:

(i) They are "close" to the homogeneous fractional domain spaces, i.e. for $\eta_1 < \theta < \eta_2$ we have continuous embeddings

$$\dot{X}_{\eta_1,A} \cap \dot{X}_{\eta_2,A} \hookrightarrow H^{\alpha}_{\theta,A}(\varphi) \hookrightarrow \dot{X}_{\eta_1,A} + \dot{X}_{\eta_2,A},$$

see Theorem 5.3.4.

- (ii) The sectorial operator $A|_{H^{\alpha}_{\theta,A}}$ induced by A on $H^{\alpha}_{\theta,A}$ has a bounded H^{∞} -calculus, see Theorem 5.3.6.
- (iii) The spaces $H^{\alpha}_{\theta,A}$ and $\dot{X}_{\theta,A}$ are isomorphic essentially if and only if A has a bounded H^{∞} -calculus (see Theorem 5.3.6). In this case the spaces $H^{\alpha}_{\theta,A}$ provide a generalized form of the Littlewood–Paley decomposition for $\dot{X}_{\theta,A}$, which enables certain harmonic analysis methods in the spectral theory of A. In particular, if $A = -\Delta$ on $L^{p}(\mathbb{R}^{d})$ with $1 , then <math>H^{\gamma}_{\theta,A} = \dot{H}^{2\theta,p}(\mathbb{R}^{d})$ is a Riesz potential space.
- (iv) They form an interpolation scale for the complex interpolation method and are realized as α -interpolation spaces of the homogeneous fractional domain spaces. (see Theorems 5.3.7 and 5.3.8).

Let us fix a framework to deal with the fractional domain spaces of a sectorial operator A on X. Let $\theta \in \mathbb{R}$ and $m \in \mathbb{N}$ with $|\theta| < m$. We define the homogeneous fractional domain space $\dot{X}_{\theta,A}$ as the completion of $D(A^{\theta})$ with respect to the norm $x \mapsto ||A^{\theta}x||_X$. We summarize a few properties of $\dot{X}_{\theta,A}$ in the following proposition. We refer to [**KW04**, Section 15.E] or [**Haa06a**, Chapter 6] for the proof.

PROPOSITION 5.3.1. Let A be a sectorial operator on X and take $\theta \in \mathbb{R}$.

- (i) $D(A^m) \cap R(A^m)$ is dense in $\dot{X}_{\theta,A}$ for $m \in \mathbb{N}$ with $|\theta| < m$.
- (ii) For $\eta_1, \eta_2 \ge 0$ we have $\dot{X}_{\eta_1,A} \cap \dot{X}_{-\eta_2,A} = D(A^{\eta_1}) \cap R(A^{\eta_2}).$
- (iii) For $\eta_1 < \theta < \eta_2$ we have the continuous embeddings

$$X_{\eta_1,A} \cap X_{\eta_2,A} \hookrightarrow X_{\theta,A} \hookrightarrow X_{\eta_1,A} + X_{\eta_2,A}$$

The spaces $H^{\alpha}_{\theta,A}(\psi)$ and their properties. Now let us turn to the spaces $H^{\alpha}_{\theta,A}$, for which we first introduce a version depending on a choice of $\psi \in H^1(\Sigma_{\sigma})$. Let A be a sectorial operator on X. Assume either of the following conditions:

• α is ideal and set $\omega_A := \omega(A)$.

• A is almost α -sectorial and set $\omega_A := \tilde{\omega}(A)$.

Let $\omega_A < \sigma < \pi$, $\psi \in H^1(\Sigma_{\sigma})$ and take $\theta \in \mathbb{R}$ and $m \in \mathbb{N}$ with $|\theta| + 1 < m$. We define $H^{\alpha}_{\theta,A}(\psi)$ as the completion of $D(A^m) \cap R(A^m)$ with respect to the norm

$$x \mapsto \left\| \psi(\cdot A) A^{\theta} x \right\|_{\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)}$$

We write

116

$$H^\alpha_{\theta,A} \mathrel{\mathop:}= H^\alpha_{\theta,A}(\varphi), \qquad \varphi(z) \mathrel{\mathop:}= z^{1/2}(1+z)^{-1}.$$

By Lemma 5.1.1 we know that $\psi(\cdot A)A^{\theta}x \in \alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)$ for any $x \in D(A^m) \cap$ $R(A^m)$ and $\psi(\cdot A)A^{\theta}x = 0$ if and only if x = 0 by (4.3), so the norm on $H^{\alpha}_{\theta,A}(\psi)$ is well-defined.

Remark 5.3.2.

- On Hilbert spaces these spaces were already studied in [AMN97]. For the γ -structure on a Banach space these spaces are implicitly used in [KKW06, Section 7] and they are studied in [KW16b] for 0-sectorial operators with a so-called Mihlin functional calculus. In [Haa06b] (see also ([Haa06a, Chapter 6]), these spaces using $L^p(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)$ -norms instead of $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$ -norms were studied and identified as real interpolation spaces. Furthermore, for Banach function spaces using $X(\ell^q)$ norms instead of $\alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X)$ -norms, these spaces were developed in [KU14, Kun15].
- For $\psi \in H^1(\Sigma_{\sigma})$ such that $\widetilde{\psi} \in H^1(\Sigma_{\sigma})$ for $\widetilde{\psi}(z) := z^{\theta}\psi(z)$, we have the norm equality

$$\|x\|_{H^{\alpha}_{\theta,A}(\psi)} = \left\| t \mapsto t^{-\theta} \widetilde{\psi}(tA) x \right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)}$$

for $x \in D(A^m) \cap R(A^m)$. Viewing $\widetilde{\psi}(tA)$ as a generalized continuous Littlewood-Paley decomposition, this connects our scale of spaces to the more classical fractional smoothness scales.

Before turning to more interesting results, we will first prove that the $H^{\alpha}_{\theta,A}(\psi)$ spaces are independent of the parameter $m > |\theta| + 1$. This is the reason why we do not include it in our notation.

LEMMA 5.3.3. The definition of $H^{\alpha}_{\theta,A}(\psi)$ is independent of $m > |\theta| + 1$.

PROOF. It suffices to show that $D(A^{m+1}) \cap R(A^{m+1})$ is dense in $H^{\alpha}_{\theta,A}(\psi)$, which is defined as the completion of $D(A^m) \cap R(A^m)$. Fix $x \in D(A^m) \cap R(A^m)$ and let φ_n as in (4.4). Then $\varphi_n(A)$ maps $D(A^m) \cap R(A^m)$ into $D(A^{m+1}) \cap R(A^{m+1})$. We consider two cases:

• If α is ideal, then since $\varphi_n(A)x \to x$ in X we have

$$\psi(\cdot A)A^{\theta}\varphi_n(A)x \to \psi(\cdot A)A^{\theta}x$$

in $\alpha(\mathbb{R}_+, \frac{dt}{t}; X)$ by Proposition 3.2.5(iii). • If A is almost α -sectorial, let $y \in D(A^{m-1} \cap R(A^{m-1}))$ be such that $x = \varphi(A)y$ with $\varphi(z) = z(1+z)^{-2}$. Since we have for any $n \in \mathbb{N}$ and $z \in \Sigma_{\sigma}$

that

$$\begin{aligned} \varphi(z)(\varphi_n(z)-1)| &= \left|\frac{z}{n+z}\frac{z}{(1+z)^2} - \frac{\frac{1}{z}}{n+\frac{1}{z}}\frac{z}{(1+z)^2} \right| \\ &\lesssim \frac{1}{n} \left(\frac{z^2}{(1+z)^2} + \frac{1}{(1+z)^2}\right) \le \frac{2}{n}, \end{aligned}$$

we deduce by Proposition 5.1.4 that

$$\begin{split} \lim_{n \to \infty} \|\psi(\cdot A)A^{\theta}\varphi_n(A)x - \psi(\cdot A)A^{\theta}x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \\ &\leq \lim_{n \to \infty} \|\varphi(\varphi_n - 1)\|_{H^{\infty}(\Sigma_{\sigma})} \|\psi(\cdot A)A^{\theta}y\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} = 0 \end{split}$$

Thus we obtain in both cases that $D(A^{m+1}) \cap R(A^{m+1})$ is dense in $H^{\alpha}_{\theta,A}(\psi)$. \Box

We start our actual analysis of the $H^{\alpha}_{\theta,A}(\psi)$ -spaces by proving embeddings that show that they are "close" to the fractional domain spaces $\dot{X}_{\theta,A}$.

THEOREM 5.3.4. Let A be a sectorial operator on X. Assume either of the following conditions:

- α is ideal and set $\omega_A := \omega(A)$.
- A is almost α -sectorial and set $\omega_A := \tilde{\omega}(A)$.

Let $\omega_A < \sigma < \pi$ and take a non-zero $\psi \in H^1(\Sigma_{\sigma})$, then for $\eta_1 < \theta < \eta_2$ we have continuous embeddings

$$\dot{X}_{\eta_1,A} \cap \dot{X}_{\eta_2,A} \hookrightarrow H^{\alpha}_{\theta,A}(\psi) \hookrightarrow \dot{X}_{\eta_1,A} + \dot{X}_{\eta_2,A}$$

PROOF. By density it suffices to show the embeddings for $x \in D(A^m) \cap R(A^m)$ for some $m \in \mathbb{N}$ with $\eta_1, -\eta_2 < m-1$. Set

$$\epsilon = \min\{\theta - \eta_1, \eta_2 - \theta\}$$

and define $\varphi(z) = z^{\epsilon}(1+z^{\epsilon})^{-2}$. Then by (4.5) we have that

$$\varphi(A)^{-1} \colon D(A^{\epsilon}) \cap R(A^{\epsilon}) \mapsto X$$

is given by $\varphi(A)^{-1} = A^{\epsilon} + A^{-\epsilon} + 2I$. For the first embedding we have by Proposition 5.1.7 and Proposition 5.3.1(iii)

$$\begin{aligned} \|x\|_{H^{\alpha}_{\theta,A}(\psi)} &= \|\psi(\cdot A)\varphi(A)\varphi(A)^{-1}A^{\theta}x\|_{\alpha(\mathbb{R}_{+},\frac{dt}{t};X)} \\ &\lesssim \|A^{\theta+\epsilon}x + A^{\theta-\epsilon}x + 2A^{\theta}x\|_{X} \\ &\lesssim \|x\|_{\dot{X}_{\eta_{1},A}\cap\dot{X}_{\eta_{2},A}} \end{aligned}$$

For the second embedding we have by $A^{\theta}x \in D(A) \cap R(A)$, Proposition 5.3.1(iii) and Proposition 5.1.7

$$\begin{aligned} \|x\|_{\dot{X}_{\eta_{1},A}+\dot{X}_{\eta_{2},A}} &\lesssim \left\|\varphi(A)(A^{\epsilon}+A^{-\epsilon}+2I)x\right\|_{\dot{X}_{\theta-\epsilon,A}+\dot{X}_{\theta+\epsilon,A}} \\ &\leq \|\varphi(A)A^{\epsilon}x\|_{\dot{X}_{\theta-\epsilon,A}} + \|\varphi(A)A^{-\epsilon}x\|_{\dot{X}_{\theta+\epsilon,A}} + 2\|\varphi(A)x\|_{\dot{X}_{\theta,A}} \\ &\lesssim \|\psi(\cdot A)A^{\theta}x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)}, \end{aligned}$$

which finishes the proof of the theorem.

The sectorial operators $A|_{\dot{X}_{\theta,A}}$ and their properties. In the scale of fractional domain spaces $\dot{X}_{\theta,A}$ one can define a sectorial operator $A|_{\dot{X}_{\theta,A}}$ on $\dot{X}_{\theta,A}$ for $\theta \in \mathbb{R}$, which coincides with A on

$$D(A^{-\theta}AA^{\theta}) = X_{\min\{\theta,0\},A} \cap X_{1+\max\{\theta,0\},A},$$

see [**KW04**, Proposition 15.24]. We would like to have a similar situation for the spaces $H^{\alpha}_{\theta,A}(\psi)$, which is the content of the following proposition.

PROPOSITION 5.3.5. Let A be a sectorial operator on X and take $\eta_1 < \theta < \eta_2$. Assume either of the following conditions:

- α is ideal and set $\omega_A := \omega(A)$.
- A is almost α -sectorial and set $\omega_A := \tilde{\omega}(A)$.

Let $\omega_A < \sigma < \pi$ and $\psi \in H^1(\Sigma_{\sigma})$. Then there is a sectorial operator $A|_{H^{\alpha}_{\theta,A}(\psi)}$ on $H^{\alpha}_{\theta,A}(\psi)$ with $\omega(A|_{H^{\alpha}_{\theta,A}(\psi)}) \leq \omega_A$ satisfying

$$A|_{H^{\alpha}_{\theta,A}(\psi)}x = Ax, \qquad x \in X_{\min\{\eta_1,0\},A} \cap X_{1+\max\{\eta_2,0\},A},$$

and for $\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\omega_A}$

$$R(\lambda, A|_{H^{\alpha}_{\theta, A}(\psi)})x = R(\lambda, A)x, \qquad x \in \dot{X}_{\min\{\eta_1, 0\}, A} \cap \dot{X}_{\max\{\eta_2, 0\}, A}$$

PROOF. Let $m \in \mathbb{N}$ such that $|\theta| + 1 < m$. Either by the ideal property of α or by Proposition 5.1.4 we have for $x \in D(A^m) \cap R(A^m)$ and $\omega_A < \nu < \pi$ that

(5.5)
$$\|\lambda R(\lambda, A)x\|_{H^{\alpha}_{\theta,A}(\psi)} \le C_{\nu} \|x\|_{H^{\alpha}_{\theta,A}(\psi)}, \qquad \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\mu}$$

Thus, since $D(A^m) \cap R(A^m)$ is dense in $H^{\alpha}_{\theta,A}(\psi)$, $R(\lambda, A)$ extends to a bounded operator $R_A(\lambda)$ on $H^{\alpha}_{\theta,A}(\psi)$ for λ in the open sector

$$\Sigma := \mathbb{C} \setminus \overline{\Sigma}_{\omega_A}$$

We will construct $A|_{H^{\alpha}_{\theta,A}(\psi)}$ from R_A , as we also did in the proof of Theorem 4.4.1. Note that for $x \in \dot{X}_{\eta_1,A} \cap \dot{X}_{\eta_2,A}$ we have

$$\lim_{t \to \infty} \|tR_A(-t)x + x\|_{\dot{X}_{\eta_1,A} \cap \dot{X}_{\eta_2,A}} = 0$$
$$\lim_{t \to 0} \|tR_A(-t)x\|_{\dot{X}_{\eta_1,A} \cap \dot{X}_{\eta_2,A}} = 0$$

and thus, by density and one of the continuous embeddings in Theorem 5.3.4, we have for all $x \in H^{\alpha}_{\theta,A}(\varphi)$

(5.6)
$$\lim_{t \to \infty} \|tR_A(-t)x + x\|_{H^{\alpha}_{\theta,A}(\psi)} = 0$$

(5.7)
$$\lim_{t \to 0} \|tR_A(-t)x\|_{H^{\alpha}_{\theta,A}(\psi)} = 0$$

Using the density of $D(A^m) \cap R(A^m)$ in $H^{\alpha}_{\theta,A}(\psi)$ we also have the resolvent equation

$$R_A(z) - R_A(w) = (w - z)R_A(z)R_A(w), \qquad z, w \in \Sigma,$$

which in particular implies that if $R_A(z)x = 0$ for some $z \in \Sigma$ and $x \in H^{\alpha}_{\theta,A}(\psi)$, then $R_A(-t)x = 0$ for all t > 0, so $R_A(z)$ is injective by (5.6).

We are now ready to define $A|_{H^{\alpha}_{\theta,A}(\psi)}$. As domain we take the range of $R_A(-1)$ and we define

$$A|_{H^{\alpha}_{\theta,A}(\psi)}(R_A(-1)x) := -x - R_A(-1)x, \qquad x \in H^{\alpha}_{\theta,A}(\psi).$$

Then by the resolvent equation we have $R(\lambda, A|_{H^{\alpha}_{\theta,A}(\psi)}) = R_A(\lambda)$ for $\lambda \in \Sigma$. Furthermore $A|_{H^{\alpha}_{\theta,A}(\psi)}$ is injective, has dense domain by (5.6) and dense range by (5.7) (see [EN00, Section III.4.a] and [HNVW17, Proposition 10.1.7(3)] for the details). So by (5.5) we can conclude that $A|_{H^{\alpha}_{\theta,A}(\psi)}$ is a sectorial operator with $\omega(A|_{H^{\alpha}_{\theta,A}(\psi)}) \leq \omega_A$.

To conclude take $x \in \dot{X}_{\min\{\eta_1,0\},A} \cap \dot{X}_{1+\max\{\eta_2,0\},A}$ and let y := (I+A)x. Then $y \in H^{\alpha}_{\theta,A}(\psi) \cap X$ by the embeddings in Proposition 5.3.1 and Theorem 5.3.4 and thus $R_A(-1)y = R(-1,A)y = -x$. Therefore we have

$$A|_{H^{\alpha}_{\theta}}(\psi)x = -x + y = Ax.$$

Similarly for $x \in \dot{X}_{\min\{\eta_1,0\},A} \cap \dot{X}_{\max\{\eta_2,0\},A}$ we have $x \in H^{\alpha}_{\theta,A}(\psi) \cap X$ and therefore

$$R(\lambda, A|_{H^{\alpha}_{\theta,A}(\psi)})x = R_A(\lambda)x = R(\lambda, A)x, \qquad \lambda \in \Sigma,$$

which concludes the proof.

If A is almost α -sectorial, then the spaces $H^{\alpha}_{\theta,A}(\psi)$ are independent of the choice of $\psi \in H^1(\Sigma_{\sigma})$ by Proposition 5.1.4 and thus all isomorphic to $H^{\alpha}_{\theta,A}$. In this case the spaces $H^{\alpha}_{\theta,A}$ and the operators $A|_{H^{\alpha}_{\theta,A}(\psi)}$ have the following nice properties:

THEOREM 5.3.6. Let A be an almost α -sectorial operator on X.

- (i) $A|_{H^{\alpha}_{\theta,A}}$ has a bounded H^{∞} -calculus on $H^{\alpha}_{\theta,A}$ with $\omega_{H^{\infty}}(A|_{H^{\alpha}_{\theta,A}}) \leq \tilde{\omega}_{\alpha}(A)$ for all $\theta \in \mathbb{R}$.
- (ii) If $H^{\alpha}_{\theta,A} = \dot{X}_{\theta,A}$ isomorphically for some $\theta \in \mathbb{R}$, then A has a bounded H^{∞} -calculus on X.
- (iii) If A has a bounded H^{∞} -calculus on X and α is unconditionally stable, then $H^{\alpha}_{\theta,A} = \dot{X}_{\theta,A}$ for all $\theta \in \mathbb{R}$.

PROOF. Fix $\theta \in \mathbb{R}$ and $m \in \mathbb{N}$ with $|\theta| + 1 < m$. Let $x \in D(A^m) \cap R(A^m)$, then by Proposition 5.3.5 we know $f(A)x = f(A|_{H^{\alpha}_{\theta,A}})x$ for $f \in H^{\infty}(\Sigma_{\sigma})$ with $\tilde{\omega}(A) < \sigma < \pi$, so by Proposition 5.1.4 we have

$$\begin{aligned} \|f(A|_{H^{\alpha}_{\theta,A}})x\|_{H^{\alpha}_{\theta,A}} &= \|f(A)\varphi(tA)x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} \\ &\lesssim \|\varphi(tA)x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} = \|x\|_{H^{\alpha}_{\theta,A}} \end{aligned}$$

with $\varphi(z) := z^{1/2}(1+z)^{-1}$. Now (i) follows by the density of $D(A^m) \cap R(A^m)$ in $H^{\alpha}_{\theta,A}$. For (ii) we set $y = A^{-\theta}x$ and estimate

$$\|f(A)x\|_{X} = \|f(A)y\|_{\dot{X}_{\theta,A}} \simeq \|f(A)y\|_{H^{\alpha}_{\theta,A}} \lesssim \|y\|_{H^{\alpha}_{\theta,A}} \simeq \|y\|_{\dot{X}_{\theta,A}} = \|x\|_{X},$$

from which the claim follows by density. Finally (iii) is an immediate consequence of Theorem 5.1.8 and another density argument. \Box

Interpolation of square function spaces. We will now show that there is a rich interpolation theory of the $H^{\alpha}_{\theta,A}$ -spaces. First of all we note that $H^{\alpha}_{\theta,A}$ is the fractional domain space of order θ of the operator $A|_{H^{\alpha}_{0,A}}$ on $H^{\alpha}_{0,A}$ and $A|_{H^{\alpha}_{0,A}}$ has a bounded H^{∞} -calculus, and thus in particular BIP, by Theorem 5.3.6. Therefore it follows from [**KW04**, Theorem 15.28] that $H^{\alpha}_{\theta,A}$ is an interpolation scale for the complex method. We record this observation in the following theorem.

THEOREM 5.3.7. Let A be an almost α -sectorial operator on X. Let $\theta_0, \theta_1 \in \mathbb{R}$, $0 < \eta < 1$ and $\theta = (1 - \eta)\theta_0 + \eta\theta_1$. Then

$$H^{\alpha}_{\theta,A} = [H^{\alpha}_{\theta_0,A}, H^{\alpha}_{\theta_1,A}]_{\eta}$$

isomorphically.

120

Our main interpolation result will be the interpolation of the fractional domain spaces using the the α -interpolation method developed in Section 3.3. We will show that this yields exactly the spaces $H^{\alpha}_{\theta,A}$. In [**KKW06**, Section 7] this result was already implicitly shown for the Rademacher interpolation method, which is connected to the γ -interpolation method by Proposition 3.4.2.

We know that $A|_{H^{\alpha}_{\theta,A}}$ has a bounded H^{∞} -calculus on $H^{\alpha}_{\theta,A}$ by Theorem 5.3.6. Therefore one can view the following theorem as an α -interpolation version of the theorem of Dore, which states that A always has a bounded H^{∞} -calculus on the real interpolation spaces $(\dot{X}_{\theta_0,A}, \dot{X}_{\theta_1,A})_{\eta,q}$ for $q \in [1, \infty]$ (see [**Dor99**] and its generalizations in [**Dor01, Haa06b, KK10**]).

THEOREM 5.3.8. Let A be an almost α -sectorial operator on X. Let $\theta_0, \theta_1 \in \mathbb{R}$, $0 < \eta < 1$ and $\theta = (1 - \eta)\theta_0 + \eta\theta_1$. Then

$$H^{\alpha}_{\theta,A} = (\dot{X}_{\theta_0,A}, \dot{X}_{\theta_1,A})^{\alpha}_{\eta}$$

isomorphically.

PROOF. Assume without loss of generality that $\theta_1 > \theta_0$, take $x \in D(A^m) \cap R(A^m)$ for $m \in \mathbb{N}$ with $|\theta| + 1 < m$ and fix $\tilde{\omega}_{\alpha}(A) < \sigma < \pi$. Let $\psi \in H^1(\Sigma_{\sigma})$ be such that $\int_0^\infty \psi(t) \frac{\mathrm{d}t}{t} = 1$ and

$$\psi_j := \left(z \mapsto z^{\theta_j - \theta} \psi(z) \right) \in H^1(\Sigma_\sigma), \qquad j = 0, 1.$$

First consider the strongly measurable function $f : \mathbb{R}_+ \to D(A^m) \cap R(A^m)$ given by

$$f(t) = \frac{t^{-\eta}}{\theta_1 - \theta_0} \psi\left(t^{\frac{1}{\theta_1 - \theta_0}} A\right) x, \qquad t \in \mathbb{R}_+$$

Then, by (4.3) and a change of variables, we have $\int_0^\infty t^\eta f(t) \frac{dt}{t} = x$ and thus, by Proposition 3.4.1 and Proposition 5.1.4, we have

$$\begin{aligned} \|x\|_{(\dot{X}_{\theta_0,A},\dot{X}_{\theta_1,A})^{\alpha}_{\eta}} &\leq \max_{j=0,1} \|t \mapsto t^j f(t)\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};\dot{X}_{\theta_j,A})} \\ &= \max_{j=0,1} \|t \mapsto \frac{1}{\theta_1 - \theta_0} \psi_j(t^{\frac{1}{\theta_1 - \theta_0}} A) A^{\theta} x \|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)} \\ &\simeq \|x\|_{H^{\alpha}_{\theta,A}}. \end{aligned}$$

Conversely, take a strongly measurable function $f \colon \mathbb{R}_+ \to D(A^m) \cap R(A^m)$ such that $t \mapsto t^j f(t) \in \alpha(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; \dot{X}_{\theta_j, A})$ for j = 0, 1 and $\int_0^\infty t^\eta f(t) \frac{\mathrm{d}t}{t} = x$. Let $\varphi \in H^1(\Sigma_\sigma)$ be such that

$$\varphi_j := (z \mapsto z^{\theta - \theta_j} \varphi(z)) \in H^1(\Sigma_{\sigma}), \qquad j = 0, 1.$$

Then, since A is almost α -sectorial, we have by Proposition 5.1.4, Proposition 4.2.3 and Theorem 3.2.6 that

$$\begin{aligned} \|x\|_{H^{\alpha}_{\theta,A}} &\simeq \left\|\varphi(tA)A^{\theta}\int_{0}^{\infty} (st^{\theta_{1}-\theta_{0}})^{\eta}f(st^{\theta_{1}-\theta_{0}})\frac{\mathrm{d}s}{s}\right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} \\ &\lesssim \int_{0}^{1} s^{\eta} \left\|\varphi_{0}(tA)A^{\theta_{0}}f(st^{\theta_{1}-\theta_{0}})\right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)}\frac{\mathrm{d}s}{s} \end{aligned}$$

$$\begin{split} &+ \int_{1}^{\infty} s^{\eta} \left\| \varphi_{1}(tA) t^{(\theta_{1}-\theta_{0})} A^{\theta_{1}} f(st^{\theta_{1}-\theta_{0}}) \right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} \frac{\mathrm{d}s}{s} \\ &\lesssim \int_{0}^{1} s^{\eta} \left\| f(st^{\theta_{1}-\theta_{0}}) \right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};\dot{X}_{\theta_{0},A})} \frac{\mathrm{d}s}{s} \\ &+ \int_{1}^{\infty} s^{(\eta-1)} \left\| st^{\theta_{1}-\theta_{0}} f(st^{\theta_{1}-\theta_{0}}) \right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};\dot{X}_{\theta_{1},A})} \frac{\mathrm{d}s}{s} \\ &\lesssim \max_{j=0,1} \left\| t \mapsto t^{j} f(t) \right\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};\dot{X}_{\theta_{j},A})}. \end{split}$$

Taking the infimum over all such f we obtain by Proposition 3.4.1

$$\|x\|_{H^{\alpha}_{\theta,A}} \lesssim \|x\|_{(\dot{X}_{\theta_0,A},\dot{X}_{\theta_1,A})^{\alpha}_{\eta}}$$

so the norms of $H^{\alpha}_{\theta,A}$ and $(\dot{X}_{\theta_0,A}, \dot{X}_{\theta_1,A})^{\alpha}_{\eta}$ are equivalent on $D(A^m) \cap R(A^m)$. As $D(A^m) \cap R(A^m)$ is dense in both spaces, this proves the theorem. \Box

In [AMN97, Theorem 5.3] Auscher, McIntosh and Nahmod proved that a sectorial operator A on a Hilbert space H has a bounded H^{∞} -calculus if and only if the fractional domain spaces of A form a interpolation scale for the complex method. As a direct corollary of Theorem 5.3.6 and Theorem 5.3.8 we can now deduce a similar characterization of the boundedness of the H^{∞} -calculus of a sectorial operator on a Banach space in terms of the α -interpolation method.

COROLLARY 5.3.9. Let A an almost α -sectorial operator on X and suppose that α is unconditionally stable. Then A has a bounded H^{∞} -calculus if and only if

$$\dot{X}_{\theta,A} = (\dot{X}_{\theta_0,A}, \dot{X}_{\theta_1,A})^{\alpha}_{\eta}$$

for some $\theta_0, \theta_1 \in \mathbb{R}$, $0 < \eta < 1$ and $\theta = (1 - \eta)\theta_0 + \eta\theta_1$.

In [**KKW06**] perturbation theory for H^{∞} -calculus is developed using the Rademacher interpolation method, which is equivalent to the γ -interpolation method on spaces with finite cotype by Proposition 3.4.2 and Proposition 1.0.1. Naturally, these results can also be generalized to the Euclidean structures framework. In particular, let us prove a version of [**KKW06**, Theorem 5.1] in our framework. We leave the extension of the other perturbation results from [**KKW06**] (see also [**Kal07**, **KW13**, **KW17**]) to the interested reader.

COROLLARY 5.3.10. Let A be an almost α -sectorial operator on X and suppose that α is unconditionally stable. Suppose that A has a bounded H^{∞} -calculus and B is almost α -sectorial. Assume that for two different, non-zero $\theta_0, \theta_1 \in \mathbb{R}$ we have

$$X_{\theta_i,A} = X_{\theta_i,B}, \qquad j = 0, 1.$$

Then B has a bounded H^{∞} -calculus.

PROOF. Let $\tilde{\theta}_0, \tilde{\theta}_1, \tilde{\theta} \in \{0, \theta_0, \theta_1\}$ be such that $\tilde{\theta}_0 < \tilde{\theta} < \tilde{\theta}_1$ and let $\eta \in (0, 1)$ be such that $\tilde{\theta} = (1 - \eta)\tilde{\theta}_0 + \eta\tilde{\theta}_1$. Then by Theorem 5.3.6 and 5.3.8 we have

$$\dot{X}_{\tilde{\theta},B} = \dot{X}_{\tilde{\theta},A} = (\dot{X}_{\tilde{\theta}_0,A}, \dot{X}_{\tilde{\theta}_1,A})^{\alpha}_{\eta} = (\dot{X}_{\tilde{\theta}_0,B}, \dot{X}_{\tilde{\theta}_1,B})^{\alpha}_{\eta},$$

so the corollary follows from Corollary 5.3.9.

5.4. Generalized square function spaces without almost α -sectoriality

In Section 5.3 we have seen that the spaces $H^{\alpha}_{\theta,A}(\psi)$ behave very nicely when A is almost α -sectorial. In this section we will take a closer look at the $H^{\alpha}_{\theta,A}(\psi)$ -spaces for sectorial operators A which are not necessarily almost α -sectorial. In this case the spaces $H^{\alpha}_{\theta,A}(\psi)$ may be different for different ψ and whether $A|_{H^{\alpha}_{\theta,A}(\psi)}$ has a bounded H^{∞} -calculus may depend on the choice of ψ . This unruly behaviour will allow us to construct some interesting counterexamples in Section 6.4.

The spaces $H^{\alpha}_{\theta,A}(\varphi_{s,r})$, and their properties. Conforming with the definition of $H^{\alpha}_{\theta,A}(\psi)$ we need it assume that α is ideal throughout this section. Let $0 < s < \frac{\pi}{\omega(A)}, 0 < r < 1, \omega(A) < \sigma < \frac{\pi}{s}$ and set

$$\varphi_{s,r}(z) := \frac{z^{sr}}{1+z^s}, \qquad z \in \Sigma_{\sigma}.$$

We will focus our attention on the spaces $H^{\alpha}_{\theta,A}(\varphi_{s,r})$, for which we have $H^{\alpha}_{\theta,A} = H^{\alpha}_{\theta,A}(\varphi_{1,\frac{1}{2}})$. We will start our analysis by computing an equivalent norm on these spaces, which will be more suited for our analysis.

PROPOSITION 5.4.1. Let A be a sectorial operator on X and assume that α is ideal. Let $0 < s < \frac{\pi}{\omega(A)}$ and 0 < r < 1. Then for $x \in D(A) \cap R(A)$ and $t \in [1, 2]$ we have

$$\begin{aligned} \|\varphi_{s,r}(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} &\simeq \|\mathrm{e}^{-\frac{\pi}{s}|\cdot|}A^{i\cdot}x\|_{\alpha(\mathbb{R};X)},\\ \left\|(\varphi_{s,r}(2^n tA)x)_{n\in\mathbb{Z}}\right\|_{\alpha(\mathbb{Z};X)} &\simeq \left\|\sum_{m\in\mathbb{Z}}\mathrm{e}^{-\frac{\pi}{s}|\cdot+2mb|}A^{i(\cdot+2mb)}x\right\|_{\alpha([-b,b];X)}\end{aligned}$$

with $b = \pi/\log(2)$ and the implicit constants only depend on s and r.

PROOF. Let $\omega(A) < \sigma < \min\{\frac{\pi}{s}, \pi\}$, then for $\xi \in \mathbb{R}$ and $z \in \mathbb{R}_+$ we have, using the change of coordinates $u^{1/s} = e^{2\pi t}z$ and the Mellin transform as in (4.18), that

$$\int_{\mathbb{R}} e^{-2\pi i t\xi} \varphi_{s,r}(e^{2\pi t}z) dt = \frac{z^{i\xi}}{2\pi s} \int_0^\infty u^{-i\xi/s} \frac{u^r}{1+u} \frac{du}{u}$$
$$= \frac{z^{i\xi}}{2s \sin(\pi (r-i\xi/s))} =: z^{i\xi}g(\xi).$$

which extends to all $z \in \Sigma_{\sigma}$ by analytic continuation. Note that

(5.8)
$$|g(\xi)| = \frac{1}{2s |\sin(\pi (r - i\xi/s))|} \simeq e^{-\frac{\pi}{s}|\xi|}, \quad \xi \in \mathbb{R}.$$

By Fourier inversion we have for all $z \in \Sigma_{\sigma}$ and $t \in \mathbb{R}$

$$\varphi_{s,r}(\mathrm{e}^{2\pi t}z) = \int_{\mathbb{R}} \mathrm{e}^{2\pi i t\xi} g(\xi) z^{i\xi} \,\mathrm{d}\xi$$

Thus, by the definition of the H^{∞} -calculus and Fubini's theorem, we have for $x \in D(A) \cap R(A)$

(5.9)
$$\varphi_{s,r}(\mathrm{e}^{2\pi t}A)x = \int_{\mathbb{R}} \mathrm{e}^{2\pi i t\xi} g(\xi) A^{i\xi} x \,\mathrm{d}\xi.$$

5.4. GENERALIZED SQUARE FUNCTION SPACES WITHOUT ALMOST α -SECTORIALITY23

as a Bochner integral, since, for $\xi \in \mathbb{R}$, $\omega(A) < \nu < \frac{\pi}{s}$, $\varphi(z) = z(1+z)^{-2}$ and $y \in X$ such that $\varphi(A)y = x$, we have

(5.10)
$$\|g(\xi)A^{i\xi}x\|_{X} = \frac{|g(\xi)|}{2\pi} \left\| \int_{\Gamma_{\nu}} z^{i\xi}\varphi(z)R(z,A)y\,\mathrm{d}z \right\|_{X} \\ \lesssim \mathrm{e}^{-\frac{\pi}{s}|\xi|} \cdot \int_{\Gamma_{\nu}} \mathrm{e}^{\nu|\xi|} \frac{|z|}{|1+z|^{2}} \|y\|_{X} \frac{|\mathrm{d}z|}{|z|} \\ \lesssim \mathrm{e}^{-(\frac{\pi}{s}-\nu)|\xi|} \|y\|_{X}.$$

Now to prove the equivalence for the continuous square function norm, define $T: L^2(\mathbb{R}_+, \frac{dt}{t}) \mapsto L^2(\mathbb{R})$ by

$$Tf(t) := \sqrt{2\pi} f(e^{2\pi t}), \qquad t \in \mathbb{R}.$$

Then T is an isometry, so by Proposition 3.2.1 we have for any $f \in L^2(\mathbb{R}_+, \frac{\mathrm{d}t}{t})$

(5.11)
$$\|f\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} = \sqrt{2\pi} \, \|t \mapsto f(e^{2\pi t})\|_{\alpha(\mathbb{R};X)}.$$

Let $x\in D(A)\cap R(A)$ and note that by the definition of the Dunford calculus and Fubini's theorem

$$(t \mapsto \varphi_{s,r}(tA)x) \in L^2(\mathbb{R}_+, \frac{\mathrm{d}t}{t}; X),$$
$$(t \mapsto \varphi_{s,r}(e^{2\pi t}A)x) \in L^1(\mathbb{R}; X).$$

So by (5.9), (5.11) and the invariance of the α -norms under the Fourier transform (see Example 3.2.3) we have

$$\begin{split} \|t \mapsto \varphi_{s,r}(tA)x\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)} &= \sqrt{2\pi} \, \|t \mapsto \varphi_{s,r}(e^{2\pi t}A)x\|_{\alpha(\mathbb{R};X)} \\ &= \sqrt{2\pi} \, \|t \mapsto \int_{\mathbb{R}} e^{2\pi i t\xi} g(\xi)A^{i\xi}x \, \mathrm{d}\xi \|_{\alpha(\mathbb{R};X)} \\ &= \sqrt{2\pi} \, \|\xi \mapsto g(\xi)A^{i\xi}x\|_{\alpha(\mathbb{R};X)}, \end{split}$$

which proves the equivalence for the continuous square function by (5.8).

For the discrete square function norm note that by (5.9) we have for $x \in D(A) \cap R(A)$ and $t \in \mathbb{R}$

$$\varphi_{s,r}(e^{2\pi t}A)x = \sum_{m\in\mathbb{Z}}\int_{-b}^{b} e^{2\pi i t(\xi+2mb)}g(\xi+2mb)A^{i(\xi+2mb)}x\,\mathrm{d}\xi.$$

The sum converges absolutely by (5.10). Thus, using $2^{in \cdot 2mb} = 1$ and setting $2^n u = e^{2\pi t}$, we have

(5.12)
$$\varphi_{s,r}(2^n u A) x = \int_{-b}^{b} 2^{in\xi} \sum_{m \in \mathbb{Z}} u^{i(\xi+2mb)} g(\xi+2mb) A^{i(\xi+2mb)} x \, \mathrm{d}\xi.$$

By Parseval's theorem and Proposition 3.2.1 for any $h \in L^1([-b,b];X)$ with $h \in \alpha([-b,b];X)$, we have

$$\|h\|_{\alpha([-b,b];X)} = \sqrt{2b} \left\| (\widehat{h}(n))_{n \in \mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)},$$

where $b = \pi / \log(2)$ and

$$\widehat{h}(n) := \frac{1}{2b} \int_{-b}^{b} h(\xi) 2^{-in\xi} \,\mathrm{d}\xi.$$

And thus, using (5.12) and the fact that $|u^{i(\xi+2mb)}| = 1$ for all $\xi \in [-b, b]$, $m \in \mathbb{Z}$ and $u \in [1, 2]$, we obtain

$$\left\| (\varphi_{s,r}(2^n uA)x)_{n \in \mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)} \simeq \left\| \sum_{m \in \mathbb{Z}} g(\cdot + 2mb) A^{i(\cdot + 2mb)} x \right\|_{\alpha([-b,b];X)}$$

which combined with (5.8) proves the equivalence for the discrete square function norm. $\hfill \Box$

From Proposition 5.4.1 we can immediately deduce embeddings between the $H^{\alpha}_{\theta,A}(\varphi_{s,r})$ -spaces.

COROLLARY 5.4.2. Let A be a sectorial operator on X and assume that α is ideal. Fix $0 < s_1 \leq s_2 < \frac{\pi}{\omega(A)}$, $0 < r_1, r_2 < 1$ and $\theta \in \mathbb{R}$. For $\nu = \pi(\frac{1}{s_1} - \frac{1}{s_2})$ and $\omega(A) < \sigma < \frac{\pi}{s_2}$ set

$$\varphi_{s_1,r_1}^{\pm\nu}(z) := \varphi_{s_1,r_1}(\mathrm{e}^{\pm i\nu}z), \qquad z \in \Sigma_{\sigma}.$$

Then we have the continuous embedding

$$H^{\alpha}_{\theta,A}(\varphi_{s_2,r_2}) \hookrightarrow H^{\alpha}_{\theta,A}(\varphi_{s_1,r_1})$$

and

$$H^{\alpha}_{\theta,A}(\varphi^{+\nu}_{s_1,r_1}) \cap H^{\alpha}_{\theta,A}(\varphi^{-\nu}_{s_1,r_1}) = H^{\alpha}_{\theta,A}(\varphi_{s_2,r_2}).$$

isomorphically.

PROOF. Without loss of generality we may assume $\theta = 0$. The claimed embedding is a direct consequence of Proposition 5.4.1 and the density of $D(A) \cap R(A)$ in $H^{\alpha}_{\theta,A}(\varphi_{s_2,r_2})$. For the isomorphism fix $x \in D(A) \cap R(A)$. Then by Proposition 5.4.1 we have

$$\begin{split} \|\varphi_{s_{1},r_{1}}^{+\nu}(\cdot A)x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} &\simeq \left\|\mathrm{e}^{-(\nu+\frac{\pi}{s_{1}})|\cdot|}A^{i\cdot}x\,\mathbf{1}_{(0,\infty)}\right\|_{\alpha(\mathbb{R};X)} \\ &+ \left\|\mathrm{e}^{-\frac{\pi}{s_{2}}|\cdot|}A^{i\cdot}x\,\mathbf{1}_{(-\infty,0)}\right\|_{\alpha(\mathbb{R};X)}, \\ \|\varphi_{s_{1},r_{1}}^{-\nu}(\cdot A)x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} &\simeq \left\|\mathrm{e}^{-\frac{\pi}{s_{2}}|\cdot|}A^{i\cdot}x\,\mathbf{1}_{(0,\infty)}\right\|_{\alpha(\mathbb{R};X)} \\ &+ \left\|\mathrm{e}^{-(\nu+\frac{\pi}{s_{1}})|\cdot|}A^{i\cdot}x\,\mathbf{1}_{(-\infty,0)}\right\|_{\alpha(\mathbb{R};X)}, \\ \|\varphi_{s_{2},r_{2}}(\cdot A)x\|_{\alpha(\mathbb{R}_{+},\frac{\mathrm{d}t}{t};X)} &\simeq \left\|\mathrm{e}^{-\frac{\pi}{s_{2}}|\cdot|}A^{i\cdot}x\,\mathbf{1}_{(0,\infty)}\right\|_{\alpha(\mathbb{R};X)} \\ &+ \left\|\mathrm{e}^{-\frac{\pi}{s_{2}}|\cdot|}A^{i\cdot}x\,\mathbf{1}_{(-\infty,0)}\right\|_{\alpha(\mathbb{R};X)}. \end{split}$$

Since $\nu + \frac{\pi}{s_1} \ge \frac{\pi}{s_2}$, the corollary now follows by density and Example 3.2.2.

From Corollary 5.4.2 we can see that the spaces $H^{\alpha}_{\theta,A}(\varphi_{s,r})$ are independent of r, which is why we will focus on the spaces $H^{\alpha}_{\theta,A}(\varphi_s)$ for

$$\varphi_s(z) := \varphi_{s,\frac{1}{2}}(z) = \frac{z^{s/2}}{1+z^s}, \qquad z \in \Sigma_{\sigma}$$

with $\omega(A) < \sigma < \frac{\pi}{s}$ for the remainder of this section. Moreover, Corollary 5.4.2 states that the $H^{\alpha}_{\theta,A}(\varphi_s)$ -spaces shrink as s increases.

5.4. GENERALIZED SQUARE FUNCTION SPACES WITHOUT ALMOST α -SECTORIALITY25

The operators $A|_{H^{\alpha}_{\theta,A}(\varphi_s)}$ and their properties. We will now analyse the properties of the operators $A|_{H^{\alpha}_{\theta,A}(\varphi_s)}$ on $H^{\alpha}_{\theta,A}(\varphi_s)$. As a first observation, we note that from Proposition 5.4.1 we immediately deduce for $0 < s < \frac{\pi}{\omega(A)}$ and $\theta \in \mathbb{R}$ that $A|_{H^{\alpha}_{\theta,A}(\varphi_s)}$ has BIP with

(5.13)
$$\omega_{\rm BIP}(A|_{H^{\alpha}_{\theta,A}(\varphi_s)}) \le \frac{\pi}{s}$$

Using the characterization of α -BIP in Theorem 4.5.6 and the transference result of Theorem 4.4.1, we can say more if s > 1.

THEOREM 5.4.3. Let A be a sectorial operator on X and assume that α is ideal. Fix $1 < s < \frac{\pi}{\omega(A)}$ and $\theta \in \mathbb{R}$. Then $A|_{H^{\alpha}_{\theta,A}(\varphi_s)}$ has a bounded H^{∞} -calculus on $H^{\alpha}_{\theta,A}(\varphi_s)$ with

$$\omega_{H^{\infty}}(A|_{H^{\alpha}_{\theta,A}(\varphi_s)}) \leq \frac{\pi}{s}.$$

We give two proofs. The first is far more elegant, relying on the transference result in Chapter 1. In particular, we will juse the characterization of α -BIP in Theorem 4.5.6. We include a sketch of a second, more direct and elementary, but highly technical proof. This leads to a proof for the angle of the H^{∞} -calculus counterexample in Section 6.4 which does not rely on the theory in Chapter 1

PROOF OF THEOREM 5.4.3. Define a Euclidean structure β on $\alpha(\mathbb{R}; X)$ by defining for $T_1, \ldots, T_n \in \alpha(\mathbb{R}; X)$

$$||(T_1,\ldots,T_n)||_{\beta} = ||T_1 \oplus \cdots \oplus T_n||_{\alpha(L^2(\mathbb{R})^n;X)},$$

where we view $T_1 \oplus \cdots \oplus T_n$ as an operator from $L^2(\mathbb{R})^n$ to X given by

$$(T_1 \oplus \cdots \oplus T_n)(h_1, \ldots, h_n) := \sum_{k=1}^n T_k h_k, \qquad (h_1, \ldots, h_n) \in L^2(\mathbb{R})^n.$$

By Proposition 5.4.1, the space $H^{\alpha}_{\theta,A}(\varphi_s)$ is continuously embedded in $\alpha(\mathbb{R}; X)$ via the map

$$x \mapsto (t \mapsto e^{-\frac{\pi}{s}|t|} A^{it+\theta} x), \qquad x \in D(A^m) \cap R(A^m)$$

with $m \in \mathbb{N}$ such that $|\theta| + 1 < m$. Therefore β can be endowed upon $H^{\alpha}_{\theta,A}(\varphi_s)$. We will show that

$$\Gamma := \left\{ e^{-\frac{\pi}{s}|t|} (A|_{H^{\alpha}_{\theta,A}(\varphi_s)})^{it} : t \in \mathbb{R} \right\}$$

is β -bounded, which combined with Theorem 4.5.6 yields the theorem. Suppose that $t_1, \ldots, t_n \in \mathbb{R}$ and $x_1, \ldots, x_n \in D(A^m) \cap R(A^m)$. Then

$$\begin{split} \left\| \left(e^{-\frac{\pi}{s} |t_{k}|} (A|_{H^{\alpha}_{\theta,A}(\varphi_{s})})^{it_{k}} x_{k} \right)_{k=1}^{n} \right\|_{\beta} \\ &= \left\| \oplus_{k=1}^{n} (t \mapsto e^{-\frac{\pi}{s} (|t|+|t_{k}|)} A^{i(t+t_{k})+\theta} x_{k}) \right\|_{\alpha(L^{2}(\mathbb{R})^{n};X)} \\ &= \left\| \oplus_{k=1}^{n} (t \mapsto e^{-\frac{\pi}{s} (|t-t_{k}|+|t_{k}|)} A^{it+\theta} x_{k}) \right\|_{\alpha(L^{2}(\mathbb{R})^{n};X)} \\ &\leq \left\| \oplus_{k=1}^{n} (t \mapsto e^{-\frac{\pi}{s} (|t|)} A^{it+\theta} x_{k}) \right\|_{\alpha(L^{2}(\mathbb{R})^{n};X)} \\ &= \left\| (x_{k})_{k=1}^{n} \right\|_{\beta} \end{split}$$

Now the β -boundedness of Γ follows by the density of $D(A^m) \cap R(A^m)$ in $H^{\alpha}_{\theta,A}(\varphi_s)$, which proves the theorem. \Box

SKETCH OF AN ALTERNATIVE PROOF OF THEOREM 5.4.3. Without loss of generality we may assume $\theta = 0$. Fix $\frac{\pi}{s} < \nu < \sigma < \pi$, take $f \in H^{\infty}(\Sigma_{\sigma})$ with $\|f\|_{H^{\infty}(\Sigma_{\sigma})} \leq 1$ and fix 0 < a, b, c < 1 such that a + b = 1 + c. Then, using a similar calculation as in the proof of Proposition 5.1.7, we can write for $\nu' = \pi - \nu$ and $x \in D(A^2) \cap R(A^2)$

$$f(A)x = \sum_{\epsilon=\pm 1} \frac{-\epsilon e^{-ia\pi}}{2\pi i} \int_0^\infty f(s^{-1}e^{\epsilon i\nu})\varphi_{1,a}(s e^{\epsilon i\nu'}A)x \frac{\mathrm{d}s}{s}.$$

To estimate $||f(A)||_{H^{\alpha}_{0,A}(\varphi_s)}$ we will first consider the integral for $\epsilon = 1$. Note that we have the identity

$$\varphi_{1,a}(\lambda A)\varphi_{1b}(\mu A) = \frac{\lambda^{1-b}\mu^b}{\mu-\lambda}\varphi_{1,c}(\lambda A) + \frac{\lambda^a\mu^{1-a}}{\lambda-\mu}\varphi_{1,c}(\mu A)$$

for $|\arg \lambda|, |\arg \mu| < \pi - \omega(A)$. Thus for s, t > 0 and $\nu'' = \pm \pi (1 - \frac{1}{s})$

$$\varphi_{1,a}(ste^{i\nu'}A)\varphi_{1,b}(te^{i\nu''}A) = \kappa_1(t)\frac{s^{1-b}}{1+s}\varphi_{1,c}(ste^{i\nu'}A) + \kappa_2(t)\frac{s^a}{1+s}\varphi_{1,c}(te^{i\nu''}A),$$

where $\kappa_1, \kappa_2 : \mathbb{R}_+ \to \mathbb{C}$ are bounded and continuous functions. Therefore

$$\begin{split} \left\| \int_0^\infty f(s^{-1}e^{i\nu})\varphi_{1,a}(se^{i\nu'}A)x\frac{\mathrm{d}s}{s} \right\|_{H^\alpha_{0,A}(\varphi_{1,b}(e^{i\nu''}\cdot))} \\ &= \left\| \int_0^\infty f(s^{-1}t^{-1}e^{i\nu})\varphi_{1,a}(ste^{i\nu'}A)\varphi_{1,b}(te^{i\nu''}A)x\frac{\mathrm{d}t}{t} \right\|_{\alpha(\mathbb{R},\frac{\mathrm{d}t}{t};X)} \\ &\lesssim \int_0^\infty \frac{s^a}{1+s} \|\varphi_{1,c}(ste^{i\nu'}A)\|_{\alpha(\mathbb{R},\frac{\mathrm{d}t}{t};X)}x\frac{\mathrm{d}s}{s} \\ &+ \int_0^\infty \frac{s^{1-b}}{1+s} \|\varphi_{1,c}(te^{i\nu''}A)\|_{\alpha(\mathbb{R},\frac{\mathrm{d}t}{t};X)}x\frac{\mathrm{d}s}{s} \\ &\lesssim \|x\|_{H^\alpha_{0,A}(\varphi_{1,c}(e^{i\nu'}\cdot))} + \|x\|_{H^\alpha_{0,A}(\varphi_{1,c}(e^{i\nu''}\cdot))}. \end{split}$$

Combining the estimates for $\nu'' = \pm \pi (1 - \frac{1}{s})$ with the isomorphism from Corollary 5.4.2 with parameters $s_1 = 1$, $s_2 = s$, $r_1 = b$, c and $r_2 = \frac{1}{2}$, we obtain

$$\begin{split} \left\| \int_0^\infty f(s^{-1}e^{i\nu})\varphi_{1,a}(se^{i\nu'}A)x\frac{\mathrm{d}s}{s} \right\|_{H^\alpha_{0,A}(\varphi_s)} \\ &\lesssim \|x\|_{H^\alpha_{0,A}(\varphi_{1,c}(e^{i\nu'}\cdot))} + \|x\|_{H^\alpha_{0,A}(\varphi_s)} \end{split}$$

and since $\nu' \leq (1 - \frac{1}{s})$, applying Corollary 5.4.2 once more yields

$$\|x\|_{H^{\alpha}_{0,A}(\varphi_{1,c}(\mathrm{e}^{i\nu'}\cdot))} \lesssim \|x\|_{H^{\alpha}_{0,A}(\varphi_{s})}.$$

Doing a similar computation for $\epsilon = -1$ yields the theorem.

If we have a strict inequality

$$\omega_{\rm BIP}(A|_{H^{\alpha}_{\theta,A}(\varphi_s)}) < \pi,$$

we can extend Theorem 5.4.3 to s = 1. So in this case $A|_{H^{\alpha}_{\theta,A}}$ on $H^{\alpha}_{\theta,A}$ "behaves" like a Hilbert space operator, as it has BIP if and only if it has a bounded H^{∞} -calculus.

THEOREM 5.4.4. Let A be a sectorial operator on X, suppose that α is ideal and fix $\theta \in \mathbb{R}$. The following are equivalent:

(i) $A|_{H^{\alpha}_{\theta,A}}$ has BIP with $\omega_{\text{BIP}}(A|_{H^{\alpha}_{\theta,A}}) < \pi$

- (ii) There is a $1 < \sigma < \frac{\pi}{\omega(A)}$ such that the spaces $H^{\alpha}_{\theta,A}(\varphi_s)$ are isomorphic for all $0 < s < \sigma$.
- (iii) $A|_{H^{\alpha}_{\theta,A}}$ has a bounded H^{∞} -calculus.

PROOF. The implication (ii) \Rightarrow (iii) follows directly from Theorem 5.4.3 and (iii) \Rightarrow (i) is immediate from (4.17). For (i) \Rightarrow (ii) let $\sigma > 1$ be such that

(5.14)
$$\left\| (A|_{H^{\alpha}_{\theta,A}})^{it} \right\| \le C \mathrm{e}^{\frac{\pi}{\sigma}|t|}, \qquad t \in \mathbb{R}.$$

Fix $x \in D(A^m) \cap R(A^m)$ with $m \in \mathbb{N}$ such that $|\theta| + 1 < m$ and take $0 < s' < s < \sigma$. Then by Proposition 5.4.1, (5.14) and the ideal property of α we have

$$\begin{aligned} \|x\|_{H^{\alpha}_{\theta,A}(\varphi_{s})} &\lesssim \left\| t \mapsto \mathrm{e}^{-\frac{\pi}{s}|t|} A^{it+\theta}(t) x \right\|_{\alpha(\mathbb{R};X)} \\ &\leq \sum_{n \in \mathbb{Z}} \left\| t \mapsto \mathrm{e}^{-\frac{\pi}{s}|t|} A^{it+\theta} \,\mathbf{1}_{[n,n+1)}(t) x \right\|_{\alpha(\mathbb{R};X)} \\ &\leq \mathrm{e}^{\frac{\pi}{s}} \sum_{n \in \mathbb{Z}} \mathrm{e}^{-\frac{\pi}{s}|n|} \left\| (A|_{H^{\alpha}_{\theta,A}})^{in} \right\| \left\| t \mapsto A^{it+\theta} \,\mathbf{1}_{[0,1)}(t) x \right\|_{\alpha(\mathbb{R};X)} \\ &\leq \mathrm{e}^{\pi(s^{-1}+s'^{-1})} \sum_{n \in \mathbb{Z}} \mathrm{e}^{-\pi(s^{-1}-\sigma^{-1})|n|} \left\| t \mapsto \mathrm{e}^{-\frac{\pi}{s'}|t|} A^{it+\theta} \,\mathbf{1}_{[0,1)}(t) x \right\|_{\alpha(\mathbb{R};X)} \\ &\lesssim \|x\|_{H^{\alpha}_{\theta,A}(\varphi_{s'})}. \end{aligned}$$

Moreover by Corollary 5.4.2 we have the converse estimate

$$\|x\|_{H^{\alpha}_{\theta,A}(\varphi_{s'})} \lesssim \|x\|_{H^{\alpha}_{\theta,A}(\varphi_{s})},$$

so by the density of $D(A^m) \cap R(A^m)$ the spaces $H^{\alpha}_{\theta,A}(\varphi_{s'})$ and $H^{\alpha}_{\theta,A}(\varphi_s)$ are isomorphic.

Using Theorem 5.4.4, we end this section with another theorem on the equivalence of discrete and continuous square functions, as treated in Proposition 5.1.2 and Corollary 5.1.5. This time for a very specific choice of ψ and under the assumption that one of the equivalent statements of Theorem 5.4.4 holds. Note that in this special case we can also omit the supremum over $t \in [1, 2]$ for the discrete square functions.

PROPOSITION 5.4.5. Let A be a sectorial operator on X and suppose that α is ideal. Assume that $A|_{H^{\alpha}_{0,A}}$ has a bounded H^{∞} -calculus on $H^{\alpha}_{0,A}$. Then there is a $1 < \sigma < \frac{\pi}{\omega(A)}$ such that for all $0 < s < \sigma$, and $x \in D(A) \cap R(A)$ we have

$$\left\|\varphi_s(\cdot A)x\right\|_{\alpha(\mathbb{R}_+,\frac{\mathrm{d}t}{t};X)} \simeq \left\|(\varphi_s(2^n A)x)_{n\in\mathbb{Z}}\right\|_{\alpha(\mathbb{Z};X)}$$

PROOF. Take $1 < \sigma < \pi/\omega(A)$ as in Theorem 5.4.4(ii), let $0 < s < \sigma$ and $0 < \delta < \frac{\pi}{s} - \omega(A)$. Then by Proposition 5.1.2 and Proposition 5.4.1 we have

$$\begin{aligned} \left\| (\varphi_s(2^n A)x)_{n \in \mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)} &\lesssim \max_{\epsilon = \pm \delta} \| t \mapsto \varphi_s(t e^{i\epsilon} A)x \|_{\alpha(\mathbb{R}_+, \frac{dt}{t};X)} \\ &\lesssim \max_{\epsilon = \pm \delta} \| t \mapsto e^{-\frac{\pi}{s}|t|} e^{-\epsilon t} A^{it} x \|_{\alpha(\mathbb{R};X)} \\ &\leq \| t \mapsto e^{-\frac{\pi}{s'}|t|} A^{it} x \|_{\alpha(\mathbb{R};X)} \\ &\lesssim \| \varphi_{s'}(\cdot A)x \|_{\alpha(\mathbb{R}_+, \frac{dt}{t};X)} \end{aligned}$$

with $s' = \frac{\pi s}{\pi - \delta s}$. So taking δ small enough such that $0 < s' < \sigma$ it follows from Theorem 5.4.4(ii) that

$$\left\| (\varphi_s(2^n A)x)_{n \in \mathbb{Z}} \right\|_{\alpha(\mathbb{Z};X)} \lesssim \|\varphi_s(\cdot A)x\|_{\alpha(\mathbb{R}_+,\frac{dt}{t};X)}.$$

For the converse inequality let $s' = \frac{\pi s}{\pi + \delta s}$ for some $0 < \delta < \frac{\pi}{s} - \omega(A)$. Then we have by Proposition 5.1.2 and Proposition 5.4.1

$$\begin{split} \|\varphi_{s}(\cdot A)x\|_{\alpha(\mathbb{R}_{+},\frac{dt}{t};X)} &\lesssim \|\varphi_{s'}(\cdot A)x\|_{\alpha(\mathbb{R}_{+},\frac{dt}{t};X)} \\ &\lesssim \sup_{|\epsilon|<\delta} \sup_{t\in[1,2]} \left\|(\varphi_{s'}(2^{n}te^{i\epsilon}A)x)_{n\in\mathbb{Z}}\right\|_{\alpha(\mathbb{Z};X)} \\ &\lesssim \sup_{|\epsilon|<\delta} \left\|\sum_{m\in\mathbb{Z}} e^{-\pi|\cdot+2mb|/s'}e^{-\epsilon(\cdot+2mb)}A^{i(\cdot+2mb)}\right\|_{\alpha([-b,b];X)} \\ &\leq \left\|\sum_{m\in\mathbb{Z}} e^{-\pi|\cdot+2mb|/s}A^{i(\cdot+2mb)}\right\|_{\alpha([-b,b];X)} \\ &\lesssim \left\|(\varphi_{s}(2^{n}A)x)_{n\in\mathbb{Z}}\right\|_{\alpha(\mathbb{Z};X)}, \end{split}$$

finishing the proof.

CHAPTER 6

Some counterexamples

In Chapter 4 we introduced various properties of a sectorial operator A on X and proved the following relations between these properties:



Moreover we noted that (1) and (2) are 'if and only if' statements if $\alpha = \ell^2$ or $\alpha = \gamma$ and X has Pisier's contraction property. Statements (3), (4) and (5) cannot be turned into 'if and only if' statements for $\alpha = \ell^2$ or $\alpha = \gamma$. Indeed, there are counterexamples on spaces admitting an unconditional Schauder basis disproving the converse of (3), (4) and (5) for \mathcal{R} -boundedness, which is equivalent to ℓ^2 - and γ -boundedness if X has finite cotype by Proposition 1.0.1. We refer to the survey of Fackler [Fac15] and the references therein for an overview of these counterexamples

In this chapter we will show that (6) can also not be turned into an 'if and only if' statement for any unconditionally stable Euclidean structure α on a Banach space admitting an unconditional Schauder basis. Moreover in the same setting we will show that even the weakest property, the almost α -sectoriality of A, does not follow from the sectoriality of A.

In Chapter 4 we have also seen that under reasonable assumptions on α the angles of (almost) α -sectoriality, (α -)BIP and of the (α -)-bounded H^{∞} -calculus are equal whenever A has these properties. Strikingly absent in this list is the angle of sectoriality of A. In Section 4.5 we already remarked that it is possible to have $\omega_{\text{BIP}}(A) \geq \pi$ and thus $\omega_{\text{BIP}}(A) > \omega(A)$, see [Haa03, Corollary 5.3]. Moreover in [Kal03] it was shown that it is also possible to have $\omega_{H^{\infty}}(A) > \omega(A)$. However, the Banach space used in [Kal03] is quite unnatural. We will end this chapter with an example of a sectorial operator with $\omega_{H^{\infty}}(A) > \omega(A)$ on a closed subspace of L^p , using the $H^{\alpha}_{\theta A}$ -spaces introduced in Chapter 5.

6.1. Schauder multiplier operators

We start by introducing the class of operators that we will use in our examples. This will be the class of so-called Schauder multiplier operators. The idea of using Schauder multiplier operators to construct counterexamples in the context of sectorial operators goes back to Clement and Baillon [**BC91**] and Venni [**Ven93**], where Schauder multipliers were used to construct examples of sectorial operators without

BIP. It has since proven to be a fruitful method to construct counterexamples in this context, see for example [AL19, CDMY96, Fac13, Fac14a, Fac15, Fac16, KL00, KL02, Lan98, LM04]. For $L^1(S)$ - and C(K)-spaces different counterexamples, connected to the breakdown of the theory of singular integral operators, are available, see e.g. [HKK04, KK08, KW05].

Schauder decompositions. Let $(X_k)_{k=1}^{\infty}$ be a sequence of closed subspaces of X. Then $(X_k)_{k=1}^{\infty}$ is called a *Schauder decomposition* of X if every $x \in X$ has a unique representation of the form $x = \sum_{k=1}^{\infty} x_k$ with $x_k \in X_k$ for every $k \in \mathbb{N}$. A Schauder decomposition induces a sequence of coordinate projections $(P_k)_{k=1}^{\infty}$ on X by putting

$$P_k\left(\sum_{j=1}^{\infty} x_j\right) := x_k, \qquad k \in \mathbb{N}$$

We denote the partial sum projection by $S_n := \sum_{k=1}^n P_k$. Both the set of coordinate and the set of partial sum projections are uniformly bounded. A Schauder decomposition is called *unconditional* if for every $x \in X$, the expansion $x = \sum_{k=1}^{\infty} x_k$ with $x_k \in X_k$ converges unconditionally. In this case the set of operators $U_{\epsilon} :=$ $\sum_{k=1}^{\infty} \epsilon_k P_k$, where $\epsilon = (\epsilon_k)_{k=1}^{\infty}$ is a sequence of signs, is also uniformly bounded.

A Schauder decomposition $(X_k)_{k=1}^{\infty}$ of X with $\dim(X_k) = 1$ for all $k \in \mathbb{N}$ is called a *Schauder basis*. In this case we represent $(X_k)_{k=1}^{\infty}$ by $\mathbf{x} = (x_k)_{k=1}^{\infty}$ with $x_k \in X_k$ for all $k \in \mathbb{N}$. Then there is a unique sequence of scalars $(a_k)_{k=1}^{\infty}$ such that $x = \sum_{k=1}^{\infty} a_k x_k$ for any $x \in X$. The sequence of linear functionals $\mathbf{x}^* = (x_k^*)_{k=1}^{\infty}$ defined by

$$x_k^*\left(\sum_{j=1}^\infty a_j x_j\right) := a_k, \qquad k \in \mathbb{N},$$

is called the biorthogonal sequence of \mathbf{x} , which is a Schauder basis of $\overline{\operatorname{span}}\{x_k^* : k \in \mathbb{N}\}$. If \mathbf{x} is unconditional, then \mathbf{x}^* is as well. If \mathbf{x} is a Schauder basis for X and \mathbf{y} is a Schauder basis for Y, then we say that \mathbf{x} and \mathbf{y} are *equivalent* if $\sum_{k=1}^{\infty} a_k x_k$ converges in X if and only if $\sum_{k=1}^{\infty} a_k y_k$ in Y for any sequence of scalars $(a_k)_{k=1}^{\infty}$. In this case X and Y are isomorphic. For a further introduction to Schauder decompositions and bases, we refer to [LT77].

Schauder mutliplier operators. Fix $0 < \sigma < \pi$ and let $(\lambda_k)_{k=1}^{\infty}$ be a sequence in Σ_{σ} . We call $(\lambda_k)_{k=1}^{\infty}$ Hadamard if $|\lambda_1| > 0$ and there is a c > 1 such that $|\lambda_{k+1}| \ge c |\lambda_k|$ for all $k \in \mathbb{N}$. Let $(X_k)_{k=1}^{\infty}$ be a Schauder decomposition of X and let $(\lambda_k)_{k=1}^{\infty}$ be either a Hadamard sequence or an increasing sequence in \mathbb{R}_+ . Consider the unbounded diagonal operator defined by

$$Ax := \sum_{k=1}^{\infty} \lambda_k P_k x,$$
$$D(A) := \Big\{ x \in X : \sum_{k=1}^{\infty} \lambda_k P_k x \text{ converges in } X \Big\}.$$

We call A the Schauder multiplier operator associated to $(X_k)_{k=1}^{\infty}$ and $(\lambda_k)_{k=1}^{\infty}$. We will first establish that this is a sectorial operator, for which we will need the following lemma. LEMMA 6.1.1. Let $(\lambda_k)_{k=1}^{\infty}$ be either a Hadamard sequence or an increasing sequence in \mathbb{R}_+ . There is a C > 0 such that for all $\lambda \in \mathbb{C} \setminus \{\lambda_k : k \in \mathbb{N}\}$

$$\sum_{k=1}^{\infty} \left| \frac{\lambda}{\lambda - \lambda_{k+1}} - \frac{\lambda}{\lambda - \lambda_k} \right| \le C \sup_{k \in \mathbb{N}} \left(\frac{\max\{|\lambda|, |\lambda_k|\}}{|\lambda - \lambda_k|} \right)^2.$$

PROOF. For $n \in \mathbb{N}$ define $\mu_n := |\lambda_1| + \sum_{k=1}^{n-1} |\lambda_{k+1} - \lambda_k|$. In both cases there exists a $C_{\mu} > 0$ such that $|\lambda_k| \le \mu_k \le C_{\mu} |\lambda_k|$ for all $k \in \mathbb{N}$. Fix $\lambda \in \mathbb{C} \setminus \{\lambda_k : k \in \mathbb{N}\}$ and define

$$C_{\lambda} := \sup_{k \in \mathbb{N}} \frac{\max\{|\lambda|, |\lambda_k|\}}{|\lambda - \lambda_k|} < \infty$$

We have for all $k\in\mathbb{N}$

$$\left|\frac{\lambda}{\lambda-\lambda_{k+1}} - \frac{\lambda}{\lambda-\lambda_k}\right| \le C_{\lambda}^2 \frac{|\lambda||\lambda_{k+1} - \lambda_k|}{\max\{|\lambda|, |\lambda_{k+1}|\} \cdot \max\{|\lambda|, |\lambda_k|\}}$$

Fix $n \in \mathbb{N}$ such that $|\lambda_n| \leq |\lambda| < |\lambda_{n+1}|$ (or take n = 0 if this is not possible). Then

$$\sum_{k=1}^{n-1} \left| \frac{\lambda}{\lambda - \lambda_{k+1}} - \frac{\lambda}{\lambda - \lambda_k} \right| \le C_{\lambda}^2 |\lambda| \sum_{k=1}^{n-1} \frac{|\lambda_{k+1} - \lambda_k|}{|\lambda|^2} \le C_{\lambda}^2 |\lambda|^{-1} |\mu_n| \le C_{\mu} C_{\lambda}^2,$$

and

$$\begin{split} \sum_{k=n+1}^{\infty} \left| \frac{\lambda}{\lambda - \lambda_{k+1}} - \frac{\lambda}{\lambda - \lambda_k} \right| &\leq C_{\lambda}^2 |\lambda| \sum_{k=n+1}^{\infty} \frac{|\lambda_{k+1} - \lambda_k|}{|\lambda_{k+1}||\lambda_k|} \\ &\leq C_{\mu}^2 C_{\lambda}^2 |\lambda| \sum_{k=n+1}^{\infty} \frac{\mu_{k+1} - \mu_k}{\mu_{k+1} \mu_k} \\ &\leq C_{\mu}^2 C_{\lambda}^2 \Big(\frac{|\lambda|}{\mu_{n+1}} + \lim_{k \to \infty} \frac{|\lambda|}{\mu_k} \Big) \leq C_{\mu}^2 C_{\lambda}^2 \end{split}$$

and finally

$$\left|\frac{\lambda}{\lambda-\lambda_{n+1}} - \frac{\lambda}{\lambda-\lambda_n}\right| \le C_{\lambda}^2 |\lambda| \frac{|\lambda_{n+1} - \lambda_n|}{|\lambda_{n+1}||\lambda|} \le 2C_{\lambda}^2.$$

Combined this proves the lemma.

To show that an operator associated to a Schauder decomposition and a Hadamard or increasing sequence is sectorial is now straightforward.

PROPOSITION 6.1.2. Let $(X_n)_{n=1}^{\infty}$ be a Schauder decomposition of X. Let $(\lambda_k)_{k=1}^{\infty}$ be either a Hadamard sequence or an increasing sequence in \mathbb{R}_+ . Let A be the operator associated to $(X_k)_{k=1}^{\infty}$ and $(\lambda_k)_{k=1}^{\infty}$. Then A is sectorial with

$$\omega(A) = \inf \left\{ 0 < \sigma < \pi : \lambda_k \in \Sigma_\sigma \text{ for all } k \in \mathbb{N} \right\}.$$

PROOF. Fix $\lambda \in \mathbb{C} \setminus \overline{\{\lambda_k : k \in \mathbb{N}\}}$ and define

$$C_{\lambda} := \sup_{k \in \mathbb{N}} \frac{\max\{|\lambda|, |\lambda_k|\}}{|\lambda - \lambda_k|} < \infty, \qquad \qquad C_S := \sup_{k \in \mathbb{N}} \|S_k\|$$

Note that for any $n \in \mathbb{N}$

(6.1)
$$\sum_{k=1}^{n} \frac{1}{\lambda - \lambda_k} P_k = \frac{1}{\lambda - \lambda_{n+1}} S_n - \sum_{k=1}^{n} \left(\frac{1}{\lambda - \lambda_{k+1}} - \frac{1}{\lambda - \lambda_k} \right) S_k.$$

So by Lemma 6.1.1 we have for all $n \in \mathbb{N}$

(6.2)
$$\left\|\sum_{k=1}^{n} \frac{1}{\lambda - \lambda_{k}} P_{k}\right\| = \left\|\frac{1}{\lambda - \lambda_{n+1}} S_{n} - \sum_{k=1}^{n} \left(\frac{1}{\lambda - \lambda_{k+1}} - \frac{1}{\lambda - \lambda_{k}}\right) S_{k}\right\|$$
$$\leq \frac{C_{\lambda}}{|\lambda|} \|S_{n}\| + \frac{CC_{\lambda}^{2}}{|\lambda|} \sup_{1 \le k \le n} \|S_{k}\|$$
$$\leq CC_{\lambda}^{2} C_{S} |\lambda|^{-1}.$$

By a similar computation we see that $\left(\sum_{k=1}^{n} \frac{1}{\lambda - \lambda_k} P_k\right)_{n=1}^{\infty}$ is a Cauchy sequence and therefore convergent. Thus

$$R(\lambda) := \sum_{k=1}^{\infty} \frac{1}{\lambda - \lambda_k} P_k,$$

is a well-defined, bounded operator on X. Moreover we have

$$(\lambda - A)R(\lambda)x = \sum_{k=1}^{\infty} \frac{\lambda}{\lambda - \lambda_k} P_k x - \sum_{j=1}^{\infty} \lambda_j P_j \sum_{k=1}^{\infty} \frac{1}{\lambda - \lambda_k} P_k x = x$$

for all $x \in X$ and similarly $R(\lambda)(\lambda - A)x = x$ for $x \in D(A)$. Therefore $\lambda \in \rho(A)$ and $R(\lambda, A) = R(\lambda)$.

Since $(X_n)_{n=1}^{\infty}$ is a Schauder decomposition, A is injective and $x_n \in D(A) \cap R(A)$ for $x_n \in X_n$, so A has dense domain and dense range. Moreover, if we fix

$$\inf \left\{ 0 < \sigma < \pi : \lambda_k \in \Sigma_\sigma \text{ for all } k \in \mathbb{N} \right\} < \sigma' < \pi,$$

then there is a $C_{\sigma'} > 0$ such that

$$C_{\lambda} = \sup_{k \in \mathbb{N}} \frac{\max\{|\lambda|, |\lambda_k|\}}{|\lambda - \lambda_k|} \le C_{\sigma'}, \qquad \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma'}.$$

So by (6.2) A is sectorial with $\omega(A) \leq \sigma'$. Equality follows since $\lambda_n \in \sigma(A)$ for all $n \in \mathbb{N}$.

From the proof of Proposition 6.1.2 we can also see that

$$\rho(A) = \mathbb{C} \setminus \{\lambda_k : k \in \mathbb{N}\}$$

and for $\lambda \in \rho(A)$ we have

(6.3)
$$R(\lambda, A) = \sum_{k=1}^{\infty} \frac{1}{\lambda - \lambda_k} P_k = \sum_{k=1}^{\infty} \left(\frac{1}{\lambda - \lambda_k} - \frac{1}{\lambda - \lambda_{k+1}} \right) S_k.$$

Indeed, this follows by taking limits in (6.1). Let $\omega(A) < \nu < \sigma < \pi$. Using (6.2) and the dominated convergence theorem we have for $f \in H^1(\Sigma_{\sigma})$

(6.4)
$$f(A) = \int_{\Gamma_{\nu}} f(z)R(z,A) \,\mathrm{d}z = \sum_{k=1}^{\infty} \int_{\Gamma_{\nu}} \frac{f(z)}{z - \lambda_k} P_k \,\mathrm{d}z = \sum_{k=1}^{\infty} f(\lambda_k) P_k.$$

To extend this to the extended Dunford calculus let $f: \Sigma_{\sigma} \to \mathbb{C}$ be holomorphic satisfying

$$|f(z)| \le C|z|^{-\delta}(1+|z|)^{2\delta}$$

for some $C, \delta > 0$ and fix $x \in X$ with $P_k x = 0$ for all $k \ge N$ for some $N \in \mathbb{N}$. Then we have by (6.4) that

(6.5)
$$f(A)x = \lim_{n \to \infty} \sum_{k=1}^{N} f(\lambda_k) \varphi_n^m(\lambda_k) P_k x = \sum_{k=1}^{N} f(\lambda_k) P_k x$$

with $m > \delta$.

(Almost) α -bounded Schauder decompositions. Let α be a Euclidean structure on X. For the operator A associated to a Schauder decomposition $(X_k)_{k=1}^{\infty}$ and a Hadamard sequence $(\lambda_k)_{k=1}^{\infty}$ we can reformulate (almost) α -sectoriality in terms of the projections associated to $(X_k)_{k=1}^{\infty}$. Motivated by the following result we call $(X_k)_{k=1}^{\infty}$ almost α -bounded if the family of coordinate projections $\{P_k : k \in \mathbb{N}\}$ is α -bounded and we call $(X_k)_{k=1}^{\infty} \alpha$ -bounded if the family of partial sum projections $\{S_k : k \in \mathbb{N}\}$ is α -bounded.

PROPOSITION 6.1.3. Let $(X_k)_{k=1}^{\infty}$ be a Schauder decomposition of X, $(\lambda_k)_{k=1}^{\infty}$ a Hadamard sequence and A the sectorial operator associated to $(X_k)_{k=1}^{\infty}$ and $(\lambda_k)_{k=1}^{\infty}$. Let α be a Euclidean structure on X. Then

- (i) A is almost α -sectorial if and only if $(X_k)_{k=1}^{\infty}$ is almost α -bounded. In this case $\tilde{\omega}_{\alpha}(A) = \omega(A)$.
- (ii) A is α -sectorial if and only if $(X_k)_{k=1}^{\infty}$ is α -bounded. In this case $\omega_{\alpha}(A) = \omega(A)$

In the proof of Proposition 6.1.3 we will need the following interpolating property of $H^{\infty}(\Sigma_{\sigma})$ -functions evaluated in the points of a Hadamard sequence.

LEMMA 6.1.4. Fix $0 < \sigma < \nu < \pi$ and let $(\lambda_k)_{k=1}^{\infty}$ be a Hadamard sequence in Σ_{σ} . For all $\boldsymbol{a} \in \ell^{\infty}$ there exists an $f \in H^{\infty}(\Sigma_{\sigma})$ such that

$$f(\lambda_k) = a_k, \qquad k \in \mathbb{N}$$

and $\|f\|_{H^{\infty}(\Sigma_{\sigma})} \lesssim \|\boldsymbol{a}\|_{\ell^{\infty}}$.

PROOF. The lemma states that $(\lambda_k)_{k=1}^{\infty}$ is an interpolating sequence for $H^{\infty}(\Sigma_{\sigma})$. On the upper half-plane a theorem due to Carleson (see for example [Gar07]) states that $(\zeta_k)_{k=1}^{\infty}$ is an interpolating sequence if and only if

$$\prod_{j\in\mathbb{N}\setminus\{k\}} \left| \frac{\zeta_k - \zeta_j}{\zeta_k - \overline{\zeta}_j} \right| > 0, \qquad k\in\mathbb{N}.$$

Since the function $z \mapsto i z^{\frac{\pi}{2\nu}}$ conformally maps Σ_{σ} onto the upper half-plane, it suffices to show

$$\prod_{j\in\mathbb{N}\setminus\{k\}}\left|\frac{\mu_k-\mu_j}{\mu_k+\overline{\mu}_j}\right|>0$$

for $\mu_k = \lambda_k^{\frac{\pi}{2\sigma}}$. Fix $k \in \mathbb{N}$, then we have

$$\prod_{j=1}^{k-1} \left| \frac{\mu_k - \mu_j}{\mu_k + \overline{\mu}_j} \right| \ge \prod_{j=1}^{k-1} \frac{|\mu_k| - |\mu_j|}{|\mu_k| + |\mu_j|} = \prod_{j=1}^{k-1} \left(1 - \frac{2|\mu_j|}{|\mu_k| + |\mu_j|} \right) \ge \prod_{j=1}^{k-1} \left(1 - \frac{2}{c^{k-j} + 1} \right),$$

where c > 1 is such that $|\mu_{k+1}| \ge c |\mu_k|$ for all $k \in \mathbb{N}$. A similar inequality holds for the product with $j \ge k+1$. Therefore, since $\sum_{j=1}^{\infty} \frac{2}{c^{j+1}} < \infty$, it follows that

$$\prod_{j\in\mathbb{N}\setminus\{k\}} \left|\frac{\mu_k - \mu_j}{\mu_k + \overline{\mu}_j}\right| \ge \left(\prod_{j=1}^{\infty} \left(1 - \frac{2}{c^j + 1}\right)\right)^2 > 0,$$

which finishes the proof.

PROOF OF PROPOSITION 6.1.3. Fix $\omega(A) < \nu < \sigma < \pi$. For statement (i) first assume that $\{P_k : k \in \mathbb{N}\}$ is α -bounded. Take $f \in H^1(\Sigma_{\nu})$, then by (6.4) we have for t > 0

$$f(tA) = \sum_{k=1}^{\infty} f(t\lambda_k) P_k$$

and by Lemma 4.3.4 we have $\sum_{k=1}^{\infty} |f(t\lambda_k)| \leq C$ for C > 0 independent of t. Therefore it follows by Proposition 1.2.3 that $\{f(tA) : t > 0\}$ is α -bounded. Thus A is almost α -sectorial with $\tilde{\omega}(A) \leq \sigma$ by Proposition 4.2.3.

Conversely assume that A is almost α -sectorial and set $t_k = |\lambda_k|$ for $k \in \mathbb{N}$. By Lemma 6.1.4 there is a sequence of functions $(f_j)_{j=1}^{\infty}$ in $H^{\infty}(\Sigma_{\sigma})$ with $||f_j||_{H^{\infty}(\Sigma_{\sigma})} \leq C$ such that $f_j(\lambda_k) = \delta_{jk}$ for all $j, k \in \mathbb{N}$. Take

$$g_j(z) = \frac{z}{(1+z)^2} \frac{(t_j + \lambda_j)^2}{t_j \lambda_j} f_j(t_j z), \qquad z \in \Sigma_\sigma,$$

then $(g_j)_{j=1}^{\infty}$ is uniformly in $H^1(\Sigma_{\sigma})$. Therefore $\{g_j(t_j^{-1}A) : j \in \mathbb{N}\}$ is α -bounded by Proposition 4.2.3. By (6.4) we have for $j \in \mathbb{N}$

$$g_j(t_j^{-1}A) = \frac{t_j^{-1}\lambda_j}{(1+t_j^{-1}\lambda_j)^2} \frac{(t_j+\lambda_j)^2}{t_j\lambda_j} P_j = P_j.$$

So the family of coordinate projections $\{P_k : k \in \mathbb{N}\}$ is α -bounded, i.e. $(X_k)_{k=1}^{\infty}$ is almost α -bounded.

For (ii) assume that $\{S_k : k \in \mathbb{N}\}$ is α -bounded and take $\lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma}$. Using the expression for the resolvent of A from (6.3), we have

$$\lambda R(\lambda, A) = \sum_{k=1}^{\infty} \left(\frac{\lambda}{\lambda - \lambda_{k+1}} - \frac{\lambda}{\lambda - \lambda_k} \right) S_k$$

Therefore the set $\{\lambda R(\lambda, A) : \lambda \in \mathbb{C} \setminus \overline{\Sigma}_{\sigma}\}$ is α -bounded by Lemma 6.1.1 and Proposition 1.2.3, so A is α -sectorial with $\omega_{\alpha}(A) \leq \sigma$.

Conversely assume that A is α -sectorial and set $t_k = |\lambda_k|$ for $k \in \mathbb{N}$. As $(\lambda_k)_{k=1}^{\infty}$ is an interpolating sequence for $H^{\infty}(\Sigma_{\sigma})$ by Lemma 6.1.4, we can find a sequence of functions $(f_j)_{j=1}^{\infty}$ in $H^{\infty}(\Sigma_{\sigma})$ such that

$$f_j(\lambda_k) = \begin{cases} 1 + \lambda_k t_j^{-1} & 1 \le k \le j \\ -1 - t_j \lambda_k^{-1} & j < k \end{cases}$$

for all $j,k \in \mathbb{N}$ with $||f_j||_{H^{\infty}(\Sigma_{\sigma})} \leq C$. Now let $g_j(z) = z(1+z)^{-2}f_j(t_jz)$. Then $(g_j)_{n=1}^{\infty}$ is uniformly in $H^1(\Sigma_{\sigma})$ and therefore $\{g_j(t_j^{-1}A) : j \in \mathbb{N}\}$ is α -bounded by

Proposition 4.2.3. Again using (6.4), we have for all $n \in \mathbb{N}$

$$g_n(t_n^{-1}A) = \sum_{k=1}^n \frac{t_n^{-1}\lambda_k(1+t_n^{-1}\lambda_k)}{(1+t_n^{-1}\lambda_k)^2} P_k - \sum_{k=n+1}^\infty \frac{t_n^{-1}\lambda_k(1+t_n\lambda_k^{-1})}{(1+t_n^{-1}\lambda_k)^2} P_k$$
$$= \sum_{k=1}^n \frac{t_n^{-1}\lambda_k}{1+t_n^{-1}\lambda_k} P_k - \sum_{k=n+1}^\infty \frac{1}{1+t_n^{-1}\lambda_k} P_k$$
$$= \sum_{k=1}^n P_k - \sum_{k=1}^\infty \frac{t_n}{t_n+\lambda_k} P_k$$
$$= S_n + t_n R(-t_n, A).$$

Since A is α -sectorial, the set $\{t_k R(-t_k, A), k \in \mathbb{N}\}$ is α -bounded. Therefore the family of partial sum projections $\{S_k : k \in \mathbb{N}\}$ is α -bounded, i.e. $(X_k)_{k=1}^{\infty}$ is α -bounded.

6.2. Sectorial operators which are not almost α -sectorial

In this section we will start our series of examples based on the sectorial operators defined in Section 6.1. In this section we will construct a Schauder basis which is not almost α -bounded for any unconditionally stable Euclidean structure α , e.g. the γ -structure on a space with finite cotype or the ℓ^2 -structure on a Banach lattice. In view of Proposition 6.1.3 this yields sectorial operators which are not almost α -sectorial. Our proof will basically be a reconstruction of the idea of Lancien and the first author [**KL00**] to construct sectorial operators that are not \mathcal{R} -sectorial. This idea has been further developed in a sequence of papers by Fackler [**Fac13, Fac14a, Fac15, Fac16**] and was recently revisited by Arnold and Le Merdy [**AL19**].

The spaces ℓ^1 and c_0 . As a warm up we consider the sequence spaces ℓ^1 and c_0 .

PROPOSITION 6.2.1. Both ℓ^1 and c_0 have a Schauder basis \mathbf{x} which is not almost α -bounded for any unconditionally stable Euclidean structure α on X.

PROOF. For c_0 we consider the so-called summing basis \mathbf{x} , given by $x_n = \sum_{k=1}^{n} e_k$, where $(e_k)_{k=1}^{\infty}$ is the canonical basis of c_0 . The biorthogonal sequence \mathbf{x}^* in ℓ^1 is given by $x_n^* = e_n^* - e_{n+1}^*$ for $n \in \mathbb{N}$, where $(e_k^*)_{k=1}^{\infty}$ is the canonical basis of ℓ^1 . Let α be an unconditionally stable Euclidean structure on c_0 and suppose that \mathbf{x} is almost α -bounded. Let $(P_k^{\mathbf{x}})_{k=1}^{\infty}$ be the coordinate projections associated to \mathbf{x} . Then we have for any $n \in \mathbb{N}$

$$\left\| (x_k)_{k=1}^n \right\|_{\alpha} = \left\| (P_k^{\mathbf{x}} e_k)_{k=1}^n \right\|_{\alpha} \lesssim \left\| (e_k)_{k=1}^n \right\|_{\alpha} \lesssim \sup_{|\epsilon_k|=1} \left\| \sum_{k=1}^n \epsilon_k e_k \right\|_{\ell^{\infty}} = 1.$$

Since $e_1^*(x_k) = 1$ for all $k \in \mathbb{N}$, we also have by Proposition 1.1.5

$$n^{1/2} = n^{-1/2} \sum_{k=1}^{n} |e_1^*(x_k)| \le n^{-1/2} \|(x_k)_{k=1}^n\|_{\alpha} \|(e_1^*)_{k=1}^n\|_{\alpha^*} = \|(x_k)_{k=1}^n\|_{\alpha},$$

a contradiction. So **x** is not almost α -bounded.

The argument for ℓ^1 is dual. We consider the basis $\mathbf{y}^* := (e_1^*, x_1^*, x_2^*, \ldots)$ with biorthogonal sequence $(e, x_1 - e, x_2 - e, \ldots)$, where $e \in \ell^\infty$ is the sequence $(1, 1, \ldots)$.

Now let β be an unconditionally stable Euclidean structure on ℓ^1 and suppose that \mathbf{y}^* is almost β -bounded. Then we have for any $n \in \mathbb{N}$

$$n = \sum_{k=1}^{n} x_{k}^{*}(e_{k}) \le \left\| (x_{k}^{*})_{k=1}^{n} \right\|_{\beta} \left\| (e_{k})_{k=1}^{n} \right\|_{\beta^{*}} \lesssim \left\| (x_{k}^{*})_{k=1}^{n} \right\|_{\beta}$$

and, denoting the coordinate projections associated to \mathbf{y}^* by $(P_k^{\mathbf{y}^*})_{k=1}^{\infty}$, we have

$$\left\| (x_k^*)_{k=1}^n \right\|_{\beta} = \left\| (P_{k+1}^{\mathbf{y}^*} e_n^*)_{k=1}^n \right\|_{\beta} \lesssim \left\| (e_n^*)_{k=1}^n \right\|_{\beta} \le n^{1/2},$$

a contradiction. So \mathbf{y}^* is not almost β -bounded.

The general case. The general case will follow from the following lemma, which is a consequence of a result by Lindenstrauss and Zippin [LZ69].

LEMMA 6.2.2. Suppose that X has an unconditional Schauder basis and is not isomorphic to ℓ^1 , ℓ^2 or c_0 . Then there is an unconditional Schauder basis **x** of X, a permutation $\pi: \mathbb{N} \to \mathbb{N}$ and a sequence of scalars $(a_k)_{k=1}^{\infty}$ such that $\sum_{k=1}^{\infty} a_k x_{2k-1}$ converges but $\sum_{k=1}^{\infty} a_k x_{\pi(2k)}$ does not converge.

PROOF. By [LZ69, Note (1) at the end] we know that X has an unconditional, non-symmetric basis x, i.e. there is a permutation $\pi: \mathbb{N} \to \mathbb{N}$ such that x and $(x_{\pi(k)})_{k=1}^{\infty}$ are not equivalent. This implies the claim by the first part of the proof of [Sin70, Chapter 2, Proposition 23.2]. \square

We are now ready to prove the main result of this section.

THEOREM 6.2.3. Suppose that X has an unconditional Schauder basis and is not isomorphic to ℓ^2 . Then X has a Schauder basis which is not almost α -bounded for any unconditionally stable Euclidean structure α .

PROOF. If X is isomorphic to ℓ^1 or c_0 , the theorem follows from Proposition 6.2.1. Otherwise we can use Lemma 6.2.2 to find an unconditional Schauder basis **x** of X, a permutation $\pi: \mathbb{N} \to \mathbb{N}$ and a sequence of scalars $(a_k)_{k=1}^{\infty}$ such that $\sum_{k=1}^{\infty} a_k x_{2k-1}$ converges but $\sum_{k=1}^{\infty} a_k x_{\pi(2k)}$ does not converge.

Define for $k \in \mathbb{N}$

$$y_k = \begin{cases} x_k + x_{\pi(k+1)} & \text{if } k \text{ is odd,} \\ x_{\pi(k)} & \text{if } k \text{ is even} \end{cases}$$

Then y is an unconditional Schauder basis of X and its biorthogonal sequence y^* is an unconditional Schauder basis for $\overline{\operatorname{span}}\{y_k^*: k \in \mathbb{N}\}$. Let α be an unconditionally stable Euclidean structure and assume that **y** is almost α -bounded. Fix $m, n \in \mathbb{N}$ and let $(b_k)_{k=m+1}^n$ be such that $\sum_{k=m+1}^n b_k y_{2k}^*$ has norm 1 and

$$\left\|\sum_{k=m+1}^{n} a_k x_{\pi(2k)}\right\|_X = \sum_{k=m+1}^{n} \left|b_k y_{2k}^*(a_k y_{2k})\right|.$$

Let $(P_k^{\mathbf{y}})_{k=1}^{\infty}$ be the coordinate projections associated to \mathbf{y} . Then since $x_{2k-1} =$ $y_{2k-1} - y_{2k}$, we obtain by Proposition 6.1.3 and the unconditionality of y, y^{*} and

$$\alpha$$

$$\begin{split} \left\|\sum_{k=m+1}^{n} a_{k} x_{\pi(2k)}\right\|_{X} &= \sum_{k=m+1}^{n} \left|b_{k} y_{2k}^{*}(a_{k} y_{2k})\right| \\ &\leq \left\|\left(-a_{k} P_{2k}^{\mathbf{y}} x_{2k-1}\right)_{k=m+1}^{n}\right\|_{\alpha} \left\|\left(b_{k} y_{2k}^{*}\right)_{k=m+1}^{n}\right\|_{\alpha^{*}} \\ &\lesssim \sup_{|\epsilon_{k}|=1} \left\|\sum_{k=m+1}^{n} \epsilon_{k} a_{k} x_{2k-1}\right\|_{X} \sup_{|\epsilon_{k}|=1} \left\|\sum_{k=m+1}^{n} \epsilon_{k} b_{k} y_{2k}^{*}\right\|_{X^{*}} \\ &\lesssim \left\|\sum_{k=m+1}^{n} a_{k} x_{2k-1}\right\|_{X}. \end{split}$$

But since $\sum_{k=1}^{\infty} a_k x_{2k-1}$ converges this implies that $\sum_{k=1}^{\infty} a_k x_{\pi(2k)}$ converges, a contradiction. So **y** is not almost α -bounded.

Theorem 6.2.3 combined with Proposition 6.1.3 yields the result we were after in this section:

COROLLARY 6.2.4. Suppose that X has an unconditional Schauder basis and is not isomorphic to ℓ^2 . Then X has a Schauder basis **x** such that for any Hadamard sequence $(\lambda_k)_{k=1}^{\infty}$ the operator associated to **x** and $(\lambda_k)_{k=1}^{\infty}$ is not almost α -sectorial for any unconditionally stable Euclidean structure α .

In particular this implies the following corollary, since the Haar basis is unconditional in $L^p(\mathbb{R})$ for $p \in (1, \infty)$.

COROLLARY 6.2.5. Let $p \in (1, \infty) \setminus \{2\}$. Then there is a sectorial operator on $L^p(\mathbb{R})$ which is not almost γ -sectorial.

6.3. Almost α -sectorial operators which are not α -sectorial

Building upon the results of the previous section, we will now construct a Schauder basis that is almost α -bounded, but not α -bounded for any unconditionally stable Euclidean structures α . In view of Proposition 6.1.3 this yields examples of sectorial operators that are almost α -sectorial, but not α -sectorial.

We start with a useful criterion for the almost $\alpha\text{-boundedness}$ of a Schauder basis.

LEMMA 6.3.1. Let \mathbf{x} be a Schauder basis of X with biorthogonal sequence \mathbf{x}^* , let α be a Euclidean structure on X and take $p \in [1, \infty]$. If there is a C > 0 such that for all sequences of scalars $(a_k)_{k=1}^n$ and $(b_k)_{k=1}^n$ we have

$$\|(a_1x_1, \dots, a_nx_n)\|_{\alpha} \le C \|(a_k)_{k=1}^n\|_{\ell_n^p}, \\ \|(b_1x_1^*, \dots, b_nx_n^*)\|_{\alpha^*} \le C \|(b_k)_{k=1}^n\|_{\ell_n^p'},$$

then \mathbf{x} is almost α -bounded.

PROOF. Fix $n \in \mathbb{N}$ and take $\mathbf{y} \in X^n$. Define $a_k = x_k^*(y_k)$ for $k = 1, \ldots, n$. Let $(b_k)_{k=1}^n$ be such that $\|(b_k)_{k=1}^n\|_{\ell_n^{p'}} = 1$ and $\sum_{k=1}^n a_k b_k = \|(a_k)_{k=1}^n\|_{\ell_n^p}$. Let $(P_k^{\mathbf{x}})_{k=1}^{\infty}$ be the coordinate projections associated to \mathbf{x} . Then

$$\left\| (P_k^{\mathbf{x}} y_k)_{k=1}^n \right\|_{\alpha} = \left\| (a_k x_k)_{k=1}^n \right\|_{\alpha} \le C \left\| (a_k)_{k=1}^n \right\|_{\ell_n^p} = C \sum_{k=1}^n b_k x_k^*(y_k) \le C^2 \|\mathbf{y}\|_{\alpha}.$$

In the same way we obtain for $m_1, \ldots, m_n \in \mathbb{N}$ distinct that

$$\left\| (P_{m_k}^{\mathbf{x}} y_k)_{k=1}^n \right\|_{\alpha} \le C^2 \, \|\mathbf{y}\|_{\alpha}.$$

To allow repetitions we consider index sets $I_j = \{k \in \mathbb{N} : m_k = j\}$ and let $N \in \mathbb{N}$ be such that $I_j = \emptyset$ for j > N. By the right ideal property of a Euclidean structure and choosing appropriate c_{jk} 's with $\sum_{k \in I_j} |c_{jk}|^2 = 1$ for $j = 1, \ldots, N$ we have

$$\begin{split} \left\| (P_{m_{k}}^{\mathbf{x}} y_{k})_{k=1}^{n} \right\|_{\alpha} &= \left\| \left(\left(\sum_{k \in I_{j}} |x_{j}^{*}(y_{k})|^{2} \right)^{1/2} x_{j} \right)_{j=1}^{N} \right\|_{\alpha} \\ &= \left\| \left(\sum_{k \in I_{j}} c_{jk} x_{j}^{*}(y_{k}) x_{j} \right)_{j=1}^{N} \right\|_{\alpha} \\ &= \left\| \left(P_{j}^{\mathbf{x}} \left(\sum_{k \in I_{j}} c_{jk} y_{k} \right) \right)_{j=1}^{N} \right\|_{\alpha} \\ &\leq C^{2} \left\| \left(\sum_{k \in I_{j}} c_{jk} y_{k} \right)_{j=1}^{N} \right\|_{\alpha} = C^{2} \left\| \mathbf{y} \right\|_{\alpha}. \end{split}$$

Therefore $\{P_k^{\mathbf{x}} : k \in \mathbb{N}\}$ is α -bounded, i.e. \mathbf{x} is almost α -bounded.

With this lemma at our disposal we can now turn to the main result of this

section. We take the example in Theorem 6.2.3 as a starting point to construct an example of a Schauder basis that is almost α -sectorial, but not α -sectorial.

THEOREM 6.3.2. Let α be an ideal unconditionally stable Euclidean structure on X. Suppose that

- X has a Schauder basis \mathbf{x} which is not α -bounded.
- X has a complemented subspace isomorphic to l^p for some p ∈ [1,∞) or isomorphic to c₀.

Then X has a Schauder basis y which is almost α -bounded, but not α -bounded.

PROOF. We will consider the ℓ^p -case, the calculations for c_0 are similar and left to the reader. By assumption there is a $p \in [1, \infty)$ and a subspace W of X for which we have the following chain of isomorphisms

$$X = W \oplus \ell^p = W \oplus \ell^p \oplus \ell^p = X \oplus \ell^p.$$

Thus we can write $X = Y \oplus Z$ where Y is isomorphic to X and Z is isomorphic to ℓ^p . We denote the projection from X onto Y by P_Y and let $V: Y \to X$ be an isomorphism.

Let $(e_k)_{k=1}^{\infty}$ be a Schauder basis of Z equivalent to the canonical basis of ℓ^p . We consider the Schauder basis **u** of X given by

$$u_{k} = \begin{cases} e_{1} & \text{if } k = 1\\ e_{k-j+1} & \text{if } 2^{j} + 1 \le k \le 2^{j+1} - 1 \text{ for } j \in \mathbb{N}\\ \frac{1}{\|V^{-1}x_{j}\|_{X}} V^{-1}x_{j} & \text{if } k = 2^{j} \text{ for } j \in \mathbb{N}. \end{cases}$$

138
For $j \in \mathbb{N}$ we define

$$v_j = \left((2^j - 1)^{-1/p} \sum_{k=2^j+1}^{2^{j+1}-1} u_k \right) - u_{2^j}$$
$$v_j^* = \left((2^j - 1)^{-1/p'} \sum_{k=2^j+1}^{2^{j+1}-1} u_k^* \right) + u_{2^j}^*.$$

If we now define the operators $T_j x = v_j^*(x)v_j$ for $x \in X$ and $j \in \mathbb{N}$, then

- $||T_j|| \le 4$, since $||v_j||_X$, $||v_j^*||_{X^*} \le 2$. $T_j^2 = 0$, since $v_j^*(v_j) = 0$. T_j leaves the subspace span $\{u_k : 2^j \le k \le 2^{j+1} 1\}$ invariant.

Therefore $I + T_j$ is an automorphism of span $\{u_k : 2^{j-1} + 1 \le k \le 2^j\}$, so we can make a new basis \mathbf{y} of X, given by

$$y_k = \begin{cases} u_1 & \text{if } k = 1, \\ \frac{1}{\|(I+T_j)u_k\|_X} (I+T_j)u_k & \text{if } 2^j \le k \le 2^{j+1} - 1 \text{ for } j \in \mathbb{N}. \end{cases}$$

Let $(S_k^{\mathbf{x}})_{k=1}^{\infty}$, $(S_k^{\mathbf{y}})_{k=1}^{\infty}$ and $(S_k^{\mathbf{u}})_{k=1}^{\infty}$ be the partial sum projections associated to \mathbf{x} , \mathbf{y} and \mathbf{u} respectively. Then $S_{2^{k+1}-1}^{\mathbf{u}} = S_{2^{k+1}-1}^{\mathbf{y}}$ and thus

$$S_k^{\mathbf{x}} = V P_Y S_{2^{k+1}-1}^{\mathbf{u}} V^{-1} = V P_Y S_{2^{k+1}-1}^{\mathbf{y}} V^{-1}$$

for all $k \in \mathbb{N}$. Since α is ideal and $(S_k^{\mathbf{x}})_{k=1}^{\infty}$ is not α -bounded, we have by Proposition 1.2.2 that $(S_k^{\mathbf{y}})_{k=1}^{\infty}$ is not α -bounded. So \mathbf{y} is not α -bounded.

Next we show that **y** is almost α -bounded. We will prove that there is a C > 0such that for all scalar sequences $(a_k)_{k=1}^n$ and $(b_k)_{k=1}^n$ we have

(6.6)
$$\left\| (a_1 y_1, \dots, a_n y_n) \right\|_{\alpha} \le C \left(\sum_{k=1}^n |a_k|^p \right)^{1/p},$$

(6.7)
$$\left\| (b_1 y_1^*, \dots, b_n y_n^*) \right\|_{\alpha^*} \le C \left(\sum_{k=1}^n |b_k|^{p'} \right)^{1/p'}$$

for all $n \in \mathbb{N}$. By Lemma 6.3.1 this implies that **y** is almost α -bounded. The calculations for (6.6) and (6.7) are similar, so we will only treat (6.6). Fix $m \in \mathbb{N}$, let $n = 2^{m+1} - 1$ and define $c_j = 2^j - 1$ for $j \in \mathbb{N}$. First suppose that $a_{2^j} = 0$ for $1 \leq j \leq m$. Then, using the triangle inequality, the unconditonal stability of α and the fact that $(e_k)_{k=1}^{\infty}$ is equivalent to the canonical basis of ℓ^p , we have

$$\begin{split} \left\| (a_k y_k)_{k=1}^n \right\|_{\alpha} &\leq \left\| (a_k u_k)_{k=1}^n \right\|_{\alpha} + \left\| \left(c_j^{-1/p'} \left(\sum_{k=2^j+1}^{2^{j+1}-1} |a_k|^2 \right)^{1/2} v_j \right)_{j=1}^m \right\|_{\alpha} \\ &\leq C \left(\sum_{k=1}^n |a_k|^p \right)^{1/p} + \sum_{j=1}^m c_j^{-1/p'} \left(\sum_{k=2^j+1}^{2^{j+1}-1} |a_k|^2 \right)^{1/2} \|v_j\|_X, \end{split}$$

If $p \in [1, 2]$, we estimate the second term by Hölder's inequality

$$\sum_{j=1}^{m} c_j^{-1/p'} \left(\sum_{k=2^j+1}^{2^{j+1}-1} |a_k|^2 \right)^{1/2} \|v_j\|_X \le 2 \sum_{j=1}^{m} c_j^{-1/p'} \left(\sum_{k=2^j+1}^{2^{j+1}-1} |a_k|^p \right)^{1/p} \le 2 \left(\sum_{j=1}^{m} c_j^{-1} \right)^{1/p'} \left(\sum_{k=1}^{n} |a_k|^p \right)^{1/p}$$

and, if $p \in (2, \infty)$, we estimate the second term by applying Hölder's inequality twice

$$\sum_{j=1}^{m} c_j^{-1/p'} \left(\sum_{k=2^j+1}^{2^{j+1}-1} |a_k|^2 \right)^{1/2} \|v_j\|_X \le 2 \sum_{j=1}^{m} c_j^{-1/p} \left(\sum_{k=2^j+1}^{2^{j+1}-1} |a_k|^p \right)^{1/p} \le 2 \left(\sum_{j=1}^{m} c_j^{-p'/p} \right)^{1/p'} \left(\sum_{k=1}^{n} |a_k|^p \right)^{1/p}$$

Combined this yields (6.6) if $a_{2^j} = 0$ for $1 \le j \le m$. Now assume that $a_k = 0$ unless $k = 2^j$ for $1 \le j \le m$. Since we have

$$y_{2^{j}} = \frac{u_{2^{j}} + v_{j}}{\|u_{2^{j}} + v_{j}\|_{X}} = \frac{c_{j}^{-1/p}}{\|u_{2^{j}} + v_{j}\|_{X}} \sum_{k=2^{j}+1}^{2^{j+1}-1} u_{k},$$

we immediately obtain

$$\left\| (a_k y_k)_{k=1}^n \right\|_{\alpha} = C \left(\sum_{k=1}^n |a_k|^p \right)^{1/p}$$

again using that α is unconditionally stable and $(e_k)_{k=1}^{\infty}$ is equivalent to the canonical basis of ℓ^p . The estimate for general $(a_k)_{k=1}^n$ now follows by the triangle inequality.

Theorem 6.3.2 combined with Theorem 6.2.3 and Proposition 6.1.3 yields examples of sectorial operators that are almost α -sectorial, but not α -sectorial:

COROLLARY 6.3.3. Let α be an unconditionally stable Euclidean structure on X. Suppose that

- X has an unconditional Schauder basis.
- X is not isomorphic to ℓ^2 .
- X has a complemented subspace isomorphic to l^p for some p ∈ [1,∞) or isomorphic to c₀.

Then X has a Schauder basis **x** such that for any Hadamard sequence $(\lambda_k)_{k=1}^{\infty}$ the operator associated to **x** and $(\lambda_k)_{k=1}^{\infty}$ is almost α -sectorial, but not α -sectorial.

Specifically for the γ -structure we have:

COROLLARY 6.3.4. Let $p \in [1, \infty) \setminus \{2\}$. There is a sectorial operator on $L^p(\mathbb{R})$ which is almost γ -sectorial, but not γ -sectorial.

PROOF. If $p \in (1, \infty) \setminus \{2\}$ this follows from Corollary 6.3.3, since the Haar basis is unconditional. Any Schauder basis of $L^1(\mathbb{R})$ is not \mathcal{R} -bounded and thus not γ -bounded by [**HKK04**, Theorem 3.4], so for p = 1 we can directly apply Theorem 6.3.2.

6.4. Sectorial operators with $\omega_{H^{\infty}}(A) > \omega(A)$

Let A be a sectorial operator on X and α a Euclidean structure on X. We have seen in Proposition 4.2.1, Proposition 4.5.1, Theorem 4.5.6 and Corollary 5.1.9 that under reasonable assumptions on α the angles of (almost) α -sectoriality, (α -)BIP and of the (α -)-bounded H^{∞} -calculus are equal whenever A has these properties. Strikingly absent in this list is the angle of sectoriality.

In general the angle of sectoriality is not equal to the other introduced angles in Chapter 4. As we already noted in Section 4.5, Haase showed in [Haa03, Corollary 5.3] that there exists a sectorial operator A with $\omega_{\text{BIP}}(A) > \omega(A)$. The first counterexample to the equality $\omega_{H^{\infty}}(A) = \omega(A)$ was given by Cowling, Doust, McIntosh and Yagi [CDMY96, Example 5.5], who constructed an operator (without dense range) with a bounded H^{∞} -calculus, such that $\omega(A) < \omega_{H^{\infty}}(A)$. Subsequently, the first author constructed a sectorial operator with $\omega(A) < \omega_{H^{\infty}}(A)$ in [Kal03]. Both these examples are on very specific (non-reflexive) Banach spaces and it is an open problem whether every infinite-dimensional Banach space admits such an example. In particular in [HNVW17, Problem P.13] it was asked whether there exists examples on L^p . In this section we will provide an example on a subspace of L^p for any $p \in (1, \infty)$.

Let us note that all known examples that make their appearance in applications actually satisfy $\omega_{H^{\infty}}(A) = \omega(A)$. This holds in particular for classical operators like the Laplacian on $L^{p}(\mathbb{R}^{d})$, but also for far more general elliptic operators as shown [Aus07], which is based on earlier results in [BK03, DM99, DR96]. More recent developments in this direction can for example be found in [CD20b, CD20a, Ege18, Ege20, EHRT19]. Also for example for the Ornstein-Uhlenbeck operator we have $\omega_{H^{\infty}}(A) = \omega(A)$ (see e.g. [Car09, CD19, GCMM⁺01, Har19]). Even in more abstract situations, like the Hörmander-type holomorphic functional calculus for symmetric contraction semigroups on L^{p} , the angle of the functional calculus, is equal to the angle of sectoriality (see [CD17]). This means that our example will have to be quite pathological.

The general idea. We will proceed as follows: We will construct a Banach space X and a Schauder multiplier operator such that $\omega(A) = 0$ and such that, on the generalized square function spaces introduced in Section 5.3, the induced operator $A^s|_{H^{\gamma}_{\theta,A^s}}$ does not have a bounded H^{∞} -calculus for s > 1. Then $\omega_{\text{BIP}}(A^s|_{H^{\gamma}_{\theta,A^s}}) = \pi$ by Theorem 5.4.4 and (5.13), so using Theorem 5.4.3 and Proposition 5.4.1 we know that $A|_{H^{\gamma}_{\theta,A}(\varphi_s)}$ with $\varphi_s(z) = \frac{z^{s/2}}{1+z^s}$ has a bounded H^{∞} calculus. Therefore

$$\omega_{H^{\infty}}\left(A|_{H^{\gamma}_{\theta,A}(\varphi_s)}\right) = \omega_{\mathrm{BIP}}\left(A|_{H^{\gamma}_{\theta,A}(\varphi_s)}\right) = \frac{1}{s}\,\omega_{\mathrm{BIP}}\left(A^s|_{H^{\gamma}_{\theta,A^s}}\right) = \frac{\pi}{s}$$

and by Proposition 5.3.5

$$\omega(A|_{H^{\gamma}_{\theta,A}(\varphi_s)}) = \omega(A) = 0.$$

Therefore $A|_{H^{\gamma}_{\theta,A}(\varphi_s)}$ on $H^{\gamma}_{\theta,A}(\varphi_s)$, which will be a closed subspace of L^p , is an example of an operator that we are looking for.

The remainder of this section will be devoted to the construction of this A. As a first guess, we could try the operators we used in Section 6.2 and Section 6.3 for our examples. The following theorem shows that this will not work.

THEOREM 6.4.1. Let **x** be a Schauder basis for X, $(\lambda_k)_{k=1}^{\infty}$ a Hadamard sequence and A the sectorial operator associated to **x** and $(\lambda_k)_{k=1}^{\infty}$. Let α be an ideal Euclidean structure on X and fix $\theta \in \mathbb{R}$. Then $A|_{H^{\alpha}_{\theta,A}}$ has a bounded H^{∞} -calculus on $H^{\alpha}_{\theta,A}$ with $\omega_{H^{\infty}}(A|_{H^{\alpha}_{\theta,A}}) = \omega(A)$

PROOF. For simplicity we assume $(\lambda_k)_{k=1}^{\infty} \subseteq \mathbb{R}_+$, i.e. $\omega(A) = 0$, and leave the case $(\lambda_k)_{k=1}^{\infty} \subseteq \Sigma_{\sigma}$ for $0 < \sigma < \pi$ to the interested reader. Let c > 1 be the constant in the definition of a Hadamard sequence and take $u > \max\{1, 1/\log(c)\}$. Define $\mu_k := u \log(\lambda_k)$, then we have

$$\inf_{j \neq k} |\mu_j - \mu_k| = \inf_{j \neq k} u \left| \log(\lambda_j / \lambda_k) \right| > 1,$$

so $(\mu_k)_{k=1}^{\infty}$ is uniformly discrete. Moreover, denoting by $n^+(r)$ the largest number of points of $(\mu_k)_{k=1}^{\infty}$ in any interval $I \subset \mathbb{R}$ of length r > 0, we have for the upper Bearling density of $(\mu_k)_{k=1}^{\infty}$

$$D^+((\mu_k)_{k=1}^{\infty}) := \lim_{r \to \infty} \frac{n^+(r)}{r} < 1.$$

Therefore, by [Sei95, Theorem 2.2], we know that $(e^{i\mu_k t})_{k=1}^{\infty}$ is a *Riesz sequence* in $L^2(-\pi,\pi)$, i.e. there exists a C > 0 such that for any sequence $\boldsymbol{a} \in \ell_n^2$ we have

$$C^{-1} \|\boldsymbol{a}\|_{\ell_n^2} \le \left\| t \mapsto \sum_{k=1}^n a_k \mathrm{e}^{i\mu_k t} \right\|_{L^2(-\pi,\pi)} \le C \|\boldsymbol{a}\|_{\ell_n^2},$$

and thus we have

$$\|\boldsymbol{a}\|_{\ell_n^2} \le C \left\| \sum_{k=1}^n a_k \mathrm{e}^{i\mu_k \cdot} \right\|_{L^2(-\pi,\pi)} \le \frac{C}{\sqrt{u}} \, \mathrm{e}^{\frac{\pi^2}{s}u} \left\| \mathrm{e}^{-\frac{\pi}{s}|\cdot|} \sum_{k=1}^n a_k \lambda_k^{i\cdot} \right\|_{L^2(-\pi u,\pi u)}.$$

Conversely we have that

$$\begin{split} \sup_{\|\boldsymbol{a}\|_{\ell_{n}^{2}} \leq 1} \left\| e^{-\frac{\pi}{s} |\cdot|} \sum_{k=1}^{n} a_{k} \lambda_{k}^{i \cdot} \right\|_{L^{2}(\mathbb{R})} \leq \sup_{\|\boldsymbol{a}\|_{\ell_{n}^{2}} \leq 1} \sqrt{u} \cdot \left\| \sum_{k=1}^{n} a_{k} e^{i\mu_{k} \cdot} \right\|_{L^{2}(-\pi,\pi)} \\ &+ \left\| e^{-\frac{\pi}{s} |\cdot|} \sum_{k=1}^{n} a_{k} \lambda_{k}^{i \cdot} \right\|_{L^{2}(\mathbb{R} \setminus (-\pi u, \pi u))} \\ \leq C \sqrt{u} + 2 \sup_{\|\boldsymbol{a}\|_{\ell_{n}^{2}} \leq 1} e^{-\frac{\pi^{2}}{s} u} \left\| e^{-\frac{\pi}{s} |\cdot|} \sum_{k=1}^{n} a_{k} \lambda_{k}^{i \cdot} \right\|_{L^{2}(\mathbb{R})}, \end{split}$$

using the change of variables $t' = t \pm \pi u$ in the second step. So for any $0 < s < \pi u$ we have

$$\left\| t \mapsto \mathrm{e}^{-\frac{\pi}{s}|t|} \sum_{k=1}^{n} a_k \lambda_k^{it} \right\|_{L^2(\mathbb{R})} \simeq \sqrt{u} \cdot \|\boldsymbol{a}\|_{\ell_n^2}.$$

Therefore $T_s \colon \ell^2 \to L^2(\mathbb{R})$ given by

$$(T_s \boldsymbol{a})(t) := \sum_{k=1}^{\infty} a_k \mathrm{e}^{-\pi |t|/u} \lambda_k^{it}, \qquad t \in \mathbb{R}$$

is an isomorphism onto the closed subspace of $L^2(\mathbb{R})$ generated by the functions $(t \mapsto e^{-\frac{\pi}{s}|t|}\lambda_k^{it})_{k=1}^{\infty}$. Its adjoint $T_s^* \colon L^2(\mathbb{R}) \to \ell^2$ is given by

$$(T_s \varphi)_k = \int_{\mathbb{R}} \varphi(t) \mathrm{e}^{-\pi |t|/u} \lambda_k^{it} \, \mathrm{d}t, \qquad k \in \mathbb{N}.$$

Now fix $\boldsymbol{a} \in \ell_n^2$ and define $x = \sum_{k=1}^n a_k x_k$. Then for $\varphi_s(z) = z^{s/2} (1+z^s)^{-1}$ we have by Proposition 5.4.1 and (6.5)

$$\|x\|_{H^{\alpha}_{\theta,A}(\varphi_s)} = \|t \mapsto \sum_{k=1}^{n} a_k \mathrm{e}^{-\frac{\pi}{s}|t|} \lambda_k^{it+\theta} x_k \|_{\alpha(\mathbb{R};X)}$$

Now if $S: L^2(\mathbb{R}) \to X$ is the operator represented by $t \mapsto \sum_{k=1}^n a_k e^{-\frac{\pi}{s}|t|} \lambda_k^{it+\theta} x_k$, then we have $S = S' \circ T_s^*$, where $S': \ell^2 \to X$ is the finite rank operator given by $S' = \sum_{k=1}^n e_k \otimes a_k \lambda_k^{\theta} x_k$ and $(e_k)_{k=1}^{\infty}$ is the canonical basis of ℓ^2 . Since T_s is an isomorphism, we see

$$\|x\|_{H^{\alpha}_{\theta,A}(\varphi_s)} = \|S\|_{\alpha(\mathbb{R};X)} \simeq \sqrt{u} \cdot \|S'\|_{\alpha} = \sqrt{u} \cdot \|(a_1\lambda_1^{\theta}x_1, \dots, a_n\lambda_n^{\theta}x_n)\|_{\alpha}$$

Thus by density we deduce that the spaces $H^{\alpha}_{\theta,A}(\varphi_s)$ are isomorphic for all $0 < s < \pi u$ and since u could be taken arbitrarily large they are isomorphic for all s > 0. In particular $H^{\alpha}_{\theta,A}$ is isomorphic to $H^{\alpha}_{\theta,A}(\varphi_s)$ for any s > 1 and thus by Theorem 5.4.3 we deduce that $A|_{H^{\alpha}_{\theta,A}}$ has a bounded H^{∞} -calculus on $H^{\alpha}_{\theta,A}$ with $\omega_{H^{\infty}}(A|_{H^{\alpha}_{\theta,A}}) = 0$

The construction of the Banach space X. Since Hadamard sequences will not work for the example we are looking for, we will construct an operator based on a Schauder basis and an increasing sequence in \mathbb{R}_+ , which is also sectorial by Proposition 6.1.2. Let us first define the Banach space X that we will work with. Fix $1 < q < p < \infty$ and denote the space of all sequences which are eventually zero by c_{00} . Let $(x_k^*)_{k=1}^{\infty}$ be a sequence in $c_{00} \cap \{x^* \in \ell^{p'} : \|x^*\|_{\ell^{p'}} \leq 1\}$ such that $\{x_k^* : k \in \mathbb{N}\}$ is dense in $\{x^* \in \ell^{p'} : \|x^*\|_{\ell^{p'}} \leq 1\}$ and each element of $\{x_k^* : k \in \mathbb{N}\}$ is repeated infinitely often. For each $k \in \mathbb{N}$ let $F_k \subseteq \mathbb{N}$ be the support of x_k^* and write $F_k = \{s_{k,1}, \ldots, s_{k,|F_k|}\}$, where $s_{k,1} < \cdots < s_{k,|F_k|}$. Define $N_0 = 0$ and $N_k = |F_1| + \cdots + |F_k|$ for $k \in \mathbb{N}$.

Let $(e_j)_{j=1}^{\infty}$ be the canonical basis of ℓ^p . For $k \in \mathbb{N}$ we define the bounded linear operator $U_k : \ell^p \to \ell^p$ by

$$U_k(e_j) := \begin{cases} e_{s_{k,(j-N_{k-1})}} & \text{if } N_{k-1} < j \le N_k \\ 0 & \text{otherwise} \end{cases}$$

and the partial inverse $V_k : \ell^p \to \ell^p$ by $V_k x = U_k^{-1}(x \mathbf{1}_{F_k})$. Now we define $X = X_{p,q}$ as the completion of c_{00} under the norm

$$||x||_{X_{p,q}} := ||x||_{\ell^p} + \left\| \left(\langle U_k x, x_k^* \rangle \right)_{k=1}^{\infty} \right\|_{\ell^q}.$$

Then X is isomorphic to a closed subspace of $\ell^p \oplus \ell^q$, which can be seen using the embedding $X \hookrightarrow \ell^p \oplus \ell^q$ given by

$$x \mapsto x \oplus \left(\langle U_k x, x_k^* \rangle \right)_{k=1}^{\infty}$$

We consider X, and therefore all parameters introduced above, to be fixed for the remainder of this section.

LEMMA 6.4.2. The canonical basis of ℓ^p is a Schauder basis of X.

PROOF. It suffices to show that the partial sum projections $(S_j)_{j=1}^{\infty}$ associated to the canonical basis of ℓ^p are uniformly bounded. Fix $m, n \in \mathbb{N}$ such that $N_{n-1} <$ $m \leq N_n$ and take $x \in X$. Then since $\|S_m\|_{\mathcal{L}(\ell^p)} \leq 1$ and $\|x_n^*\|_{\ell^{p'}} \leq 1$ we have

$$\begin{split} \left\| \left(\langle U_k S_m x, x_k^* \rangle \right)_{k=1}^{\infty} \right\|_{\ell^q} &= \left\| \left(\langle U_k S_m x, x_k^* \rangle \right)_{k=1}^n \right\|_{\ell_n^q} \\ &\leq \left(\sum_{k=1}^{n-1} |\langle U_k x, x_k^* \rangle|^q \right)^{1/q} + \left| \langle U_n S_m x, x_n^* \rangle \right| \\ &\leq \left\| \left(\langle U_k x, x_k^* \rangle \right)_{k=1}^{\infty} \right\|_{\ell^q} + \|x\|_{\ell^p}. \end{split}$$

Therefore $||S_m||_{\mathcal{L}(X)} \leq 2$ for all $m \in \mathbb{N}$.

The construction of the sectorial operator A. Next we construct the sectorial operator on X, for which $A^s|_{H^{\alpha}_{\theta,A^s}}$ with s > 1 will not have a bounded H^{∞} -calculus. Define for $j \in \mathbb{N}$

(6.8)
$$\lambda_j = 2^k \left(2 - \frac{1}{s_{k,(j-N_{k-1})}} \right), \qquad N_{k-1} < j \le N_k.$$

Then $(\lambda_j)_{j=1}^{\infty}$ is an increasing sequence in \mathbb{R}_+ , so by Proposition 6.1.2 and Lemma 6.4.2 the operator associated to $(\lambda_j)_{j=1}^{\infty}$ and the canonical basis $(e_j)_{j=1}^{\infty}$ is sectorial with $\omega(A) = 0$. The following technical lemma will be key in our analysis of this operator.

LEMMA 6.4.3. Let A be the sectorial operator associated to $(e_j)_{j=1}^{\infty}$ and $(\lambda_j)_{j=1}^{\infty}$. Let $0 < \sigma < \pi$ and suppose that $f, g \in H^1(\Sigma_{\sigma})$ such that

$$\left\| (g(2^{j}A)x)_{j\in\mathbb{Z}} \right\|_{\gamma(\mathbb{Z};X)} \lesssim \left\| (f(2^{j}A)x)_{j\in\mathbb{Z}} \right\|_{\gamma(\mathbb{Z};X)}, \qquad x \in c_{00}.$$

Then there is a sequence $\mathbf{a} \in \ell^2(\mathbb{Z})$ so that

$$g(z) = \sum_{j \in \mathbb{Z}} a_k f(2^j z), \qquad z \in \Sigma_{\sigma}$$

PROOF. Fix $x \in \ell^p$ with all entries non-zero and $x^* \in \{x_k^* : k \in \mathbb{N}\}$. Let $d_1 < d_2 < \cdots$ be such that $x^* = x_{d_k}^*$ for all $k \in \mathbb{N}$, which is possible since each element of $\{x_k^* : k \in \mathbb{N}\}$ is repeated infinitely often. Define $T : \ell^p \to \ell^p$ by $Te_j = (2 - \frac{1}{j})e_j$, then

$$A = \sum_{k=1}^{\infty} 2^k V_k T U_k.$$

Fix $n \in \mathbb{N}$ and define

$$y_n = V_{d_1}x + \dots + V_{d_n}x \in c_{00}.$$

Then for all $j \in \mathbb{N}$ we have for h = f, g

$$h(2^{j}A)y_{n} = \sum_{k=1}^{\infty} h(2^{j+k}V_{k}TU_{k})y_{n} = \sum_{k=1}^{n} h(2^{j+d_{k}}T)V_{d_{k}}x$$

Noting that the vectors $V_{d_k}x$ are disjointly supported shifts of $x \mathbf{1}_F$ for $F = F_{d_1} = F_{d_2} = \cdots$, we obtain

$$\left\| \left(\sum_{j \in \mathbb{Z}} |h(2^{j}A)y_{n}|^{2} \right)^{1/2} \right\|_{\ell^{p}} = \left\| \left(\sum_{k=1}^{n} \sum_{j \in \mathbb{Z}} |h(2^{j+d_{k}}T)V_{d_{k}}x|^{2} \right)^{1/2} \right\|_{\ell^{p}}$$
$$= n^{1/p} \left\| \left(\sum_{j \in \mathbb{Z}} |h(2^{j}T)(x \mathbf{1}_{F})|^{2} \right)^{1/2} \right\|_{\ell^{p}}$$

and similarly

$$\left\| \left(\sum_{j \in \mathbb{Z}} \left(|\langle U_k h(2^j A) y_n, x_k^* \rangle|^2 \right)^{1/2} \right)_{k=1}^{\infty} \right\|_{\ell^q} = n^{1/q} \left(\sum_{j \in \mathbb{Z}} |\langle h(2^j T) x, x^* \rangle|^2 \right)^{1/2} \right)_{k=1}^{\infty} \|_{\ell^q}$$

Now since X is a closed subspace of a Banach lattice with finite cotype, we have by Proposition 1.1.3 and our assumption on f and g that there is a C > 0 such that for all $n \in \mathbb{N}$

$$n^{1/p} \left\| \left(\sum_{j \in \mathbb{Z}} |g(2^{j}T)(x \mathbf{1}_{F})|^{2} \right)^{1/2} \right\|_{\ell^{p}} + n^{1/q} \left(\sum_{j \in \mathbb{Z}} |\langle g(2^{j}T)x, x^{*} \rangle|^{2} \right)^{1/2} \\ \leq C \left(n^{1/p} \left\| \left(\sum_{j \in \mathbb{Z}} |f(2^{j}T)(x \mathbf{1}_{F})|^{2} \right)^{1/2} \right\|_{\ell^{p}} + n^{1/q} \left(\sum_{k \in \mathbb{Z}} |\langle f(2^{j}T)x, x^{*} \rangle|^{2} \right)^{1/2} \right).$$

Since q < p we obtain by dividing by $n^{1/q}$ and taking the limit $n \to \infty$ that

$$\left(\sum_{k\in\mathbb{Z}} |\langle g(2^kT)x, x^*\rangle|^2\right)^{1/2} \le C\left(\sum_{k\in\mathbb{Z}} |\langle f(2^kT)x, x^*\rangle|^2\right)^{1/2}.$$

In particular we have

(6.9)
$$|\langle g(T)x, x^* \rangle| \le C \left(\sum_{k \in \mathbb{Z}} |\langle f(2^k T)x, x^* \rangle|^2\right)^{1/2}.$$

Since T is bounded and invertible, we know by Lemma 4.3.4 that

$$\sum_{|k|\ge n} \|f(2^k T)x\|_{\ell^p} \to 0, \qquad n \to \infty.$$

Therefore (6.9) extends to all $x^* \in \ell^{p'}$ of norm one by density. Define the closed (compact!) convex set

$$\Gamma := \left\{ \sum_{k \in \mathbb{Z}} a_k f(2^k T) x : \|\boldsymbol{a}\|_{\ell^2} \le C \right\}$$

and suppose that $g(T)x \notin \Gamma$. Using the Hahn–Banach separation theorem [**Rud91**, Theorem 3.4] on Γ and $\{g(T)x\}$, we can find an $x^* \in \ell^{p'}$ such that

$$C\left(\sum_{k\in\mathbb{Z}} |\langle f(2^kT)x, x^*\rangle|^2\right)^{1/2} = \sup_{\|\boldsymbol{a}\|_{\ell^2} \le C_1} \operatorname{Re}\left(\left\langle\sum_{k\in\mathbb{Z}} a_k f(2^kT)x, x^*\right\rangle\right)$$
$$< \operatorname{Re}\left(\langle g(T)x, x^*\rangle\right)$$
$$\le |\langle g(T)x, x^*\rangle|,$$

a contradiction with (6.9). Thus $g(T)x \in \Gamma$, so there is an $a \in \ell^2$ with $||a||_{\ell^2} \leq C_1$ such that

$$g(T)x = \sum_{k \in \mathbb{Z}} a_k f(2^k T)x.$$

Since every coordinate of x is non-zero, this implies that

$$g\left(2-\frac{1}{j}\right) = \sum_{k \in \mathbb{Z}} a_k f\left(2^k \left(1-\frac{1}{j}\right)\right)$$

for all $j \in \mathbb{N}$. As $\sum_{k \in \mathbb{Z}} a_k f(2^k z)$ converges uniformly to a holomorphic function on compact subsets of Σ_{σ} , by the uniqueness of analytic continuations this implies that

$$g(z) = \sum_{k \in \mathbb{Z}} a_k f(2^k z), \qquad z \in \Sigma_\sigma$$

which completes the proof.

Using Lemma 6.4.3 we can now prove the main theorem of this section, which concludes our study of Euclidean structures.

THEOREM 6.4.4. Let $p \in (1, \infty) \setminus \{2\}$ and $\sigma \in (0, \pi)$. There exists a closed subspace Y of $L^p([0, 1])$ and a sectorial operator A on Y such that A has a bounded H^{∞} -calculus, has BIP and is (almost) γ -sectorial with $\omega(A) = 0$ and

$$\omega_{H^{\infty}}(A) = \omega_{\mathrm{BIP}}(A) = \omega_{\gamma}(A) = \tilde{\omega}_{\gamma}(A) = \sigma.$$

PROOF. If $p \in (1,2)$ take $X = X_{2,p}$ and if $p \in (2,\infty)$ take $X = X_{p,2}$. Let A be the sectorial operator associated to $(e_j)_{j=1}^{\infty}$ and $(\lambda_j)_{j=1}^{\infty}$, with $(\lambda_j)_{j=1}^{\infty}$ as in (6.8) and $(e_j)_{j=1}^{\infty}$ the canonical basis of ℓ^p . Set $\nu = \pi/\sigma$ and define $B = A^{\nu}$, which is a sectorial operator with

$$\omega(B) = \nu \,\omega(A) = 0$$

Suppose that the operator $B|_{H_{0,B}^{\gamma}}$ as in Proposition 5.3.5 has a bounded H^{∞} calculus. Then by Theorem 5.4.4 there is an $1 < s' < \infty$ such that for 0 < s < s'the spaces $H_{0,B}^{\gamma}(\varphi_s)$ with $\varphi_s(z) = z^{s/2}(1+z^s)^{-1}$ are isomorphic. In particular by
(6.5) and a change of variables we have for $x \in c_{00}$

$$\|\varphi_{\nu s}(\cdot A)x\|_{\gamma\left(\mathbb{R}_{+},\frac{dt}{t};X\right)} = \sqrt{\nu}\|x\|_{H^{\gamma}_{0,B}(\varphi_{s})} \simeq \sqrt{\nu}\|x\|_{H^{\gamma}_{0,B}} = \|\varphi_{\nu}(\cdot A)x\|_{\gamma\left(\mathbb{R}_{+},\frac{dt}{t};X\right)}$$

and thus by Proposition 5.4.5 there is a 1 < s < s' such that

$$\left| (\varphi_{\nu s}(2^k A) x)_{k \in \mathbb{Z}} \right\|_{\gamma(\mathbb{Z};X)} \le C \left\| (\varphi_{\nu}(2^k A) x)_{k \in \mathbb{Z}} \right\|_{\gamma(\mathbb{Z};X)}$$

This implies by Lemma 6.4.3 that there is a $a \in \ell^2$ such that we have

(6.10)
$$\varphi_{\nu s}(z) = \sum_{k \in \mathbb{Z}} a_k \varphi_{\nu}(2^k z), \qquad z \in \Sigma_{\mu}$$

for any $0 < \mu < \pi/\nu s$. Thus (6.10) holds for all $z \in \Sigma_{\pi/\nu s}$. But $\varphi_{\nu} \in H^1(\Sigma_{\pi/\nu s})$ and $\varphi_{\nu s}$ has a pole of order 1 on the boundary of $\Sigma_{\pi/\nu s}$, a contradiction. So $B|_{H^{\gamma}_{0,B}}$ does not have a bounded H^{∞} -calculus. By Theorem 5.4.4 and (5.13) this implies $\omega_{\text{BIP}}(B|_{H^{\gamma}_{0,B}}) = \pi$.

Now we have by Proposition 5.3.5 that the operator $A|_{H^{\gamma}_{0,A}(\varphi_{\nu})}$ on $Y = H^{\gamma}_{0,A}(\varphi_{\nu})$ is sectorial with

$$\omega(A|_{H^{\gamma}_{0,A}(\varphi_{\nu})}) = \omega(A) = 0.$$

and by Theorem 5.4.3 and Corollary 5.1.9 we know that $A|_{H^{\gamma}_{0,A}(\varphi_{\nu})}$ has a bounded H^{∞} -calculus with

$$\omega_{H^{\infty}}(A|_{H^{\gamma}_{0,A}(\varphi_{\nu})}) = \omega_{\mathrm{BIP}}(A|_{H^{\gamma}_{0,A}(\varphi_{\nu})}) = \frac{1}{\nu}\,\omega_{\mathrm{BIP}}(B|_{H^{\gamma}_{0,B}}) = \pi\nu = \sigma.$$

Here we used that by Proposition 5.3.5 we have for all $x \in c_{00}$

$$B|_{H_{0,B}^{\gamma}}x = (A|_{H_{0,A}^{\gamma}(\varphi_{\nu})})^{1/\nu}x$$

It remains to observe that Y is a closed subspace of $\gamma(\mathbb{R}_+, \frac{dt}{t}; X)$, which is isomorphic to a closed subspace of $L^p(\Omega; \ell^p \oplus \ell^2)$ for some probability space (Ω, \mathbb{P}) , which in turn is isomorphic to a closed subspace of $L^p([0, 1])$, see e.g. [AK16,

146

Section 6.4]. Finally note that Y has Pisier's contraction property and therefore $A|_{H_{0,A}^{\gamma}(\varphi_{\nu})}$ is γ -sectorial by Theorem 4.3.5 and the angle equalities follow from Proposition 4.2.1 and Corollary 5.1.9.

Acknowledgements

The authors would like to thank Martijn Caspers, Christoph Kriegler, Jan van Neerven and Mark Veraar for their helpful comments on the draft version of this memoir and Tuomas Hytoñen for allowing us to include Theorem 2.4.6, which he had previously shown in unpublished work. The authors would also like to thank Mitchell Taylor for bringing the recent developments regarding p-multinorms under our attention. Moreover the authors express their deep gratitude to the anonymous referee, who read this memoir very carefully and provided numerous insightful comments. Finally the authors would like to thank everyone who provided feedback on earlier versions of this manuscript to N.J. Kalton.

Bibliography

- [AFM98] D. Albrecht, E. Franks, and A. McIntosh. Holomorphic functional calculi and sums of commuting operators. Bull. Austral. Math. Soc., 58(2):291–305, 1998.
- [AK16] F. Albiac and N.J. Kalton. *Topics in Banach space theory*, volume 233 of *Graduate Texts in Mathematics*. Springer, second edition, 2016.
- [AL19] L. Arnold and C. Le Merdy. New counterexamples on Ritt operators, sectorial operators and *R*-boundedness. *Bull. Aust. Math. Soc.*, 100(3):498–506, 2019.
- [Alb94] D.W. Albrecht. Functional calculi of commuting unbounded operators. PhD thesis, Monash University Clayton, 1994.
- [ALV19] A. Amenta, E. Lorist, and M.C. Veraar. Rescaled extrapolation for vector-valued functions. Publ. Mat., 63(1):155–182, 2019.
- [AMN97] P. Auscher, A. McIntosh, and A. Nahmod. Holomorphic functional calculi of operators, quadratic estimates and interpolation. *Indiana Univ. Math. J.*, 46(2):375–403, 1997.
- [Aus07] P. Auscher. On necessary and sufficient conditions for L^p -estimates of Riesz transforms associated to elliptic operators on \mathbb{R}^n and related estimates. *Mem. Amer. Math. Soc.*, 186(871):xviii+75, 2007.
- [BC91] J.-B. Baillon and P. Clément. Examples of unbounded imaginary powers of operators. J. Funct. Anal., 100(2):419–434, 1991.
- [Bd92] K. Boyadzhiev and R. deLaubenfels. Semigroups and resolvents of bounded variation, imaginary powers and H^{∞} functional calculus. Semigroup Forum, 45(3):372– 384, 1992.
- [BK03] S. Blunck and P.C. Kunstmann. Calderón-Zygmund theory for non-integral operators and the H^{∞} functional calculus. *Rev. Mat. Iberoamericana*, 19(3):919–942, 2003.
- [BL04] D.P. Blecher and C. Le Merdy. Operator algebras and their modules—an operator space approach, volume 30 of London Mathematical Society Monographs. New Series. The Clarendon Press, Oxford University Press, Oxford, 2004.
- [Bou84] J. Bourgain. Extension of a result of Benedek, Calderón and Panzone. Ark. Mat., 22(1):91–95, 1984.
- [BRS90] D.P. Blecher, Z. Ruan, and A.M. Sinclair. A characterization of operator algebras. J. Funct. Anal., 89(1):188–201, 1990.
- [Cal64] A.P. Calderón. Intermediate spaces and interpolation, the complex method. Studia Math., 24:113–190, 1964.
- [Car09] A. Carbonaro. Functional calculus for some perturbations of the Ornstein-Uhlenbeck operator. Math. Z., 262(2):313–347, 2009.
- [CD17] A. Carbonaro and O. Dragičević. Functional calculus for generators of symmetric contraction semigroups. Duke Math. J., 166(5):937–974, 2017.
- [CD19] A. Carbonaro and O. Dragičević. Bounded holomorphic functional calculus for nonsymmetric Ornstein-Uhlenbeck operators. Ann. Sc. Norm. Super. Pisa Cl. Sci. (5), 19(4):1497–1533, 2019.
- [CD20a] A. Carbonaro and O. Dragičević. Bilinear embedding for divergence-form operators with complex coefficients on irregular domains. *Calc. Var. Partial Differential Equations*, 59(3):Paper No. 104, 36, 2020.

[CD20b] A. Carbonaro and O. Dragičević. Convexity of power functions and bilinear embedding for divergence-form operators with complex coefficients. J. Eur. Math. Soc. (JEMS), 22(10):3175-3221, 2020. [CDMY96] M. Cowling, I. Doust, A. McIntosh, and A. Yagi. Banach space operators with a bounded H^{∞} functional calculus. J. Austral. Math. Soc. Ser. A, 60(1):51-89, 1996. T. Coulhon and D. Lamberton. Régularité L^p pour les équations d'évolution. In [CL86] Séminaire d'Analyse Fonctionelle 1984/1985, volume 26 of Publ. Math. Univ. Paris VII, pages 155-165. Univ. Paris VII, Paris, 1986. [CMP11] D.V. Cruz-Uribe, J.M. Martell, and C. Pérez. Weights, extrapolation and the theory of Rubio de Francia, volume 215 of Operator Theory: Advances and Applications. Birkhäuser/Springer Basel AG, Basel, 2011. [CMP12] D.V. Cruz-Uribe, J.M. Martell, and C. Pérez. Sharp weighted estimates for classical operators. Adv. Math., 229(1):408-441, 2012. [Con90] J.B. Conway. A course in functional analysis, volume 96 of Graduate Texts in Mathematics. Springer-Verlag, New York, second edition, 1990. [CP01] P. Clément and J. Prüss. An operator-valued transference principle and maximal regularity on vector-valued L_p -spaces. In Evolution equations and their applications in physical and life sciences (Bad Herrenalb, 1998), volume 215 of Lecture Notes in Pure and Appl. Math., pages 67-87. Dekker, New York, 2001. [CPSW00]P. Clément, B. de Pagter, F.A. Sukochev, and H. Witvliet. Schauder decompositions and multiplier theorems. Studia Math., 138(2):135-163, 2000. [CV11] S. Cox and M.C. Veraar. Vector-valued decoupling and the Burkholder-Davis-Gundy inequality. Illinois J. Math., 55(1):343-375 (2012), 2011. [DHP03] R. Denk, M. Hieber, and J. Prüss. *R*-boundedness, Fourier multipliers and problems of elliptic and parabolic type. Mem. Amer. Math. Soc., 166(788):viii+114, 2003. [DJT95] J. Diestel, H. Jarchow, and A. Tonge. Absolutely summing operators, volume 43 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 1995. [DK19] L. Deleaval and C. Kriegler. Dimension free bounds for the vector-valued Hardy-Littlewood maximal operator. Rev. Mat. Iberoam., 35(1):101-123, 2019. [DKK18] L. Deleaval, M. Kemppainen, and C. Kriegler. Hörmander functional calculus on UMD lattice valued L^p spaces under generalised Gaussian estimates. To appear in J. Anal. Math. arXiv:1806.03128, 2018. [DLOT17] H.G. Dales, N.J. Laustsen, T. Oikhberg, and V.G. Troitsky. Multi-norms and Banach lattices. Dissertationes Math., 524:115, 2017. [DM99] X.T. Duong and A. McIntosh. Singular integral operators with non-smooth kernels on irregular domains. Rev. Mat. Iberoamericana, 15(2):233-265, 1999. G. Dore. H^∞ functional calculus in real interpolation spaces. Studia Math., [Dor99] 137(2):161-167, 1999. [Dor01] G. Dore. H^{∞} functional calculus in real interpolation spaces. II. Studia Math., 145(1):75-83, 2001. [DP12] H.G. Dales and M.E. Polyakov. Multi-normed spaces. Dissertationes Math., 488:165, 2012.[DR96] X.T. Duong and D.W. Robinson. Semigroup kernels, Poisson bounds, and holomorphic functional calculus. J. Funct. Anal., 142(1):89-128, 1996. [DV87] G. Dore and A. Venni. On the closedness of the sum of two closed operators. Math. Z., 196(2):189-201, 1987. [Ege18]M. Egert. L^{p} -estimates for the square root of elliptic systems with mixed boundary conditions. J. Differential Equations, 265(4):1279-1323, 2018. [Ege20]M. Egert. On *p*-elliptic divergence form operators and holomorphic semigroups. J. Evol. Equ., 20(3):705-724, 2020. [EHRT19] T. ter Elst, R. Haller-Dintelmann, J. Rehberg, and P. Tolksdorf. On the L^p -theory for second-order elliptic operators in divergence form with complex coefficients. arXiv:1903.06692, 2019. [Eid40] M. Eidelheit. On isomorphisms of rings of linear operators. Studia Math., 9:97–105, 1940.[EN00] K.-J. Engel and R. Nagel. One-parameter semigroups for linear evolution equations, volume 194 of Graduate Texts in Mathematics. Springer-Verlag, New York, 2000.

[ER00] E.G. Effros and Z.-J. Ruan. Operator spaces, volume 23 of London Mathematical Society Monographs. New Series. The Clarendon Press, Oxford University Press, New York, 2000. [Fac13] S. Fackler. An explicit counterexample for the L^p -maximal regularity problem. C. R. Math. Acad. Sci. Paris, 351(1-2):53-56, 2013. [Fac14a] S. Fackler. The Kalton-Lancien theorem revisited: maximal regularity does not extrapolate. J. Funct. Anal., 266(1):121-138, 2014. [Fac14b] S. Fackler. On the structure of semigroups on L_p with a bounded $H^\infty\text{-calculus}.$ Bull. Lond. Math. Soc., 46(5):1063-1076, 2014. [Fac15] S. Fackler. Regularity properties of sectorial operators: counterexamples and open problems. In Operator semigroups meet complex analysis, harmonic analysis and mathematical physics, volume 250 of Oper. Theory Adv. Appl., pages 171–197. Birkhäuser/Springer, Cham, 2015. [Fac16] S. Fackler. Maximal regularity: positive counterexamples on UMD-Banach lattices and exact intervals for the negative solution of the extrapolation problem. Proc. Amer. Math. Soc., 144(5):2015-2028, 2016. [FM98] E. Franks and A. McIntosh. Discrete quadratic estimates and holomorphic functional calculi in Banach spaces. Bull. Austral. Math. Soc., 58(2):271-290, 1998. [FT79] T. Figiel and N. Tomczak-Jaegermann. Projections onto Hilbertian subspaces of Banach spaces. Israel J. Math., 33(2):155-171, 1979. [FW06] A.M. Fröhlich and L. Weis. H^{∞} calculus and dilations. Bull. Soc. Math. France, 134(4):487-508, 2006.[Gar90] D.J.H. Garling. Random martingale transform inequalities. In Probability in Banach spaces 6 (Sandbjerg, 1986), volume 20 of Progr. Probab., pages 101–119. Birkhäuser Boston, Boston, MA, 1990. [Gar07] J.B. Garnett. Bounded analytic functions, volume 236 of Graduate Texts in Mathematics. Springer, New York, first edition, 2007. $[GCMM^+01]$ J. García-Cuerva, G. Mauceri, S. Meda, P. Sjögren, and J.L. Torrea. Functional calculus for the Ornstein-Uhlenbeck operator. J. Funct. Anal., 183(2):413-450, 2001. [Gei99] S. Geiss. A counterexample concerning the relation between decoupling constants and UMD-constants. Trans. Amer. Math. Soc., 351(4):1355-1375, 1999. [GKT96] J. García-Cuerva, K.S. Kazarian, and J.L. Torrea. On the Fourier type of Banach lattices. In Interaction between functional analysis, harmonic analysis, and probability (Columbia, MO, 1994), volume 175 of Lecture Notes in Pure and Appl. Math., pages 169-179. Dekker, New York, 1996. [GM01] A.A. Giannopoulos and V.D. Milman. Euclidean structure in finite dimensional normed spaces. In Handbook of the geometry of Banach spaces, Vol. I, pages 707-779. North-Holland, Amsterdam, 2001. [GMT93] J. García-Cuerva, R. Macías, and J.L. Torrea. The Hardy-Littlewood property of Banach lattices. Israel J. Math., 83(1-2):177-201, 1993. [GR85] J. García-Cuerva and J.L. Rubio de Francia. Weighted norm inequalities and related topics, volume 116 of North-Holland Mathematics Studies. North-Holland Publishing Co., Amsterdam, 1985. [Gra14] L. Grafakos. Classical Fourier analysis, volume 249 of Graduate Texts in Mathematics. Springer, New York, third edition, 2014. [Haa03] M.H.A. Haase. Spectral properties of operator logarithms. Math. Z., 245(4):761–779, 2003[Haa06a] M.H.A. Haase. The functional calculus for sectorial operators, volume 169 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 2006. M.H.A. Haase. Operator-valued H^{∞} -calculus in inter- and extrapolation spaces. [Haa06b] Integral Equations Operator Theory, 56(2):197-228, 2006. S. Harris. Optimal angle of the holomorphic functional calculus for the Ornstein-[Har19] Uhlenbeck operator. Indag. Math. (N.S.), 30(5):854-861, 2019. [HH14] T.S. Hänninen and T.P. Hytönen. The A_2 theorem and the local oscillation decomposition for Banach space valued functions. J. Operator Theory, 72(1):193–218, 2014.[HKK04] M. Hoffmann, N. Kalton, and T. Kucherenko. R-bounded approximating sequences and applications to semigroups. J. Math. Anal. Appl., 294(2):373-386, 2004.

[HL19] T.S. Hänninen and E. Lorist. Sparse domination for the lattice Hardy-Littlewood maximal operator. Proc. Amer. Math. Soc., 147(1):271-284, 2019. [HLN16] T.P. Hytönen, S. Li, and A. Naor. Quantitative affine approximation for UMD targets. Discrete Anal., 37(6), 2016. [HNVW16] T.P. Hytönen, J.M.A.M. van Neerven, M.C. Veraar, and L. Weis. Analysis in Banach Spaces. Volume I: Martingales and Littlewood-Paley Theory, volume 63 of Ergebnisse der Mathematik und ihrer Grenzgebiete. Springer, 2016. [HNVW17] T.P. Hytönen, J.M.A.M. van Neerven, M.C. Veraar, and L. Weis. Analysis in Banach spaces. Volume II: Probabilistic methods and operator theory, volume 67 of Ergebnisse der Mathematik und ihrer Grenzgebiete. Springer, 2017. [HV09] T.P. Hytönen and M. Veraar. *R*-boundedness of smooth operator-valued functions. Integral Equations Operator Theory, 63(3):373-402, 2009. [Hyt15] T.P. Hytönen. A quantitative Coulhon-Lamberton theorem. In Operator semigroups meet complex analysis, harmonic analysis and mathematical physics, volume 250 of Oper. Theory Adv. Appl., pages 273–279. Birkhäuser/Springer, Cham, 2015. [Jam87] G.J.O. Jameson. Summing and nuclear norms in Banach space theory, volume 8 of London Mathematical Society Student Texts. Cambridge University Press, Cambridge, 1987. [JJ78] W.B. Johnson and L. Jones. Every L_p operator is an L_2 operator. Proc. Amer. Math. Soc., 72(2):309-312, 1978. [Kal03] N.J. Kalton. A remark on sectorial operators with an H^{∞} -calculus. In Trends in Banach spaces and operator theory (Memphis, TN, 2001), volume 321 of Contemp. Math., pages 91–99. Amer. Math. Soc., Providence, RI, 2003. N.J. Kalton. A remark on the H^{∞} -calculus. In CMA/AMSI Research Symposium [Kal07] "Asymptotic Geometric Analysis, Harmonic Analysis, and Related Topics", volume 42 of Proc. Centre Math. Appl. Austral. Nat. Univ., pages 81-90. Austral. Nat. Univ., Canberra, 2007. [KK08] N. J. Kalton and T. Kucherenko. Rademacher bounded families of operators on L_1 . Proc. Amer. Math. Soc., 136(1):263-272, 2008. [KK10] N.J. Kalton and T. Kucherenko. Operators with an absolute functional calculus. Math. Ann., 346(2):259-306, 2010. [KKW06] N. Kalton, P. Kunstmann, and L. Weis. Perturbation and interpolation theorems for the $H^\infty\text{-}\mathrm{calculus}$ with applications to differential operators. Math. Ann., 336(4):747-801. 2006. [KL00] N.J. Kalton and G. Lancien. A solution to the problem of L^p -maximal regularity. Math. Z., 235(3):559-568, 2000. N.J. Kalton and G. Lancien. L^p -maximal regularity on Banach spaces with a [KL02] Schauder basis. Arch. Math. (Basel), 78(5):397-408, 2002. [KL10] C. Kriegler and C. Le Merdy. Tensor extension properties of C(K)-representations and applications to unconditionality. J. Aust. Math. Soc., 88(2):205-230, 2010. [KU14] P. Kunstmann and A. Ullmann. \mathcal{R}_s -sectorial operators and generalized Triebel-Lizorkin spaces. J. Fourier Anal. Appl., 20(1):135-185, 2014. [Kun15] P.C. Kunstmann. A new interpolation approach to spaces of Triebel-Lizorkin type. Illinois J. Math., 59(1):1-19, 2015. [KVW16] S. Kwapień, M.C. Veraar, and L. Weis. R-boundedness versus γ -boundedness. Ark. Mat., 54(1):125–145, 2016. [KW01] N.J. Kalton and L. Weis. The H^{∞} -calculus and sums of closed operators. Math. Ann., 321(2):319-345, 2001. [KW04] P.C. Kunstmann and L. Weis. Maximal L_p -regularity for parabolic equations, Fourier multiplier theorems and H^{∞} -functional calculus. In Functional analytic methods for evolution equations, volume 1855 of Lecture Notes in Math., pages 65–311. Springer, Berlin, 2004. [KW05] T. Kucherenko and L. Weis. Real interpolation of domains of sectorial operators on L_p-spaces. J. Math. Anal. Appl., 310(1):278–285, 2005. [KW13] P.C. Kunstmann and L. Weis. Erratum to: Perturbation and interpolation theorems for the $H^\infty\text{-}\mathrm{calculus}$ with applications to differential operators [MR2255174]. Math.

Ann., 357(2):801-804, 2013.

[KW16a] N.J. Kalton and L. Weis. The H^{∞} -functional calculus and square function estimates. In Selecta. Volume 1., pages 716–764. Basel: Birkhäuser/Springer, 2016. [KW16b] C. Kriegler and L. Weis. Paley–Littlewood decomposition for sectorial operators and interpolation spaces. Math. Nachr., 289(11-12):1488-1525, 2016. [KW17] P.C. Kunstmann and L. Weis. New criteria for the H^{∞} -calculus and the Stokes operator on bounded Lipschitz domains. J. Evol. Equ., 17(1):387-409, 2017. [Kwa72] S. Kwapień. Isomorphic characterizations of inner product spaces by orthogonal series with vector valued coefficients. Studia Math., 44:583-595, 1972. [Lan98] G. Lancien. Counterexamples concerning sectorial operators. Arch. Math. (Basel), 71(5):388-398, 1998. F. Lancien and C. Le Merdy. Square functions and H^∞ calculus on subspaces of [LL05] L^p and on Hardy spaces. Math. Z., 251(1):101–115, 2005. [LL21] N. Lindemulder and E. Lorist. The sequence structured interpolation method. In preparation, 2021. F. Lancien, G. Lancien, and C. Le Merdy. A joint functional calculus for sectorial [LLL98] operators with commuting resolvents. Proc. London Math. Soc. (3), 77(2):387-414, 1998.[LM96] C. Le Merdy. On dilation theory for c_0 -semigroups on Hilbert space. Indiana Univ. Math. J., 45(4):945-959, 1996. [LM98] C. Le Merdy. The similarity problem for bounded analytic semigroups on Hilbert space. Semigroup Forum, 56(2):205-224, 1998. [LM04] C. Le Merdy. On square functions associated to sectorial operators. Bull. Soc. Math. France, 132(1):137-156, 2004. [LM10] Christian Le Merdy. γ -Bounded representations of amenable groups. Adv. Math., 224(4):1641-1671, 2010. [LM12] C. Le Merdy. A sharp equivalence between H^{∞} functional calculus and square function estimates. J. Evol. Equ., 12(4):789-800, 2012. [LN19] E. Lorist and B. Nieraeth. Vector-valued extensions of operators through multilinear limited range extrapolation. J. Fourier Anal. Appl., 25(5):2608-2634, 2019. [LN20] E. Lorist and Z. Nieraeth. Sparse domination implies vector-valued sparse domination. arXiv:2003.02233, 2020. [LNR04] A. Lambert, M. Neufang, and V. Runde. Operator space structure and amenability for Figà-Talamanca-Herz algebras. J. Funct. Anal., 211(1):245–269, 2004. [Lor16] E. Lorist. Maximal functions, factorization, and the \mathcal{R} -boundedness of integral operators. Master's thesis, Delft University of Technology, Delft, the Netherlands, 2016. [Lor19] E. Lorist. The ℓ^s -boundedness of a family of integral operators on UMD Banach function spaces. In Positivity and Noncommutative Analysis: Festschrift in Honour of Ben de Pagter on the Occasion of his 65th Birthday, pages 365–379, Cham, 2019. Springer International Publishing. [Loz69] G.Ya. Lozanovskii. On some Banach lattices. Siberian Mathematical Journal, 10(3):419-431, 1969. [LP64] J.-L. Lions and J. Peetre. Sur une classe d'espaces d'interpolation. Inst. Hautes Études Sci. Publ. Math., 19(1):5-68, 1964. [LP74] W. Linde and A. Pietsch. Mappings of Gaussian measures of cylindrical sets in Banach spaces. Teor. Verojatnost. i Primenen., 19:472-487, 1974. [LS02] C. Le Merdy and A. Simard. A factorization property of R-bounded sets of operators on L^p-spaces. Math. Nachr., 243:146–155, 2002. [LT77] J. Lindenstrauss and L. Tzafriri. Classical Banach spaces. I. Springer-Verlag, Berlin-New York, 1977. [LT79] J. Lindenstrauss and L. Tzafriri. Classical Banach spaces. II, volume 97 of Ergebnisse der Mathematik und ihrer Grenzgebiete. Springer-Verlag, Berlin-New York, 1979[LZ69] J. Lindenstrauss and M. Zippin. Banach spaces with a unique unconditional basis. J. Functional Analysis, 3:115-125, 1969. [Mar01] J.L. Marcolino Nhani. La structure des sous-espaces de treillis. Dissertationes Math. (Rozprawy Mat.), 397:50, 2001.

[Mau73]	B. Maurey. Théorèmes de factorisation pour les opérateurs linéaires à valeurs dans un espace $L^p(U, \mu)$, $0 . In Séminaire Maurey-Schwartz Année 1972–1973: Espaces L^p et applications radonifiantes, Exp. No. 15, page 8. Centre deMath., École Polytech., Paris, 1973.$
[Mau74]	B. Maurey. Un théorème de prolongement. C. R. Acad. Sci. Paris Sér. A, 279:329–332, 1974.
[McI86]	A. McIntosh. Operators which have an H_{∞} functional calculus. In <i>Miniconference</i> on operator theory and partial differential equations (North Ryde, 1986), volume 14 of <i>Proc. Centre Math. Anal. Austral. Nat. Univ.</i> , pages 210–231. Austral. Nat. Univ., Canberra, 1986.
[Mon97]	S. Monniaux. A perturbation result for bounded imaginary powers. Arch. Math. (Basel), 68(5):407–417, 1997.
[MS86]	V.D. Milman and G. Schechtman. Asymptotic theory of finite-dimensional normed spaces, volume 1200 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 1986.
[Nik70]	E.M. Nikišin. Resonance theorems and superlinear operators. Uspehi Mat. Nauk, $25(6(156)):129-191, 1970.$
[Oik18]	T. Oikhberg. Injectivity and projectivity in <i>p</i> -multinormed spaces. <i>Positivity</i> , 22(4):1023–1037, 2018.
[OY19]	A. Osękowski and I.S. Yaroslavtsev. The Hilbert transform and orthogonal martin- gales in Banach spaces. <i>Int. Math. Res. Not. IMRN</i> , 2019. Online first.
[Pau02]	V. Paulsen. Completely bounded maps and operator algebras, volume 78 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2002.
[Pee69]	J. Peetre. Sur la transformation de Fourier des fonctions à valeurs vectorielles. <i>Rend.</i> Sem. Mat. Univ. Padova, 42:15–26, 1969.
[Pet08]	S. Petermichl. The sharp weighted bound for the Riesz transforms. <i>Proc. Amer. Math. Soc.</i> , 136(4):1237–1249, 2008.
[Pie80]	A. Pietsch. Operator ideals, volume 20 of North-Holland Mathematical Library. North-Holland Publishing Co., Amsterdam-New York, 1980.
[Pis78]	G. Pisier. Some results on Banach spaces without local unconditional structure. <i>Compositio Math.</i> , 37(1):3–19, 1978.
[Pis82]	G. Pisier. Holomorphic semigroups and the geometry of Banach spaces. Ann. of Math. (2), 115(2):375–392, 1982.
[Pis89]	G. Pisier. The volume of convex bodies and Banach space geometry, volume 94 of Cambridge Tracts in Mathematics. Cambridge University Press, Cambridge, 1989.
[Pis03]	G. Pisier. Operator spaces. In <i>Handbook of the geometry of Banach spaces, Vol. 2</i> , pages 1425–1458. North-Holland, Amsterdam, 2003.
[Pis16]	G. Pisier. Martingales in Banach spaces, volume 155 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2016.
[PR07]	B. de Pagter and W.J. Ricker. $C(K)$ -representations and R -boundedness. J. Lond. Math. Soc. (2), 76(2):498–512, 2007.
[PS90]	J. Prüss and H. Sohr. On operators with bounded imaginary powers in Banach spaces. <i>Math. Z.</i> , 203(3):429–452, 1990.
[Rub82]	J.L. Rubio de Francia. Weighted norm inequalities and vector valued inequalities. In <i>Harmonic analysis (Minneapolis, Minn., 1981)</i> , volume 908 of <i>Lecture Notes in</i> <i>Math.</i> marga 26, 101 Company, Backy New York, 1982
[Rub86]	<i>Math.</i> , pages 86–101. Springer, Berlin-New York, 1982. J.L. Rubio de Francia. Martingale and integral transforms of Banach space valued functions. In <i>Probability and Banach spaces (Zaragoza, 1985)</i> , volume 1221 of
[Rud91]	Lecture Notes in Math., pages 195–222. Springer, Berlin, 1986. W. Rudin. Functional analysis. International Series in Pure and Applied Mathe-
[Sei95]	matters. McGraw-rini, Inc., New York, second edition, 1991. K. Seip. On the connection between exponential bases and certain related sequences is $L^2(-\pi, \sigma) = L$. Expect Appl. 120(1):121–160
[Sim99]	In $L_{(-\pi,\pi)}$. J. Funct. Anu., 150(1):151–160, 1995. A. Simard. Factorization of sectorial operators with bounded H^{∞} -functional calculus. Houston L Math. 25(2):351–370, 1000
[Sin70] [SN47]	 I. Singer. Bases in Banach spaces. I. Springer-Verlag, New York-Berlin, 1970. B. Sz-Nagy. On uniformly bounded linear transformations in Hilbert space. Acta

[SW06]	J. Suárez and L. Weis. Interpolation of Banach spaces by the γ -method. In <i>Methods</i>
	in Banach space theory, volume 337 of London Math. Soc. Lecture Note Ser., pages
	293–306. Cambridge Univ. Press, Cambridge, 2006.
[SW09]	J. Suárez and L. Weis. Addendum to "Interpolation of Banach spaces by the γ -method" [MR2326391]. <i>Extracta Math.</i> , 24(3):265–269, 2009.
[Tit86]	E.C. Titchmarsh. Introduction to the theory of Fourier integrals. Chelsea Publishing
[]	Co., New York, third edition, 1986.
[Tom89]	N Tomczak-Jaegermann. Banach-Mazur distances and finite-dimensional operator
	ideals, volume 38 of Pitman Monographs and Surveys in Pure and Applied Mathe-
	matics. Longman Scientific & Technical, Harlow; copublished in the United States
	with John Wiley & Sons, Inc., New York, 1989.
[Ven93]	A. Venni. A counterexample concerning imaginary powers of linear operators. In
	Functional analysis and related topics, 1991 (Kyoto), volume 1540 of Lecture Notes
	in Math., pages 381–387. Springer, Berlin, 1993.
[Ver07]	M.C. Veraar. Randomized UMD Banach spaces and decoupling inequalities for sto-
	chastic integrals. Proc. Amer. Math. Soc., 135(5):1477-1486, 2007.
[Wei01a]	L. Weis. A new approach to maximal L_p -regularity. In Evolution equations and
	their applications in physical and life sciences (Bad Herrenalb, 1998), volume 215
	of Lecture Notes in Pure and Appl. Math., pages 195–214. Dekker, New York, 2001.
[Wei01b]	L. Weis. Operator-valued Fourier multiplier theorems and maximal L_p -regularity.
	Math. Ann., 319(4):735–758, 2001.
[Wit00]	J. Wittwer. A sharp estimate on the norm of the martingale transform. Math. Res.
	Lett., 7(1):1-12, 2000.
[Yar20]	I.S. Yaroslavtsev. Local characteristics and tangency of vector-valued martingales.
	Probab. Surv., 17:545–676, 2020.